

# An investigation of the impacts of unrestricted cattle access to

# watercourses on freshwater physicochemical and microbial

# parameters

Patrícia Oliveira Antunes

B.Sc. Biology, M.Sc. Ecology



# March 2021

A thesis presented to the School of Health and Science, Dundalk Institute of

Technology in fulfilment of the requirements of the degree of Ph. D.

School of Health and Science, Centre for Freshwater and Environmental Studies,

Dundalk Institute of Technology

Supervisors: Prof. Eleanor Jennings, Dr Daire O'hUallacháin (Teagasc)

## Declaration

I hereby certify that this material, which I now submit for assessment on the programme of study leading to the award of Ph. D, is entirely my own work, and that I have exercised reasonable care to ensure that the work is original, and does not to the best of my knowledge breach any law of copyright, and has not been taken from the work of others save to the extent that such work has been cited and acknowledged within the text of my work.

Patueia Olivei a Quamez. Signed:

ID No.: D00189016 Date: 14/09/2021 We, the undersigned declare that this thesis entitled *An investigation of the impacts of unrestricted cattle access to watercourses on freshwater physicochemical and microbial parameters* is entirely the author's own work and has not been taken from the work of others, except as cited and acknowledged within the text.

The thesis has been prepared according to the regulations of Dundalk Institute of Technology and has not been submitted in whole or in part for an award in this or any other institution.

Author Name: Patrícia Oliveira Antunes

Author Signature:

Patueia Olivei a Quamez.

Date: 14/09/2021

Supervisor Name: Eleanor Jennings

Supervisor Signature:

Date: 17/09/2021

#### Acknowledgements

This study was funded as part of the project *COSAINT – Cattle exclusion from watercourses: environmental and socio-economic implications,* by the Environmental Protection Agency under their EPA Research 2014 – 2020 Programme.

Firstly, I would like to thank my supervisors, Dr Eleanor Jennings and Dr Daire O hUallacháin, for their guidance, infinite patience, and kindness. You have been extremely supportive and understanding and I am very grateful to you.

I would also like to thank other members of the COSAINT team who have provided me with advice and assistance, in particular to Dr Matt O'Sullivan, Dr Paul O'Callaghan, Dr Mary Kelly-Quinn from University College Dublin; Dr Fiona Regan, Dr Ciprian Briciu-Burghina and Maria O'Neill from Dublin City University; and Dr Paul Kilgarriff from Teagasc. Special thanks to the members of the EPA Steering Committee for their valuable comments and advice throughout the project.

Sincere thanks to Dr Raymond Flynn for the comments provided during my transfer examination. Thanks also to Dr Orla Sherlock and Dr Joseph Lynch for feedback on some of the written material.

I would also like to thank Dr Lyubov Bragina for guidance and support during laboratory work. Sincere thanks also to everyone who has assisted me in laboratory work and field work, in particular to Allison Murdock, Noelle Dunne, Shannon Dixon, Tom Curray, Tom Cooney, Ewan Geffroy and Amruta Patel. I would also like to thank to the Teagasc staff that have assisted me during visits to sampling sites and liaised with landowners.

iv

Massive thanks to all my friends and colleagues who have supported me in many different ways, in particular Maria Caldero Pascual, Tadhg Moore and Victor Carbajal Perello. Thank you for being present when I most needed, for listening to me, and for your kind words that kept me going – you are amazing friends. A special thanks to Harriet Wilson, who became a very close, precious friend and who listened to me countless times, offered me immense love and support, and really made a difference in the most difficult times.

Thank you to all the incredible people who I have met in Dundalk during my years in this country, who have offered me kindness and encouraged me to keep going. Lastly, I would like to thank my parents, Antonio (Toze) and Odete, my brother Joao, and my grandparents Daniel and Carminda, Armando and Ilda, for your never-ending love and support.

# **Table of Contents**

1.	Intr	oduction	1
	1.1.	Aims and objectives of this research	5
	1.2.	Thesis structure	6
2.	Lite	rature Review	8
	2.1. A	griculture is one of the major threats to freshwater systems globally	9
	2.2 Pa	astoral agriculture in Europe and in Ireland	10
	2.3.	Key contaminants from agriculture	11
	2.3.1.	Phosphorus	11
	2.3.2.	Nitrogen	14
	2.3.3.	Excess suspended solids	17
	2.3.4.	Contamination of surface waters with faecal material	18
	2.4. H	ow does cattle-based agriculture impact on freshwater systems?	22
	2.5. U	nrestricted cattle access to watercourses	23
	2.5.1.	Impacts on stream hydromorphology and sedimentation	24
	2.5.2.	Impacts on stream water and sediment nutrient levels	26
	2.5.3.	Faecal contamination of stream waters and sediments	30
	2.6. W	ater quality protection in European and Irish policy	32
	2.6.1.	Agri-environment schemes	35
	2.6.2.	Recent policy developments	37
	2.6.3. protec	Measures to restrict cattle access to watercourses for water q	uality 38
	2.7.	Relevance of the current study	38
3.	Site	Selection and Description	40
	3.1.	Site selection criteria	41
	3.2.	Study catchments and experimental design	43
	3.3.	Site description	49
	3.3.1.	Munster Blackwater catchment	51
	3.3.2.	Douglas River catchment	55
	3.3.3.	Brackan River catchment	59
	3.3.4.	Commons River catchment	63
	3.3.5.	Milltown Lake catchment	67
4. fa	Imp lecal c	acts of unrestricted cattle access to watercourses on streambed sedimen ontamination in agricultural streams	i <b>t</b> 71
	4.1.	Introduction	72
	4.2.	Methods	76
	4.2.1.	Site selection and experimental design	76
	4.2.2.	Site description	78

4.2.3.	Sediment sampling	78
4.2.4.	E. coli enumeration	79
4.2.5.	Sediment characterisation	79
4.2.6.	Statistical analysis	82
4.3.	Results	83
4.3.1.	Background levels of faecal contamination	83
4.3.2. seaso	Impact of cattle access to streams on sediment faecal contamination	on and 86
4.3.3.	Cumulative downstream gradient	89
4.4.	Discussion	92
4.4.1.	Background levels of stream sediment contamination	94
4.4.2. conta	Impacts of cattle direct access to watercourses on sediment mination and seasonal variation	faecal 95
4.4.3.	Downstream cumulative pattern of faecal contamination at the study site	s97
4.5.	Implications of stream sediment faecal contamination	98
4.6.	Conclusions	98
5. Imp nutrien	bacts of unrestricted cattle access to watercourses on streambed sedime t concentrations in agricultural streams	<b>nt</b> 100
5.1.	Introduction	101
5.2.	Methods	107
5.2.1.	Site selection and experimental design	107
5.2.2.	Sediment sampling	108
5.2.3.	Sediment analysis	109
5.2.4.	Estimation of stream nutrient loadings	111
5.2.5.	Statistical analysis	111
5.3.	Results	113
5.3.1.	Particle size distribution of the samples	113
5.3.2. fractio	Nutrient concentrations in the <63 µm and <2mm particle size se	diment 114
5.3.3.	Effects of direct cattle access on stream sediment nutrient levels	122
5.3.4.	Sediment nutrient loads	126
5.3.5. levels	Effects of cattle density and intensity of impact on stream sediment r	nutrient 130
5.4.	Discussion	133
5.4.1.	Enrichment of the <63µm fraction	134
5.4.2.	Impacts of cattle access on sediment nutrient concentrations	135
5.4.3.	Impacts of cattle density on sediment nutrient concentrations	136
5.4.4.	Variation in sediment nutrient concentrations between sampling times	138
5.5. l	mplications	138
6. A n physico	ear-real time assessment of the effects of cattle in-stream activity on wat ochemical and microbial parameters	: <b>er</b> 141

vii

6.1.	Introduction	142
6.2.	Methods	146
6.2.1.	Study site	146
6.2.2.	Water sampling	150
6.2.3.	Water analysis	151
6.2.4.	Data processing	153
6.2.5.	Statistical analysis	154
6.3.	Results	155
6.3.1.	Background variation in water quality parameters	155
6.3.2.	Changes in water quality parameters during cattle access to the stream	160
6.3.3.	Loads of nutrients, TSS and E. coli bacteria during cattle access	184
6.4.	Discussion	193
6.4.1.	Impact of cattle access on total suspended solids in waters	193
6.4.2.	Impact of cattle access on <i>E. coli</i> bacteria in waters	195
6.4.3.	Impact of cattle access on $NH_4$ -N in waters	197
6.4.4.	Impact of cattle access on SRP and TP in waters	199
6.4.5.	Impact of cattle access on NO $_3$ -N in waters	200
6.5.	Implications	200
7. A sl	hort study investigating the effectiveness of fencing as mitigation measur	<b>e</b> 202
7.1.	Introduction	203
7.2. S	ite description and methods	206
7.2.1.	Site description	206
7.2.2.	Sample collection	213
7.2.3.	Laboratory analysis	214
7.2.4.	Data analysis	214
7.3.	Results	215
7.3.1.	Before-after study on the impacts of fencing on short-term	215
7.3.2.	Paired treatment-control study on the effects of fencing on long-term	219
7.4. D	iscussion	225
7.5.	Conclusions	228
8. Fina	al Discussion	229
8.1.	Key findings	233
8.1.1. contar	Unrestricted cattle access to watercourses contributed to sediment f	aecal 235
8.1.2. reserv	Cattle access to watercourses potentially contributes to stream sed oirs of phosphorus	iment 238
8.1.3. suspe	Cattle in-stream activity consistently results in increased concentrations o nded solids and <i>E. coli</i> bacteria in stream waters	f total 239
8.1.4. <i>E. coli</i>	Fencing of watercourses can have positive impacts on sediment nutrient <i>i</i> reservoirs	s and 241

8.2.	Implications for management	241
8.3.	Implications for policy	
8.4.	Limitations of the study and future research recommendations	247
8.4.1.	Uncertainty associated with the study	247
8.4.2.	Influence of site-specific characteristics and management	
8.4.3.	Sediment faecal contamination and public health risk	
8.4.4.	Effectiveness of fencing as a mitigation measure	
8.4.5.	Studies on phosphorus fractions at cattle access sites	250
8.4.6.	Studies on cattle behaviour	250
8.5.	Overall conclusion	250
Appendices		
References		

# List of Tables

## Chapter 3

**Table 3.1.** Quality Rating System (Q-value) classification and relationship to the WaterFrameworkDirectiveDirective(WFD)waterqualityclassification(takenhttp://www.epa.ie/QValue/webusers/).

**Table 3.2.** Summary of the characteristics of the study catchments.

**Table 3.3.** Description of the Munster Blackwater (BW) catchment. Water quality status information was derived from Edenireland.ie; information at time of sampling (2016/2017).

**Table 3.4.** Summary characteristics of the study sites (BWA, BWB, and BWC) in the Munster Blackwater (BW) catchment.

**Table 3.5.** Description of the Douglas River (DG) catchment. Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

**Table 3.6.** Summary characteristics of the study sites in the Douglas River (DG) catchment.

**Table 3.7.** Description of the Brackan River (BK) catchment. Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

**Table 3.8.** Summary characteristics of the study sites in the Brackan River (BK) catchment.

**Table 3.9.** Description of the Commons River (CM) catchment. Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

 Table 3.10.
 Summary characteristics of the study sites in the Commons River (CM) catchment.

**Table 3.11.** Description of the Milltown Lake (MT) catchment. Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

**Table 3.12.** Summary characteristics of the study sites in the Milltown Lake (CM) catchment.

### Chapter 4

**Table 4.1** Concentrations of *E. coli* bacteria in cattle faeces reported in literature.\*figures presented are calculated from reported results.

Table 4.2. Allocation of study sites in Study 1 and Study 2 of the experimental design.

**Table 4.3.** Moisture content of the sediment samples and sediment organic carbon at the study sites (mean  $\pm$  S.E.).

**Table 4.4.** Mean *E. coli* concentrations in sediments at the sites in the five study catchments in both sampling times (n = 3).\*n = 2. nd: not detected. Grey shading = nine headwater sites.

**Table 4.5.** Analysis of variance of *E. coli* in sediment in Study 1 (repeated measures ANOVA; n = 36) and Study 2 (linear mixed model analysis; n=36). Treatment = US and CAS. Time = mid-grazing season and post-grazing season. Figures in bold denote statistical significance at the 0.05 level.

**Table 4.6.** Post-hoc comparisons (Tukey test) for the analysis of *E. coli* in sediment in Study 2 (interactions between factor Treatment (US and CAS and Time (mid-grazing season [MG] and post-grazing season [PG]). Figures in bold denote statistical significance at the 0.05 level.

#### Chapter 5

**Table 5.1.** Nitrogen and phosphorus concentrations in cattle urine and faeces reported in literature. \*Figures presented here are averaged from the published data.

**Table 5.2.** Summary of some of the studies that have reported nutrient concentrations in soils, bank sediments or bed sediments in agricultural catchments.TP and TN concentrations are in mg.kg dry wt<sup>-1</sup> unless stated otherwise.

**Table 5.3.** Particle size distribution and coverage of sand and silt+clay particles at the study sites.

**Table 5.4.** Sediment nutrient concentrations in the <2mm and <63 $\mu$ m particle size fraction for each catchment for each sampling time (n = 60), location sampled (n = 15) and catchment (n = 12) (mean ± S.E.).

**Table 5.5.** Statistical parameters for the effect of sediment fraction (GLS analysis) and for the effect of Location, Time and interaction between the two factors (mixed effects modelling) on sediment nutrient concentrations in the <  $63\mu$ m fraction. Significant effects are shown in bold.

**Table 5.6.** Statistical parameters for the effect of Location, Time and interaction between the two factors (mixed effects modelling) on sediment nutrient concentrations in the <2mm fraction. Significant effects are shown in bold.

**Table 5.7.** Sediment nutrient loads in each study catchment and at each location in late grazing season (mean  $\pm$  S.E.).

**Table 5.8.** Pairwise comparisons within factor Location for sediment nutrient loadings in late grazing season. Significant factors are shown in bold.

**Table 5.9.** Generalised additive model parameters for the effect of cattle density on sediment nutrient concentrations (n = 120) (ECD = estimated cattle density; edf = estimated degrees of freedom). Note that the optimum model for TP <63 $\mu$ m used an ordered factor approach, and therefore includes a parametric term for the factor Location, where upstream (US) was set as the reference level. Only values for significant effects are shown except for ECD\*Location (INT) effect.

**Table 5.10.** Estimation of cattle density in EU Member States. Cattle density was calculated by dividing an estimation of the total number of cattle by the area devoted to permanent grassland in each country.

### Chapter 6

**Table 6.1.** Water quality parameters at the study site at Dunleer (CM3) in 2016 (data for one composite sample based on five subsamples).

**Table 6.2.** Characteristics of the bed sediment at the study site at the study site (CM3) in 2016.

**Table 6.3.** List of sampling events conducted in this study. Events discussed in this chapter are in bold.

**Table 6.4.** Description of the sampling events presented in this chapter.

**Table 6.5.** Concentrations of nutrients, total suspended solids (TSS) and *E. coli* bacteria measured upstream of the cattle access site during the sampling events (mean  $\pm$  S.D.).

**Table 6.6.** Peak concentrations of nutrients, total suspended solids and *E. coli* bacteria measured downstream (DS) of the access site during periods of cattle access to the stream, and concentrations measured upstream (US) for the same sampling interval (15 minute composite samples).

**Table 6.7.** Calculated differences in nutrient, TSS and *E. coli* bacteria loads at the downstream of the site relative to upstream loads during periods of cattle in-stream activity.

**Table 6.8.** Statistical parameters for the Kolmogorov-Smirnov test with bootstrap resampling (n =1000). Significant effects are shown in bold.

### Chapter 7

**Table 7.1.** Study sites in the Slaney catchment (not described previously in Chapter 3). \*Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017)..

**Table 7.2.** Summary characteristics of the study sites in the Slaney catchment.

**Table 7.3.** Characteristics of the streams used in the paired control-treatment study, in the Milltown Lake catchment (from Veerkamp, 2019).

**Table 7.4.** Description of the sites sampled in the fenced (TV) and unfenced (TH) tributaries in the Milltown Lake catchment (from Bragina, 2017).

**Table 7.3.** Bed sediment concentrations of nutrients (mean  $\pm$  S.E., n = 6) and *E. coli* (mean  $\pm$  S.E., n = 5) at the study sites between upstream (US) and cattle access site (CAS): prefencing (2016) and post- fencing (2017).

**Table 7.5.** Bed sediment (mean  $\pm$  S.E., n = 6) data for the study sites at the upstream control (US) and the interface (INT) at the cattle access site: pre-fencing (2017) and postfencing (2018); data in bold were significantly different US versus INT.

**Table 7.6.** Stream water nutrients and *E. coli* concentrations in the fenced and unfenced streams in the Milltown Lake catchment (mean  $\pm$  S.E.).

**Table 7.7.** Stream water nutrients and *E. coli* concentrations in the fenced and unfenced streams in the Milltown Lake catchment (mean  $\pm$  S.E.). Mean monthly discharge is calculated from daily mean discharge available from the EPA monitoring station in the Drumleek river.

# List of Figures

## Chapter 2

**Fig.2.1.** Potential impacts of unrestricted cattle access to streams on different physicochemical and biotic parameters, as reviewed by O'Callaghan et al. (2018). Arrows in gauges indicate consensus in the reviewed literature regarding the effectiveness of fencing on the mitigation of cattle access impacts (Y = consensus that there was mitigation; N = consensus that there was no mitigation; ? = the studies were inconclusive). Figure taken from O'Callaghan et al. (2018),

**Fig.2.2.** Left: streambank erosion and trampling at a cattle access site. Right: streambank collapse. Both sites are in the Brackan River catchment (Co Wexford).

Fig.2.3. Cow urinating directly in waters at access point.

## Chapter 3

Fig.3.1. Site selection criteria for the COSAINT project.

Fig.3.2. Map of the study catchments and selected sites.

**Fig.3.3.** Location of the sampled sites in the catchments. Red circles highlight the headwater sites (9 in total), and yellow circles highlight the sites used to assess potential downstream cumulative impacts of cattle access to the streams (9 in total).

**Fig.3.4.** Sampling scheme showing the different locations of sample collection at each cattle access site.

**Fig.3.5.** Cattle access sites in the Munster Blackwater (BW) catchment. From the top to the bottom: BWA, BWB, BWC.

Fig. 3.6. Cattle access sites in the Douglas River (DG) catchment. From the top to the bottom: DGA, DGB.

**Fig.3.7.** Cattle access sites in the Bracken River (BK) catchment. From the bottom, clockwise: BK3A, BK2A/BK3B, BK1A, BK1B BK2B.

Fig.3.8. Cattle access sites in the Commons River (CM) catchment. From the top to the bottom: CM1, CM2, CM3.

**Fig.3.9.** Cattle access sites in the Milltown Lake (MT) catchment. From the top to the bottom: MT1, MT2, MT3.

## Chapter 4

**Fig.4.1.** Boxplot of *E. coli* ( $\log_{10}$ CFU.g dry wt<sup>-1</sup>) concentrations in the sediments of the five study catchments in mid-grazing season and post-grazing season (n = 18).

**Fig.4.2.** Boxplot of average sediment concentrations of *E. coli* at cattle access sites (CAS) and at upstream areas with no cattle access (US) at the nine headwater sites (study 1) (n = 9).

**Fig.4.3.** Boxplots of average sediment concentrations of E. coli per type (n = 9), site (n = 6) and catchment assessed in study 2 (n = 18), in mid-grazing season and post-grazing season. Note the different scales for different sampling times.

**Fig.4.4.** Scatterplot and smoother for RHI scores in the GAM with levels of *E. coli* in sediments as response variable. The central solid line is the smoother and the grey area is 95% confidence bands. Y axis units are the scaled smoother (s) for RHI with edf of 1.73.

**Fig.4.5.** Scatterplot and smoother for estimated cattle density (ECD; cattle.ha<sup>-1</sup>) in the GAM with levels of *E. coli* in sediments in post-grazing season as response variable. The central solid line is the smoother and the grey area is 95% confidence bands. Y axis units are the scaled smoother (s) for RHI with estimated degrees of freedom (edf) of 3.43.

### Chapter 5

**Fig. 5.1.** Boxplots of sediment concentrations in the <63  $\mu$ m and <2 mm sediment fractions in the five study catchments, in early grazing season (EG) and late grazing season (LG) (n = 12).

**Fig.5.2.** Scatterplots showing the relationships between nutrients in the  $<63\mu$ m (top row) and the <2mm (bottom row) particle size fractions.

**Fig.5.3.** Boxplots of sediment concentrations in the  $<63\mu$ m particle size sediment fractions at each of the locations sampled at the cattle access sites, in early grazing season (EG) and late grazing season (LG) (n = 15).

**Fig.5.4.** Boxplots of sediment concentrations in the <2mm particle size sediment fractions at each of the locations sampled at the cattle access sites, in early grazing season (EG) and late grazing season (LG) (n = 15).

**Fig. 5.5**. Nutrient loads in the silt+clay fraction (top row) and <2mm fraction (bottom row) in late grazing season at each study location.

**Fig.5.6.** Top row: fine sediment (<63µm) total phosphorus (TP), organic carbon (OC) and total nitrogen (TN) concentrations versus estimated cattle density (ECD) (including data for the early and late grazing season and for upstream, interface, cattle access points, and downstream locations); bottom row: generalised additive models using these data. The central solid line is the smoother, the grey area is the 95% confidence bands, and the Y axis units are the scaled smoother (s) for ECD with estimated degrees of freedom (edf).

**Fig5.7.** A. smoother for the effect of estimated cattle density (ECD) on TP for upstream (US) sites (reference level), and difference smooths reflecting the estimated differences between the upstream sites and the CAP sites (CAS, top, right), interface sites (INT, bottom, left), and downstream (DS, bottom, right) respectively. The central solid line is the smoother, the grey area is the 95% confidence bands, and the Y axis units are the scaled smoother (s) with estimated degrees of freedom (edf).

# Chapter 6

**Fig 6.1.** Schematic representation of the experimental site and position relatively to the ACP hydrometric station.

**Fig. 6.2.** Images taken during sampling events. a) fence pole view showing 3 animals in the stream in Event 2; b) side view of at least 4 animals in the stream in Event; c) in-stream urination and d) defecation in Event 9; e) one animal in Even10; f) side view of two animals and g) several animals in the stream in Event 16; h) front view of one animal in Event 15, after the sampling period.

**Fig.6.3** Variation in mean nutrients, TSS and *E. coli* bacteria concentrations upstream and downstream of the cattle access site during Event 8 (May 24, 2017), when cattle were present in the fields but did not access the stream during the sampling period. The dashed red line represents the limit of the detection for the analysis of the parameter.

**Fig.6.4.** Variation in nutrients, TSS and *E. coli* bacteria concentrations upstream and downstream of the cattle access site during Event 12 (August 30, 2017), when a maximum of three animals accessed the stream simultaneously. The dashed red line represents the limit of the detection for the analysis of the parameter.

**Fig.6.5.** Variation in nutrients, TSS and *E. coli* bacteria concentrations upstream and downstream of the cattle access site during Event 7 (May 10, 2017), when a maximum of six animals accessed the stream simultaneously. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. The dashed red line represents the limit of the detection for the analysis of the parameter

**Fig.6.6.** Variation in TSS concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales.

**Fig.6.7.** Variation in TSS concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

**Fig.6.8.** Variation in TP concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales.

**Fig.6.9.** Variation in TP concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the

**Fig.6.10.** Variation in SRP concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales.

**Fig.6.11.** Variation in SRP concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

**Fig.6.12.** Variation in  $NH_4$ -N concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales. The dashed red line represents the limit of the detection for the analysis of the parameter

**Fig.6.13.** Variation in NH<sub>4</sub>-N concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. The dashed red line represents the limit of the detection for the analysis of the parameter. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

**Fig.6.14.** Variation in NO<sub>3</sub>-N concentrations upstream and downstream of the cattle access during events with no cattle access to the stream.

**Fig.6.15.** Variation in NO<sub>3</sub>-N concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

**Fig.6.16.** Variation in *E. coli* bacteria concentrations upstream and downstream of the cattle access during events with no cattle access to the stream.

**Fig.6.17.** Variation in *E. coli* bacteria concentrations upstream and downstream of the cattle access during the sampling events. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

**Fig.6.18.** Violin plots for the differences in loads downstream in relation to upstream of nutrients, TSS and *E. coli* bacteria during cattle access to the stream and during periods of no access.

**Fig.6.19.** Scatterplots showing the relationship between the differences in loads of nutrients, TSS and *E. coli* bacteria downstream relative to upstream during periods of cattle access to the stream (orange) and periods of no access (blue), intensity of access expressed in cattle-minutes.

**Fig.6.20.** Boxplots and empirical cumulative distribution functions (ECDF) for the difference in loads of a) TSS, b-e) nutrients and f) *E. coli* bacteria downstream of the access site relative to upstream during periods of cattle access to the stream and periods of no access (grey line on ECDF = Access, black = No Access); points are the values where there is a maximum difference between the two distributions.

# Chapter 7

**Fig.7.1.** Map of the study sites in the *before-after* study of fencing effectiveness as a water quality protection measure.

**Fig.7.2.** Map of the study sites in the paired *treatment - control* study of fencing effectiveness as a water quality protection measure.

**Fig.7.3.** Change in the mean sediment organic carbon (OC, row 1), total nitrogen (TN, row 2), and total phosphorus (TP, row 3) concentration at the upstream (US) control sites and at the cattle access site (CAS) and from the US sites versus the interface (INT) for the six sites (columns 1 and 3 = pre-fencing (2016) and columns 2 and 4 = post-fencing (2017).

**Fig.7.4.** Change in the mean sediment *E. coli* concentration at the upstream (US) control sites and at the cattle access site (CAS) for the five sites (left = pre-fencing (autumn 2016) and right = post-fencing (autumn 2017).

**Fig.7.5.** Monthly total precipitation (mm) in the Milltown Lake catchment from December 2017 to December 2018 (data from the Met Eireann station Coose – Castleblayney), discharge (m<sup>3</sup>.s<sup>-1</sup>) in the Drumleek River (data from the nearest EPA monitoring station) and nutrient concentration variation in the TV tributary (fenced) (black) and TH tributary (unfenced) (grey).

**Fig.7.6.** Stream water nutrient concentrations in the fenced and unfenced streams. Top row: (December 2017 to November 2018, no sample in February and September 2018) (SRP: n = 60; NH<sub>4</sub>-N: n = 48; TP and NO<sub>3</sub>-N: n = 47). Bottom row: during the period of cattle grazing outdoors (April to October 2018, no sample in February 2018) (SRP: n = 36; TP: n = 35; NH<sub>4</sub>-N and NO<sub>3</sub>-N: n = 24).

**Fig.7.7.** Stream water *E coli* concentration the fenced and unfenced streams. a) and c) data obtained using the membrane filtration technique ( $\log_{10}$  CFU.100 ml<sup>-1</sup>); b) and d) data obtained using the Colilert method (Most Probable Number (MPN).100 ml<sup>-1-</sup>); a) and b) data collected in November and December 2017, January, May and November 2018 (MF: n = 59; Colilert method: n = 48, no data for November 2017); c) and d) data for May 2018 when cattle were grazing outdoors (n = 12).

# Chapter 8

**Fig.8.1.** Diagram showing how unrestricted cattle access to watercourses can affect stream geochemical and microbial parameters. Parameters in bold are impacts demonstrated in this study.

## List of Publications and Presentations

## **Publications**

O'Callaghan, P., Kelly-Quinn, M., Jennings, E., Antunes, P., O'Sullivan, M., Fenton, O., and Ó hUallacháin, D., 2018. The environmental impact of cattle access to watercourses: a review. J. Environ. Qual. 48(2), 340 - 351

O'Sullivan, M., Ó hUallacháin, D., Antunes, P.O., Jennings, E., Kelly-Quinn, M., 2019. The impacts of cattle access points on deposited sediment levels in headwater streams in Ireland. River Res. Appl. 1–13. <u>https://doi.org/10.1002/rra.3382</u>

O'Sullivan, M., Ó hUallacháin, D., Antunes, P., Jennings, E., Kelly-Quinn, M., 2019. The impacts of cattle access to headwater streams on hyporheic zones. Biology and Environment: Proceedings of the Royal Irish Academy 119B(1), 13 – 27. https://doi.org/10.3318/bioe.2019.02

Ó hUallacháin, D., Jennings, E., Antunes, P. Green, S., Kilgarriff, P., Linnane, S., O'Callaghan, P., O'Sullivan, M., Regan, F., Ryan, M., Kelly-Quinn, M., 2020. COSAINT: Cattle exclusion from watercourses: environmental and socio-economic implications.EPA Research Report No.330. Available online at http://www.epa.ie/pubs/reports/research/water/Research\_Report\_330.pdf

Antunes, P., Ó hUallacháin, D., Dunne, N, Kelly-Quinn, M, O'Sullivan, M, and Jennings, E. (In prep).Unrestricted cattle access to watercourses increases streambed sediment concentrations of *Escherichia coli*.

Antunes, P., Ó hUallacháin, D., Kelly-Quinn, M., O'Sullivan, M., and Jennings, E. (In prep). Assessing the contribution of unrestricted cattle access to watercourses to streambed sediment nutrient reservoirs in agricultural catchments.

### Oral Presentations

Antunes, P., Bragina, L., Dunne, N., Van Rossum, A.J., Sherlock, O., Jennings, E., Ó'hUallacháin, D., Kelly-Quinn, M., O'Sullivan, M., 2017. Stream sediment reservoirs of E. coli – hidden contamination in the stream sediment? Annual Irish Freshwater Biology Meeting 2017, 10<sup>th</sup> March 2017, Dundalk Institute of Technology, Ireland/

Antunes, P., Jennings, E., Ó hUallacháin, D., Dunne, N., Kelly-Quinn, M., O'Sullivan, M., Regan, F., 2018. Assessing the impacts of cattle access to watercourses on water and sediment faecal contamination in farmland streams. 28<sup>th</sup> Irish Environmental Research Colloquium (ENVIRON), 27<sup>th</sup> – 29t<sup>h</sup> March 2018, Cork Institute of Technology, Ireland.

Antunes, P., Jennings, E., Ó hUallacháin, D., Kelly-Quinn, M., O'Sullivan, M., Regan, F., 2018. A real-time sampling experiment to assess the impacts of cattle access on freshwater

biogeochemical parameters in a farmland stream.34<sup>th</sup> Congress of the International Society of Limnology, 19<sup>th</sup> – 24<sup>th</sup> August 2018, Nanjing, China.

Antunes, P., Jennings, E., Ó hUallacháin, D., Dunne, N., Bragina, L., Kelly-Quinn, M.,O'Sullivan, M., 2019. If cattle drink, water might stink. Pint of Science Festival, 22<sup>nd</sup> May 2019, Dundalk, Ireland.

Antunes, P., Ó hUallacháin, Kelly-Quinn, M., O'Sullivan, M., Regan, F., Jennings, E., 2021. The impacts of cattle in-stream activity on water physicochemical and microbial parameters. 12<sup>th</sup> Symposium for European Freshwater Sciences, 25<sup>th</sup> – 30<sup>th</sup> July 2021.

#### Poster Presentations

Antunes, P., Jennings, E., Ó hUallacháin, D., ,Kelly-Quinn, M., Linanne, S., Murnaghan, S., O'Callaghan, P., O'Sullivan, M., Regan, F., 2016. The impact of cattle access and exclusion from watercourses on freshwater geochemical and microbial parameters. 26<sup>th</sup> Irish Environmental Research Colloquium (ENVIRON), March 22<sup>nd</sup> – 24<sup>th</sup> 2016, University of Limerick, Ireland.

Antunes, P., Jennings, E., Ó hUallacháin, D., ,Kelly-Quinn, M., Linanne, S., Murnaghan, S., O'Callaghan, P., O'Sullivan, M., Regan, F., 2016. The impact of cattle access and exclusion from watercourses on freshwater geochemical and microbial parameters. Global Lake Ecological Survey Network (GLEON) 18<sup>th</sup> All-Hands Meeting, 4<sup>th</sup> – 8<sup>th</sup> July 2016, Lunz & Gaming, Austria.

Antunes, P., Jennings, E., Ó hUallacháin, D., Dunne, N., Bragina, L., Kelly-Quinn, M.,O'Sullivan, M., 2017. Assessing the impact of cattle access to watercourses in stream faecal contamination. 27<sup>th</sup> Irish Environmental Research Colloquium (ENVIRON),10t<sup>h</sup> – 12<sup>th</sup> April 2017, Athlone Institute of Technology, Ireland.

# List of abbreviations

AEOS	Agri-Environment Options Scheme
AES	Agri-Environment Scheme
AIC	Akaike Information Criterion
AI	Aluminium
ANOVA	Analysis of Variance
ATCC	American Type Culture Collection
BK	Brackan River catchment
BMP	Best Management Practice
BW	Munster Blackwater catchment
BWD	Bathing Water Directive
Са	Calcium
CaCO <sub>3</sub>	Calcium carbonate
CAP	Common Agricultural Policy
CAS	Cattle Access Site
CFU	Colony forming units
CI	Chloride
СМ	Commons River catchment
COSAINT	Cattle exclusion from watercourses: Environmental and Socio-economic Implications
CSA	Critical source area
CSO	Central Statistics Office
D	Defecation
DAFM	Department of Agriculture, Food and the Marine
DED	District Electoral Division
DF	Degrees of freedom
DG	Douglas River catchment
DO	Dissolved Oxygen
DoHPLG	Department of Housing, Planning and Local Government
DWD	Drinking Water Directive
DS	Downstream
E. coli	Escherichia coli
EC	European Commission
ECD	Estimated Cattle Density
ECDC	European Centre for Disease Prevention
edf	Estimated degrees of freedom
EEA	European Environment Agency
EEA	European Economic Area
EFSA	European Food Safety Authority
EG	Early grazing season
EPA	Environmental Protection Agency
EU	European Union
Fe	Iron
FIB	Faecal Indicator Bacteria

FST	Faecal Source Trackers
GAM	General Additive Models
GF/C	Glass Fibre Filters Grade C
GLAS	Green Low carbon Agri-environment Scheme
GLS	General Least Squares
GO	Gentle Owen
HCI	Hydrochloric acid
IMS	Industrial Methylated Spirits
INT	Interface
ISO	International Organisation for Standardisation
KH <sub>2</sub> PO <sub>4,</sub>	Potassium phosphate
KS	Kolmogorov-Smirnov
LG	Late grazing season
LOI	Loss on Ignition
LSU	Livestock Units
MARS	Managing Aquatic ecosystems and water Resources under multiple Stress
MG	Mid-grazing season
Mn	Manganese
MPN	Most Probable Number
MS	Member State
MT	Milltown Lake catchment
Ν	Nitrogen
N2	Dinitrogen
$Na_3PO_4$	Sodium phosphate
NaNO <sub>3</sub>	Sodium nitrate
NAP	Nitrates Action Plan
ND	Nitrates Directive
NE	North-East
NH4	Ammonium nitrogen
NH <sub>4</sub> CI	Ammonium chloride
NO <sub>3</sub>	Nitrate
NO <sub>x</sub> -N	Oxidised Nitrogen
NSPPP	National Source Protection Pilot Project
NVZ	Nitrates Vulnerable Zone
OC	Organic Carbon
OPW	Office of Public Works
Р	Phosphorus
PCR	Polymerase Chain Reaction
PG	Post-grazing season
RBMP	River Basin Management Plan
REPS	Rural Environmental Protection Scheme
RHI	Riparian Habitat Index
SAC	Special Area of Conservation
SD	Standard deviation
SE	Standard error

SRP	Soluble reactive phosphorus
TH	Tullycaghney
TKN	Total Kjeldahl Nitrogen
TN	Total Nitrogen
TON	Total Oxidised Nitrogen
TP	Total Phosphorus
TRP	Total Reactive Phosphorus
TSS	Total suspended solids
TV	Tievnamara
U	Urination
UK	United Kingdom
US	Upstream
USA	United States of America
USEPA	United States Environmental Protection Agency
UV	Ultraviolet
VBNC	Viable but non-culturable
VTEC	Verocytotoxigenic <i>E. coli</i>
WFD	Water Framework Directive
WHO	World Health Organisation

#### Abstract

Where pastoral agriculture dominates, the practice of allowing livestock access to farmland watercourses as a cheap and low maintenance source of drinking water has been shown to have an adverse impact on water quality. In Ireland, agriculture, which is predominantly cattle-based, has been linked to the downward trend in water quality observed in recent decades, which conflicts with the goals of the Water Framework Directive. However, the research investigating the potential impacts of cattle access to watercourses on freshwater systems has predominantly been conducted in the USA and Australasia. In these regions, climate and farming practices typically contrast with those observed in Ireland and in Europe, thus making comparison difficult. This study aimed at contributing to an understanding of the impacts of cattle access to watercourses on freshwater systems in the Irish setting. Specifically, the current study investigated the impacts of cattle access to streams on freshwater abiotic parameters (i.e. nutrients, sediment) and on freshwater faecal contamination. It also addressed the efficacy of streamside fencing as a mitigation measure for such impacts.

The findings of this study show that faecal contamination in watercourses draining agricultural areas in Ireland is widespread. *Escherichia coli* (*E. coli*) concentrations in the streambed sediment compartment were in the order of  $10^3$  to  $10^4$  CFU g dry wt<sup>-1</sup> at stream reaches with no cattle access, during grazing season. However, at stream reaches with unrestricted cattle access, *E. coli* sediment concentrations were significantly higher, with maximum average concentrations of  $1.6 \times 10^7$  CFU g dry wt<sup>-1</sup> in mid-grazing season. Sediment *E.* coli concentrations were found to decrease but persist in post-grazing season, with concentrations at upstream reaches of up to  $10^3$  CFU g and up to  $10^5$  CFU g dry wt<sup>-1</sup> at cattle access sites. Furthermore, the study found a significant negative correlation between the Riparian Habitat Index scores of the sites, which reflect the intensity of cattle access impact (and whereby a lower score indicates higher site degradation) and *E. coli* sediment concentrations in mid-grazing season.

XXV

Nutrient (TP, TN and OC) reservoirs in sediments at cattle access sites were assessed. Results here indicate that cattle access to watercourses does not generally result in localised nutrient accumulation in streambed sediments. However, this study found a significant positive relationship between cattle density at the access sites and all three nutrient concentrations in the silt and clay fraction of the sediments. The results suggest that while cattle access may contribute to sediment nutrient reservoirs, rapid flushing downstream of nutrients at access sites may occur. Additionally, results suggest that sediment nutrient concentrations in agricultural streams are mainly driven by diffuse pollution at the catchment scale.

Potential changes in water quality parameters during cattle in-stream activity were assessed in a near-real time experiment which showed that cattle access led to significant increases in water loads of *E. coli* bacteria, TSS and  $NH_4$ -N. While increases were observed in TP loads, these were not significant in the context of the natural variation at the experimental site.  $NO_3$ -N and SRP did not show significant variation in response to cattle in-stream activity.

A short study on the effects of streamside fencing showed a positive effect in streambed sediment concentrations of *E. coli* and nutrients. However, this study also highlighted the need of adopting a site-specific, holistic approach that combines cattle exclusion measures with other measures to control diffuse pollution if cattle-based agriculture pressures are to be successfully managed.

The current research has thus demonstrated that cattle access to watercourses contributes to both faecal and nutrient contamination of freshwaters, and that while microbial pollution is mainly governed by field-scale management, nutrient pollution is driven by catchment-scale practices. Results presented here support the implementation of fencing to exclude cattle from watercourses; however, this needs to be considered as part of wider, integrated catchment management plans. The study contributes to the literature describing agricultural pressures in headwater streams and provides relevant information for stakeholders and policy makers.

xxvi

# 1. Introduction

### **Chapter 1. Introduction**

Freshwater accounts for only 0.01% of the water volume in the world and 0.8% of the land surface cover (Pham et al., 2019), but supports almost 6% of all species described to date (Dudgeon et al., 2006). Additionally, freshwaters provide a wide range of ecosystem services, such as provisional services (e.g. drinking water, irrigation water for agriculture, food), regulating services (e.g. water purification, flood buffering), supporting services (e.g. nutrient cycling) and recreational services (e.g. spiritual, leisure and aesthetic value) (Pham et al., 2019). However, largely due to their vital role in supporting human populations, freshwater systems are one of the most impacted systems in the world (Dudgeon et al., 2006), threatened by urbanisation and industry activities, agriculture intensification, water abstraction, flow diversion and damning, introduction of invasive species and diseases, and climate change (Malmqvist and Rundle, 2002).

Agriculture has been long recognised as a major pressure on freshwater systems globally (Vörösmarty et al., 2010). In the 1990s, agriculture was the main cause of pollution of freshwater resources in the USA (Cooper, 1993), and has been identified as the main source of contaminants to freshwater systems in New Zealand (Howard-Williams et al., 2010) and throughout Europe (Ulén et al., 2007). Much of this pollution, which includes excess nitrogen (N) and phosphorus (P) from fertiliser and slurry application, excess fine sediment loadings and faecal contaminants, reaches surface waters through diffuse pathways of contamination (Deakin et al, 2016; Heathwaite, 2010; Muirhead and Monaghan, 2012), which, by definition, do not have a single origin, but are rather scattered in the landscape.

Excess nutrients can result in eutrophication of freshwater resources (Dodds and Oakes, 2008; Smith et al., 1999), thereby impacting aquatic biota and affecting drinking water supplies (Smith et al., 1999). Excess fine sediments can reduce water clarity and affect

primary producers and aquatic foodwebs (Hickey et al., 1994; Davies-Colley et al., 2008, Izagirre et al. 2009), and also smother bed substrates with resulting implications for benthic fauna (Braccia and Voshell, 2006), as well as potentially interfering with biogeochemical cycles in the hyporheic zone (Jones et al., 2015). Faecal material can be associated with pathogens and with nutrient enrichment and can therefore affect drinking water supplies and preventing recreational use of surface waters (Howard-Williams et al., 2010), and may also contribute to eutrophication (James et al., 2007).

Recognising the need to reduce the pressures on freshwater systems, several pieces of legislation have been developed to protect freshwaters and ensure their sustainable use. In Europe, water protection legislation began in 1975, with the introduction of standards for rivers and lakes used for drinking water abstraction, and saw a major development with the introduction of the Water Framework Directive (2000/60/EC) (WFD) in 2000 (European Commission [EC], 2020a). The WFD aggregated previous legislation aimed at water quality protection, expanded their scope to all waterbodies, introduced water management based at river basin scale, and defined a concrete goal: to achieve at least good ecological status of all European waterbodies. Such developments at the policy and management level were fruitful in improving water quality, having, for instance, led to a reduction of N surpluses by 18% between 2000 and 2015 (European Environmental Agency [EEA], 2018a). Nevertheless, the recent State of Environment Report (SOER) 2020 of the EEA stated that, despite such recent improvements, environmental pressures on European freshwaters remain substantial, with diffuse pollution from agriculture and hydromorphological degradation being by far the major issues (EEA, 2019). Currently, approximately 50% of the European surface waterbodies are classified as moderate or bad status, thereby failing to meet the WFD goals (Hering and Birk, 2018). The MARS – Managing Aquatic ecosystems and water Resources under multiple Stress Project (2014 – 2018), launched by the EU with the aim of understanding how multiple stressors affect surface waters and informing management and policy, acknowledged the complexity the environmental pressures

currently faced by European freshwaters. In its recommendations for mitigation, the MARS report indicated that, in particular for rivers, a consistent and broad scale implementation of riparian buffer strips was necessary to address both diffuse pollution and hydromorphological deterioration (Hering and Birk, 2018).

In regions where pastoral agriculture dominates, as is the case with Ireland, additional sources of pollution can include discrete areas in streams or rivers that are used by livestock to gain access to drinking water and/or and to cross between parcels of land. Such practice has been reported as having negative effects on riparian vegetation (Platts and Nelson, 1985), streambank stability and resistance to erosion and stream channel morphology (Kauffman et al. 1983; Trimble 1994; Trimble and Mendel 1995; Sovell et al. 2000), leading to increased sedimentation, loss of in-stream habitat and alterations in benthic fauna communities (Braccia and Voshell, 2007; Zaimes and Schultz, 2011). Moreover, where livestock have unrestricted access to watercourses, they often void faeces and urine within the channel or in the proximity to waters. This has been linked to increases in water and sediment faecal contamination (Davies-Colley et al., 2004), with potential implications for both human and animal health.

Despite the evidence that unrestricted cattle access to watercourses can have detrimental impacts on water quality (O'Callaghan et al., 2018), the body of literature addressing the topic is relatively small, and the extent of the impacts as well as the mechanisms involved remain unclear (O'Callaghan et al., 2018). Factors such as stocking density and hydroclimatic conditions are likely to influence the extent to which unrestricted cattle access can impact the aquatic system, making it challenging to quantify such impacts and implement appropriate mitigation measures (Madden et al., 2019). To date, the majority of the studies addressing the topic has been conducted in the USA, Canada and New Zealand, where management, hydrological and climate conditions differ from Europe. There is, therefore, a lack of detailed understanding of the contribution of cattle access to

watercourses on freshwater systems in a European context, and of the effectiveness of mitigation measures that address this issue in protecting water quality. Despite this, in Ireland, mitigation measures that exclude cattle from watercourses have been incorporated in agri-environmental policy since the implementation in 1994, almost three decades ago, of the first Irish agri-environmental scheme (AES), the Rural Environment Protection Scheme (REPS).

### 1.1. Aims and objectives of this research

The work presented in this thesis aimed at contributing to the limited body of literature on the topic by investigating how cattle access to watercourses affects freshwater physicochemical parameters and potentially contributes to excess nutrients and faecal contamination of waters. The levels of these contaminants were assessed in the stream sediment compartment given its widely recognised ability to act as both a reservoir to nutrients and bacteria and a source of these pollutants to overlying waters. Despite this important function, little attention has been given specifically to the sediment compartment in studies focusing on the impacts of livestock in-stream activity on freshwater quality. Therefore, the specific objectives of this study were:

- 1. To investigate the impact of unrestricted cattle access to watercourses on stream sediment concentrations of *Escherichia coli*
- To assess the contribution of unrestricted cattle access to stream sediment levels of nitrogen (N), phosphorus (P) and organic carbon (OC)
- 3. To quantify the near real time (i.e. through 3 minutes interval continuous sampling) changes in water physicochemical parameters (i.e. dissolved phosphorus and nitrogen, total phosphorus, total suspended solids) and faecal contamination (*E. coli* bacteria) before, during and after cattle in-stream activity

4. To assess the effectiveness of cattle exclusion fencing as a mitigation measure in terms of key sediment and freshwater parameters

#### 1.2. Thesis structure

This thesis aims to address the objectives identified above through four integrated data chapters, followed by an overall discussion chapter. Chapter 1 provides a background for the study and outlines its specific objectives. A detailed literature review is presented in Chapter 2, focusing on the impacts of agriculture on freshwater systems globally and in Ireland, and how cattle-based agriculture and unrestricted cattle access to watercourses can further contribute to water quality degradation. Chapter 3 provides a description of the selected sites sampled in this study. Sites were clustered at the catchment scale incorporating both intensively and extensively managed catchments. Inclusion of a variety of sites facilitated the evaluation of local/reach scale effects in relation to catchment-scale effects. Chapter 4 examines the impacts of cattle access to watercourses on bed sediment concentrations of E. coli bacteria, commonly used as indicators of faecal contamination and consequent health risk to human populations. The results provide estimates of the level of E. coli contamination of stream bed sediment from cattle access. Similar to Chapter 4, Chapter 5 investigates how cattle access to watercourses contributed to enriched nutrient concentrations in stream sediments. Chapter 6 describes a high temporal resolution monitoring experiment to assess the impacts of cattle in-stream activity on a number of water quality parameters, such as nutrients, total suspended solids and E. coli bacteria. Intensive sampling of stream water was undertaken at a selected cattle access point capturing both periods of cattle in-stream activity and periods of cattle absence. Chapter 7 describes a short study with the aim of assessing the efficacy of fencing watercourses (to exclude cattle) as a mitigation measure. The study investigated the potential benefits of fencing in the short-term (after ~ one year), in

terms of the parameters analysed in Chapters 4 and 5, and also presents an assessment of the effectiveness of fencing on water quality by analysing freshwater parameters (nutrients, *E. coli* bacteria) in two streams (one fenced ~ nine years and one unfenced for that time) over a one year study. Each of these four chapters contains a more targeted literature review on their specific topic. Finally, Chapter 8 reviews the findings of the previous chapters, presents conclusions and discusses the implications of this research in agricultural management and policy.

# 2. Literature Review

Sections of this literature review contributed to the following review paper:

O'Callaghan, P., Kelly-Quinn, M., Jennings, E., Antunes, P., O'Sullivan, M., Fenton, O., Ó hUallacháin, D., 2018. The environmental impact of cattle access to watercourses: a review. J. Environ. Qual. 48(2), 340 - 351

#### **Chapter 2. Literature Review**

#### 2.1. Agriculture is one of the major threats to freshwater systems globally

Deterioration of freshwater resources, as a result of physical alteration and degradation, habitat loss, nutrient enrichment, introduction of alien species, overexploitation, pollution and climate change is a global primary concern (Malmqvist and Rundle, 2002;Dudgeon et al., 2006; Heathwaite, 2010; Vörösmarty et al., 2010). Degradation of freshwater systems has been documented in many parts of the world (Malmqvist and Rundle, 2002). In Europe, freshwater ecosystems and their associated ecosystem services have been deteriorating since the 1950s, mainly due to a combination of physical modification of rivers and streams, water abstraction, drainage and eutrophication caused by diffuse pollution sources such as agriculture (Follett and Hatfield, 2001; Malmqvist and Rundle, 2002; Harrison et al., 2010; Ulén et al., 2007; Vidon et al., 2008; Heathwaite, 2010; Smolders et al., 2015).

One of the most concerning threats to freshwater systems is eutrophication, which has been defined by Nixon (1995) as 'an increase in the supply of organic matter to an ecosystem'. High concentrations of nutrients in freshwater systems can result in an increase in primary producer biomass, particularly algae or cyanobacteria in the system, with subsequent high rates of decomposition and decreases in dissolved oxygen (Smith et al., 1999). This in turn impacts aquatic biota, whereas hypoxic conditions can cause sediment-bound contaminants to be released into waters (Correll, 1998). The addition of high loads of organic matter from increased primary production or other sources, for example as animal or human faeces, and its subsequent decomposition, can also lead to reductions in dissolved oxygen in waters and therefore affect oxygen-sensitive macroinvertebrate communities (Braccia and Voshell, 2006). Separately, high ammonium inputs from animal or human sources can pose a direct

toxicity threat to aquatic communities (Camargo et al., 2005). Excess nutrients, particularly phosphorus, generally the main limiting nutrient in freshwater systems, can also accumulate in the stream sediments, potentially causing internal chronic pollution effects (Sharpley et al., 2013; Fox et al., 2016).

#### 2.2 Pastoral agriculture in Europe and in Ireland

In 2016, 173 million ha in the European Union (EU-28) were used for agricultural production, representing about 47.1% of the total land area (Eurostat, 2019). In Ireland, agriculture accounts for approximately 67.4% of the land area (Central Statistics Office [CSO], 2020). Livestock production accounts for approximately 65% of the EU agricultural land (Leip et al., 2015), with 92% of the agricultural area of Ireland dedicated to grassland and rough grazing (CSO, 2020). There were 87 million bovine animals in EU in 2018 (Eurostat, 2019), with the majority of these animals kept in seven Member States (MS): France (21.2 %), Germany (13.7 %), the United Kingdom (11.0 %), Ireland (7.5 %), Spain (7.4 %), Italy (7.2 %) and Poland (7.1 %) (Eurostat, 2019). Cattle constitute the majority of livestock in two MS: Luxembourg (84%) and Ireland (82%) (Eurostat, 2020).

Leip et al. (2015) estimated that livestock agriculture in Europe was responsible for 73% of water pollution (both N and P) caused by agriculture. Diffuse sources of agricultural pollution affected 38% of the waterbodies surveyed by the European Environment Agency (EEA) (EEA, 2018b). The EEA reported in 2018 that nutrient enrichment from agriculture and loss of habitat due to hydromorphological changes were amongst the main pressures on European surface waters, with less than half (40%) of the surface waters were in good or high ecological status or potential (EEA, 2018b).
In Ireland, water quality has been declining in recent decades. In its most recent water quality monitoring programme (2013 – 2018), the Environmental Protection Agency (EPA), reported that one-third of river and lakes were failing to meet nutrient environmental quality standards and a quarter of rivers and lakes showing increasing nutrient concentrations (EPA, 2019). Indeed, this assessment showed an overall net decline in water quality in comparison to the previous full assessment (2010-2015), with 47.2% of surface water bodies currently in less than good ecological status (EPA, 2019). This change was nearly entirely driven by a decrease in river water quality, with 4.4% of the monitored river water bodies declining in status (EPA, 2019). Additionally, the EPA highlighted a concerning steady deterioration of the highest quality Irish river bodies over the past decades, with watercourses considered pristine falling from 13.4% (575 sites) in 1987 – 1990 to only 0.7% (20 sites) in 2016 – 2018. This downward trend in quality conflicts with the goals of the EU Water Framework Directive (WFD; Directive 2000/60/EC, 2000), whereby all EU Member States are required to achieve at least good chemical and ecological water quality in all surface water bodies, as well as maintaining high water quality sites, preventing deterioration of these systems. Moreover, surface waters constitute the main source of drinking water in Ireland (81.5%) (EPA, 2016).

# 2.3. Key contaminants from agriculture

#### 2.3.1. Phosphorus

Phosphorus is generally the limiting nutrient for primary production in freshwater systems (Smith, 2003; Carpenter, 2008), and, as such, plays a major role in eutrophication of surface waters. Agriculture is considered the major source of P to surface waters in agricultural catchments (Neidhart et al., 2019). Phosphorus is applied to agricultural soils in the form of slurries and animal manures, which contain predominantly organic P forms, or chemical

fertilisers, which contain mainly inorganic soluble or bound forms of P. Soluble inorganic P is readily available to plants; however P is also a highly particle-reactive element (McGechan et al., 2005), particularly in the presence of clays and calcium, aluminium and iron ions (Reddy et al., 1999; Withers and Jarvie, 2008). Soil inorganic P is in equilibrium with P that is loosely bound to soil particles, so that bound inorganic P is converted to soluble P in soil pore water when plant uptake reduces the concentrations of soluble P. However, continuous P applications can result in P-saturated soils, with increased total P pools and higher soluble P levels (Fox et al., 2016).

Delivery of phosphorus to surface waters typically occurs in particulate forms in surface runoff (Deakin et al., 2016), particularly in poached or eroded soils. The greatest agricultural losses of P typically occur during storm events (Palmer-Felgate et al., 2009; Sharpley et al., 2013). In P-saturated soils, significant quantities of dissolved phosphorus can also be transported to surface waters through subsurface pathways (Jennings et al., 2003; Deakin et al., 2016). Both dissolved and particulate forms of P can be transported through subsurface preferential flowpaths (Records et al., 2016). At times of low flows, when the proportion of P contributed to freshwaters by diffuse catchment sources is lower, point sources of P can be relatively important (Heathwaite, 2010).

In aquatic systems, P exists as a dissolved inorganic molecule, adsorbed onto particulate material, incorporated in biomass or incorporated within organic molecules of varying complexity (Jennings et al., 2003). Bioavailable P, defined as phosphorus in those fractions 'that can be that are readily assimilated by organisms, or can be made assimilable through the activities of organisms, and that portion which has already been assimilated' (Reynolds and Davies, 2001), is in the orthophosphate form (PO<sub>4</sub><sup>3-</sup>). The biogeochemical cycling in freshwater systems is complex and depends on several factors including climate, hydrology, and reactivity of soils and sediments (House, 2003). Processes of P retention within lotic systems include biological uptake (by macrophytes, periphyton and microorganisms), P

adsorption and precipitation onto sediments (e.g., precipitation with Fe or Mn hydroxides, coprecipitation of phosphate with Ca), and deposition of particulate forms of P during stable or falling stream discharge (House, 20039). Conversely, processes through which P is released into the water sediment include remobilisation of P-rich sediments and associated release of dissolved P from sediment pore water, desorption of P from sediments, and organic P hydrolysis (Reddy et al., 1999; House, 2003;).

Phosphorus interaction with sediments is a major process in the P cycling in freshwater systems, particularly in rivers and streams where the ratio of bed sediment surface to water volume is relatively high (House and Denison, 2002; House, 2003). In such systems, this interaction of P with sediments is an important factor regulating their productivity (Reddy et al., 1999). The capacity of bed sediments to retain P is influenced by several factors including the overlying waters composition, local redox conditions, water:bed sediment ratio, water residence time, and sediment properties such as particle size, presence of metal oxide coatings, and concentration of exchangeable phosphate adsorbed onto the sediment (Palmer-Felgate et al., 2009; Fox et al., 2016;). Additionally, the availability of other elements, such as carbon (C) and N, and their ratios to P can also strongly influence P immobilisation in sediments (Records et al., 2016). Organic matter can also inhibit P sorption to aluminium and iron oxides, thereby reducing the sediment's capacity to retain P (Records et al., 2016).

The incorporation of P in bed sediments results in changes in P forms in the system, which has implications for P bioavailability (Palmer-Felgate et al., 2009). For example, highly bioavailable soluble inorganic P is converted in less bioavailable particulate forms following uptake by Fe-rich sediments (Palmer-Felgate et al., 2009). In impacted systems, however, the sediments can store large pools of TP, which can be converted into bioavailable forms when, for instance, redox conditions change, or the diffusion gradient across the sediment-water interface is reversed due to a reduction in waters concentrations of available P

resulting from pollution control measures or land use changes (Palmer-Felmate et al., 2009; Jarvie et al., 2013; Neidhart et al., 2019). Thus, this stored P can be released from the sediments back to the water column over time, thereby increasing P loadings to downstream waterbodies with a lag time that can be of years or decades (Neidhart et al., 2019). Furthermore, these P reservoirs can be mobilised in particulate forms when sediment disturbance occurs, for example during high flows, with implications to downstream waterbodies (Palmer-Felgate et al., 2009). This 'legacy P' effectively becomes a chronic source of pollution to waterbodies, and can hinder the effectiveness of P pollution mitigation measures (Jarvie et al., 2013; Sharpley et al., 2013), thus representing a challenge for successful nutrient management in agricultural catchments.

#### 2.3.2. Nitrogen

Agriculture is widely recognised as the largest single source of N inputs to the freshwater environment (Birgand et al., 2007). In addition to causing problems in freshwater systems (Follett and Hatfield, 2001), increasing riverine nitrogen loads to coastal waters can altered receiving ecosystems (Jones et al., 2018). Nitrate is a weakly negatively charged ion which is highly soluble and leaches readily from agricultural soils if it is not incorporated into plant biomass (Deakin et al., 2016). It is often delivered to freshwater systems via subsurface pathways, particularly in well-drained systems (Deakin et al., 2016). In streams, the nitrogen cycle is governed by biogeochemical reactions strongly associated with stream sediments and other substrata, which occur through interactions between the surface water, subsurface water and the hyporheic zone (Durand et al., 2011; Trimmer et al., 2012; Corner-Warner et al., 2020). The nitrate concentration in stream waters is influenced by a number of retentionrelease mechanisms, such as biotic uptake, abiotic adsorption, remineralisation and burial of nitrogen associated with organic matter, and removal mechanisms, including denitrification, ammonia volatilisation and export downstream (Bernot and Dodds, 2005; Trimmer et al., 2012; Welsh et al., 2017). Ammonia volatilisation, however, is generally negligible as the pH

of most freshwaters is not high enough (> pH 8) for  $NH_4^+$  to be converted to  $NH_3$  and for this reaction to occur (Bernot and Dodds, 2005).

In agricultural streams, the majority of inorganic nitrogen is in the form of nitrate (van Kessel et al., 2009), although ammonium (and organic nitrogen) can also be abundant in streams where animal slurries are applied (Birgand et al., 2007). Nitrate and ammonium can be taken up directly by aquatic plants and algae, as well as low molecular weight dissolved organic nitrogen (DON) (Durand et al., 2011). Studies investigating the relative contribution of macrophyte assimilation in nitrogen retention in streams have reported that macrophyte uptake can account for between 5 and 70% of nitrogen retention (e.g. House et al., 2001; Howard-Williams et al., 1982; Jansson et al., 1994). The proportion of nitrogen that is removed from the water column by macrophyte uptake is influenced by several factors including macrophyte density and spatial repartition, nitrate affinity and time of the year (Birgand et al., 2007). Few studies have been conducted on nitrogen removal from streams by algae. Microbial biofilms (i.e. microbial communities embedded in a self-produced matrix of extracellular polymeric substances attached on stream substrates) influence the nitrogen cycle in streams by removing nitrogen from the water column via assimilation, and recycling N into the water column by mineralisation. Additionally, biofilms can alter biogeochemical conditions locally thus affecting microbial transformation rates (Bernot and Dodds, 2005).

As with biotic uptake, storage and burial of nitrogen within bed sediments is an important mechanism of nitrogen removal from the water column in agricultural streams (Birgand et al., 2007). Some clays incorporate nitrogen as fixed NH<sub>4</sub> (Bernot and Dodds, 2005), which can act as a mechanism of nitrogen removal from the water column. In sediments, anaerobic conditions tend to exist at different depths, driven by the oxygen demand of decomposing organic matter, and the limited diffusion of oxygen from overlying waters (Birgand et al., 2007). In anaerobic conditions, the mineralisation of buried organic nitrogen does not proceed from the process of ammonification, which produces NH<sub>4</sub> (Kuypers et al., 2018).

Thus, organic-rich sediments in agricultural streams are often characterised by an accumulation of  $NH_4$  in interstitial pore water of the anoxic zone. This creates a concentration gradient with overlying waters which drives the upward diffusion of  $NH_4$  from sediments to waters (Birgand et al., 2007). In the aerobic zone of the sediment, however, nitrification can occur, whereby  $NH_4$  is converted to  $NO_3$  by chemoautotrophic organisms.

The nitrification process is an important mechanism in retention and removal of nitrogen in streams because it results in the production of NO<sub>3</sub> from NH<sub>4</sub>, which is very soluble and more likely to be mobilised and transported downstream. This is because ammonium can bind to organic and inorganic particles through ion exchange, which decreases its mobility (Bernot and Dodds, 2005). Moreover, nitrification is also often coupled with the denitrification process, whereby nitrate is reduced to gaseous nitrogen oxide which can then be further oxidised to dinitrogen (N<sub>2</sub>). Denitrification is carried out by facultative anaerobic microorganisms which utilise nitrate as an electron acceptor in their respiratory chain in the absence of oxygen (Canfield et al., 2010). It is controlled by a number of factors including the availability of and quality of organic carbon (functioning as electron source), redox conditions, temperature and pH (Corner-Warner et al., 2020). The coupling of nitrification denitrification was first hypothesised by Patrick and Reddy (1976). They suggested that the nitrification of ammonium in the aerobic layer of the sediments created a concentration gradient that caused the upward diffusion of ammonium in the underlying anaerobic layer, which would then undergo nitrification. The resulting nitrate would then diffuse downwards to the anaerobic layer driven by a nitrate concentration gradient between the overlying water, the aerobic surface of the sediments and the anaerobic zone; there, it would be denitrified (Birgand et al., 2007). The existence of adjacent nitrification and denitrification zones in the sediments was later experimentally proven by Sweerts and de Beer (1989) and Jensen et al (1993). Jensen et al (1993) further suggested that the proportion of nitrate produced by nitrification that undergoes denitrification rather than being lost to the water column is influenced by diffusion distances from the aerobic layer of the sediment to the water column

and the anaerobic layer. They hypothesised that when oxygen penetrates deeper into the sediment, nitrification occurs further from the surface of the sediment, favouring nitrate diffusion downward to the denitrification layer over diffusion to overlying waters (Jensen et al., 1993). Both nitrification and denitrification therefore take place primarily in the stream sediments (Butturini et al., 2000). Rooted macrophytes can favour the coupling of nitrification and denitrification by increasing organic matter contents and creating aerobic conditions in the rhizosphere (Birgand et al., 2007; Forshay and Dodson, 2011). Aquatic fauna also directly and indirectly influence the nitrogen cycle, particularly those that burrow vertically in the sediment profile (Nickerson et al., 2019).

Other biochemical reactions in the nitrogen cycle include the dissimilatory nitrate reduction to ammonium (DRNA) and anaerobic ammonia oxidation (annamox). DRNA, however, has rarely been measured directly in freshwaters (Trimmer et al., 2012). Annamox is the combination of ammonium with nitrite, in anaerobic conditions, to produce N<sub>2</sub>. The process is carried out by chemolithoautotrophic organisms and, as with denitrification, leads to the permanent removal of nitrogen from the aquatic system (Burgin and Hamilton, 2007).

#### 2.3.3. Excess suspended solids

The effects of excess suspended solids pressure on aquatic systems resulting from intensification of agriculture are a global concern (Nader et al., 2016). Suspended solids, defined as fine organic and inorganic particulate matter (<62µm) (Bilotta and Brazier, 2008) can be delivered in excessive quantities to watercourses in areas where these systems and sediments are hydrologically connected and agriculture has resulted in increased soil erosion (Sherriff et al., 2019). This sediment connectivity is controlled by factors such as climate, lithology and land use (Sherriff et al., 2019). Poorly drained catchments, for

example, might have higher sediment connectivity because increased surface runoff can result in higher sediment quantities transferred to watercourses (Mellander et al., 2012).

Augmented supplies of suspended solids to freshwater systems can have several detrimental effects. They can cause reduced light penetration, thereby affecting primary producers, cause temperature changes, and smothering of watercourse substrates upon deposition, resulting in habitat loss for benthic taxa (Kemp et al., 2011). Suspended particles can also cause abrasion on aquatic organisms and clog respiratory and feeding organs of invertebrates and fish (Bilotta and Brazier, 2008). Additionally, suspended solids can release contaminants into waters, such as pesticides or phosphorus (Bilotta and Brazier, 2008). Where suspended solids are rich in organic matter, decomposition may lead to oxygen depletion in waters with implications for aquatic organisms (Bilotta and Brazier, 2008).

## 2.3.4. Contamination of surface waters with faecal material

Faecal contamination is a major cause of water impairment in many countries, including developing nations, but also countries with advanced water treatment systems (Smolders et al., 2015) such as the USA (Rehmann and Soupir, 2009) and New Zealand (Muirhead et al., 2004; Collins et al., 2007). In Ireland, a report for the year 2016 stated that 42% of the groundwater WFD sites were contaminated with faecal indicator organisms, highlighting the need for testing and adequately treating groundwater drinking supplies (EPA, 2018). Indeed, faecal contamination is an on-going issue in drinking water, particularly in small private water supplies (EPA, 2020).

Faecal contamination of water resources is concerning because a large number of infections can be transmitted through the consumption of contaminated water. The main groups of microorganisms that can cause waterborne infection include protozoa, bacteria and viruses (Gray, 2008). Bacteria are the most important group of faecal pathogens, accounting for the

majority of waterborne disease outbreaks (Gray, 2008). Faecal pathogenic bacteria include Salmonella sp., Campylobacter sp., and Escherichia. coli (Gray, 2008). Salmonella sp. (Gray, 2008) and Campylobacter sp. (Evans et al., 2003) are common causes of gastroenteritis in Europe. Escherichia. coli is present in the normal human gastrointestinal flora (Kaper et al., 2004). However there are several distinct pathogenic serotypes, including enterohaemorrhagic *E. coli* (EHEC; also referred to as verocytotoxigenic serotypes), of which *E. coli* O157:H7 is considered the most important serotype (Kaper et al., 2004). *E. coli* O157:H7 causes haemorrhagic colitis, haemolytic uraemic syndrome and kidney disease in children. Several outbreaks of *E. coli* O157:H7 have been documented; in the United States; for instance, 4928 cases of infection were reported between 2003 and 2012 (Heiman et al., 2015). In Europe, the average incidence of *E. coli* O157:H7 infection 2018 was 2.4 per 100,000 population, a sharp increase in comparison to the previous four years (1.7 - 1.8) (European Centre for Disease Control [ECDC], 2020).

In addition to potential human and animal health risks, it has been observed that cattle frequently avoid or limit consumption of water contaminated with faecal matter due to poor water palatability (Willms et al., 2002). A decrease in water consumption has been associated with a decrease in forage consumption, and consequently in weight gains (Willms et al., 2002; Lardner et al., 2005), suggesting that failure in providing animals with adequate drinking water sources may lead to economic losses.

In waters, faecal bacteria are usually found in lower concentrations in the water column when compared to bed sediments (Pachepsky and Shelton, 2011; Ling et al., 2012). It has been suggested that, in open waters, bacteria are less able to survive due to nutrient deprivation, predation, inactivation by sunlight and competition with native organisms (Alm et al., 2003). In contrast, association with sediments may favour bacterial persistence as a result of higher nutrient availability and protection from predation (Desmarais et al., 2002) and UV radiation (Kim et al., 2010) Factors influencing bacterial survival in sediments

include temperature, salinity and sediment characteristics (Pachepsky and Shelton, 2011). *E. coli* decay rates have been observed to be lower in low salinity (Anderson et al., 2005; Pachepsky and Shelton, 2011) and lower temperature conditions (Pachepsky and Shelton, 2011) and in sediments with high contents of fine particles and organic matter (Desmarais et al., 2002; Craig et al., 2004; Pachepsky and Shelton, 2011). *E. coli* is generally found in the upper sediment layers (0 – 5 cm; Desmarais et al., 2002) and bacterial distribution in the sediments is usually patchy (Pachepsky and Shelton, 2011).

Because sediments favour bacteria accumulation and persistence, they may act both as sinks and sources of water faecal contamination. Bacteria populations in sediments may be mobilised into the water column when sediments are disturbed. Craig et al. (2004) observed a dramatic increase of faecal coliforms concentrations in both waters and sediments at a recreational coastal site in Australia following a significant rainfall event, from 17 ± 11 CFU.100 ml<sup>-1</sup> and 143  $\pm$  57 CFU.100 mg<sup>-1</sup> to more than 10<sup>6</sup> CFU.100 ml<sup>-1</sup> and 10<sup>6</sup> CFU.100 mg<sup>-1</sup>, respectively. The authors observed that two days after the peak, bacterial concentrations in waters had decreased to 2.2 x 10<sup>3</sup>.100 ml<sup>-1</sup>, whereas bacteria concentrations in sediments remained at 1.2 x 10<sup>5</sup>.100 g<sup>-1</sup>; they suggested that bacteria decay rates in sediments are lower than in the water column and thus microorganisms were able to accumulate in the sediments. Muirhead et al, (2004) reported a 2- to 3-fold increase in faecal indicator bacteria concentrations in waters after storm events when compared to baseflow levels, and it has been suggested that sediment agitation may be a more important mechanism of water faecal bacteria concentrations increase than overland runoff (Davies-Colley et al., 2008; Pachepsky and Shelton, 2011). Similarly, other events that cause sediment disturbance can lead to the resuspension of bacterial cells from sediment reservoirs, such as the disturbance caused by cattle crossing unbridged streams in farming areas (Davies-Colley et al., 2004).

Other faecal pathogenic organisms commonly transmitted through water consumption are Cryptosporidium and Giardia species, which are protozoa (Gray, 2008). Cryptosporidium species are parasitic organisms capable of infecting humans and a wide range of animals (Mendonca et al., 2007). One of the most common species affecting humans and cattle is C. parvum (Ryan et al., 2005). It causes a gastrointestinal illness in humans and neonatal livestock (Wells et al., 2015). Neonatal disease in cattle due to cryptosporidiosis can lead to significant economic losses (Mendonça et al., 2007). In waters, Cryptosporidium species exist as highly resistant cells known as oocysts (Lucy et al. 2008; Wells et al., 2015), which can remain viable for months (EPA, 2011). Studies have suggested that 1 - 10 oocysts are generally sufficient to cause infection (Gray, 2008). Because infected animals and humans typically excrete large quantities of oocysts (up to 10<sup>10</sup> cells; Gray, 2008), infection may spread rapidly in farming areas and into the environment (Wells et al., 2015). Giardia sp. is also found in the environment as highly resistant cysts and can infect cattle and humans, causing diarrhoeal disease (Mendonça et al., 2007; Lucy et al., 2008). Several outbreaks of cryptosporidiosis have been documented in the UK, with contamination origins traced to livestock grazing nearby water reservoirs and runoff from fields following slurry application (Gray, 2008). In Ireland, cryptosporidiosis became notifiable in 2004, meaning that any outbreaks of the disease must be notified to the government authorities by law; since then, a total of 3552 cases of infection were recorded (EPA, 2015).

## 2.4. How does cattle-based agriculture impact on freshwater systems?

By trampling the topsoil of the grazing fields as well as reducing the vegetation at a site, cattle can alter the hydrology and drainage pathways of these areas. This in turn can result in a decrease of the soil water infiltration capacity and a consequent increase in surface runoff (Line, 2003; Kurz et al., 2006). Kurz et al. (2006) observed that this alteration in infiltration capacity led to an increase in the concentration of particulate N and organic P in surface runoff in grazed pastures, and that this effect persisted during winter time, when cattle were absent from the fields. Other studies have shown that faecal matter deposited onto grazing fields can be transported to streams during run-off forming rainfall events, resulting in water contamination with organic matter, particulate forms of nutrients and faecal microorganisms (e.g. Line, 2003; James et al., 2007). Other potential diffuse source of stream water pollution related to cattle farming include the application of cattle slurry and manure on agricultural fields and subsequent wash-off into adjacent streams and rivers (i.e. incidental transfers) (Vidon et al., 2008; Bragina et al., 2017).

Point sources of water pollution related to cattle-based production, such as effluent discharges from farmyards, can also play an important role. For instance, Edwards et al. (2012) measured nutrients, faecal indicator organisms and suspended sediment concentrations loads from field drains as well as from a dairy farm effluent to a small catchment in NE Scotland, and concluded that the farmyard effluent delivered a large proportion of ammonium, phosphate and faecal indicator organisms to the waters.

Additional potential point sources of pollutants include direct livestock access to watercourses, which consist of sites used by the animals to drink, and those used as crossing points between adjacent fields. Some authors have suggested that cattle are characteristically attracted to water (Davies-Colley et al., 2004; Collins et al., 2007) and while they do not seem to spend a disproportional amount of time in the watercourse itself (Bond

el at., 2012; Hann et al., 2010), it has been observed that they tend to congregate in the riparian area more than elsewhere (James et al, 2007; Bond et al., 2012; Haan et al., 2010;: Kay et al., 2018). This has been attributed to the fact that riparian areas provide not only drinking water, but also shade and better quality forage to the animals (James et al., 2007).

# 2.5. Unrestricted cattle access to watercourses

Cattle access to watercourses has been demonstrated to have the potential to impact water quality in a variety of ways. Cattle grazing in close proximity to watercourses prevent riparian vegetation from growing (O'Callaghan et al., 2018). Riparian vegetation has been shown to have a buffering effect by protecting stream waters from receiving contaminants in run-off from the farmland, and can also stabilise streambanks, reducing bank erosion (Stutter et al., 2012). Frequent traffic of cattle into or across streams causes stream bank trampling and erosion, which leads to increased sediment inputs to the streams at and nearby access points (McKergow et al., 2001; Miller et al., 2011). Elevated loads of inorganic sediment in streams alter stream habitat quality due to infilling of interstitial spaces in streambed sediment, impacting macroinvertebrate communities (Conroy et al., 2016). Increased sediment loads may also lead to associated increases in particulate forms of nutrients, particularly phosphorus (Fox et al., 2016). Finally, excretions in or nearby the watercourses can directly add nutrients and potentially pathogenic faecal organisms to the aguatic system (Larsen et al., 1994). However, the body of literature investigating the extent to which unrestricted cattle access can be detrimental to water quality is limited, and studies assessing the effectiveness of cattle exclusion in preventing negative impacts are often conflicting (O'Callaghan et al., 2018) (Figure 2.1). The following sections discuss how unrestricted cattle access to watercourses can contribute to the deterioration of aquatic systems.



**Fig.2.1.** Potential impacts of unrestricted cattle access to streams on different physicochemical and biotic parameters, as reviewed by O'Callaghan et al. (2018). Arrows in gauges indicate consensus in the reviewed literature regarding the effectiveness of fencing on the mitigation of cattle access impacts (Y = consensus that there was mitigation; N = consensus that there was no mitigation; ? = the studies were inconclusive). Figure taken from O'Callaghan et al. (2018),

# 2.5.1. Impacts on stream hydromorphology and sedimentation

Where cattle have direct access to watercourses, their incisional and erosional potential, associated with overgrazing of protective riparian vegetation, can result in reduction of stream bank stability. This can lead to increased erosion and sedimentation at cattle access sites (Zaimes and Schultz, 2012) and alterations to stream channel morphology, with stream channels often becoming shallower and wider (Harrison and Harris, 2002). For instance, in a two year study in NE Oregon, USA, Kauffman et al. (1983) reported significantly greater streambank erosion in stream reaches with uncontrolled cattle grazing than in ungrazed reaches. Similarly, Trimble (1994) observed that unrestricted cattle access to a stream reach in Tennessee, USA, caused approximately six times more gross streambank erosion in

comparison to a control protected reach, which was attributed mainly to streambank breakdown by cattle and consequent reduction of the bank's geomorphic resistance.

Parallel to the enhanced streambank degradation and erosion, cattle in-stream movements while crossing or drinking also cause streambed sediment resuspension (Terry et al., 2014). Vidon et al. (2007) monitored water quality upstream and downstream of an area grazed by 25 cows with unrestricted access to the stream over a 12 month period, and reported a dramatic increase in water turbidity (13-fold) and total suspended solids (TSS) (11-fold) from the upstream to downstream points in the summer period, when cattle were often near or in the stream. In their study in in the Sherry River, New Zealand, Davies-Colley et al. (2004) estimated that two crossing events with the total duration of 19 minutes led a to a 54% increase in TSS water concentrations. Elevated inputs of sediment to waters caused by increased erosion of exposed banks and disturbance of the substrate during cattle in-stream activity can lead to habitat alteration as the fine sediment smoothers the substrate or clogs interstitial spaces and the hyporheic zone (Boulton et al., 2003). This can alter the composition of macroinvertebrate communities, and has been highlighted as the main ecological impact of excessive sediment pollution (Braccia and Voshell, 2006). Some studies have reported that cattle activity can cause bank failure, leading to streams becoming shallower and wider (Harrison and Harris, 2010). This in turn can lead to increased water temperatures (Herbst et al., 2012) and impact temperature-sensitive aquatic biota (Braccia and Voshell, 2006).



**Fig.2.2.** Left: streambank erosion and trampling at a cattle access site. Right: streambank collapse. Both sites are in the Brackan River catchment (Co Wexford).

A number of studies have reported significant improvements in stream morphology and stream bank condition reported improvements in streambank and channel conditions following the adoption of cattle exclusion measures, including increased bank stability and decreased erosion (e.g. Clary, 1999; Laubel et al., 2003; Scrimgeour and Kendall, 2003), decreased channel width (e.g. Magilligan and McDowell, 1997; Harrison and Harris, 2002;), increased stream depth (e.g. Ranganath et al., 2009) and higher riparian vegetation biomass (e.g. Scrimgeour and Kendall, 2003; Ranganath et al., 2009). Significant decreases in TSS concentrations following the adoption of cattle exclusion measures have also been observed (e.g. Line et al., 2000; Georgakakos et al., 2018). Other studies have reported no improvement in stream morphology after the implementation of cattle exclusion measures, and it has been suggested that such benefits might only become apparent after at least a decade following implementation (Kondolf, 1995).

# 2.5.2. Impacts on stream water and sediment nutrient levels

Cattle access points can act as localised sources of nutrients to watercourses through two main mechanisms: contribution of particulate phosphorus and nitrogen to waters as a result of streambank degradation and erosion (Fox et al., 2016) and excretion directly in waters or in the vicinity of the watercourse (James et al., 2007). Streambank soil P concentrations

ranging from 200 mg.kg<sup>-1-</sup> (Purvis et al., 2016) to 1400 mg.kg<sup>-1</sup> (Kronvang et al., 2012) have been reported in literature. Fox et al. (2016) postulated that streambanks in catchments which are affected by excessive nutrient inputs should be considered a significant source of TP to waters where streambank erosion and failure are systematic. Thus, streambank TP concentrations and streambank erosion rates are the primary variables controlling P loadings contributed to streams from banks (Fox et al., 2016), both of which can be exacerbated by cattle activity. Indeed, McDowell and Wilcock (2007) have stated that when livestock is allowed access to watercourses, damaged streambanks, rather than topsoil in agricultural fields, may become the primary source of PP in streamflow.

At cattle access sites, frequent defecation and urination directly into stream waters or in trampled and exposed banks can contribute nutrients and organic matter to the aquatic system. A number of studies have reported that, in addition to preferentially congregating in riparian areas (James et al., 2007; Bond et al., 2012) cattle tend to defecate more often when in proximity to water. For instance, in their study in the Sherry River, New Zealand, Davies-Colley et al. (2004), reported that the animals defecated ca. 50 times more per metre when crossing the river than elsewhere. Similarly, while studying the behaviour of four pastured herds of dairy cattle in the Cannonsville Watershed (New York, USA), James et al. (2007) observed that a significant number of animals (average across four herds of 21.4% of the herd) concentrated within 0 - 10 m of the stream at any given time, and tended to defecate more often while in this area.

Cattle faeces are the main pathway of excretion of non-utilised phosphorus (Ternouth, 1990). Faecal TP concentrations for cattle are determined by factors such as the type of diet, feed intake and animal reproductive status (Dou et al., 2002; James et al., 2007), with studies reporting concentrations from 4.9 mg TP.kg dry wt<sup>-1</sup> (Orr et al., 2012) to 12.65 mg TP.kg dry wt<sup>-1</sup> (Dou et al., 2002). Dou et al. (2002) reported that a substantial amount of the faecal TP was readily soluble inorganic P, and that this fraction increased with increasing P

intake, representing 30.3% to 49.6% of the total. On the other hand, nitrogen is mostly excreted in urine (Selbie et al., 2015). Again, total nitrogen concentrations in cattle urine vary with type of diet, feed and water intake, and even time of the day (Selbie et al., 2015). Studies have reported TN concentrations in cattle urine ranging from 3.0 g.L<sup>-1</sup> (Spek et al., 2012) to 20.5 g.L<sup>-1</sup> (Bristow et al., 1992), with the dominant form being urea-N, which in these studies represented 52.1% to 93.5% of TN. Ammonium has been reported to account for a small proportion of urine TN (average 0.9% to 2.9%; Bristow et al., 1992; Gonda and Lindberg, 1994). In a study assessing the defecation and urination behaviour of 24 beef cattle in the UK, Orr et al. (2012) reported average loads of P and N in faeces of 0.7 and 4.1 g.event<sup>-1</sup>, and of 0.003 and 4.5 g.event<sup>-1</sup> in urine, respectively.



Fig. 2.3. Cow urinating directly in waters at access point (Ireland).

In their study in the Cannonsville Watershed, James et al. (2007) estimated that approximately 10% of the phosphorus loading at the watershed level attributable to agriculture originated from direct deposits of faecal matter by cattle in stream waters. The authors also estimated that the adoption of cattle exclusion measures in approximately onethird of the dairy farms in the watershed had resulted in a reduction of 32% of the P loadings to waters (James et al., 2007). Likewise, Byers at al. (2005) reported significantly higher loads of dissolved reactive phosphorus (DRP) and TP in an unfenced stream where cattle spent 9% of the time in the riparian area, compared to a stream where cattle spent 5% of the time in the riparian area. Vidon et al. (2007) reported increases in average water concentrations of NH<sub>4</sub>-N (fourfold), TKN (fourfold) and TP (fivefold) resulting from unrestricted cattle access to the study stream. Similarly, Meals (2001) observed reductions in TP and TKN following the implementation of a number of cattle access mitigation measures, including fencing, in an experimental catchment in Vermont, US, whist observing increases in nutrient levels in a control catchment. Galeone (2000) also observed reductions in N and P exports after introduction of cattle exclusion measures in Pennsylvania, USA. Line et al. (2000) reported statistically significant reductions in TKN (78%), TP (76%) in a stream following the implementation of cattle exclusion measures. In a more recent study, the authors involving paired watersheds, Line et al. (2016) reported significant reductions in TKN (34%), NH<sub>3</sub>–N (54%), and total P (47%) in the treatment relative to a control watershed after the implementation of cattle exclusion measures, but no change in oxidised nitrogen (NO<sub>x</sub>-N) loads. Sheffield et al. (1997) reported that water column concentrations of total N and total P decreased by 54% and 81%, respectively, when cattle were offered an alternative water source, which in turn led to a reduction of the time cattle spent in the stream causing stream bank erosion to decrease by more than 70%. However, the authors noted that the concentrations of nitrate and orthophosphate were negatively impacted by the mitigation measure (Sheffield et al., 1997). More recently, Georgakakos et al. (2018) observed significant reductions in TP concentrations and loadings in a stream following the implementation of a cattle exclusion area, but not in SRP levels. In contrast, McKergow et al. (2003) did not find significant reductions in total P or total N loads, or concentrations, in relation to cattle exclusion in a catchment in Western Australia, whereas Davies-Colley et al. (2004) reported only a modest increase (10%) in TN concentrations in their study following two cattle crossing events of a stream.

The impacts of unrestricted cattle access on nutrient levels in the freshwater sediment compartment have received less attention. Palmer-Felgate et al. (2009) reported higher TP sediment concentrations (ranging on average from 1429 to 2480 mg.kg<sup>-1</sup>) in a UK stream located closer to farms where cattle had direct access at a number of points, when compared to a control site within the same catchment in an area grazed by only a small number of animals (ranging on average from 657 to 1060 mg.kg<sup>-1</sup>). The same authors also observed relatively high TP sediment concentrations (ranging on average from 155 to 636 mg.kg<sup>-1</sup>) at a control site with low agriculture intensity, which they hypothesised was caused by unrestricted cattle access to the stream (Palmer-Felgate et al., 2009). In contrast, in a study where ion-exchange membranes where used to assess nutrient dynamics in streambanks, no significant impacts of cattle grazing intensity and access to the stream were observed for either NO<sub>3</sub>-N or P (Miller et al., 2017).

## 2.5.3. Faecal contamination of stream waters and sediments

Defecation in or near the streams can add high levels of faecal microorganisms to waters. In their study in the Sherry River (New Zealand), Davies-Colley et al. (2004) estimated one herd to have deposited around 230 billion colony forming units (CFU) of the faecal coliform bacteria *E. coli* to waters in one single stream crossing, following 25 defecation events. Direct deposition of fresh faecal matter in waters is particularly important because animal faeces contain *E. coli* concentrations that may be as high as 10<sup>9</sup> cells g<sup>-1</sup> (Murphy et al., 2015), and, contrary to diffuse transport of faecal matter, where mechanisms of retention and inactivation operate, during direct defecation there are no opportunities of bacteria

immobilisation or die-off before reaching waters (Collins et al., 2007). Cattle are an important reservoir for many pathogens, including *E. coli* O157:H7 (Williams et al., 2008). Although infection with *E. coli* O157:H7 in animals is generally asymptomatic, infected animals excrete large quantities of bacteria, typically  $10^2 - 10^5$  CFU.g<sup>-1</sup>, but possibly as high as  $10^7$  CFU.g<sup>-1+</sup> (Williams et al., 2008). *E. coli* O157:H7 can survive for prolonged periods in the environment (Williams et al., 2008), and it has been observed to be more resistant to environmental stressors that non-pathogenic *E. coli* strains (Jenkins et al., 2012, 2015), while also having a very low infectious dose (<100 bacterial cells) (Brehony et al., 2018).

Several studies have investigated the impacts of cattle exclusion measures on freshwater faecal contamination. Line (2003) investigated the effects of excluding cattle from a 340 m long stretch of a small stream in North Carolina, USA, which was crossed at half-length by a farm road. In a long term study, the author sampled one point at the start of the fenced stretch (upstream) and at the end (downstream), both before and after fencing. The author reported a reduction in the average concentrations of *E. coli* in waters of 58.6% and 91.0% at the upstream and downstream sites following fencing, respectively, as well as a 95.9% reduction in the difference observed between average levels at the upstream and downstream sites prior to fencing. Smolders et al. (2015) compared faecal coliform levels in waters across a longitudinal gradient in two streams draining a grazing area in the Lake Baroon catchment, Australia, which were both crossed daily by cattle, following installation of a culvert bridge in one of the streams (i.e. treatment stream). The authors demonstrated that concentrations downstream of the crossing site in the unmodified stream (i.e. control stream) were significantly higher than those observed upstream, whereas in the treatment stream, differences between both sites were not significant after cattle exclusion through bridge improvement. However, concentrations in the bridged stream were higher after stream modification, which the authors attributed to a cumulative effect originating at the unmodified stream. McKergow et al. (2001) carried out a 10 year monitoring experiment to investigate the impacts of installing a fence and creating a riparian buffer on sediment and nutrient

loadings in an agricultural stream in Western Australia. The authors reported that while there was a noticeable decrease in sediment loadings in the stream, improvements in nutrient exports were limited, which was attributed to the specific geology of the studied site. More recently, Bragina et al. (2017) showed that streamside exclusion fencing effectively reduced stream sediment contamination with *E coli* in a catchment impacted by intensive cattle farming in NE Ireland. The authors reported that sediment average concentrations of *E. coli* in a fenced stream were in the order of  $10^2$  CFU.g dry wt<sup>-1</sup> whereas in an unfenced stream of the same catchment, they were significantly higher and up to  $10^5$  CFU.g dry wt<sup>-1</sup> (Bragina et al., 2017).

#### 2.6. Water quality protection in European and Irish policy

In 1962, the European Common Agricultural Policy (CAP) was launched with the aim of ensuring food supply in Europe (EC, 2015a). However, although the CAP had been very effective in achieving European Union's self-sufficiency, its strong productivist approach and the resulting rapid expansion of agriculture led to increased concerns on negative effects of agriculture on the environment. In the mid-1980, this increasing concern on environmental sustainability of agricultural practices led the Community and national governments to work towards implementing agri-environmental measures (Latacz-Lohmann and Hodge, 2003).

One of the first European Commission (EC) environmental laws with potential impacts on agricultural activity was the Drinking Water Directive of 1980 (80/778/EEC), which introduced limits for concentrations of nitrates and pesticides in waters intended for human consumption (Latacz-Lohmann and Hodge, 2003). In 1991, the Nitrates Directive (91/676/EEC; ND) was introduced, which was aimed at specifically protecting water bodies and drinking water quality from nitrate pollution derived from agricultural activity (Latacz-Lohmann and Hodge,

2003; EC, 2010a). The ND required Member States (MS) to identify all surface freshwaters and groundwaters with nitrate concentrations above 50 mg L<sup>-1</sup> or in risk of exceeding this threshold, as well as targeting all eutrophic water bodies (EC, 2015b). Furthermore, it imposed the identification of areas of land which drain into polluted waters or waters at risk of pollution, thus potentially contributing to nitrate contamination, as Nitrate Vulnerable Zones (NVZ) (EC, 2015b). The Directive required MS to implement compulsory programmes of measures in NVZ to tackle nitrate pollution, and adopt Codes of Good Agricultural Practice (GAP) to be applied by farmers on a voluntary basis elsewhere (EC, 2015b).

The ND became a complimentary measure to the Water Framework Directive (2000/60/EC; WFD) since the introduction of that over-arching directive in 2000. The WFD resulted from the need to address water quality in a more integrated manner, and established the ambitious goal of restoring and protecting the quality of all waters across Europe (EC, 2010b). The Directive introduced water management through an integrated river basin management approach, regardless of political or administrative boundaries (EC, 2010b), and defined a concrete objective: to achieve or maintain at least "good" and non-deteriorating chemical and ecological status in all waters by 2015 (EC, 2010b; Kallis and Butler, 2001).

Following the WFD establishment, the MS defined river basin districts, which would become the basic units of water management. In total, 110 river basin districts were delimited across the EU (EC, 2010b). In addition, water monitoring networks were established (EC, 2010b). In 2009, EU MS presented individual River Basin Management Programmes (RBMPs), which included programmes of measures to achieve the WFD goals, and were required to be fully operational by 2012. The first management cycle of the WFD ended in 2015, and MS presented their second RBMPs, based on the results and lessons learned from their predecessors. The second and third management cycles of the WFD will end in 2021 and 2027, respectively, with the latter being now the revising deadline to the achievement of the WFD goal.

In addition to the ND, the WFD is supplemented by other daughter directives regarding water quality protection, of which the Drinking Water Directive (Directive 98/83/EC) (DWD hereafter), and the Bathing Water Directive (Directive 2006/7/EC) (BWB hereafter) are perhaps more relevant in the context of agricultural pollution. The DWD regulates the quality intended for human consumption and applies to all distribution systems serving more than 50 people or supplying more than 10 m<sup>3</sup>.day<sup>-1</sup>, and also distribution systems serving less than 50 people/supplying less than 10 m<sup>3</sup>.day<sup>-1</sup> if the water is supplied as part of an economic activity drinking water from tankers, and water used in the food-processing industry (EC, 2021a). It requires a total of 48 microbiological, chemical and indicator parameters to be monitored (EC, 2021a), including *E. coli* and intestinal enterococci (0 CFU.100 ml<sup>-1</sup>). The directive's latest amendments were adopted in 2020 (Directive (EU) 2020/2184).

The BWD was first introduced in 1976 (Directive 76/160/EEC) and later reviewed and replaced by the current New Bathing Water Directive (2006/7/EC) (EC, 2021b). It is intended to protect human health by preserving and protecting the quality of coastal and inland surface water bodies that are commonly used as bathing areas (EC, 2021b). Under the BWD, the MS are required to monitor bathing areas every year during the bathing season (typically from May to September), mainly for microbial parameters (intestinal enterococci and E. coli) although other parameters such as cyanobacteria and microalgae can also be monitored (EC, 2021b). The BWD introduces four categories of bathing waters in relation to numerical standards of bacteriological quality; these values vary for inland and coastal or transitional waters, ranging, for E. coli, from 250 - 500 CFU.100 ml<sup>-1</sup> ("Excellent"), 500 -1000 CFU.100 ml<sup>-1</sup> ("Good") (values based upon a 95-percentile evaluation) or 500 – 900 CFU.100 ml<sup>-1</sup> (values based upon a 90-percentile evaluation) ("Sufficient") (Directive 2006/7/EC). The category of "Poor" refers to failure to comply which such bacteriological standards, in which case the MS must take measures such as banning or advising against bathing, declassifying the area as bathing area (when it has failed to meet mandatory standards for five consecutive years), and taking corrective measures (EC, 2021b).

#### 2.6.1. <u>Agri-environment schemes</u>

The first agri-environmental measures in Europe were initially statutory regulations which focused on controlling pollution from agriculture, mainly by regulating nitrate contamination, intensive livestock farming, the use of pesticides and slurry application on land (Latacz-Lohmann and Hodge, 2003). These regulations were non-compulsory and were defined and applied within individual MS, with little articulation across the European Community. However, government attempts to extend these measures beyond pollution control, aiming at coping with habitat deterioration and habitat loss, were largely contested by farmers, who saw such measures as undue interferences in their property rights and demanded monetary compensations for the loss of opportunities and profits resulting from their implementation (Latacz-Lohmann and Hodge, 2003). In 1981, Britain established the Wildlife and Countryside Act, which worked on the basis of compensating farmers for not undertaking potentially damaging operations in protected land. The scheme was opposed to due to the financial overburden imposed to conservation associations that were responsible for farmers' compensation, eventually leading to the novel idea of offering payments to farmers in protected areas for the provision of environmental goods, rather than for not taking damaging actions. In 1984, the British government launched the Broads Grazing Marshes Scheme in order to protect the threatened area of the Halvergate Marshes (Norfolk Broads, East England), which offered an annual payment to farmers in return for a commitment to farm at low intensities in the area. It marked a shift from a negative compensation policy to a more positive approach while highlighting the wider role of agriculture as a means for both food and fibre production and environmental conservation. The Broads Grazing Marshes Scheme led to the definition of EU agri-environmental schemes as flat-rate payments offered to farmers in return for voluntarily adopt environmentally friendly farming practices. Such schemes were established in all EU MS with the MacSharry reform of the CAP in 1992, and have become a keystone in agricultural and environmental management (Latacz-Lohmann and Hodge, 2003). With the later Agenda 2000 and 2003 reforms, AES become part of the

second pillar of the CAP, aimed at improving economic and social conditions of rural areas. Currently, AES are implemented as part of Rural Development Plans (RDP) designed individually by each MS, and are the only mandatory rural development measure for MS.

## 2.6.1.1. Agri-environment policy in Ireland

In Ireland, the Nitrate Action Programme required by the ND is given legal effect through the European Union (Good Agricultural Practice for Protection of Waters) (GAP) regulations, of which the most recent are GAP Regulations S.I. No. 65/2018. These regulations include amendments to the previous S.I. No. 605/2017. The key components of the Irish NAP include limits on stocking rates, legal limits for nitrogen and phosphorus application rates, prohibition of application of organic and chemical fertilisers at more environmentally vulnerable times (e.g. before periods of heavy rain), minimum livestock manure storage requirements, and set-back distances from waters (Government of Ireland, 2021). Ireland is currently operating the 4<sup>th</sup> NAP which will run until the end of 2021.

Ireland has also had agri-environmental schemes in place since 1994, when the first Rural Environmental Protection Scheme (REPS) was implemented (Whelan and Fry, 2010). REPS was a voluntary five year programme which was applied horizontally i.e. at a national level (Whelan and Fry, 2010), rather than in specific areas of the country. Its stated goals were to establish farming practices and production methods which reflected a concern for conservation and landscape protection; to protect wildlife habitats and endangered species; and to produce quality food in an environmentally friendly manner (Lenihan and Brasier, 2009). In order to subscribe to REPS, farmers had to implement, among other measures, an individual agri-environmental plan produced by qualified planners which included a Nutrient Management Plan based on the farm conditions (Emerson and Gillmor, 1999; Lenihan and Brasier, 2009). REPS had four incarnations (I-IV), and was superseded by the Agri-Environment Options Scheme (AEOS) in 2010. Both schemes represented a top-down, horizontal approach; however, while REPS involved applying measures to the whole farm

with payments made on a per hectare basis, AEOS required farmers to select specific actions for delimited areas in the farm, and, apart from cross-compliance, there was no other requirement regarding the remaining area.

The AEOS was followed by the on-going Green Low Carbon Agri-Environment Scheme (GLAS), launched in 2015. GLAS is a large-scale top-down plan that differs from its predecessors by attempting at a targeted approach towards individual farms in areas of environmental concern (farms with "Priority and Environmental Assets and Actions") such as Natura 2000 sites and areas with vulnerable catchments or high status waterbodies (Cullen et al., 2018). Famers in these areas had priority access to the scheme. This targeted approach has led, in comparison with the previous AES, to a greater superimposition of regions with high farmer participation rates and regions with high concentrations of environmental public goods, thus contributing to the scheme's financial effectiveness (Cullen et al., 2018).

#### 2.6.2. <u>Recent policy developments</u>

In June 2018, the EU presented legislative proposals for the post-2020 CAP (2021-2027), aimed at delivering a higher level of climate and environmental ambition while placing greater emphasis in the achievement of results at the regional and national scale, thus moving away from a focus on compliance (EC, 2019a). The new policy comprises nine specific goals to be achieved by each MS, three of which are directly related to protection and environment and climate (EC, 2019a). Each MS is responsible to delineate a CAP Strategic Plan, which will include specific targets and objectives for its territory and present actions to achieve them (EC, 2019a).

The CAP 2021-2027 is interlinked with other EU policies for the protection of the environment and the agri-food sector such as the WFD and the recent Farm to Fork Strategy and Biodiversity Strategy 2030. These are part of the European Green Deal, an action plan

presented in 2019 with the over-arching goal of making Europe climate-neutral by 2050 (EC, 2019b). The Farm to Fork Strategy specifically requires the development of integrated nutrient management action plans in MS CAP Strategic Plans to tackle "nutrient pollution at source and increase the sustainability of the livestock sector" and the application of precise fertilisation techniques and sustainable agricultural practices particularly in areas of intensive livestock farming (EC, 2020b).

## 2.6.3. <u>Measures to restrict cattle access to watercourses for water quality protection</u>

In terms of measures to protect water quality, both REPS and AEOS included, among others, fencing of watercourses to restrict cattle access. The current GLAS, and indeed the Irish RDP 2014-2020 (of which GLAS is the largest scheme) have a strategic focus on water quality objectives (DoHPLG, 2018a). In GLAS, cattle are only allowed access to streams as a means of crossing between parcels of land when no other option is available. Furthermore, in response to the continuous general decline in water quality in Ireland, the 4<sup>th</sup> Nitrates Action Plan of the Nitrates Directive, launched in 2018, requires farms with an allowance (derogation: hereafter referred to as derogation farms) to farm at a grassland stocking rate over 170 kg N/ha, to prevent cattle from accessing watercourses from January 2021.

# 2.7. Relevance of the current study

Despite the inclusion of cattle exclusion measures in all Irish AES to date, little is known regarding cattle impacts and exclusion benefits in streams in the Irish or even European context. Most studies conducted on this topic have been carried out in areas of the globe where conditions (e.g. climate conditions, geological features) are much different from the Irish conditions. Such studies suggest that cattle exclusion measures can have beneficial effects on water quality in agricultural fields. However, they have also highlighted that the

extent to which such measures can have positive impacts is dependent on both site specific characteristics and farmland management. The current work has therefore focused on quantifying the potential impacts of such practice on the described parameters – freshwater nutrient levels, excess suspended solids, and faecal bacteria - specifically in the Irish context. The study aims at both contributing to the literature describing livestock agriculture pressures on freshwater systems and to substantiate Irish agri-environment policy.

# 3. Site Selection and Description

# 3.1. Site selection criteria

The selection of the experimental sites studied for this thesis was carried out as part of the COSAINT Project – *Cattle access to watercourses: environmental and socio-economic implications*, funded by the Irish Environmental Protection Agency (EPA) under the Research Program 2014 – 2020, a project which also supported the work presented in this thesis. Specifically, the experimental site selection and characterisation was conducted under Work Package (WP) 2 the of the COSAINT project. The work presented in this thesis was developed under WP3, aimed at analysing geochemical and microbial parameters relative to cattle access and exclusion from watercourses. The project also included the publication of a literature review on the impacts of cattle access to watercourses on freshwater systems and the efficacy of fencing as a water quality protection measure (WP1); the analysis of the effects of cattle access at a national level (WP5) and an analysis of the cost-effectiveness and socio-economic implications of cattle exclusion measures (WP6).

In Ireland, biological water quality is assessed using the Quality Rating System, developed by the EPA, which relates the abundance of five key groups of macroinvertebrates to water quality. The system uses a five point scale (Q-values, or Q) of water quality rating, with intermediate scores obtainable, with Q1 representing bad water quality and Q5 representing high water quality (McGarrigle et al., 2002) (see Table 3.1). In the current study, five catchments were selected for sampling. In order to reflect the predominantly grass-based agricultural conditions found in the Irish countryside, the majority of the selected catchments (three) were of poor/moderate water quality and represented higher agriculture intensity (see

Table 3.2). Two of the five selected catchments were of high Q-value status and had more extensive agriculture (Table 3.2). Selected sites included a mixture of catchments with waterways fed by ground and surface sources to encompass the different low-flow conditions, sediment dynamics and nutrient pathways represented by each. For all catchments, all sites were on first and second order streams, as these are more commonly used as drinking water sources for cattle, are more vulnerable to the related pressures due to lower dilution rates (Lassaletta et al., 2010), influence downstream water quality (Freeman et al., 2007), are less affected by confounding factors than higher order streams/rivers which may be subject to further unrelated pressures (e.g. wastewater treatment works) and are more easily and safely sampled due to their lower average depths and velocities.



Fig. 3.1. Site selection criteria for the COSAINT project.

Q-value	Water quality status	WFD status			
Q5, Q4-5	Unpolluted	High			
Q4	Unpolluted	Good			
Q3-4	Slightly polluted	Moderate			
Q3, Q2-3	Moderately polluted	Poor			
Q2, Q1-2, Q1	Seriously polluted	Bad			

**Table 3.1.** Quality Rating System (Q-value) classification and relationship to the Water Framework Directive (WFD) water quality classification

Information taken from http://www.epa.ie/QValue/webusers/.

## 3.2. Study catchments and experimental design

The five catchments selected for sampling under the COSAINT Project were the Muster Blackwater catchment (BW; Co. Cork); the Douglas River catchment (DG; Co. Laois); the Brackan River catchment (BK; Co. Wexford); the Commons River catchment (CM; Co. Louth); and the Milltown Lake catchment (MT; Co. Monaghan) (see Fig 3.2). Table 3.2 summarises the characteristics of these catchments including ecological status and dominant hydrology pathways (surface flow or subsurface flow, i.e. water flow beneath soil surface (Hu and Li, 2018).

A total of 15 cattle access sites were selected for sampling. Three cattle access sites were selected in the headwaters of the BW and DG catchments, with each cattle access site located on a separate tributary, and labelled A, B and C. These sites represented the uppermost cattle access sites on each of those tributaries. In the BK, CM and MT catchments, three cattle access points were selected that were located longitudinally along a tributary of each catchment, and were labelled 1, 2 and 3. This approach allowed an investigation of the overall effects of cattle access across a range of catchments and aimed at giving insight into the following (see Fig. 3.3):

- a. The impacts, if any, of cattle access to watercourses on the selected bed sediment stream microbial (i.e. *E. coli*; Chapter 4) and physicochemical (Chapter 5) parameters at a single access site, i.e. when there are no significant access pressures upstream. To investigate this, nine sites were selected in the headwater zones of the catchments, which had no cattle access upstream (Fig. 3.2). Six of these sites were in the BW and DG catchments with one site on each (the uppermost site) in the BK, CM and MT catchments (see Fig. 3.3);
- b. The potential cumulative downstream effects of cattle access sites. Here, the upper sites in BK, CM and MT were used along with two additional sites per stream along a downstream gradient (Fig. 3.2 and 3.3) to investigate potential downstream cumulative effects of cattle access to watercourses on bed sediment freshwater and geochemical parameters.

In Ireland, cattle graze outdoors for the summer months, typically from April to November. For the work presented in this thesis, the stream bed sediments were sampled at each site in April/May and September/October 2016 for sediment characterisation and analysis of sediment nutrient concentrations, June and November/December 2016, for quantification of *E. coli* bacteria sediment concentrations The times of sampling are referred to in this thesis as early grazing season (EG) and late grazing season (LG) in the study investigating the impacts of cattle access on sediment nutrient concentrations (Chapter 5) and as mid-grazing season (MG) and post-grazing season (PG) in the study assessing sediment *E. coli* concentrations (Chapter 4).

Catchment	Munster Blackwater (BW)	Douglas River (DG)	Brackan River (BK)	Commons River (CM)	Milltown Lake (MT)
County	Cork	Laois	Wexford	Louth	Monaghan
Hydrology	Surface	Surface	Sub-surface/Surface	Surface	Surface
Stream Order	1	2	2	2/3	1/2
Predominant soil type <sup>1</sup>	Blanket peat, mineral alluvium	Limestone, sandstone and shale	Poorly drained gleys and some well drained brown earths	Typical and stagnic luvisols, brown earths	Poorly drained mineral soils and peat
Ecological Status (based on Q value)	Good	High/Good	Moderate	Poor	Moderate
Water quality trend (based on Q values)	No change	No change	Deteriorating	No change	No change/Deteriorating
At risk of not meeting WFD objectives?	No	No	Yes	Yes	Yes
Is agriculture a significant pressure?	No	No	Yes	Yes	Yes

Table 3.2. Summary of the characteristics of the study catchments.

Information derived from WFD assessment reports for the study catchments made available by the EPA at https://www.catchments.ie/data/#/?\_k=43w5ni. <sup>1</sup>Information on predominant soil types retrieved from ACP programme for BK and CM catchments; from the National Source Protection Pilot Project (NSPPP) report for the MT catchment; and from Teagasc and GSI maps for the BW and DG catchments.



Fig.3.2. Map of the study catchments and selected sites.
Following the first sampling campaign (April – June 2016), sites BK1 and BK3 in the Brackan River catchment became acutely impacted with severe collapse of streambanks such that these sites were no longer suitable for inclusion in the COSAINT project. Therefore, site BK1 and BK2 were replaced by two new sites, while site BK2 in early grazing season became the lower-most site BK3 in late grazing season (Fig.3.2). Given the similarity between the new sites and the replaced sites, it was considered that this substitution did not affect the purposes of this work.



**Fig.3.3.** Location of the sampled sites in the catchments. Red circles highlight the headwater sites (9 in total), and yellow circles highlight the sites used to assess potential downstream cumulative impacts of cattle access to the streams (9 in total).

Fig 3.4. shows a scheme of the sampling locations at each cattle access site. For the sediment characterisation, samples were collected at the cattle access site (CAS), upstream of the access site at a reach to which cattle had no access, either due to fencing or natural physical barriers (US), downstream (DS) and at the interface (edge) of the stream water level, at the access path used by cattle to enter the stream (INT) (Fig. 3.4). Sampling of this interface area was included (for sediment nutrients only) because it was hypothesis that this area would be subject to a high localised impact of cattle. For assessing sediment faecal contamination with *E. coli*, only US and the cattle access site. At one of the sites, CM3, a high temporal resolution monitoring experiment was conducted between 2016 and 2018 to assess the impacts of direct cattle access to watercourses on water physicochemical and microbial parameters (Chapter 6). Each study catchment and study sites within catchments are described in more detail in the following sections.



**Fig.3.4.** Sampling scheme showing the different locations of sample collection at each cattle access site.

# 3.3. Site description

Catchment land use across sites was principally grassland, although some arable land and plantation forestry were present in the Brackan (Co. Wexford) and Blackwater (Co. Cork) catchments, respectively. Information on the study catchments' ecological, biological and chemical status and stream order were derived from the nearest EPA water quality monitoring stations. Information on flow discharge was retrieved from the nearest flow monitoring station available at https://www.epa.ie/hydronet/#Water%20Levels. Total annual rainfall for each catchment in the year of sampling was calculated using monthly data from the nearest Met Eireann weather station available at the Met Eireann website (https://www.met.ie/climate/available-data). This information is presented in Table 3.3, Table 3.5, Table 3.7, Table 3.9 and Table 3.11.

Stream substrates at all study sites were characterised by visually assessing the mean substrate composition of substrates within each geomorphic unit and standardising these means to the reach scale based on the proportional representation of each geomorphic unit at said reach scale. Stream widths ranged from 0.75 m to 3.5 m. Stream reach gradients at individual sites were calculated using Google Earth tools over a stretch of 600 – 700 m of each stream. These were variable, but there was a tendency towards steeper channels at more upstream sites (particularly in the CM and BW catchments). Data on the type of geological formations and soil type were collected online from the Geological Survey Ireland website (https://www.gsi.ie/en-ie/data-and-maps/Pages/default.aspx) and Teagasc Soil Map (http://gis.teagasc.ie/soils/map.php), as well as from the Agricultural Catchments Programme (ACP) website (https://www.teagasc.ie/environment/water-quality/agricultural-catchments/) and from the National Source Protection Pilot Project (NSPPP) conducted in the MT catchment.

An attempt to obtain animal numbers and stocking rates for the study sites, was well on duration of grazing season, slurry spreading information and other agricultural management, was made through a survey conducted with the farmers within the COSAINT project (results of the survey are presented in Appendix B1). However, it was not possible to gather clear information for all sites. Hence, for the studies presented in Chapters 4 and 5, total cattle numbers in each catchment were estimated by dividing total cattle numbers per district electoral division (DED), available at the Irish Central Statistics Office website (https://www.cso.ie/en/databases/; figures from the last Agricultural Census, conducted in 2010) by the total pastoral area of each DED. This assumed equal distribution of the cattle within each DED. This is referred to in this study as estimated cattle density (ECD; number of cattle.ha<sup>-1</sup>). This information is presented in Table 3.4, Table 3.6, Table 3.8, Table 3.10 and Table 3.12.

Cattle access points at the majority of sites were constricted to narrow points of access with fencing or other natural barriers limiting the ability of the cattle to roam within the stream or on stream banks. At some sites however, such as at DGA and CM2, cattle had access to longer stretches of channel and stream bank. All sites were characterised by a Riparian Habitat Index (RHI), developed by O'Sullivan et al. (2019). This index is a metric of the condition of the riparian habitat and can reflect the intensity of cattle access in this study. It was developed by calculating several subindices of qualitative habitat assessment published in literature and adapting the results to reflect the Irish landscape (see O'Sullivan et al., 2019 for details). The maximum achievable RHI score is 80. Here, the RHI of the streams at the cattle access sites and at areas upstream of these sites, to which cattle had no access, are provided to enable a comparison of general habitat quality between the sites.

#### 3.3.1. Munster Blackwater catchment

The selected sites in the Munster Blackwater were part of the River Allow sub-catchment. The total population of the Munster Blackwater catchment is approximately 109 030 with a population density of 33 people per km<sup>2</sup> (EPA, 2020). The River Allow sub-catchment is 310 km<sup>2</sup> and 70% of the soils are poorly drained mineral soils. The dominant hydrology pathway is surface runoff. Blanket peat covers 5% with mineral alluvium being associated with the river channels. The main agricultural land use is grassland for beef and dairy.

River Blackwater and its tributaries, in particular the River Allow, are classed as Special Areas of Conservation (SAC) due to presence of many species and habitats of European importance. These include a number of EU Habitats Directive Annex II listed species, such as the freshwater pearl mussel (*Margaritifera margaritifera*), the salmon (*Salmo salar*) and otter (*Lutra lutra*), and the EU Birds Directive Annex I listed species, the kingfisher (*Alcedo atthis*).

The nearest EPA stations to the study sites were stations RS18B020075 and RS18O090400, which reported good water quality status for the sampling period (Table 3.3). However, pre-sampling (invertebrate kick sampling) of the sites under the COSAINT project indicated that these study sites were in high-status streams.

Catchment	Munster Blackwater	
River	Munster Blackwater (Duhallow Region)	
County	Cork	
Mean annual flow (m <sup>3</sup> .s <sup>-1</sup> ) <sup>1</sup>	0.059	
Total annual precipitation in 2016 (mm) <sup>2</sup>	1540.9	
Hydrology	Surface	
Ecological status (Sampling period)	Good	
Biological status (Sampling period)	Good	
Chemistry conditions (Sampling period)	Pass	
Nutrient Condition	Pass	
Ortho-P status quality (trend)	High (downwards)	
WFD Risk	Not At Risk	
Waterbody trend	No change	
Significant Pressures	None	

**Table 3.3.** Description of the Munster Blackwater (BW) catchment. Water quality statusinformation was derived from Edenireland.ie; information at time of sampling (2016/2017).

<sup>1</sup>Flow data retrieved from EPA Islandbrack station. <sup>2</sup>Precipitation data retrieved from the Ballydesmond Met Éireann station.

**Table 3.4.** Summary characteristics of the study sites (BWA, BWB, and BWC) in theMunster Blackwater (BW) catchment.

		BWA	BWB	BWC	
Stream order		1	1	1	
Stream	width (m)	1.95	1.57	1.42	
Reach gr	adient (%)	-2.70	-4.60	-3.00	
Soil	type	Blanket peat soils with interspersed poorly drained mineral soils			
Geologic	formations	Car	boniferous limestone a	nd shale	
	Boulder	4	6	10	
Substrate	Cobble	78	60	81	
(%)	Gravel	16	32	8	
	Sand+Silt	2	2	1	
Site description		Open access site, nearby a bridge; ditch draining opposite the site; vegetated banks	Open access site, other cattle access points downstream; vegetated banks	Open access site; few ditches draining into the stream; vegetated banks	
RHI s (upstream;	scores access site)	66; 48	50; 36	48; 21	
Estimated cattle density (animals.ha <sup>-1</sup> )		1.47	0.92	1.32	







**Fig.3.5.** Cattle access sites in the Munster Blackwater (BW) catchment. From the top to the bottom: BWA, BWB, BWC.

#### 3.3.2. Douglas River catchment

The Douglas river catchment is located in the east of Ireland (55°12'N, 6°42'W) (Fig. 3.2). The catchment is a sub-catchment of the Barrow River catchment, which has a total population of approximately 188 117 and population density of 62 people per km<sup>2</sup> (EPA, 2020).

The DGA site is on the Fuer River and is upstream of the EPA monitoring station 14DO30100, which reported a high water quality status (Table 3.5). Sites DGB and DGC are on the Douglas River and are upstream of EPA monitoring station 14DO30040, which reported a good water quality status (Table 3.5). However, pre-sampling (invertebrate kick sampling) of the sites under the COSAINT project indicated that all sites were in high-status streams (O'hUallachain et al., 2020).

Catchment	Barrow
River	Douglas
County	Laois
Flow (m <sup>3</sup> .s <sup>-1</sup> ) <sup>1</sup>	0.015
Total annual precipitation in 2016 (mm) <sup>2</sup>	669.9
Hydrology	Surface
Ecological status (Sampling period)	Good (DGB, DGC); High (DGA)
Biological status (Sampling period)	High
Chemistry conditions (Sampling period)	Pass (DGB, DGC): NA (DGA)
Nutrient Condition	Pass (DGB, DGC): NA (DGA)
Ortho-P status Quality (trend)	Moderate (Downwards) (DGB, DGC; NA (DGA)
WFD Risk	Not At Risk
Waterbody trend	No change
Significant Pressures	None

**Table 3.5.** Description of the Douglas River (DG) catchment. Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

<sup>1</sup>Flow data retrieved from the EPA Clonagh station.<sup>2</sup> Precipitation data etrieved from the Athy (Levitstown) Met Eireann station. NA – Not Assigned

		DGA	DGB	DGC	
Stream	ream order 2 2 2		2		
Stream v	width (m)	1.93	2.00 2.00		
Reach gra	adient (%)	-2.10	-2.60	-3.40	
	Boulder	0	0	0	
Substrate	Cobble	79	67	81	
(%)	Gravel	15	28	18	
	Sand + Silt	6	5	1	
Soil	type	Well drained mineral soils Poorly drained mineral soils		ed mineral soils	
Geologic f	formations	Carbonifer	ferous limestone, sandstone and shale		
Site des	scription	Access site is a crossing point between fields; vegetated banks	Open access site; multiple access points in the field		
RHI scores acces	(upstream; s site)	56; 28	62; 50 80; 69		
Estimated cattle density (animals.ha <sup>-1</sup> )		1.81	1.81	1.81	

**Table 3.6.** Summary characteristics of the study sites in the Douglas River (DG) catchment.



Fig.3.6.Cattle access sites in theDouglasRiver (DG) catchment.the top to the bottom:DGA, DGB.

#### 3.3.3. Brackan River catchment

The Brackan River sub-catchment (11.0 km<sup>2</sup>) is located in the south-east of Ireland (co. Wexford) (52°36'N, 6°20'W) (Figure 3.2). It is part of the Owenavorragh catchment which has a total population of approximately 27 319 and population density of 69 people per km<sup>2</sup> (EPA, 2020). The catchment is part of the Agricultural Catchments Program (Teagasc, 2017a), which operates a hydrometric station nearby the catchment outlet that collects high temporal resolution data on a variety of water physicochemical parameters.

The soil type is predominantly poorly-drained groundwater gleys in the catchment lowlands with a clay loam texture in A- and B-horizons resulting from a clayey calcareous Irish Sea till subsoil. The uplands contain smaller areas of well-drained brown earths; these soils are underlain by drift deposits with siliceous stones. The underlying geology is permeable, dominated by Ordovician volcanics and metasediments of the Campile formation (Tietzsch-Tyler et al., 1994), which form a productive aquifer with faults (Mellander et al., 2012). Artificial drainage is a key feature including open drains, defined here as ditches, and closed, sub-surface piped drains (predominantly 80 mm diameter). This catchment is considered to be dominated by overland flow pathways (Mellander et al., 2012; Shore et al., 2013) except for areas of well-drained soils featuring sub-surface transport pathways. Land is predominantly grass-based for dairy and beef cattle grazing, and also sheep enterprises (Shore et al., 2013). Arable crops such as spring barley are common on the well-drained soils which are unmanaged between harvest and ploughing for following crop. The stream where the study sites were located is a tributary of the Owenavorragh river, with the nearest downstream EPA monitoring station being 110010400, reporting a likely water quality status of moderate (Table 3.7)

Catchment	Owenavorragh
River	Brackan River
County	Wexford
Flow (m <sup>3</sup> .s <sup>-1</sup> ) <sup>1</sup>	0.200
Total annual precipitation in 2016 (mm) <sup>2</sup>	985.7
Hydrology	Sub-surface/ Surface
Ecological status (Sampling period)	Moderate
Biological status(Sampling period)	Good
Chemistry conditions (Sampling period)	Moderate
Nutrient Condition	Fail
Ortho-P status quality (trend)	Moderate (downwards)
WFD Risk	At Risk
Waterbody trend	Downward
Significant Pressure	Agriculture

**Table 3.7.** Description of the Brackan River (BK) catchment. Information derivedpredominantly from Edenireland.ie at time of sampling (2016/2017).

<sup>1</sup>Flow data retrieved from Office of Public Works (OPW) station Boleany. <sup>2</sup>Precipitation data retrieved from Met Eireann station Monamolin.

		BK1A	BK1B	BK2A/BK3B	BK2B	BK3A
Strea	am order	2	2	2	2	2
Stream width (m)         0.67         0.93         2.40         2.00			2.34			
Reach g	gradient (%)	-3.83	-2.60	-1.40	-1.50	-0.51
	Boulder	0	0	2	2	0
Substrate	Cobble	26	23	72	55	5
(%)	Gravel	56	60	10	23	80
	Sand + Silt	19	17	16	10	14
Sc	oil type		Mineral soils, mostly	well drained with poorly	/ drained soils intersper	sed
Geologi	c formations	Ordovician, rhyolitic grey and brown shales				
Site descriptionVery shallow and narrow stream; vegetated banks; sheep graze in surrounding fieldsNarrow stream; open access site; ditch draining upstream of the access siteOpen access site with vegetation impeding access upstream; vegetated banks and ditches draining into the stream		Discrete site, next to bridge; vegetated banks	Open access crossing site; vegetated banks, sheep graze in surrounding fields			
RHI score	es (upstream; ess site)	NA	48; 43	28; 41	68; 54	NA
Estimated (anim	cattle density nals.ha <sup>-1</sup> )	1.68	1.72	1.98	1.72	1.98

Table 3.8. Summary characteristics of the study sites in the Brackan River (BK) catchment.

A – site sampled during Spring/Summer 2016; B – Sites sampled during Autumn/Winter 2016, following site replacement. NA – Not assigned





**Fig.3.7.** Cattle access sites in the Brackan River (BK) catchment. From the bottom, clockwise: BK3A, BK2A/BK3B, BK1A, BK1B BK2B.

#### 3.3.4. Commons River catchment

The Commons River catchment (9.5 km<sup>2</sup>) is located in north-east Ireland (53°49'N, 6°27'W), and, similarly to the Brackan River catchment, it is monitored under the Agricultural Catchments Program (Teagasc, 2017a). The Commons River catchment is part of the Newry, Fane, Glyde and Dee catchment which is a cross border catchment with a total population in the Republic of Ireland of 115 900 people, with a population density of 83 people per km<sup>2</sup> (EPA, 2020). It has a complex pattern of poor- to moderately-drained soils (Melland et al., 2012) with a loam soil texture dominating the A-horizon whereas clay loams are dominant in the B-horizon. The subsoil is dominated by fine till containing siliceous stones with fluvioglacial sediments located near-channel. Soils are underlain by calcareous greywacke and banded mudstone geology and produce a poorly productive aquifer (Mellander et al., 2012). Hydrologically, surface pathways dominate; however, below-ground pathways may also be important especially during winter (Melland et al., 2012; Mellander et al., 2012). Artificial drainage is dominant, particularly in the poorly-drained catchment areas. Arable land is dominated by winter-sown cereals, but also comprises maize and potatoes. These areas are unmanaged between cropping cycles; however, crop rotation is common. Additional areas of permanent grassland are utilised for dairy cattle, beef cattle, and sheep grazing. The river is a tributary of the White River, with the nearest downstream EPA monitoring station being 06W010400, reporting a likely poor water quality status (Table 3.9).

Catchment	Newry, Fane, Glyde and Dee
River	Commons River
County	Louth
Flow (m <sup>3</sup> /s) <sup>1</sup>	0.019
Total annual precipitation in 2016 (mm) <sup>2</sup>	715.9
Hydrology	Surface
Ecological status (Sampling period)	Poor
Biological status (Sampling period)	Poor
Chemistry conditions (Sampling period)	Pass/Good
Nutrient Condition	Pass/Fail
Ortho-P status quality (trend)	Moderate/Poor
WFD Risk	At Risk
Waterbody trend	No change/ Deteriorating
Significant Pressure	Channelisation; Agriculture; Urban pressures

**Table 3.9.** Description of the Commons River (CM) catchment. Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

<sup>1</sup>Flow data derived from the EPA station Coneyburrow Br.. <sup>2</sup>Precipitation data retrieved from the Met Eireann stationTogher (Barmeath Castle).

		CM1	CM2	CM3
Stream order		2	3	3
Stream v	vidth (m)	1.52	1.96	2.90
Reach gra	adient (%)	-5.90	-2.60	-3.20
	Boulder	0	0	0
Substrate	Cobble	77	57	79
(%)	Gravel	13	22	14
	Sand+Silt	10	21	7
Soil	typo			Deep well drained
Son type			mineral soils	
Geologic formations		Si	lurian greywacke and m	nudstone
Site description		Fenced upstream	Open access site;	
		of the site;	crossing point	Discrete access site:
		vegetated banks;	between two adjacent	vogotated banks: only
		other cattle	fields; one access	access site in the field
		access points	point upstream;	
		downstream	vegetated banks	
RHI scores (upstream; cattle access)		80:35	46:28	61:15
		00, 00	40, 20	04, 40
Estimated c	attle density	2.06	2 12	2.04
(animals/ha)		2.00	2.12	2.04

**Table 3.10.** Summary characteristics of the study sites in the Commons River (CM) catchment.



**Fig.3.8.** Cattle access sites in the Commons River (CM) catchment. From the top to the bottom: CM1, CM2, CM3.

#### 3.3.5. Milltown Lake catchment

The Milltown lake catchment (30.6 km<sup>2</sup>) is located in north-east Ireland (55°12'N, 6°42'W). It is a sub-catchment of the Newry, Fane, Glyde and Dee catchment. The Drumleek River is the only inflow to Milltown Lake. The catchment total area, excluding Milltown Lake, is 28.8 km<sup>2</sup> (Linnane et al. 2011).

The river branches into three inflowing tributaries 440 m above the lake. Two of the larger tributaries are themselves fed by small lakes: a small lake on the western tributary in Carnagh Forest, Tievenamara, and Gentle Owen's Lake on the middle tributary (Carson, 2010). The area is comprised of small hills with poorly drained alluvial gleys, peaty gleys and inter-drumlin peats with extensive blanket bog. Drainage improvements mean that flashy storm flows and suppressed base flows now characterise the hydrology of the catchment (Wynne, 2012). Land use is predominantly agricultural, with grassland accounting for 90% of agricultural land. However, small areas of other agriculture, forestry and peat bog are also present in the upper section of the catchment. The nearest EPA monitoring station to the study sites is RS06G040080, reporting a likely moderate water quality status (Table 3.11)

Catchment	Newry, Fane, Glyde and Dee
River	Gentle Owen's Lake Stream
County	Monaghan
Flow (m <sup>3</sup> /s) <sup>1</sup>	0.033
Total annual precipitation (mm) <sup>2</sup>	906.8
Hydrology	Surface
Ecological status (Sampling period)	Moderate
Biological status (Sampling period)	Moderate
Chemistry conditions (Sampling period)	Pass
Nutrient Condition	Pass
WFD Risk	At Risk
Ortho-P status quality (trend)	NA
Waterbody trend	No change
Significant Pressures	Agriculture; wastewater

**Table 3.11.** Description of the Milltown Lake (MT) catchment. Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

<sup>1</sup>Flow data derived from the EPA Drumleek station. <sup>2</sup>Precipitation data derived from the Met Eireann Coose – Castleblayney station. NA – Not Assigned

		MT1	MT2	MT3
Stream	n order	1	1	2
Stream v	vidth (m)	1.90	2.28	2.96
Reach gra	adient (%)	-2.20	-2.10	-1.40
	Boulder	4	0	3
Substrate	Cobble	85	73	75
(%)	Gravel	10	25	30
	Sand + Silt	1	2	2
Soil	type	Poorly dra	ined peaty soils	Shallow mineral soils
Geologic formations		Silurian shale	Ordovician black shale	Silurian shale
Site description		Discrete access site; steep field; vegetated banks, some shrub coverage	Open access site, with other small access points upstream; vegetated banks	Open access site located by bridge; another (partially restricted) access site approximately 20 m upstream; vegetated banks
RHI scores acces	(upstream; s site)	72; 53	37; 18	43; 27
Estimated c (anima	attle density ls.ha <sup>-1</sup> )	2.21	2.21	2.00

**Table 3.12.** Summary characteristics of the study sites in the Milltown Lake (MT) catchment.







**Fig.3.9.** Cattle access sites in the Milltown Lake (MT) catchment. From the top to the bottom: MT1, MT2, MT3.  Impacts of unrestricted cattle access to watercourses on streambed sediment faecal contamination in agricultural streams

# Chapter 4. Impacts of unrestricted cattle access to watercourses on streambed sediment faecal contamination in agricultural streams

# 4.1. Introduction

Agriculture is recognised as a significant contributor to pollution and impairment of surface waters throughout the world (Smith et al., 1999; Malmqvist and Rundle, 2002; Pärn et al., 2012). While many of these impacts arise from diffuse pathways of contamination (e.g Crowther et al., 2002; Collins et al., 2005; Davies-Colley et al., 2008; Douglas et al., 2007; Ulén et al., 2007; Miller et al., 2010), point sources can also play an important role. These include sites where livestock directly access the watercourses for drinking and/or crossing between fields (Vidon et al., 2008; Smolders et al., 2015; O'Callaghan et al., 2018; O'Sullivan et al., 2019). Such activity can result in the removal of riparian vegetation, bank erosion (Kauffman et al., 1983), stream channel degradation (Trimble and Mendel, 1995; Herbst et al., 2012), increased sedimentation (Sheffield et al., 1997; Sovell et al., 2000; O'Sullivan et al., 2019), changes in biotic communities (Conroy et al., 2016) and contamination of waters with potentially pathogenic faecal organisms as a consequence of direct excretion into the stream channel (Eyles et al., 2003; Davies-Colley et al., 2004; Smolders et al., 2015).

A common indicator of faecal contamination is the bacterium *Escherichia coli*, which constitutes part of the normal gastrointestinal flora of humans and warm-blooded animals. *Escherichia coli* (*E. coli*) bacteria are generally commensal, however there strains capable of causing a variety of diseases (Kaper et al., 2004; Croxen et al., 2013), that are generally divided in three clinical syndromes: enteric or diarrhoeal diseases, urinary tract infections and sepsis/meningitis (Kaper et al., 2004). These pathogenic variants, or pathotypes, are an

important cause of morbidity and mortality worldwide (Croxen et al. 2013). Among the pathotypes that cause enteric disease is verocytotoxigenic *E. coli* (VTEC) e.g. the O157 serotype (Karmali et al., 2010; Petit et al., 2017). *E. coli* O157 can cause severe illness in humans, particularly young children (Brehony et al., 2018; Pennington, 2010; Rice et al., 2016), and, although it is asymptomatic in cattle (Murphy et al., 2016), infected animals can excrete the microorganism in numbers as high as  $10^9$  colony forming units (CFU) per g in their faeces (McCabe et al., 2018). The potential severity of illness associated with VTEC, together with its very low infectious dose (10 – 100 bacterial cells) (Brehony et al., 2018; Murphy et al., 2016; Rice et al., 2016) and its reported ability to survive in the environment for prolonged periods of time (Money et al., 2010) make VTEC infection a major health concern associated with cattle-based agriculture (Óhaiseadha et al., 2016; Brehony et al., 2018) (see Table 4.1 for concentrations of *E. coli* bacteria in cow faeces reported in literature).

Reference	E. coli concentration in cattle faeces	
Avery et al. (2004)	7.70 log <sub>10</sub> CFU .g <sup>-1</sup>	
Davies-Colley et al. (2004)	1.2 x 10 <sup>7</sup> CFU.g <sup>-1</sup> (7.08 log <sub>10</sub> CFU.g <sup>-1</sup> )	
Weavers and Graves (2005)	5.88 log <sub>10</sub> CFU.g <sup>-1</sup> (pasture cattle)	
Weavers and Graves (2003)	7.27 log <sub>10</sub> CFU.g <sup>-1</sup> (penned cattle)	
Oliver et al. (2010)	7.12 log <sub>10</sub> CFU.g dry wt <sup>-1</sup>	
Olandeinde et al. (2014)	2.20 x 10 <sup>6</sup> MPN.g dry wt <sup>-1</sup>	
Oliver and Page (2016)*	6.90 log <sub>10</sub> CFU dry wt <sup>-1</sup>	

**Table 4.1** Concentrations of *E. coli* bacteria in cattle faeces reported in literature.\*figures presented are calculated from reported results.

In Ireland, agricultural land covers approximately 72% of the total land area (EC, 2016), of which 80% is permanent grassland (EC, 2015c). Bovines constitute approximately 82% of

the total livestock units (LSU) (Eurostat, 2019). The Irish Environmental Protection Agency (EPA) has identified agriculture as a major source of pollution of Irish waterbodies (Fanning et al., 2017). Faecal contamination of drinking water supplies in rural areas in Ireland has continuously been recorded (EPA, 2018), and the country has consistently reported the highest incidence rate of VTEC infection in the EU (Garvey et al., 2015; Óhaiseadha et al., 2016; Brehony et al., 2018), which was 20.0 cases per 100 000 population in 2018, compared to an average incidence of 2.4 per 100,000 in the rest of the EU (ECDC, 2020). The consumption of contaminated water from rural household wells, which are exempt from regulations and monitoring, has been identified as the most significant primary transmission pathway of *E. coli* O157 infection (Garvey et al., 2015; Óhaiseadha et al., 2016; Óhaiseadha et al., 2015; Óhaiseadha et al., 2016).

The use of E. coli to assess recent environmental faecal contamination (Pachepsky and Shelton, 2011; Cloutier and Mclellan, 2017) is based on the assumptions that these bacteria do not multiply or persist for long periods of time outside of the host (Pachepsky and Shelton, 2011). However, as discussed in previous chapters, it is now widely recognised that E. coli can persist in the environment, particularly in sediments. In stream waters, bacteria often become attached to sediment particles which are then deposited on the stream bed, where they can accumulate in concentrations that are generally orders of magnitude higher than those of the overlying waters (Pachepsky and Shelton, 2011) and where they have been observed to persist for weeks to months (Davies et al., 1995; Craig et al., 2004;). The role of sediments in favouring bacterial persistence has been demonstrated in different environments, including coastal sediments (e.g. Craig et al., 2004; Ishii et al., 2007), and freshwater sediments (e.g. Muirhead et al., 2004; Garzio-Hadzick et al., 2010). Thus, bed sediments can act as sinks for faecal contamination, but they can also act as sources (Ishii et al., 2007; Hassard et al., 2016) when disturbance of contaminated sediments as a result of cattle in-stream movements (Collins and Rutherford, 2004) or of high flow conditions (Nagels et al., 2002; Muirhead et al., 2004) causes resuspension of viable faecal organisms into the water column (Jamieson et al., 2005a; Ouattara et al., 2011). Furthermore, studies

have indicated that *E. coli* can become naturalised and integrated in environmental indigenous microbial communities (Ishii and Sadowsky, 2008; Perchec-Merien and Lewis, 2012; Jang et al., 2017). Therefore, the bed sediment *E. coli* population would represent a combination of recent contamination and naturalised communities.

Water quality protection measures that limit or restrict cattle access to watercourses (e.g. fencing), have been demonstrated to have positive effects on water quality (McKergow et al., 2001; Agouridis et al., 2005; O'Callaghan et al., 2018), including reducing faecal contamination (e.g. Line, 2003; Wilcock et al., 2013; Smolders et al., 2015; Bremner et al., 2016; Bragina et al., 2017), and have been frequently included in agri-environmental policy. In Ireland, under the current Green Low Carbon Agri-Environment Scheme (GLAS), approximately 20,100 farmers have committed to implement cattle exclusion measures (GLAS Farm Planners [GFP], 2020). Restricting cattle access to watercourses through fencing will also become compulsory in 2021 (DoHPLG, 2018b) for approximately 12 000 derogation farmers under the fourth Nitrates Action Programme. Nevertheless, despite the known impacts of cattle-based agriculture on faecal contamination of surface waters and the wide application of cattle exclusion measures in agricultural policy, the contribution of direct cattle access to stream bed sediment E. coli contamination, and how this contamination can persist, has rarely been addressed in a European context. Bragina et al. (2017) recently showed that the levels of *E. coli* in stream sediments in one of the catchments in the current study (Milltown Lake catchment – see Chapter 3) were significantly higher where cattle had access to the stream than where access was restricted through fencing. In that study, average *E. coli* in sediments in a fenced stream ranged from 1.0 x 10<sup>2</sup> to 3.7 x 10<sup>2</sup> CFU.g dry wt<sup>-1-</sup>, whereas in unfenced streams these ranged from 4.8 x  $10^2$  to 5.2.x  $10^5$  CFU.g dry wt<sup>-1</sup> (Bragina et al. 2017). Interestingly, the authors also reported average E. coli concentrations of 1.9 x 10<sup>4</sup> CFU.g dry wt<sup>-1</sup> at one cattle access site than remained unfenced in their fenced study stream (Bragina et al., 2017). The present study builds on those findings and aims to garner a better understanding of whether unrestricted cattle access can have similar impacts

on sediment faecal contamination across Irish catchments generally, and across a range of water quality status and agricultural intensities. It is the first study attempting to quantify this contamination at multiple sites, in Ireland or elsewhere, while also contributing to a general understanding of the effects of unrestricted cattle access on faecal contamination of rural watercourses. The specific objectives of this study were:

- a. to quantify background levels (i.e. with no cattle access) of *E. coli* in bed sediments in first to third order streams in five agricultural catchments across Ireland;
- b. to investigate the impacts of cattle access on stream bed sediment *E. coli* contamination;
- c. to establish whether this contamination varied from mid-grazing season to post grazing season; and
- d. to determine whether there was a downstream cumulative effect of sediment *E. coli* contamination in streams with several access sites.

## 4.2. Methods

## 4.2.1. Site selection and experimental design

The study on sediment *E. coli* concentrations was separated into two study designs, described in detail in Chapter 3 (Table 4.2). Study 1 aimed to assess the impact of cattle access on *E. coli* levels in sediment in the headwater sites. For this purpose, nine sites were selected in the headwater zones of the catchments. Six of these sites were in the BW and DG catchments (labelled BWA, BWB, BWC and DGA, DGB and DGC), with one site on each of the BK, CM and MT catchments (labelled BK1, CM1 and MT1). The levels of sediment *E. coli* upstream of these sites, where cattle did not have access to the water, were used to assess background levels of *E. coli* contamination. Study 2 investigated potential

downstream cumulative effects of cattle access to watercourses on sediment faecal pollution. Here the upper sites in BK, CM and MT were used along with two additional sites per stream along a downstream gradient (sites labelled 1 - 3) (See Chapter 3).

Study	Objective	Sites included
Ctudy	Investigate the impacts of cattle access to	MT1, CM1, BK1
Sludy	watercourses on stream sediment E. coli	DGA, DGB, DGC
I	levels in headwater sites (i.e. uppermost sites)	BWA, BWB, BWC
	Assess potential cumulative effects of cattle	MT1, MT2, MT3
Study	access to watercourses on sediment E. coli	CM1, CM2, CM3
2	contamination (i.e. downstream gradient along	
sites)		BK1, BK2, BK3

**Table 4.2.** Allocation of study sites in Study 1 and Study 2 of the experimental design.

In Ireland, cattle graze outdoors during the spring and summer months (generally April to October/November) and are housed over the winter months. The 15 sites (Study 1 and Study 2) were sampled in mid-grazing season (June) and in post-grazing season (late November – early December) after cattle had been housed for the winter period, in 2016. These sampling times were selected to ensure that the impacts of cattle access to watercourses on sediment faecal contamination were captured, in accordance with the findings of Bragina et al. (2017), who sampled in three key points of the agricultural management cycle in Ireland (April, July and October) in two consecutive years. The authors reported a recurrent seasonal pattern in which *E. coli* sediment concentrations were the lowest in April, after cattle had been absent from the grazing fields for the winter period, and the highest in July, during grazing season. Thus, June was selected in the current study to assess sediment *E. coli* concentrations when the impact of cattle access would have been apparent for several months. This was to capture faecal contamination during grazing

season while combining microbial sample collection with sampling for other studies within the COSAINT project.

# 4.2.2. Site description

The study sites are described in detail in Chapter 3.

# 4.2.3. Sediment sampling

On each sampling occasion, three sediment samples were collected randomly at the cattle drinking sites and three were collected upstream (10 – 300 m) of these sites (i.e. a total of six samples per site). The samples were collected into new Petri dishes (1.2 cm depth, 8.5 cm diameter) by placing the dishes upside down onto the stream sediment and lifting them with a help of a metal scrapper that had been introduced under the Petri dish (Bragina el al., 2017). To avoid cross contamination, the metal scrapper was sterilised between samples using a solution of 70% industrial methylated spirit (IMS) and rinsed with deionised water (Bragina et al., 2017). The Petri dishes containing sediment samples were sealed using Parafilm and transported to the laboratory in a cooler box for subsequent analysis.

All samples were stored in the dark at the temperature of 4°C and analysed within 24 hours where possible; in a small number of instances, where long distance travelling did not allow the completion of analysis within this time period, the analysis was concluded within a maximum of 48 hours. Analysis of *E. coli* densities for samples held at less than 10°C within 48 hours of collection has been shown to yield comparable results to those obtained in analyses undertaken within 8 hours of sample collection (Pope et al. 2003).

#### 4.2.4. *E. coli* enumeration

For bacterial extraction from sediment, a technique previously described by Boehm et al. (2010) and adapted by Bragina et al. (2017) was used. The sediment samples were thoroughly mixed prior to extraction to ensure homogeneity. Approximately 10 grams of wet sediment were added to 90 ml of Ringer diluent (Oxoid, Hampshire, England) in a sterilised 100 ml Duran bottle. Each bottle was hand shaken for one minute and allowed to settle for a further minute before a set of sequential dilutions was prepared. Each dilution was filtered through a sterile cellulose esters membrane with 0.45  $\mu$ m pore size with grids (British Standard Institution, 2000). The membranes were placed onto Petri dishes containing Harlequin<sup>TM</sup> *E. coli*/Coliform medium (LabM, Lancashire, UK) and incubated at 37 °C for 18 – 24 hours according to the manufacturer's instructions. All green-blue colonies were counted as presumptive *E. coli* bacteria.

#### 4.2.5. Sediment characterisation

The sediment moisture content was determined as described by Hanlon et al. (2000). Approximately 20 g of fresh sediment from each sample were weighed in crucibles, dried at 105 °C for 24 hours and placed in a dried atmosphere (desiccator) to allow the crucibles to cool down. The samples were then re-weighed, with the difference in the sediment sample weight giving the moisture content. Sediments at the 15 study sites were further characterised as part of the study described in Chapter 5. The upper 3 – 5 cm of bed sediment were collected using a core sampler with a diameter of approximately 73 mm at each cattle access site and upstream, with a total of 6 samples collected at each location. The samples were oven-dried at 105°C for at least 24 hours and sieved to < 2 mm particle size. Composite samples for each location were then made using a quartering technique to allow homogeneous sectioning of each individual sample, and further analysed. Sediment Sediment organic matter content has been reported as a significant factor controlling the

survival of *E. coli* in sediment (e.g. Bragina et al., 2017). For sediment characterisation, organic carbon concentrations were determined using an Elementar El Vario Cube elemental analyser (Elementar Analysensysteme GmbH, Hanau, Germany), following removal of inorganic carbon by exposing dry sediments to concentrated HCI (37%) fumes (Harris et al., 2001). Samples were calibrated against a standard (acetanilide) (Merck, Darmstadt, Germany) with known concentrations of carbon (71.09%). Sediment characterisation data are presented in Table 4.3.

	Site	Moisture (%)		Organic carbon (mg g dry wt <sup>-1</sup> )	
Catchment		Mean ± S.E.	n¹	Mean ± S.E.	n¹
BW	BWA	27.86 ± 2.51	12	23.97 ± 1.28	8
	BWB	29.18 ± 4.90	12	28.31 ± 8.29	8
	BWC	32.39 ± 3.34	12	24.01 ± 3.01	8
DG	DGA	29.45 ± 3.42	12	$30.69 \pm 2.66$	8
	DGB	24.93 ± 1.74	12	21.21 ± 1.50	8
	DGC	21.75 ± 0.48	12	21.86 ± 2.46	8
ВК	BK1MG	34.79 ± 4.32	6	21.44 ± 5.98	4
	BK2MG / BK3PG	27.70 ± 3.14	12	24.94 ± 1.91	8
	BK3MG	16.49 ± 0.99	6	18.82 ± 1.23	4
	BK1PG	29.93 ± 5.46	6	16.12 ± 2.23	4
	BK2PG	33.15 ± 4.12	6	$16.88 \pm 4.65$	4
СМ	CM1	19.66 ± 2.16	12	34.29 ± 3.94	8
	CM2	21.16 ± 1.29	12	31.22 ± 5.68	8
	CM3	29.92 ± 3.71	12	36.28 ± 5.57	8
MT	MT1	41.98 ± 6.88	12	44.83 ± 7.66	8
	MT2	29.03 ± 1.80	11	40.78 ± 4.19	8
	MT3	47.71 ± 7.34	10	29.20 ± 1.97	8

**Table 4.3.** Moisture content of the sediment samples and sediment organic carbon at the study sites (mean  $\pm$  S.E.).

<sup>1</sup>Moisture content (%) was determined in sediment samples used in microbiological analysis (collected in the two sampling times, except in BK sites sampled in only one sampling time); OC was determined in sediment samples for the same sites analysed for nutrients (discussed in Chapter 5).

#### 4.2.6. Statistical analysis

All *E. coli* data were log<sub>10</sub> transformed prior to all analyses, to account for the extreme range in the data. All analyses were carried out using R software (version 3.5.1; R Core Team, 2016). Assumptions of equal variance and independence were checked for in all statistical analyses using the methods described in Zuur et al. (2009).

Differences in background sediment bacteria concentrations (i.e. at upstream sites) in each of the nine headwater sites (Study 1) and between the two sampling times were assessed using a two-way analysis of variance (ANOVA). For the remaining analyses where comparisons of bacterial concentrations at cattle access sites (CAS) and at upstream sites (US) were conducted, mean values for bacterial concentrations from the three samples taken (n=3) were used to avoid pseudo-replication. For Study 1, a general assessment of the differences in sediment bacteria concentrations between US and CAS (factor Treatment) and between sampling times (factor Time) was performed using a repeated measures two-way ANOVA, with interactions included. Further comparisons of bacteria concentrations between US and CAS for each sampling time were conducted using a post-hoc test (Tukey test) as well as paired t-tests.

For Study 2, differences in *E. coli* concentrations between US and CAS, sampling times, and sites were assessed using a mixed effects model analysis where factor Catchment was included in the model as a random component, and a variance structure for factor Treatment was added to account for unequal variance in residuals (Zuur et al., 2009). This model was selected by comparing different models using the Akaike Information Criterion (AIC). The model was then optimized by starting with all factors and interactions in the fixed component of the model, systematically dropping non-significant interactions and factors, and comparing nested models according to Zuur et al. (2009). The final model included factors Time,
Treatment and an interaction between factors in its fixed component. Post-hoc comparisons were conducted using a Tukey test.

Lastly, generalised additive modelling (GAM) analysis was used to assess the relationship between general site degradation and *E. coli* sediment levels. GAM analysis is a more flexible analysis than traditional methods as it allows for non-linear relationships between explanatory and response variables. The analysis was conducted for all sites separately for each sampling time, using average *E. coli* concentrations and RHI scores for US and CAS. For mid-grazing season data, a variance structure was added to the model to account for unequal variances. In addition, since O'Sullivan et al. (2019) conducted the site assessment in early autumn, the two sites that were later replaced in the BK catchment were not included in the analysis of the data collected in mid-grazing season. GAM analysis was also used to assess potential relationships between the estimated cattle density (cattle.ha<sup>-1</sup>) at each of the study sites *E. coli* bed sediment concentrations, in each sampling time.

#### 4.3. Results

#### 4.3.1. Background levels of faecal contamination

Sediment *E. coli* concentrations at the five study catchments are shown in Fig, 4.1 and are presented in Table 4.4 (mean and S.E.) for each of the sampled locations. The mean *E. coli* sediment concentrations in the study catchments ranged from  $5.7 \times 10^3$  CFU.g dry wt<sup>-1</sup> (BW catchment) to 2.7 x 10<sup>6</sup> CFU.g dry wt<sup>-1</sup> (DG catchment) in mid-grazing season, whereas in post-grazing season, mean value ranged from  $3.9 \times 10^3$  CFU.g dry wt<sup>-1</sup> (BK catchment) to  $3.4 \times 10^4$  CFU.g dry wt<sup>=1</sup> (MT catchment). In bed sediment at the upstream locations (US) of the nine headwater sites, which was considered to represent the background concentration

in the absence of cattle, the mean *E. coli* concentrations measured ranged from  $6.3 \times 10^2$  CFU g dry wt<sup>-1</sup> (site BWA) to  $1.1 \times 10^4$  CFU g dry wt<sup>-1</sup> (site DGA) in mid-grazing season (Table 4.4). The concentrations in mid-grazing season were significantly higher than those observed in post-grazing season (F-value = 27.875, *p* < 0.001; Fig. 4.2), when site CM1 in the Commons River catchment was the most contaminated site (9.5 x  $10^3$  g dry wt<sup>-1</sup>), while at site BWC *E. coli* was not detected (Table 4.4).



**Fig.4.1.** Boxplot of overall *E. coli* ( $\log_{10}$ CFU.g dry wt<sup>-1</sup>) concentrations in the sediments of the five study catchments in mid-grazing season and post-grazing season (n = 18) (including both US and CAS locations).

			<i>E. coli</i> (CFU g dry wt <sup>-1</sup> )				
Catchment	Site	Treatment	Mid-grazin	g season	Post-grazing season		
		-	Mean	S. <i>E.</i>	Mean	S.E.	
		US	6.3 x 10 <sup>2</sup>	3.0 x 10 <sup>2</sup>	1.4 x 10 <sup>2</sup>	74	
	DVVA	CAS	1.1 x 10 <sup>3</sup>	7.6 x 10 <sup>2</sup>	6.9 x 10 <sup>2</sup>	5.3 x 10 <sup>2</sup>	
	DWD	US	7.1 x 10 <sup>3</sup>	4.0 x 10 <sup>2</sup>	7.1 x 10 <sup>2</sup>	92	
BW	BAAR	CAS	2.3 x 10 <sup>4</sup>	2.0 x 10 <sup>4</sup>	5.5 x 10 <sup>4</sup>	5.3 x 10 <sup>4</sup>	
	DWO	US	1.2 x 10 <sup>3</sup>	5.5 x 10 <sup>2</sup>	nd	-	
	BWC	CAS	1.5 x 10 <sup>3</sup>	1.1 x10 <sup>3</sup>	1.3 x 10 <sup>2</sup>	1.1 x 10 <sup>2</sup>	
	DGA	US	5.9 x 10 <sup>3</sup>	2.9 x 10 <sup>3</sup>	1.1 x 10⁵	1.0 x 10 <sup>5</sup>	
	DGB	CAS	1.6 x 10 <sup>7</sup>	9.0 x 10 <sup>6</sup>	2.2 x 10 <sup>4</sup>	1.7 x 10 <sup>4</sup>	
DO		US	2.5 x 10 <sup>3</sup>	3.9 x 10 <sup>2</sup>	4.6 x 10 <sup>2</sup>	2.7 x 10 <sup>2</sup>	
DG	DGC	CAS	4.4 x 10 <sup>3</sup>	1.4 x 10 <sup>3</sup>	1.5 x 10⁴	8.3 x 10 <sup>3</sup>	
		US	1.3 x 10 <sup>3</sup>	5.8 x 10 <sup>2</sup>	3.7 x 10 <sup>2</sup>	55	
		CAS	2.5 x 10 <sup>2</sup>	1.5 x 10 <sup>2</sup>	3.5 x 10 <sup>2</sup>	38	
	BK1MG/	US	1.1 x 10 <sup>4</sup>	2.1 x 10 <sup>3</sup>	2.2 x 10 <sup>3</sup>	1.2 x 10 <sup>3</sup>	
	BK1PG	CAS	1.9 x 10 <sup>6</sup>	1.5 x 10 <sup>6</sup>	4.5 x 10 <sup>3</sup>	1.9 x 10 <sup>3</sup>	
	BK2MG/	US	1.5 x 10 <sup>4</sup>	6.2 x 10 <sup>3</sup>	3.8 x 10 <sup>3</sup>	7.2 x 10 <sup>2</sup>	
BK	BKPG	CAS	1.3 x10⁵	7.3 x 10 <sup>4</sup>	7.5 x 10 <sup>3</sup>	7.9 x 10 <sup>2</sup>	
	BK3MG/	US	8.5 x 10 <sup>3</sup>	6.5 x 10 <sup>3</sup>	2.3 x 10 <sup>3</sup>	3.0 x 10 <sup>2</sup>	
	BK3PG	CAS	2.8 x 10⁵	1.4 x 10 <sup>5</sup>	3.1 x 10 <sup>3</sup>	96	

**Table 4.4.** Mean *E. coli* concentrations in sediments at the sites in the five study catchments in both sampling times (n = 3).\*n = 2. nd - not detected. Grey shading = nine headwater sites.

	Site		<i>E. coli</i> (CFU g dry wt <sup>-1</sup> )				
Catchment		Treatment	Mid-grazin	g season	Post-grazing season		
			Mean	S.E.	Mean	S. <i>E.</i>	
	CM1	US	2.8 x10 <sup>3</sup>	8.3 x 10 <sup>2</sup>	9.5 x 10 <sup>3</sup>	5.8 x 10 <sup>3</sup>	
	CIVIT	CAS	1.9 x 10 <sup>6</sup>	4.6 x 10⁵	7.9 x 10 <sup>2</sup>	1.0 x 10 <sup>2</sup>	
<u>CM</u>	0140	US	1.0 x 10 <sup>3</sup>	5.4 x 10 <sup>2</sup>	1.1 x 10 <sup>3</sup>	1.6 x 10 <sup>2</sup>	
СМ	CIVIZ	CAS	2.5 x 10 <sup>6</sup>	1.7 x 10 <sup>6</sup>	8.2 x 10 <sup>4</sup>	6.6 x 10 <sup>4</sup>	
	CM3	US	2.5 x 10 <sup>3</sup>	9.0 x 10 <sup>2</sup>	4.1 x 10 <sup>3</sup>	9.9 x 10 <sup>2</sup>	
		CAS	7.3 x 10 <sup>6</sup>	2.7 x 10 <sup>6</sup>	3.0 x 10 <sup>4</sup>	1.3 x 10 <sup>4</sup>	
MT	MT1	US	1.3 x 10 <sup>3</sup>	3.5 x 10 <sup>2</sup>	2.1 x 10 <sup>2</sup>	27	
		CAS	3.8 x 10 <sup>4</sup>	2.2 x 10 <sup>4</sup>	2.0 x 10⁵	5.5 x 10 <sup>4</sup>	
	MTO	US	2.8 x 10 <sup>3</sup>	2.4 x 10 <sup>2</sup>	7.0 x 10 <sup>2</sup>	2.6 x 10 <sup>2</sup>	
	MT2	CAS	*1.1 x	x 10⁵	2.8 x 10 <sup>2</sup>	1.8 x 10 <sup>2</sup>	
		US	1.6 x 10 <sup>3</sup>	8.2 x 10 <sup>2</sup>	1.9 x 10 <sup>3</sup>	5.4 x 10 <sup>2</sup>	
	IVI I 3	CAS	1.8 x 10 <sup>3</sup>	2.2. x10 <sup>3</sup>	2.0 x 10 <sup>3</sup>	9.2 x 10 <sup>2</sup>	

## 4.3.2. Impact of cattle access to streams on sediment faecal contamination and seasonal variation

Study 1

In the nine headwater sites in Study 1, sediments at cattle access sites (CAS) were significantly more contaminated with *E. coli* than those at the upstream sites (Table 4.5). While post-hoc tests did not allow further conclusions, paired t-tests conducted separately

for each sampling season indicated that this difference between US and CAS was only marginally significant in mid-grazing season (t = -2.443, p = 0.0404) and was non-significant in post-grazing season (t = -1.882, p = 0.0966). Nevertheless, it should be noted that the highest *E. coli* concentration observed in this study overall (1.6 x 10<sup>7</sup> CFU g dry wt<sup>-1</sup>) was found at the cattle access site at site DGA in the mid-grazing season. At this site, the *E. coli* concentration was four orders of magnitude higher that the concentration found immediately upstream (5.9 x 10<sup>3</sup> CFU g dry wt<sup>-1</sup>). Similarly, concentrations at sites BK1 and CM1 also increased from 1.1 x 10<sup>4</sup> and 2.8 x 10<sup>3</sup> CFU g dry wt<sup>-1</sup> upstream to both having concentrations of 1.9 x 10<sup>6</sup> CFU g dry wt<sup>-1</sup> at the access site, respectively.



**Fig.4.2.** Boxplot of average sediment concentrations of *E. coli* at cattle access sites (CAS) and at upstream areas with no cattle access (US) at the nine headwater sites (study 1) (n = 9).

#### Study 2

Similarly, in Study 2, for 9 sites in the three catchments with moderate water quality status, a significant difference in sediment *E. coli* concentrations was also found between the US and CAS sites (Table 4.5). Post-hoc tests showed that this difference was again only significant in mid-grazing season (Table 4.6), when bed sediment *E. coli* concentrations were generally one to three orders of magnitude higher at cattle access points than at sites immediately upstream (Fig. 4.3). In the post-grazing season sampling period, this effect was not observed, as sediment *E. coli* levels at cattle access sites significantly decreased, while not varying significantly between mid- and post-grazing season at upstream sites (Table 4.6).



# **Fig.4.3.** Boxplots of average sediment concentrations of E. coli per type (n = 9), site (n = 6) and catchment assessed in study 2 (n = 18), in mid-grazing season and post-grazing season. Note the different scales for different sampling times.

Finally, generalised additive model analysis of sediment *E. coli* levels at all sites in midgrazing season and the RHI scores revealed a significant negative relationship in midgrazing season (F value = 6.772, p = 0.008), with an R-sq. (adj) of 0.24 (Fig. 4.4), indicating that higher bed sediment concentrations of *E. coli* occurred at those sites where a general degradation of the riparian habitat had been recorded. There was no significant relationship between *E.* coli concentrations and RHI for the post-grazing season data (F-value = 0.118, p= 0.668). In contrast, there was as a significant relationship between *E. coli* bed sediment concentrations and estimated cattle density (ECD; cattle.ha<sup>-1</sup>) in post-grazing season (Fvalue = 3.727, p = 0.035), with an R-sq. (adj) of 0.26, but not in mid-grazing season (F-value = 1.161, p = 0.291) (Fig. 4.5).

#### 4.3.3. Cumulative downstream gradient

Analysis of *E. coli* sediment levels at the three sampled sites located along a downstream gradient in the BK, CM and MT catchments revealed no significant difference between the sites (F-value = 0.063, p = 0.939), indicating that there was no significant cumulative downstream pattern of faecal contamination of sediments for either sampling times (Table 4.4, Fig. 4.3).

**Table 4.5**. Analysis of variance of *E. coli* in sediment in Study 1 (repeated measures ANOVA; n = 36) and Study 2 (linear mixed model analysis; n=36). Treatment = US and CAS. Time = mid-grazing season and post-grazing season. Figures in bold denote statistical significance at the 0.05 level.

Significant factors	DF	F-value	p
Study 1			
Treatment	1,8	18.406	0.003
Time	1,8	9.636	0.0146
Time*Treatment	8	0.137	0.721
Study 2			
Treatment	1	16.168	<0.001
Time	1	55.896	<0.001
Time*Treatment	1	17.696	<0.001

**Table 4.6.** Post-hoc comparisons (Tukey test) for the analysis of *E. coli* in sediment in Study 2 (interactions between factor Treatment (US and CAS and Time (mid-grazing season [MG] and post-grazing season [PG]). Figures in bold denote statistical significance at the 0.05 level.

Contrasts	DF	t ratio	p
CAS – US, MG	101	8.242	<0.001
CAS – US, PG	101	2.378	0.088
CAS, MG – CAS, PG	101	5.466	<0.001
US MG – US,PG	101	2.096	0.161



**Fig.4.4.** Scatterplot and smoother for RHI scores in the GAM with levels of *E. coli* in sediments in mid-grazing season as response variable. The central solid line is the smoother and the grey area is 95% confidence bands. Y axis units are the scaled smoother (s) for RHI with estimated degrees of freedom (edf) of 1.73.



**Fig.4.5.** Scatterplot and smoother for estimated cattle density (ECD; cattle.ha<sup>-1</sup>) in the GAM with levels of *E. coli* in sediments in post-grazing season as response variable. The central solid line is the smoother and the grey area is 95% confidence bands. Y axis units are the scaled smoother (s) for RHI with estimated degrees of freedom (edf) of 3.43.

#### 4.4. Discussion

Contamination of freshwater waters and sediments with faecal organisms has implications for both human and animal health. In this study, it was notable that bed sediments at all 15 sites upstream of cattle access points in this study were contaminated with *E. coli*, with only one site where no *E. coli* were detected on one occasion (upstream of cattle access site BWC, in post-grazing season). These results highlight the widespread contamination of stream sediments with *E. coli* in agricultural catchments. Additionally, they show that cattle access to streams clearly exacerbates this contamination.

While studies investigating E. coli contamination in sediments of agricultural streams are scarcer in the literature that those on, for example, coastal sands, the concentrations from other studies provide a context for the resulted presented. Hassard et al. (2017) assessed the levels of faecal pollution in estuarine sediments of two areas in the UK impacted by a variety of pollution sources, including agricultural land used for livestock grazing, urban areas and wastewater discharges, and found average E. coli levels above 10<sup>4</sup> CFU 100 g dry wt<sup>-1</sup>. Bonilla et al. (2007) also reported similar average *E. coli* concentrations of 10<sup>4</sup> CFU 100 g<sup>-1</sup> in sand samples of three recreational beaches in south Florida, USA, over a two-year sampling period. Ishii et al. (2007) observed *E. coli* concentrations of 10<sup>2</sup> to 10<sup>3</sup> CFU g dry wt<sup>-1</sup> in sediment and sand in a recreational lake beach affected by sewage treatment effluents, while Abdelzaher et al. (2010) reported maximum enterococci concentrations of 10<sup>3</sup> CFU a dry wt<sup>-1</sup> in sand samples from a recreational beach impacted by diffuse sources of pollution. The levels found in the current study, which ranged from 1.3 x 10<sup>2</sup> CFU g dry wt<sup>-1</sup> up to 1.6 x 10<sup>7</sup> CFU g dry wt<sup>-1</sup>, are therefore at the higher end of those reported in literature and are likely to have serious implications for the quality of the overlying water if these microorganisms are resuspended.

Bed sediment levels of *E. coli* were highly variable within sites in our study, highlighting the patchy nature of bacteria distribution in sediments (Pachepsky and Shelton, 2011) and also within catchments, suggesting that stream faecal contamination is also governed by field-scale management practices. This would include field-specific factors that were not assessed in the current study, such as stocking rate, period of grazing, grazing rotation practices, existence of other access sites or alternative drinking water sources, and slurry spreading. In this study, however, there was no apparent relationship between estimated cattle density per hectare for each site and the average *E. coli* concentrations measured at the access points, or with the difference between the US and CAS subsites. Unfortunately, despite designing a questionnaire and getting some individual results, it was not possible to collect additional reliable data on agricultural management practices from all farmers and

landowners involved in this study to better understand the influence of individual management practices on sediment faecal contamination. The importance of farm-level conditions in understanding faecal pollution dynamics and the difficulties in accessing such information has been highlighted by Winter et al. (2011).

#### 4.4.1. Background levels of stream sediment contamination

Sediment faecal contamination in reaches of the stream with no cattle access at the nine headwater sites, which also did not have significant cattle access pressure upstream or any visible immediate visible influence of point discharges, are assumed to be mostly a result of diffuse pathways of pollution as well as possibly some residual contribution of wildlife faecal contamination. Septic tanks within the catchments can also contribute to faecal contamination; a study by Arnscheidt et al (2007) conducted in three agricultural catchments Ireland reported that although faecal contamination in the sediments was predominantly of herbivore origin, between 7% and 27% of contamination was of human origin. While it is surprising that stream sediments at these headwater reaches (to which cattle had no access) had relatively high concentrations of E. coli, these levels of faecal contamination are comparable to those measured by Bragina et al. (2017) in the Milltown Lake catchment, who reported a median *E. coli* sediment concentration of 9.6 x 10<sup>2</sup> CFU g dry wt<sup>-1</sup> in a fenced tributary. The same authors also reported a sediment *E. coli* median value of 43 CFU g dry wt<sup>-1</sup> at a first-order stream in an agricultural area grazed by small numbers of sheep, with no cattle production or human settlement (Bragina et al., 2017). These findings, together with the observation that sediment faecal contamination at the headwater sites was significantly more pronounced in the grazing season, support the evidence that catchment-scale activities can have significant impacts on stream sediment faecal contamination regardless of whether animals have access to the watercourses (Collins et al., 2005; Davies-Colley et al., 2008; Smolders et al., 2015; Bragina et al., 2017). Mechanisms of faecal contamination of watercourses arising from cattle-based agriculture include runoff from grazing fields (Frey

et al., 2015; Cho et al., 2016;) and spreading of organic fertilisers such as slurry and farmyard manure (Hodgson et al., 2016), which in Ireland is permitted from approximately mid-January - February to mid-October - November, depending on the region (DAFM and Department of Housing, Planning and Local Government [DHPLG], 2017) and takes place predominantly during the summer months.

## 4.4.2. <u>Impacts of cattle direct access to watercourses on sediment faecal contamination</u> and seasonal variation

Sediment E. coli concentrations in this study were significantly higher at the access sites than at upstream sites in mid-grazing season, with a subsequent general decrease of E. coli levels at the access sites in post-grazing season. These findings confirm that direct cattle access to watercourses can exacerbate already high stream sediment faecal pollution. The relationship found in this study between the Riparian Habitat Index scores and the levels of sediment E. coli in mid-grazing season indicated that increased faecal contamination is associated with increased general degradation of the sites, which likely also relates to intensity of cattle usage. A similar relationship was found between RHI scores and resuspendable sediment levels at the same sites by O'Sullivan et al. (2019). However, this impact seems to be relatively short-lived. Similarly, in their study in the Milltown Lake catchment, Bragina et al. (2017) observed significantly higher sediment E. coli concentrations at sites with unrestricted cattle access in mid-grazing season, but noted that E. coli levels decreased at all study tributaries over the winter period and were at their lowest levels in April, with no significant differences observed between any of the study tributaries at this time. In the post-grazing season, in the absence of cattle access and associated frequent direct inputs of faecal contamination, sediment reservoirs of E. coli at access sites may be quickly reduced due to a combination of bacteria die-off and episodes of sediment flushing as a result of increased flows in this time of the year (Nagels et al., 2002; Jamieson et al., 2005a). Bacteria wash-off, along with sediment particles, during high flow episodes is

likely to be more pronounced at cattle access sites than at sites with no access due to the general stream channel degradation and typical absence of aquatic vegetation at these sites (O'Sullivan et al., 2019). Nevertheless, it is of note that *E. coli* concentrations at both upstream and cattle access sites in these catchments remained at levels of  $10^2$  to  $10^3$  CFU g dry wt<sup>-1</sup>. Presumably, this sediment *E. coli* contamination is a result of pollution from various sources (e.g. septic tanks and other human wastes, together with slurry spreading and surface run-off) and prolonged persistence of *E. coli* in the sediment matrix. *E. coli* persistence in sediments has been widely reported (Craig et al., 2002; Ishii et al., 2007; Garzio-Hadzick et al., 2010; Pachepsky and Shelton, 2011), and has indeed been shown to be favoured by factors such as lower temperatures (i.e. 4°C) (e.g. Garzio-Hadzick et al., 2010).

Although faecal contamination was significantly higher at access sites than at sites with no access in both Study 1 and Study 2, in the sites in Study 1 this impact was apparently less pronounced. This might be explained by the fact that these sites are located in areas that typically have less intensive agriculture when compared to the sites in Study 2 (See Chapter 3). Interestingly, in this study, there was a relationship between estimated cattle density at each site and sediment *E. coli* concentrations only in post-grazing season. This might indicate that, during grazing season, cattle access to watercourses has a strong effect on stream bed sediment *E. coli* contamination, whereas in the absence of cattle, diffuse pathways of pollution at the catchment scale dominate. The observation that the highest sediment levels of *E. coli* in this study ( $1.6 \times 10^7$  CFU g dry wt<sup>-1</sup>) were found at the DGA site, which had one of the lowest RHI scores, further supports the evidence that faecal contamination is largely determined by localised agricultural practices when cattle is present.

No evidence of a cumulative effect on sediment E. coli concentrations along a downstream gradient was found in this study. Similar results have been reported for the Milltown Lake catchment by Bragina et al. (2017), and indeed for resuspendable sediment levels at the same sampled sites in a study by O'Sullivan et al. (2019). Bacteria can be transported in streams as free floating individuals, but it is widely recognised that in aquatic systems, bacteria are generally associated with sediment particles (Jamieson et al., 2005b); Drummond et al., 2014), which are typically less than 60 µm in diameter (Jamieson et al., 2005b), and that this association influences bacteria transport along the stream channel. Drummond et al. (2014) investigated the transport of synthetic fluorescent fine particles and E. coli bacteria in a stream and observed that both fine particles and bacteria migrated similarly in the stream through a series of deposition and resuspension events, with retention occurring mostly near the sediment-water interface in macrophyte strands or at the top 3 cm of the streambed sediments, due to bulk advection, turbulent diffusion and hyporheic exchange processes. The authors observed that macrophyte strands and streambed sediments could act both as short and long-term (i.e. months) reservoirs for microorganisms and fine particles, and that the extent of remobilisation was influenced by the structure of the stream environment, the delivery of water-borne material to depositional areas and the frequency of disturbance events (Drummond et al., 2014). Given that the distance between the sites investigated in the present study is relatively large (approximately 700 to 5000 m), it is likely that, although sediment bacteria resuspension during disturbance events generally play an important role in faecal bacteria dynamics in streams, mechanisms of bacteria deposition in storage zones between sites and of microorganisms die-off can prevent the build-up of a cumulative gradient of contamination along the stream channel.

#### 4.5. Implications of stream sediment faecal contamination

This study shows that where cattle access watercourses, they contribute to reservoirs of E. coli in sediments that are able to persist after cattle removal from the grazing fields, with implications for water quality and both human and animal health. This effect can be particularly concerning in small order streams where the low water volume to sediment area ratio and the limited amount of submerged aquatic vegetation make bed sediments the largest available substrate for faecal bacteria accumulation (Badgley et al., 2011). Resuspension of accumulated faecal organisms following sediment disturbance has been widely reported (Jamieson et al., 2005a; Cho et al., 2010; Kim et al., 2010; O'Mullan et al., 2019). Moreover, it has been shown that a significant release of faecal bacteria from sediments to waters can also occur under baseflow conditions via hyporheic exchange (Pachepsky et al., 2017). Moreover, it should be noted that in this study, faecal bacteria were quantified using a culture-based method, however some bacteria, including E. coli, have been shown to persist in the environment in a viable but non-culturable (VBNC) state, in which they remain metabolically active but do not grow in microbiological growth media (Hassard et al., 2017, 2016). These bacteria cannot be detected by these conventional methods, but retain the ability to resuscitate if suitable conditions are provided, and are potentially pathogenic (Hassard et al., 2017, 2016). Thus, the sediment concentrations of faecal indicator bacteria reported in this study likely underestimate the real levels of faecal contamination and the animal and human health risk at the study sites.

#### 4.6. Conclusions

The findings of this study show that stream sediments in agricultural catchments can be important reservoirs of *E. coli* and likely therefore other faecal pathogens, and that this contamination can be substantially increased when cattle are allowed access to watercourses. Given that the practice of allowing cattle access to agricultural watercourses is widespread in Ireland, it is likely to have a significant overall impact on water quality and animal and human health, particularly in rural areas. The results of this work emphasise the need to adopt agricultural practices that protect human health and therefore support the integration of cattle exclusion measures in future domestic and international agricultural policy.

5. Impacts of unrestricted cattle access to watercourses on streambed sediment nutrient concentrations in agricultural streams

# Chapter 5. Impacts of unrestricted cattle access to watercourses on streambed sediment nutrient concentrations in agricultural streams

#### 5.1. Introduction

Livestock agriculture has been highlighted as an major contributor of excess nutrients (phosphorus and nitrogen) to waterbodies in Ireland (Mockler et al., 2016; O'Boyle et al., 2017). For instance, in a paleolimnological study in Milltown Lake, NE Ireland, Carson et al. (2015) reported indicators in lake sediments of a rapid deterioration of the water quality coinciding with the intensification of livestock agriculture in the catchment since the 1970s. In another paleolimnological study, Foy et al. (2003) reported that the intensification of agriculture and the associated increased loss of phosphorus (P) from agricultural soils were the main drivers of increased P levels in Lough Neagh, Northern Ireland, while Taylor et al. (2006) reached similar conclusions for five out of six lakes in the Irish Ecoregion. The most recent water quality report published by the Environmental Protection Agency (EPA), referring to the period of 2013 – 2018, indicated that 47.2% of monitored surface waterbodies are in less than good ecological status, representing a decrease of 2.6% of waterbodies in satisfactory status in relation to the period of 2010 – 2015, mostly driven by a general decrease water quality in Irish rivers (EPA, 2019). Excess nutrient loadings from agriculture were stated to be the main cause of this surface water pollution (EPA, 2019).

Contamination of watercourses occurs mainly through diffuse mechanisms such as transport of dissolved and particulate nutrients from agricultural soils in surface runoff, subsurface flow and drains and ditches (Douglas et al., 2007). This can represent residual (i.e. release of excess nutrients accumulated in soils) or incidental (e.g. nutrient loss from recently applied mineral or organic fertilisers) nutrient transfers (Shore et al., 2017). However, point source

pathways, including points of direct cattle access to watercourses for drinking and crossing between grazing fields, can also play an important role (O'Callaghan et al., 2018).

Cattle can increase water nutrient concentrations by directly defecating and urinating in watercourses (see table 5.1 for nutrient concentrations measured in cattle faeces and urine available in literature) and by causing nutrient release from sediments by grazing and trampling activity (Capece et al., 2007). In addition, where cattle have access to watercourses, they can cause stream banks to become exposed and susceptible to accelerated erosion (Fox et al., 2016), thereby significantly increasing the risk of loss of particulate nutrients to the stream (Fox et al., 2016; McDowell and Wilcock, 2007). Once in the aquatic system, nutrients can accumulate in bed sediments, which can then act as nutrient sources to the water column through resuspension, desorption and remineralisation mechanisms (House, 2003). Within sediments, the finer fraction (<2mm) is generally considered the most reactive due to its higher surface to volume ratio (Lucci et al., 2010). Phosphorus in particular interacts strongly with sediments as P compounds are 'reactive solutes' (McGechan et al., 2005) with high affinity for alumino-silicates (clays) and metal (particularly Fe and AI) oxides and hydroxides (Withers and Jarvie, 2008). This causes P to generally accumulate in solid phases in much higher concentrations than in solution phases (Sharpley et al., 2013). The mobilisation of accumulated P from stream sediment reservoirs is determined by geochemical factors such as dissolved P concentrations in the overlying waters and redox conditions, as well as sediment mineralogy and particle size distribution, presence of metal oxides, and the concentration of exchangeable phosphate adsorbed to the sediment particles (Palmer-Felgate et al., 2009; Neidhart et al., 2019). This potential for sediments to serve as both sinks and sources for P has been shown to lead to legacy P effects, whereby stored P is remobilised and acts as a source of contamination for long periods of time (e.g. decades or centuries) after deposition, even after the contamination source has been removed or controlled (Sharpley et al., 2013; Fox et al., 2016). This process is particularly difficult to manage, as the retention and remobilisation processes are

slow and ubiquitous along the aquatic system, thus limiting the effectiveness of best management practices and water quality protection measures (Jarvie et al., 2013; Wironen et al., 2018).

The available literature on the contribution of direct cattle access to excess nutrient loadings in watercourses is limited, especially in European sites (see Chapter 2), and very few studies report data on sediment nutrients in the context of livestock impacts (see Table 5.2 for an overview nutrient concentrations in stream banks, sediments and agricultural soils reported in literature). In a study where the effects of cattle activity on nutrient dynamics in streambanks were assessed using ion exchange membranes, Miller et al. (2017) found no significant impacts of cattle access to the stream or cattle grazing intensity on either NO<sub>3</sub>-N or P. These findings contrast with those of Palmer-Felgate et al. (2009), who compared streambed sediment TP concentrations in headwater systems with low agriculture intensity (named control systems) and high agricultural intensity within three lowland catchments in the UK. The authors reported higher TP sediment concentrations (ranging on average from 1429 to 2480 mg.kg<sup>-1</sup>) at a site located in the proximity of farmsteads and to which cattle had direct access at a number of points, in comparison to the control site (within the same study catchment) which was in an area grazed by only a few animals (where they ranged on average from 657 to 1060 mg.kg<sup>-1</sup>) (Palmer-Felgate et al., 2009). Additionally, the authors observed relatively high TP sediment concentrations (ranging on average from 155 to 636 mg.kg<sup>-1</sup>) at their control site in one of their study catchments (Wye catchment), which they hypothesised was caused by unrestricted cattle access to the stream (Palmer-Felgate et al., 2009).

Deference	Туре	Urine		Faeces		
Reference	cattle	N	Р	Ν	Р	
Lantinga et al. (1987)	Dairy	6.1 – 9.7 g.L <sup>-1</sup> TN				
Bristow et al. (1992)	Dairy	6.8 – 20.5 g.L <sup>-1</sup> TN				
Kirchmann and Witter (1992)	Dairy				9.0 g.kg dry wt <sup>-1</sup>	
Gonda and Lindberg (1994)	Dairy	5.8 – 10.7 g.L <sup>-1</sup> TN				
van Vuuren and Smits (1997)	Dairy	3.9 – 7.6 g.L⁻¹ TN				
Dou et al. (2002)	Dairy				5.21 – 12.65 TP g.kg dry wt <sup>-1</sup>	
Sorensen et al. (2003)	Dairy			18.1 – 37.8 g.kg dry wt <sup>-1</sup> TN		
Kool et al. (2006)	Dairy	9.0 – 10.3 g.L <sup>-1</sup> TN				
Orr et al. (2012)*	Beef	5.9 g.kg <sup>-1</sup> TN	0.0039 g.kg dry wt <sup>-1</sup>	3.2 g.kg <sup>-1</sup> TKN	4.9 g.kg dry wt <sup>-1</sup>	
Spek et al. (2012)	Dairy	3.0 – 10.4 g.L <sup>-1</sup> TN				
Dai and Karring (2014)	Beef	261 mmol.L <sup>-1</sup> TKN 15.9 mmol.L <sup>- 1</sup> TAN 152.7 mmol.L <sup>-1</sup> UN		337.8 mmol.kg <sup>-1</sup> TKN 21.2 mmol.kg <sup>-1</sup> TAN		
Misselbrook et al. (2016)	Dairy and beef	0.6 – 34.4 g.L <sup>-1</sup> TN				

**Table 5.1.** Nitrogen and phosphorus concentrations in cattle urine and faeces reported in literature.\*Figures presented here are averaged from the published data.

**Table 5.2.** Summary of some of the studies that have reported nutrient concentrations in agricultural field soils, stream bank sediments or streambed sediments in agricultural catchments.TP and TN concentrations are in mg.kg dry wt<sup>-1</sup> unless stated otherwise.

Reference	Location	Substrate	ТР	TN	OC
Murphy et al., 2000	England	Field soils		26 – 37 kg.ha <sup>-1</sup> (mineral N)	
McDowell and Sharpley, 2001	New Zealand	Bank sediment	259		
McDowell and Sharpley, 2001	New Zealand	Streambed sediment	214		
Thoma et al., 2005	USA	Bank sediment	249 - 452		
Thoma et al., 2005	USA	Field soil	622		
Falkengren-Grerup et al., 2006	Sweden	Forested, formerly cultivated soil		32.0 mol.m <sup>-2</sup>	
Taylor et al., 2006	Ireland	Lake surface sediments	2000- 4000+		100-150 g.kg <sup>-1</sup> (20-30% LOI)*
McDowell and Wilcock, 2007	New Zealand	Field soil (topsoil; subsoil)	3291; 1578		20.9; 8.1 g.kg <sup>-1</sup>
McDowell and Wilcock, 2007	New Zealand	Bank sediment	2738		26.1 g.kg <sup>-1</sup>
McDowell and Wilcock, 2007	New Zealand	Streambed sediment	921		11.1 g.kg <sup>-1</sup>

\*OC concentration is based on reported % loss-on-ignition (LOI) as 50% LOI (Dean, 1974).

Table 5.2	(continued)

Reference	Location	Substrate	ТР	TN	OC
Zaimes et al., 2008	USA	Bank sediment	303 - 555		
Palmer-Felgate et al., 2009	UK	Streambed sediment	394 - 2678		
Tufekcioglu, 2010	USA	Bank sediment	246 - 349		
Kronvang et al., 2012	Denmark	Bank sediment	400-1400		
Carson et al., 2015	Ireland	Lake surface sediment	940 - 1690		110 g.kg <sup>-1</sup> (22% LOI)*
Ishee et al., 2015	USA	Bank sediment	138 - 1140		
Shore et al., 2016	Ireland	Ditch sediment	200-1800		47 g.kg <sup>-1</sup> (9.40% LOI)*
Shore et al., 2016	Ireland	Bank sediment	200-400		21 g.kg <sup>-1</sup> (4.20% LOI)*
Shore et al., 2016	Ireland	Field soil	500-1400		62.5 g.kg <sup>-1</sup> (12.50% LOI)*
Neidhart et al., 2019	Germany	Streambed sediment	449 - 1392		

\*OC concentration is based on reported % loss-on-ignition (LOI) as 50% LOI (Dean, 1974).

Most of the literature available to date referring to cattle access to watercourses has focused on assessing the effectiveness of mitigation measures, rather than quantifying the impact of unrestricted cattle access to watercourses on nutrient levels within the aquatic system. Additionally, few studies have considered the stream sediment compartment and its potential role as a sink and source of excess nutrients. Notably, to the knowledge of the authors, there is no study specifically investigating the influence of cattle density on stream sediment nutrient concentrations. This factor is particularly relevant in Ireland as the size of the national herd increases to respond to changes in agricultural policy (e.g. the abolition of EU's milk production quotas in 2015) and agricultural development plans aimed at expanding the Irish agri-food sector such as Food Wise 2025 (O'Boyle et al., 2017; DAFM, 2020).

The present study aimed to contribute to these gaps in the literature. The aims of the study were:

- a. To determine whether concentrations of total phosphorus, total nitrogen and organic carbon in stream bed sediments were higher at reaches of the stream where cattle had access (cattle access points) compared to upstream reaches of the stream not used by cattle;
- b. To assess whether nutrient concentrations were greater the smaller silt + clay fraction (< $63\mu$ m) of the sediment than in the <2mm fraction;
- c. To evaluate whether cattle access points in areas with higher cattle density (expressed as estimated animal numbers per hectare, calculated at the locality level) had higher sediment nutrient concentrations than cattle access points in areas with lower cattle density.

#### 5.2. Methods

#### 5.2.1. Site selection and experimental design

A total of 15 active cattle access sites located in five agricultural catchments in Ireland were selected for this study. These sites and the study catchments are described in detail in Chapter 3.

#### 5.2.2. Sediment sampling

At each site, the stream bed sediment was sampled at four locations: 1. at the cattle access site, i.e. where cattle actively used the stream (CAS); 2. immediately upstream) of the access site, where animals had no access to the stream either due to fencing or natural physical barriers (this could be 20 – 322 m depending on the site (US); 3. immediately downstream of the cattle access site, where cattle did not have access (18 – 144 m depending on the site) (DS); and 4. at the interface (edge) of the stream water level, at the access path used by cattle to enter the stream (INT) (see Chapter 3). Sampling of this interface area was included because this area was hypothesised to be subject to a higher localised impact of cattle as they stand and drink from the stream.

Six sediment samples were collected randomly at each of the four locations using a clear Plexiglas corer with a bevelled edge (73 mm diameter) (Hedrick et al., 2013). The corer was used to ensure consistent sample size within and between sites. A metal scrapper, which had been previously acid-washed and rinsed with stream water to prevent sample contamination, was placed under the core base once inserted to the desired depth to aid in core removal. The upper 3 to 5 cm of the bed sediment were collected depending on the sediment depth, giving an approximate sample of 126 cm<sup>3</sup> to 209 cm<sup>3</sup>. Sediment samples were placed in individual plastic bags and transported to the lab in cool boxes, where they were kept in the dark at 4°C until further processing.

In Ireland, cattle graze outdoors during the spring and summer months (generally April to October/November) and are housed over the winter months. The 15 sites were sampled in early grazing season (mid-April and May) and in late grazing season (October), before cattle were housed for the winter period, in 2016. This was to capture potential differences in sediment nutrient concentrations related to the agricultural management cycle. i.e. following the winter/spring period when cattle would be absent from grazing fields, and following the

summer months during which cattle would have been grazing and therefore using the study cattle access site.

#### 5.2.3. Sediment analysis

Sediment samples were oven-dried at 105°C for a minimum of 24 hours and sieved to <2mm particle size fraction. The weights of both the >2mm and <2mm fractions of each sample were recorded before discarding the former fraction. Composite samples for each of the four locations at each site were then prepared using a quartering technique to ensure homogenous sectioning of each <2mm original sample, yielding a total of 120 composite samples. Each composite <2mm sample was then subsampled to a <63µm particle size sample using the same quartering technique. Both fractions were subsequently analysed to determine sediment particle size distribution, total phosphorus (TP), organic carbon (OC), and total nitrogen (TN) concentrations.

Sediment particle size distribution was determined in each sample using the hydrometer method, as described in Carter and Gregorich (1993). The samples were soaked overnight in a solution containing 100 ml of 50 g.L<sup>-1</sup> sodium hexametaphosphate (Fisher Scientific, UK), a dispersing agent, and 300 ml of ultrapure water, and shaken for one hour on a mechanical shaker, to prevent flocculation and ensure dispersion of particles.

The samples were then transferred to graduated cylinders and the volume was brought to 1 L by adding ultrapure water. The suspensions were allowed to equilibrate to room temperature (20 - 25°C) and shaken thoroughly to mix the contents. The density of each suspension was then determined after 40 seconds using a hydrometer (ASTM E100 152H, GH Zeal Ltd.). This reading gives the percentage of sand in each sample according to the equation

Sand (%) = 
$$100 - (R40s - RL)x \frac{100}{oven - dried \ soil \ weight \ (g)}$$

where *R40s* is the reading taken after 40 seconds and *RL* is the reading obtained from a suspension of sodium hexametaphosphate (ThermoFisher Scientific) (5  $g.L^{-1}$ ). The percentage of <63µm particle size sediment (silt+clays) was then determined as the remaining portion of each sample.

Sediment total phosphorus was extracted separately for the <2mm fraction and the < 63µm fraction using a microwave assisted acid digestion method (adapted from the USEPA 3051a method (USEPA, 2007). Using this method, 0.5 ± 0.01 g of well mixed dry sediment was placed in Teflon Xpress vessels (CEM Corporation, USA) and sequentially predigested with 1 ml of concentrated sulphuric acid (98%) for 15 minutes and 10 ml of concentrated nitric acid (70%) for further 15 minutes. The samples were then digested using a Mars 230/60 Microwave Digester (CEM Corporation, USA). Each digestate was then diluted to a 100 ml solution in acid-washed volumetric flasks and left overnight to allow residual sediment particles to settle. Total phosphorus was then measured in each sample using a manual procedure adapted from the colorimetric method described by Murphy and Riley (1962) as modified by Watanabe and Olsen (1965). Calibration standards and quality control standards were prepared using a phosphate certified standard solution (1000 mgL<sup>-1</sup>; Inorganic<sup>TM</sup> Ventures) and a stock solution prepared using an analytical grade KH<sub>2</sub>PO<sub>4</sub> salt (1000 mgL<sup>-1</sup>; ThermoFisher Scientific), respectively. The limit of detection for this analysis was calculated as 10 µg P.g<sup>-1</sup>.

Sediment organic carbon and total nitrogen were determined simultaneously using an Elementar El Vario Cube elemental analyser (Elementar Analysensysteme GmbH, Hanau, Germany), following removal of inorganic carbon by exposing dry sediments to concentrated hydrochloric acid (37%) fumes (Harris et al., 2001). The instrument was calibrated against a

standard (acetanilide) (Merck, Darmstadt, Germany) with known concentrations of carbon (71.09%), nitrogen (10.36%) and hydrogen (6.71%). Drift calibration standards were also included every ten samples to account for any drift over each run. The limit of detection for this analysis was estimated as < 40 ppm (0.004%) for both TN and OC.

#### 5.2.4. Estimation of stream nutrient loadings

To estimate the mass sediment per metre squared for each site, the average mass of <2mm sediment contained in each composite sample was extrapolated up to a square meter by multiplying by 238.9 (number of Plexiglas corer areas in 1 m<sup>2</sup>). The mass of the <63µm fraction was then calculated using the percentage of silt+clay in each composite sample (obtained in the particle size distribution analysis). To give a more accurate estimate of sediment cover, the mass of each fraction was then corrected for the heterogeneous composition of the substrate at each location using estimations of the coverage of fine sediment at each site (% of substrate). These were visually assessed by O'Sullivan et al. (2019) during site characterisation for the determination of Riparian Habitat Index values. Nutrient bed sediment nutrient concentrations by the estimated mass of sediment (g.m<sup>-2</sup>). Since O'Sullivan et al. (2019) conducted their assessment of percentage of fine sediment the late grazing season, nutrient loadings were estimated for this sampling time only.

#### 5.2.5. Statistical analysis

All analyses were carried out using R software (version 3.5.1; R Core Team, 2016), and the packages *nlme* (Pinheiro et al., 2019) and *mgcv* (Wood, 2015). Differences in sediment nutrient concentrations between the <63µm and the <2mm particle size fractions were assessed individually for each nutrient using generalised least squares (GLS) analysis. A

compound symmetry correlation structure was included to account for correlation between observations obtained from the same composite samples, and a variance structure was included for the factor Catchment to account for heterogeneity in the model residuals (Zuur et al., 2009). Potential correlations between the concentrations of all nutrients in both fractions were assessed using Spearman rank correlation test. Prior to the assessment of potential differences in sediment nutrient concentrations due to direct cattle access, the independence of the 15 sites was confirmed through spatial correlation analysis (Zuur et al., 2009).

The impact of direct cattle access on sediment nutrients in each of the sediment fractions was assessed separately for each nutrient using linear mixed effects modelling. Linear mixed models allow for the inclusion of variables as random components of the linear models and are useful for nested or hierarchical designs (Zuur et al., 2009). The analysis included factors Time (two levels: EG (early grazing season), LG (late grazing season), Location (four levels: US, CAS, INT, DS) and an interaction between these factors (Time x Location) as fixed terms, with the factor Catchment modelled as a random intercept. This approach allows the intercept of the model to change per catchment, and induces a correlation structure for all observations from the same catchment (Zuur et al., 2009). The factor Catchment was included as a random term in all models except for the analysis for OC concentrations in the <2mm particle size fraction, where the addition of a random intercept did not improve the model in comparison to a GLS model with only fixed terms.

The analysis was performed on log-transformed data for OC and TN in the <2mm fraction to account for heterogeneity problems that were not appropriately resolved using various variance structures in previous attempts. Additionally, different variance structures were added to the models to eliminate issues caused by heterogeneity of variances. The addition of a random intercept and of a compound symmetry correlation structure for factor Site was also tested but did not improve the models.

Finally, the effect of estimated cattle density (ECD) (described in Chapter 3) for each sampled site on sediment nutrient concentrations as well as on nutrient loadings was assessed separately for each nutrient using generalised additive modelling, which allows for the modelling of non-linear relationships between variables. In this analysis, a smoother for ECD and the factor Location were included, as well as an interaction between the two variables. Variance structures were applied as necessary. Additionally, GAM analysis was used to assess the relationship between general site degradation and nutrient sediment levels. The analysis was conducted for all sites separately for each sampling time, using sediment nutrient concentrations, nutrient loadings, and RHI scores for the US and INT subsites. Since O'Sullivan et al. (2019) conducted the site assessment in early autumn, the two sites that were later replaced in the BK catchment were not included in the statistical analysis of the data collected in early grazing season. All models were selected validated and optimized according to the protocols described in Zuur et al. (2009). The optimum model was that with the lowest AIC (Akaike Information Criterion) value.

#### 5.3. Results

#### 5.3.1. Particle size distribution of the samples

The particle size distribution of streambed sediment at the study sites is presented in Table 5.3. The relative proportions of sand and silt+clay varied across the sites. However, the sediments were dominated by sand size particles across all catchments and sites, with average silt+clay fraction proportions of  $9.38\% \pm 1.17\%$  for BW,  $5.42\% \pm 0.94\%$  for DG,  $10.30\% \pm 2.19\%$  for BK,  $5.82\% \pm 0.86\%$  for CM and  $9.23\% \pm 2.02\%$  for MT (mean  $\pm$  S.E.). Although the DG and CM catchments had lower average values for the silt+clay fraction, differences between the five catchments were not statistically significant (F-value = 0.654, *p* 

= 0.637). There was also no statistically significant difference for the particle size distribution between early grazing season and late grazing season (F-value = 0.137, p = 0.713). The average percentage for the silt+clay fraction at the different locations were 10.35% ± 2.07% for US, 6.93% ± 1.64% for DS sites, 5.70% ± 0.75% for CAS and 9.34% ± 0.92% at the interface areas (INT) between the stream and the field. Particle size distribution, however, did vary significantly with Location (F-value = 7.019, p < 0.001). Pairwise tests revealed that the silt+clay proportion was significantly lower at CAS areas than at INT areas (t-ratio = - 4.197, p < 0.001).

#### 5.3.2. Nutrient concentrations in the <63 µm and <2mm particle size sediment fractions

Total phosphorus concentrations in the <63  $\mu$ m fraction ranged from 137 to 2191  $\mu$ g.g dry wt<sup>-1</sup> (median value of 683.5  $\mu$ g.g dry wt<sup>-1</sup>), organic carbon concentrations ranged from 2.90 to 85.05 mg.g dry wt<sup>-1</sup> (median value of 25.99 mg.g dry wt<sup>-1</sup>) and total nitrogen concentrations ranged from 0.39 to 7.54 mg.g dry wt<sup>-1</sup> (median value of 2.73 mg.g dry wt<sup>-1</sup>). In the <2mm particle size fraction (which it should be noted included the <63 $\mu$ m fraction) concentrations ranged from 98 to 1540  $\mu$ g.g dry wt<sup>-1</sup> for TP (median value of 464  $\mu$ g.g dry wt<sup>-1</sup>), 2.87 to 57.17 mg.g dry wt<sup>-1</sup> for OC (median value of 15.12 mg.g dry wt<sup>-1</sup>) and 0.33 to 5.11 mg.g dry wt<sup>-1</sup> for TN (median value of 1.51 mg.g dry wt<sup>-1</sup>).

The <63  $\mu$ m sediment fraction had significantly higher concentrations of all three nutrients than the <2mm fraction (Fig. 5.1.; Table 5.4). In the former fraction, consistent patterns were observed when comparing across the five study catchments, with sediments in the CM and the MT catchments having the highest nutrient concentrations, and sediments from the BW catchment generally having the lowest concentrations. This pattern of difference between the five catchments was not apparent in the <2mm fraction (Fig. 5.1).

TN and OC concentrations were significantly correlated in both the <63  $\mu$ m and <2mm fraction (p < 0.001, r = 0.93; p < 0.001, r = 0.81, respectively) (Fig. 5.2), indicating that most TN present in the sediment was incorporated in organic material. TP concentrations were also correlated with both TN and OC for the <63 $\mu$ m fraction but with a lower Peason's coefficient value (p < 0.001, r = 0.54; p < 0.001, r = 0.49), but were not correlated for the <2mm fraction (Fig. 5.2).

	Site				%Estimated	Sand	Silt+clay	
Catchment		Location	%Sand %	%Silt+clay	fine sediment	(a.m <sup>-2</sup> )	(a m <sup>-2</sup> )	
					coverage	(9)	(9)	
		US	93	8	10	1196.8	97.0	
	1	CAS	83	18	52	6934.4	1470.9	
	I	INT	73	28	89	11014.1	4177.8	
		DS	93	8	58	10173.6	824.9	
		US	58	43	48	4233.5	3129.1	
PK	2	CAS	95	5	59	17248.7	907.8	
DK	Z	INT	93	8	50	11919.7	966.5	
		DS	95	5	48	7118.1	374.6	
		US	95	5	29	3802.2	200.1	
	2	CAS	97	3	70	16574.6	512.6	
	3	INT	88	13	92	12434.0	1776.3	
		DS	99	1	42	5209.3	77.4	
	A	US	83	18	4	248.0	52.6	
		CAS	95	5	25	7712.2	405.9	
		INT	85	15	80	15937.0	2812.4	
		DS	100	0	2	158.7	0.0	
		US	90	10	37	5438.9	604.3	
	В	CAS	95	5	9	1084.2	57.1	
DVV		INT	93	8	20	1338.9	108.6	
		DS	94	6	7	1113.6	67.9	
		US	70	30	<1	41.8	17.9	
	C	CAS	95	5	<1	19.5	1.0	
	C	INT	93	8	74	8551.0	693.3	
		DS	98	3	<1	36.2	0.9	
		US	95	5	25	1798.8	94.7	
	1	CAS	90	10	64	8085.5	898.4	
CM	I	INT	89	11	55	2978.9	369.2	
CIVI		DS	88	12	100	2821.6	369.2	
	C	US	89	11	62	3031.7	360.3	
	Z	2	CAS	95	5	24	4299.8	226.3

**Table 5.3.** Particle size distribution and coverage of sand and silt+clay particles at the study sites.

### Table 5.3 (continued).

				%Silt+Clay	%Estimated	Sand	Siltualay	
Catchment	Site	Location	%Sand		fine sediment	(a m <sup>-2</sup> )	$(a m^{-2})$	
					coverage	(g.m.)	(g.m.)	
	2	INT	93	8	85	8232.8	667.5	
	2	DS	90	10	52	7262.7	807.0	
CM		US	99	1	15	1870.4	25.5	
CM	З	CAS	95	5	55	8142.6	428.6	
	5	INT	93	8	90	12641.8	1025.0	
		DS	98	3	54	6590.2	169.0	
	-	US	95	5	15	1856.1	97.7	
	Λ	CAS	97	3	45	4206.6	128.9	
	~	INT	94	6	50	5472.5	364.8	
		DS	97	3	30	3656.3	113.1	
	В	US	76	24	17	3595.2	1119.8	
		CAS	98	3	53	7112.6	182.4	
DG		INT	94	6	60	7151.7	476.8	
		DS	98	2	37	5038.4	122.7	
	С	US	97	3	32	8233.6	232.8	
		CAS	98	2	42	4231.3	82.3	
		INT	93	7	NA	NA	NA	
		DS	98	2	7	984.5	22.6	
		US	88	13	6	553.4	79.1	
		CAS	95	5	8	673.0	35.4	
	I	INT	93	8	89	7635.4	619.1	
		DS	50	50	8	1216.3	1216.3	
		US	98	3	14	235.6	6.0	
NAT	0	CAS	96	4	19	5030.2	196.0	
IVI I	2	INT	98	3	20	1976.5	50.7	
		DS	88	13	<1	47.8	6.8	
		US	96	4	11	2045.2	79.7	
	6	CAS	98	3	31	5340.9	136.9	
	3	INT	95	5	68	7544.9	397.1	
			DS	96	4	8	2004.4	78.1



sediment concentrations in the <63  $\mu$ m and <2 mm sediment fractions in five study the catchments, in early grazing season (EG)

and late grazing season (LG) (n = 12).
	TP		٦	TN		C		
	(µg.g	dr wt⁻¹)	(mg.g	dr wt⁻¹)	(mg.g dr wt <sup>-1</sup> )		C:N	atio
	<2mm	<63µm	<2 <i>mm</i>	<63µm	<2 <i>mm</i>	<63µm	<2mm	<63 µm
Early grazing seas	son							
Catchment								
BW	317±8	409±15	1.575±0.110	2.150±0.141	16.973±3.226	21.820±1.491	10.6±1.9	10.2±0.4
DG	511±29	596±52	1.522±0.233	2.369±0.186	17.328±2.704	22.341±1.621	11.8±1.1	9.5±0.3
BK	342±29	603 <b>±</b> 38	2.079±0.169	2.555±0.199	19.820±1.992	21.333±2.211	9.5±0.5	8.2±0.5
СМ	524±19	844±24	1.936±0.091	3.318±0.148	16.496±1.703	28.366±1.856	8.4±0.7	8.5±0.3
MT	666 ± 84	1233±142	2.041±0.337	3.819±0.299	21.837±4.719	40.199±3.943	9.9±0.6	10.4±0.3
Location								
US	440±38	663±68	1.843±0.192	2.699±0.258	17.179±2.412	25.398±3.320	9.1±0.6	9.2±0.5
CAS	435±27	691±66	1.848±0.219	2.697±0.234	18.327±3.269	25.749±3.173	9.5±0.8	9.4±0.4
INT	561±80	903±152	1.906±0.221	3.083±0.271	21.514±3.336	28.031±2.590	11.3±1.5	9.2±0.3
DS	453±34	691±63	1.726±0.131	2.889±0.199	16.943±1.428	28.069±2.115	10.2±0.7	9.7±0.3
Average for EG	472±25	737±48	1.831±0.095	2.842±0.120	18.491±1.348	26.812±1.393	10.0±0.5	9.4±0.2

**Table 5.4.** Sediment nutrient concentrations in the <2mm and <63 $\mu$ m particle size fraction for each catchment for each sampling time (n = 60), location sampled (n = 15) and catchment (n = 12) (mean ± S.E.).

# Table 5.4 (continued)

ТР		TP	٦	٢N	0	C	C <sub>1</sub> N <sub>rotio</sub>	
	(µg.g	dr wt <sup>-1</sup> )	(mg.g	dr wt⁻¹)	(mg.g dr wt <sup>-1</sup> )		CIN	ratio
	<2 <i>mm</i>	<63µm	<2 <i>mm</i>	<63µm	<2 <i>mm</i>	<63µm	<2 <i>mm</i>	<63µm
Late grazing seas	son							
Catchment								
BW	359±33	496±48	1.184±0.173	2.519±0.279	12.770±2.398	29.042±5.462	10.2±0.6	11.1±0.8
DG	681±83	682±68	0.972±0.121	2.487±0.252	15.450±2.196	26.830±2.526	16.3±1.8	10.9±0.4
BK	310±22	735±34	1.582±0.149	2.048±0.203	17.391±2.480	19.714±2.152	10.5±0.8	9.4±0.4
СМ	669±64	895±56	1.658±0.300	3.744±0.418	20.031±3.791	39.491±5.017	12.3±0.8	10.3±0.3
MT	745 ±82	1195±135	1.645±0.424	3.551±0.533	19.671±5.194	39.313±6.074	11.4±0.5	10.7±0.4
Location								
US	514 ±72	776±96	1.310±0.173	2.691±0.384	15.823±2.670	29.394±4.628	12.3±1.8	10.5±0.4
CAS	547 ±67	798±98	1.128±0.104	2.789±0.245	13.552±1.504	28.919±3.063	12.0±0.7	10.2±0.3
INT	617 ±79	892±106	1.798±0.297	3.401±0.453	21.222±3.393	37.664±5.855	11.9±0.6	10.9±0.6
DS	533 ±72	737±58	1.396±0.297	2.598±0.294	17.653±3.931	27.535±3.413	12.3±0.8	10.4±0.3
Average for LG	553 <b>±</b> 36	801±45	1.408±0.118	2.870±0.177	17.063±1.513	30.878±2.195	12.1±0.5	10.5±0.2



**Fig. 5.2.** Scatterplots showing the relationships between nutrients in the <63µm (top row) and the <2mm (bottom row) particle size fractions.

#### 5.3.3. Effects of direct cattle access on stream sediment nutrient levels

For the <63µm fraction, all nutrient concentrations were generally higher at INT sites than at the remaining Locations (US, CAS and DS) (Fig. 5.3), however the factor Location was only statistically significant for TN (Table 5.5). Pairwise tests revealed that overall TN concentrations at INT areas were significantly higher than at US areas (t-ratio = 2.734, p = 0.036). There was also a significant increase in sediment TP concentrations in late grazing season compared to early grazing season, but no significant differences were observed for OC and TN between the two sampling times (Table 5.5). No significant interactions between factors Time and Location were found (Table 5.5)

For the <2mm fraction, no significant differences were found between the four different levels of Location for any of the three nutrients (Fig. 5.4, Table 5.6). However, *p* values for this comparison were on the boundary of significance for all three nutrients (Table 5.6). In contrast with the <63 $\mu$ m fraction, TP concentrations in this fraction did not change significantly between the early grazing season and the late grazing season. Conversely, TN sediment concentrations decreased significantly from early grazing season to late grazing season (Table 5.6). Organic C concentrations did not vary significantly between sampling times. Similarly to what had been observed for <63 $\mu$ m fraction, there were no significant interactions between the factors Location and Time for any of the nutrients (Table 5.6).



**Fig.5.3.** Boxplots of sediment concentrations in the  $<63\mu$ m particle size sediment fractions at each of the locations sampled at the cattle access sites, in early grazing season (EG) and late grazing season (LG) (n = 15).



Fig.5.4. Boxplots of sediment concentrations in the <2mm particle size sediment fractions at each of the locations sampled at the cattle access sites, in early grazing season (EG) and late grazing season (LG) (n = 15).

**Table 5.5.** Statistical parameters for the effect of sediment fraction (GLS analysis) and for the effect of Location, Time and interaction between the two factors (mixed effects modelling) on sediment nutrient concentrations in the <  $63\mu$ m fraction. Significant effects are shown in bold.

Analysis	ТІ	2	Т	Ν	OC	
Andiysis	F-value	р	F-value	p	F-value	p
Fraction	158.138	< 0.001	118.412	< 0.001	48.157	< 0.001
Location	1 997	0 119	2 722	0 048	1 857	0 141
(<63µm)	1.557	0.115		0.040	1.007	0.141
Time	6 221	0.014	1 007	0 1 1 0	2 700	0 102
(<63µm)	0.231	0.014	1.997	0.119	2.700	0.103
Location x						
Time	0.469	0.704	0.405	0.750	0.162	0.922
(<63µm)						

**Table 5.6.** Statistical parameters for the effect of Location, Time and interaction between the two factors (mixed effects modelling) on sediment nutrient concentrations in the <2mm fraction. Significant effects are shown in bold.

Analysis	TF	)	т	'N	0	С
	F-value	p	F-value	р	F-value	p
Location (<2mm)	2.673	0.051	2.392	0.073	2.525	0.061
Time (<2mm)	1.956	1.645	22.501	<0.001	1.298	0.257
Location x Time ( <2mm)	0.067	0.977	1.016	0.389	0.126	0.945

#### 5.3.4. Sediment nutrient loads

Nutrient loads in the stream bed sediment based on the particle size distribution of the sediment at the study sites and the fine sediment coverage estimates are presented in Table 5.7 (for late grazing season only). For TP, TN and OC, sediment nutrient loads in the silt+clay fraction were significantly higher at INT sites than at the three remaining locations (TP: F-value = 7.590, p < 0.001; TN: F-value = 6.629, p = 0.001; OC: F-value = 6.630, p < 0.001) (Fig 5.5; Table 5.8). In the <2mm fraction, sediment loads of all three nutrients varied significantly with Location (TP: F-value = 7.482, p < 0.001; TN: F-value = 7.226, p < 0.001; OC: F-value = 8.214, p < 0.001). Pairwise comparisons showed that sediment nutrient loads were higher at INT sites than at US and DS sites, but not significantly higher than sediment loads at CAS (Fig.5.5; Table 5.8).



**Fig. 5.5**. Nutrient loads in the silt+clay fraction (top row) and <2mm fraction (bottom row) in late grazing season at each study location.

	TP (mg.m <sup>-2</sup> )		TN (mg	g.m⁻²)	OC (mg.m <sup>-2</sup> )		
	<2 <i>mm</i>	<63µm	<2 <i>mm</i>	<6 µm	<2 <i>mm</i>	<63µm	
Catchment <sup>1</sup>							
BW	1449.8 ± 610.8	163.6 ± 84.9	6078.1 ± 2937.7	1066.4 ± 575.6	67541.5 ± 33039.7	11060.0 ± 5681.6	
DG	3322.5 ± 664.3	162.0 ± 44.5	4929.0 ± 800.7	586.2 ± 159.8	75935.1 ± 12789.3	6079.3 ± 1488.0	
BK	3222.6 ± 524.1	830.1 ± 230.8	16494.6 ± 3222.9	1924.5 ± 486.0	188272.0 ± 42421.4	18016.2 ± 4844.2	
СМ	3828.0 ± 614.7	383.9 ± 71.5	9710.5 ± 1934.1	1600.8 ± 387.9	116942.0 ± 24487.5	16361.6 ± 3964.7	
МТ	2787.9 ± 1066.2	268.7 ± 109.6	5947.0 ± 3393.8	714.7 ± 374.4	66461.3 ± 37397.3	7611.2 ± 4075.4	
Location <sup>2</sup>							
US	1606.7 ± 424.2	266.2 ± 1 36.5	3490.8 ± 734.9	597.7 ± 178.9	39244.6 ± 7815.2	5718.5 ± 1539.2	
CAS	3131.5 ± 534.6	262.3 ± 68.9	8553.2 ± 2430.5	852.8 ± 186.5	98852.4 ± 28522.7	8053.9 ± 1592.8	
INT	5279.8 ± 818.0	755.1 ± 174.9	17739.7 ± 3347.6	2883.9 ± 568.4	206801.0 ± 36286.0	29552.1 ± 5623.4	
DS	1801.1 ± 378.3	202.6 ± 62.1	5597.7 ± 1657.3	532.9 ± 155.0	75948.0 ± 26348.4	5543.0 ± 1704.3	

 Table 5.7. Sediment nutrient loads in each study catchment and at each location in late grazing season (mean ± S.E.).

<sup>1</sup>n = 12; <sup>2</sup>n = 15

**Table 5.8.** Pairwise comparisons within factor Location for sediment nutrient loadings in late grazing season. Significant factors are shown in bold.

Comparison	ТР		P	T	Ν	0	OC	
	df	t-ratio	p	t-ratio	p	t-ratio	p	
Silt+clay fraction								
CAS – DS	51	0.160	0.999	1.866	0.255	1.505	0.442	
CAS – INT	51	-4.019	0.001	-3.591	0.004	-3.852	0.002	
CAS – US	51	-0.456	0.968	1.104	0.688	1.131	0.672	
DS – INT	51	-4.171	0.001	-4.200	0.001	-4.293	<0.001	
DS – US	51	-0.616	0.926	-0.301	0.990	-0.083	0.100	
INT - US	51	3.585	0.004	3.940	0.001	4.163	0.001	
<2mm fraction								
CAS – DS	51	2.495	0.073	1.201	0.629	0.604	0.930	
CAS – INT	51	-2.281	0.116	-2.354	0.099	-2.366	0.097	
CAS – US	51	2.421	0.086	2.281	0.116	2.061	0.180	
DS – INT	51	-4.083	0.001	-3.412	0.007	-2.497	0.024	
DS – US	51	0.413	0.976	1.363	0.528	1.363	0.528	
INT - US	51	4.015	0.001	4.198	0.001	4.525	<0.001	

The generalised additive modelling (GAM) analysis revealed a highly significant effect of estimated cattle density (ECD) on sediment concentrations for all three nutrients in the <63µm particle size fraction across the 15 sites (Fig. 5.6, Table 5.9). ECD explained 23%, 28% and 41% of the variance for OC, TN and TP, respectively. When similar analysis was undertaken for the <2mm particle size fraction, ECD had a significant effect on TP only, explaining 23% of the variance (Table 5.9). Interestingly, a model which included ECD and factor Location revealed a significant effect of factor Location for TP concentrations in the <63µm fraction. This model indicated a significant interaction there between INT and factor ECD when different variance structures where applied. However this interaction became non-significant at the 5% level when applying the variance structure that gave the lowest AIC value (p = 0.079; Table 5.9). Figure 5.7 shows the smoother for the effect of ECD on sediment TP concentrations in this fraction at US areas (set as a reference), while the remaining three plots show the additional change in TP concentrations explained by each of the other levels of factor Location. TP concentrations show a further increase at INT areas compared to US areas, whereas no further changes were observed for CAS and DS sites. The same analysis conducted on nutrient loadings at each sampled locations did not reveal any significant effect of ECD on these parameters.

In contrast, no effect of the intensity of impact at cattle access sites, as expressed by the RHI scores attributed by O'Sullivan et al. (2019) were found in this study for bed sediment nutrient concentrations or loads.



**Fig.5.6.** Top row: fine sediment (<63µm) total phosphorus (TP) organic carbon (OC) and total nitrogen (TN) concentrations versus estimated cattle density (ECD; animals.ha<sup>-1</sup>) (including data for the early and late grazing season and for upstream, interface, cattle access points, and downstream locations); bottom row: generalised additive models using these data. The central solid line is the smoother, the grey area is the 95% confidence bands, and the Y axis units are the scaled smoother (s) for ECD with estimated degrees of freedom (efd).



**Fig 5.7.** A. smoother for the effect of estimated cattle density (ECD) on TP for upstream (US) sites (reference level), and difference smooths reflecting the estimated differences between the upstream sites and the CAP sites (CAS, top, right), interface sites (INT, bottom, left), and downstream (DS, bottom, right) respectively. The central solid line is the smoother, the grey area is the 95% confidence bands, and the Y axis units are the scaled smoother (s) with estimated degrees of freedom (efd).

**Table 5.9.** Generalised additive model parameters for the effect of cattle density on sediment nutrient concentrations (n = 120) (ECD = estimated cattle density; edf = estimated degrees of freedom). Note that the optimum model for TP <63 $\mu$ m used an ordered factor approach, and therefore includes a parametric term for the factor Location, where upstream (US) was set as the reference level. Only values for significant effects are shown except for ECD\*Location (INT) effect.

Fraction	Nutrient	Driver	edf	Statistics	p	R² adj
<2mm	TP	ECD	2.65	F = 16.70	<0.0001	0.23
<63µm	OC	ECD	3.23	F = 10.81	<0.0001	0.23
<63µm	TN	ECD	3.21	F = 14.81	<0.0001	0.28
<63µm	TP	ECD	2.97	F = 32.32	<0.0001	0.41
		Location (INT)		t = 2.51	0.0136	
		ECD* Location (INT)	2.56	F = 2.15	0.079	

## 5.4. Discussion

The practice of allowing cattle access to watercourses has been shown to negatively impact freshwater systems in a number of studies (Trimble, 1994; Vidon et al., 2008; Herbst et al., 2012; Terry et al., 2014); however, its effects on sediment nutrient levels have been largely overlooked in literature. Furthermore, while high stocking intensity has been linked to exacerbated diffuse nutrient pollution (Chalar et al., 2017), it has rarely been considered in the context of cattle access to watercourses and indeed in relation to contamination of streambed sediments. In the present study, although direct cattle access did not cause a measurable accumulation of phosphorus or organic matter at a local level, with only TN varying significantly with Location within cattle access sites, ECD was a significant driver of stream sediment nutrient concentrations in the  $<63\mu$ m particle size fraction. For TP sediment concentrations only, this effect was accentuated at the interface locations at the cattle access site.

#### 5.4.1. Enrichment of the <63µm fraction

Overall, the nutrient concentrations measured in this study fell within the range of what has been observed by other authors that have examined streambed sediment, stream bank and soils nutrient concentrations in areas impacted by agriculture (Table 5.2). The current study also found a strong nutrient enrichment effect for the smaller <63 $\mu$ m (silt+clay) proportion of the sediment in comparison to the <2 mm particle size fraction. This was expected as this fraction is considered to be the most chemically reactive due to its higher surface to volume ratio (Sharpley et al., 2013), and would be particularly apparent for phosphorus, due to its affinity for aluminium-silicate minerals (Fox et al., 2016). This enrichment likely played a role in the overall pattern that was observed across the five study catchments that was not apparent in the <2mm fraction and that was consistent for all three nutrients. More importantly the exploration of the effect of cattle numbers on streambed sediment nutrient levels showed that this pattern was strongly related to cattle density for the sites, as shown by the GAM analysis. The highest nutrient concentrations were found for the five study catchments were in the CM and MT catchments, which are characterised by intensive agriculture, while the lowest nutrient concentrations were the BW catchment, which has more extensive agriculture. To the knowledge of the authors, such relationship between cattle agriculture intensity and bed sediment nutrient concentrations has not been previously reported.

#### 5.4.2. <u>Impacts of cattle access on sediment nutrient concentrations</u>

Sediment nutrient concentrations did not differ significantly at stream reaches used by cattle compared to areas not used by cattle in the <2mm size fraction whereas in the <63 $\mu$ m particle size fraction, only TN differed significantly with Location, with concentrations at INT areas being significantly higher than at US areas. However, *p* values obtained for all three nutrients in the <2mm particle size fraction were on the boundary of the significance limit, which suggests that there is a weak effect of cattle access on sediment nutrient levels in this fraction.

Increased sediment nutrient concentrations at access sites would result from in-stream defecation and urination, addition of faecal matter attached to the animals' legs and washing-off of faecal material and urine deposited nearby the stream. Bond et al. (2014) estimated that cattle faeces are mostly composed of water (89.4%), containing on average 0.79% nitrogen and 0.43%% phosphorus by wet mass. The authors suggested that cattle faeces would be particularly soluble and susceptible to transport and dispersion within the stream following deposition into stream waters. Furthermore, within the stream, cattle faeces can be quickly colonized and utilized by benthic invertebrate taxa (Mesa et al., 2016). Indeed, a study carried out at the same sites sampled in the present work based on isotope analysis of organic carbon sources indicated that invertebrate taxa may consume cattle faeces at cattle access sites (Ó hUallacháin et al., 2020). Cattle urine contains mainly N in the form of urea, which hydrolises rapidly upon excretion (Dijkstra et al., 2013). Following deposition, urine is subject to dilution and transport within the stream. Both major dissolved forms of N, NO<sub>3</sub><sup>-</sup> and NH<sub>4</sub><sup>+</sup>, can be readily absorbed by microoorganisms, periphyton and macrophytes or can be transported downstream before it becomes incorporated in the bed sediment (Kronvang et al., 1999; Butturini et al, 2000). This may help to explain the observation that OC and TP did not accumulate significantly in sediment at areas used by cattle despite frequent in-stream defecation and urination. Furthermore, the systematic

disturbance of the sediments by cattle while using the stream would cause its resuspension and subsequent transport downstream, particularly for the smaller particles (Naden et al, 2010; Sharpley et al., 2013), which can also prevent the accumulation of nutrient-enriched sediments at access sites.

When looking at nutrient levels at cattle access sites in terms of loads, all three nutrients where present in significantly higher loads at interface areas. This is related to the fact these areas had higher fine sediment coverage than the remaining areas. These results indicate that these interface areas that are created with cattle repeatedly accesses a watercourse, and where animals likely congregate to drink, and which are rich in nutrients and fine sediment, likely to act as critical source areas of these contaminants to the watercourse (Thompson et al., 2013).

#### 5.4.3. Impacts of cattle density on sediment nutrient concentrations

One of the most important results from this study was that there was a significant effect of estimated cattle density on streambed sediment concentrations of all three nutrients in the <63µm fraction, and for TP in the <2mm fraction. For both OC and TN, there seemed to be a threshold at an estimated 1.6 animals per ha, after which sediment nutrient concentrations increased sharply (Fig. 5.6). For TP, this threshold seemed to be lower at 1.2 animals per ha (Fig. 5.6). In Europe, dairy cattle densities are highest in Western Germany, Bavaria, the Netherlands, Northern Italy, Ireland and the Brittany region in France, whereas beef cattle densities are highest in Scotland and North England, Ireland and Central France (Neumann et al., 2009). An estimation of cattle density in ten European countries with strong cattle-based agriculture is shown in Table 5.10. According to this estimation, only two of these EU Member-States have cattle densities lower than 1.6 cattle.ha<sup>-1</sup> (Table 5.9). Thus, the findings of the current study have implications in the context of livestock agriculture management in Europe.

**Table 5.10.** Estimation of cattle density in EU Member States. Cattle density was calculated by dividing an estimation of the total number of cattle by the area devoted to permanent grassland in each country.

	Total	% EU	Number of	Permanent	
Country	bovine	bovine		grassland	Cattle.ha <sup>-1</sup>
	population <sup>1</sup>	population <sup>1</sup>	Cattle	(ha)⁵	
France		0.212	18444000	9593990	1.92
Germany		0.137	11919000	4713400	2.53
UK		0.11	9570000	11277000	0.85
Ireland		0.075	6525000	4064210	1.61
Spain	9700000	0.074	6438000	7037370	0.91
Italy	87000000	0.072	6264000	3659630	1.71
Poland		0.071	6177000	3149870	1.96
The Netherlands		-	3721000 <sup>2</sup>	763790	4.87
Luxembourg		-	196127 <sup>3</sup>	67710	2.90
Belgium		-	2500000 <sup>4</sup>	1356080	1.84

<sup>1</sup>Information retrieved from the most recent EU Agriculture, Forestry and Fishery Statistics report (Eurostat, 2019). <sup>2</sup>Data retrieved from Statistics Netherlands (www.cbs.nl). <sup>3</sup>Data retrieved from https://statistiques.public.lu. <sup>4</sup>Data retrieved from https://www6.inrae.fr/sustainbeef/Publications/Beef-production-in-the-EU/Beef-production-in-Belgium. <sup>5</sup>Data retrieved from https://ec.europa.eu/eurostat/web/products-datasets/-/tag00025.

In addition to the effect of cattle density on sediment nutrient concentrations in the silt+clay fraction, there was also an increase in TP concentrations at interface (INT) areas of the cattle access sites with estimated cattle density, indicating a possible effect of direct cattle access at these more impacted areas and cattle density on stream bed sediment phosphorus reservoirs. The fact that this effect was only apparent for TP again illustrates the higher chemical reactivity of phosphorus with sediments particles (McGechan et al., 2005) in comparison to other nutrients. This impact of diffuse contamination likely reflects the nature of the soils at the study sites, which are predominantly poorly drained, and thus would favour

the transport of particulate forms of P to the aquatic streams in surface pathways (Deakin et al., 2016).

## 5.4.4. Variation in sediment nutrient concentrations between sampling times

In general, sediment nutrient concentrations might be expected to have increased in late grazing season compared to early grazing season as a result of cattle activity. However, there was only a marginal increase in TP in the overall sediment concentrations in the silt+clays fraction between early and late season, whereas TN significantly decreased in the late grazing season in the <2mm particle size fraction. A closer examination of the sediment nutrient concentrations at each sampled location within access sites showed that the general increase in TP levels in late grazing season was mainly seen at interface areas. Conversely, the decrease in TN sediment concentrations in late grazing season occurred at all locations within access sites, but remained high in comparison at interface areas. The higher TN concentrations in early grazing season may have been associated with slurry spreading, which has been reported to occur mostly in the spring months (roughly from January to end of April) (Hennessy et al, 2011). The higher TN concentrations at INT sites in late grazing season suggest that cattle access to watercourses influences TN sediment levels at access sites. The fact that this was not reflected in the statistical analysis of the data could be related to the small size of the sample when examining interactions between sampling time and location.

# 5.5. Implications

The results of this study highlight the role that cattle-based agriculture plays in streambed nutrient contamination. Although diffuse pollution was not specifically monitored in this study,

results indicate that streambed nutrient contamination is influenced by diffuse pathways of pollution. Cattle-based agriculture can result in increased nutrient losses from agricultural fields via the following mechanisms: 1) wash-out of nutrients from application of animal slurries, usually associated with intensive livestock production, on grazing fields, 2) wash-out of nutrients from inorganic fertilisers applied to pastoral areas, and 3) nutrient loss associated with runoff from fields where cattle faeces and urine have been deposited. Diffuse nutrient transfers from grazing fields are primarily controlled by rainfall runoff events and can be separated in residual transfers, whereby nutrient losses are derived from nonutilised nutrients stored in agricultural soils, and incidental transfers, whereby nutrients are lost from recently applied slurries or fertilisers (Shore et al., 2017). Cattle can also contribute to diffuse pollution indirectly by causing changes in soil structure, particularly in wet soils, which can promote surface runoff generation (Bilotta et al., 2007). The mechanic action of cattle hooves on soils can cause soil treading, pugging (i.e. livestock treading on wet soft soil creating deep hoof imprints) and poaching, resulting in reduced soil porosity and increased soil bulk density, and associated reduced water infiltration capacity (Bilotta et al., 2007). This deformation of agricultural soils has been shown to increase with stocking density (Willatt and Pullar, 1983; Mulholland and Fullen, 1991), due to the cumulative impact of the animals on the soils, and to the lower protective vegetation cover that is generally available at higher stocking rates (Bilotta et al., 2007).

In addition to this larger scale, diffuse pollution caused by cattle-based agriculture, the findings of this study suggest that this effect can be further exacerbated by direct cattle access to watercourses as cattle creates nutrient-rich, erodible areas in streams, that can act as critical source areas of pollution (Thompson et al., 2013). This contamination will negatively impact water quality, while also contributing to potential legacy phosphorus issues (Sharpley et al., 2013). This can hinder the effect of mitigation measures and water quality protection practices (Jarvie et al., 2013), with potential implications for the achievement of the Water Framework Directive requirement of achieving at least good ecological status of

all waterbodies by 2027 (Government of Ireland, 2018). The results of this study are particularly important in the context of the current agricultural and agri-environmental policies, such as Green Low Carbon Agri-Environment Scheme (GLAS) which contains measures to restrict cattle access to watercourses. However, adherence to this scheme and the implementation of such measures is voluntary, and limited by financial and other practical constrains. Further research should therefore focus on the cost-effectiveness of cattle exclusion measures on reducing excess nutrient inputs to waterbodies, and measures aimed at engaging the farmer community on water quality protection are recommended.

6. A near-real time assessment of the effects of cattle in-stream activity on water physicochemical and microbial parameters

Chapter 6. A near-real time assessment of the effects of cattle in-stream activity on water physicochemical and microbial parameters

# 6.1. Introduction

Cattle seek watercourses and riparian areas for drinking water, shade and palatable vegetation (McKergow et al., 2003; James et al., 2007;), and may also use them as crossing points between grazing fields (Davies–Colley et al., 2004; O'Callaghan et al., 2018). It has been reported that where access to watercourses is unrestricted, cattle tend to preferentially congregate in the riparian area (e.g. James et al, 2007; Haan et al., 2010; Bond et al., 2012: Kay et al., 2018), although some studies have reported that cattle do not favour the watercourse itself (e.g. Hann et al., 2010; Bond et al., 2012). Nevertheless, as discussed in previous chapters, unrestricted cattle access to watercourses has been shown to have a wide range of direct detrimental effects including streambank deterioration (Braccia and Voshell, 2007; Zaimes and Schultz 2011), increased sedimentation (O'Sullivan et al., 2019), loss of riparian vegetation and habitats (Belsky et al., 1999) and streambed faecal contamination (Bragina et al., 2017).

A number of studies have reported that cattle tend to defecate more often when in proximity to water. For example, in their study of the water quality impacts of a dairy herd (246 animals) crossing the Sherry River in New Zealand, Davies-Colley et al. (2004) estimated that the herd defecated c. 50 times more per metre of stream then elsewhere. Similarly, Bond et al. (2012), studying a herd of 68 bullocks in the UK, observed that the animals defecated in waters five times more frequently than the average frequency of defecation overall. A study of four dairy cow herds in the US also reported that the animals defecated 1.3 to 7.8 times more often in the stream and riparian area than elsewhere (James et al.,

2007). There is no similar study investigating the frequency of urination by cattle in or nearby streams, however it has been observed that the spatial incidence of urination is positively correlated to the time cattle spend in certain areas (White et al., 2001; Draganova et al., 2015). Thus, given the spatial preference of cattle for riparian areas (James et al, 2007; Haan et al., 2010; Bond et al., 2012; Kay et al., 2018), it is likely that cattle urinate more often in these areas, potentially increasing the risk of nutrient loss to waters. Additionally, as highlighted in previous chapters, stirring up of stream bed sediments during cattle activity within the stream channel can also increase concentrations of nutrients (House, 2003) and faecal organisms in the water column (Collins and Rutherford, 2004).

The volume of cattle excretal output is determined by factors such as type of diet, feed and water intake, animal liveweight and reproductive status (Dijkstra et al., 2013; Smith and Frost, 2000). Studies have reported average numbers of urination events per animal per day ranging from 6.5 (dairy cattle) to 13.5 (beef cattle) (Oudshoorn et al., 2008; Orr et al., 2012; Selbie et al., 2015; Misselbrook et al., 2016), but typically 10 urinations.day<sup>-1..</sup> Defecation events are typically around 10 per day (Oudshoorn et al., 2008; Orr et al., 2012). Urination events consisting of 0.4 L to 6.4 L (mean of 1.8 L) have been reported (Misselbrook et al., 2016), whereas defecation events typically consist of deposits of 1 – 2 kg (fresh weight) (Davies-Colley et al., 2004; James et al., 2007; Bond et al., 2014). Dry matter in cow faeces generally makes up between 11% (Bond et al., 2012; Orr et al., 2012) to 18% (James et al., 2007) of the fresh faeces.

Urine is the main pathway of excretion of nitrogen in cattle, containing more readily available forms of N than faeces (Selbie et al., 2015). Studies have reported TN concentrations in cattle urine ranging from 3.0 g.L<sup>-1</sup> to 20.5 g.L<sup>-1</sup> (see Table 5.1). The dominant form of nitrogen in urine is urea-N, which in these studies represented 52.1% to 93.5% of TN. Ammonia was measured in two of these studies, corresponding to an average of 0.9% (Gonda and Lindberg, 1994) to 2.9% (Bristow et al., 1992) of the TN in urine. Phosphorus,

on the other hand, is mostly excreted in faeces and is present in cattle urine in much lower concentrations, with Orr et al. (2012) reporting an average of 0.0039 mg.kg<sup>-1</sup> urine TP in their study. TP concentrations in cattle faeces ranging from 4.9 mg.kg dry wt<sup>-1</sup> to 12.65 mg.kg dry wt<sup>-1</sup> have been reported in literature (see Table 5.1). Furthermore, in a study investigating the effects of dietary P on faecal P excretion by dairy cows, Dou et al. (2002) observed that a substantial amount of the faecal TP was readily soluble inorganic P, and that this fraction increased with increasing P intake, representing 30.3% to 49.6% of the total. The addition of phosphorus to stream waters is particularly relevant as phosphorus is generally the limiting nutrient for plant and algal growth in freshwater systems (Reddy et al., 1999; Jennings et al., 2003). Nitrogen concentrations in faeces ranging from 18.1 g.kg dry wt<sup>-1</sup> to 37.8 g.kg dry wt<sup>-1</sup> have been reported (Table 5.1), with 3 g.kg dry wt<sup>-1</sup> to 28 g.kg dry wt<sup>-1</sup> corresponding to NH<sub>4</sub>-N (Sorensen et al., 2003).

Concentrations of the faecal bacteria *E. coli* in fresh cattle faeces reported in literature are typically in the range of  $10^6 - 10^7$  CFU.g dry wt<sup>-1</sup> (see Table 4.1), with some studies suggesting that *E. coli* populations in faeces can grow in the days following faecal deposition (Oliver et al., 2010; Oliver and Page, 2016). Furthermore, as discussed in Chapters 2 and 4, it has been widely reported that once bacteria are deposited in the aquatic system, they can be incorporated into the sediment, where they can persist for prolonged periods of time (Garzio-Hadzick et al., 2010; Ishii et al., 2007; Ishii and Sadowsky, 2008; Badgley et al., 2011; Shelton et al., 2014). Indeed, significantly higher concentrations of *E. coli* bacteria have been reported in sediments at sites with unrestricted cattle access compared with sites with no access by Bragina et al. (2017), and have been observed in the study described in Chapter 4 of this thesis.

As discussed in earlier chapters, studies have shown that unrestricted cattle access to watercourses can lead impact water concentrations of nutrients (e.g. Davies-Colley et al., 2004; Byers et al., 2005; Vidon et al., 2007). Demal (1982) conducted an early pilot real-time

study on the impacts of cattle access to watercourses and in-stream activity on water concentrations of nutrients and faecal bacteria in the Avon River basin, Canada, and observed increases in TSS, TP, free ammonium and faecal coliform concentrations during cattle access, but not in SRP or nitrate concentrations. More recently, Terry et al. (2014) conducted a high-temporal resolution study using sensors to investigate the impacts of direct cattle access on water TSS concentrations, and concluded that 57.9% of the events that caused TSS concentrations to exceed a threshold of 25 mg.L<sup>-1</sup> were attributable to cattle activity. However, the authors also noted that these events corresponded to only 3.6% of the total SS exports in the stream for the period of study (Terry et al., 2014). Wilson and Everard (2017) also reported cattle in-stream activity to be strongly correlated with increases in water turbidity and faecal coliforms concentrations, but the study was inconclusive regarding its effects on SRP concentrations.

The present study aimed at addressing the general paucity in previous research regarding the impacts of direct cattle access to watercourses on the aquatic system by providing a more in-depth analysis of the impacts of direct cattle access on a set of water quality parameters (i.e. SRP, TP, NH<sub>4</sub>-N, NO<sub>3</sub>-N, TSS, and *E. coli* bacteria), in the Irish context. To the knowledge of the authors, only three other studies have aimed at assessing the impacts of cattle in-stream activity on water quality at a high temporal resolution (Demal, 1982; Terry et al., 2014: Wilson and Everard, 2017). The early study by Demal (1982) was published as a project report, but measured a comprehensive set of water quality parameters, including nutrients (SRP, TP, TKN, nitrate and nitrite, and free ammonium), suspended solids and faecal contaminants (faecal coliforms, faecal streptococci, *Pseudomonas aeruginosa, Salmonella* sp.). However, this study was based on a rather small number of manual samples (four upstream and downstream of the site for physicochemical analysis, one at each location for microbiological analysis) collected for only two cattle access events although in five access sites with varying land use and physical characteristics. Terry et al. (2014) and Wilson and Everard (2017) conducted studies with more intensive sampling,

however Terry et al., (2014) focused their study on suspended solids concentrations, whereas Wilson and Everard (2017) reported turbidity, SRP and *E. coli* only, at a lower temporal resolution (5 minutes – 30 minutes interval samples)and in a rather limited number of events (on 8 cattle access events data was collected on turbidity, on 7 events for SRP, and on 4 events for *E. coli*).

The specific aim of the current study was to quantify upstream to downstream concentrations changes in the selected variables at a single cattle access site at a high temporal frequency and for multiple events. These data were then used to estimate the changes in loads both when cattle where in the stream and when no cattle were present. The research described in this chapter is the first study that has measured a wide range of water quality parameters during multiple events at a fine temporal resolution, and that has also included events of no cattle access, thereby acknowledging the background variability of the measured parameters at the study site. The study hypothesis was that in-stream cattle activity would result in increased concentrations of total suspended solids, faecal bacteria and nutrients resulting from streambed sediment disturbance and excretion directly to and nearby waters.

# 6.2. Methods

## 6.2.1. Study site

This study was conducted in the Commons River catchment, Co. Louth (in site CM3, see Chapter 3). The cattle access site was selected due to the site fulfilling a number of required criteria (including that is used utilised by cattle, isolated from other cattle access points in the near vicinity and there was access permission from the landowner). This site also had availability of high temporal resolution water quality and discharge data, collected by a

hydrometric station located approximately 280 m downstream from the site (Fig 6.1) as part of the Agricultural Catchments Programme (ACP) (Teagasc, 2017a). Meteorological data for this site were also available from the ACP which operates a weather station located centrally in the catchment. The study site was on a third order stream. It had a width of approximately 2.90 m, and consisted of a stretch of approximately 8 m in the stream with electric fencing at the upstream and downstream boundaries preventing cattle from further dwelling in the stream (Fig. 6.1). The streambanks were dominated by alder (Alnus sp.), ash (Fraxinus excelsior) and sycamore (Acer pseudoplatanus) trees. Fencing as well as riparian vegetation prohibited the animals of crossing the stream to the adjacent field. The site was the only access site to the stream in a field of approximately 5.46 ha, grazed by between 20 and 25 beef heifers during each grazing season. The stream was the only drinking water source available to the animals, i.e. there were no drinkers provided. Additionally, there were no feeders or other structures located in proximity to the access site that could have had an added influence on the amount of time the animals congregated in the area. In Ireland, the grazing season generally takes place between April and October - November. In 2017, the period when most of the sampling for the current study took place, animals were first seen in the fields on 15<sup>th</sup> April, and their presence was last registered on the 8<sup>th</sup> October. Prior to the start of this study, stream water and stream bed sediment samples were collected at the site in Spring and Summer and repeated in Autumn 2016, at the same time as the work described in the previous chapters (Tables 6.1 and 6.2). This provided the general ranges for the study parameters prior to the current study.



**Fig 6.1** Schematic representation of the experimental site and position relatively to the ACP hydrometric station.

Timo	Temperature	DO	DO	Conductivity	ъЦ	SRP	TP	NH₄-N	TON
Inne	(°C)	(mg.L <sup>-1</sup> )	(%)	(µS.cm⁻¹)	рп	(µg.L <sup>-1</sup> )	(µg.L⁻¹)	(mg.L⁻¹)	(mg.L <sup>-1</sup> )
Spring 2016	12.17	10.23	97.86	317	8.24	44	91	0.04	3.14
Autumn 2016	11.87	9.62	87.83	637	8.14	174	224	0.03	2.50

Table 6.1. Water quality parameters at the study site at Dunleer (CM3) in 2016 (data for one composite sample based on five subsamples).

Table 6.2. Characteristics of the bed sediment at the study site at the study site (CM3) in 2016.

Time	TN (mg.g dry wt⁻¹)	OC (mg.g dry wt⁻¹)	TP (µg.g dry wt⁻¹)	<i>E. coli</i> (CFU.g dry wt⁻¹)	Particle size
Spring – Summer 2016	2.15	19.26	521	7.3 x 10 <sup>6</sup>	90% sand, 10% silt+clay
Autumn 2016	1.34	16.72	556	3.0 x 10 <sup>4</sup>	94% sand, 6% silt+clay

TN – Total Nitrogen, OC – Organic Carbon, TP – Total Phosphorus. Particle size and nutrients analysed on composite samples based on 6 subsamples; n = 3 for microbiological analysis).

# 6.2.2. Water sampling

Two autosamplers (HACH AS950 Portable Autosamplers, with 24 x 1 L capacity bottles) were placed approximately 20 m upstream and 20 m downstream of the boundaries of the cattle access site (Figure 6.1). Water quality monitoring was carried out during set time periods referred to as "events". During each of these events, both autosamplers were set to collect composite samples every 15 minutes, with 200 ml of stream water collected every 3 minutes to yield one 1 L composite sample for each 15 minute interval. This sampling continued for between 2.5 and 5 hours, with most events lasting 3.7 hours (222 min). There was a 3 minute delay between the start time used for the upstream autosampler and the start time for the downstream autosampler, with the aim of capturing the same stream water body at both sampling sites. This time interval was calculated based on an average stream flow rate at the site. Three motion-activated cameras (Bushnell Trophy Cam™ HD) were placed at the cattle access site to register cattle presence in the stream as well as to allow estimation of animal numbers and in-stream defecation and urination events (see Figure 4.1 for locations). The cameras were set to take three consecutive images following triggering of the motion-activated sensors, with intervals of 4 to 13 seconds in between shots. The cameras had been trialled on a number of occasions to investigate possible patterns in the timing of cattle access and to estimate the best time of the day for sampling. After each sampling event, the autosampler water bottles and the camera data were collected and immediately transported to the DkIT laboratory for analysis. Flow measurements were taken manually upstream and downstream of the site at the start and the end of each sampling event using a flow meter (Global Water, Xylem Inc.).

# 6.2.3. Water analysis

In preparation for each sampling event, the autosampler bottles and water collection tubes were acid-washed with a 10% HCl solution, sterilised with a solution of 70% industrial methylated spirit (IMS) and rinsed several times with Milli-Q water. Following collection, the samples were analysed for concentrations of dissolved nutrients (SRP, NH<sub>4</sub>-N, NO<sub>3</sub>-N), total reactive phosphorus (TRP), total phosphorus (TP), total suspended solids (TSS), *E. coli,* Other Coliforms and conductivity. Chloride concentrations were also measured to rule out possible influences of other sources e.g. septic tank discharges or dairy washings, on changes in concentrations of dissolved nutrients and faecal bacteria.

## 6.2.3.1. Analysis of physicochemical parameters

All glassware used in nutrient analysis was acid-washed with 10% HCI, rinsed with Milli-Q water and dried prior to the analysis. An aliquot of each sample was filtered through a nitrocellulose 0.45 µm filter (Sartorius Stedim Biotech, Germany) for analysis of dissolved nutrients. SRP was determined on the filtrate using a manual colorimetric method based on phosphate ascorbic acid reduction (Murphy and Riley, 1962). NH<sub>4</sub>-N was determined using a manual colorimetric method based on ammonia reaction with salvcilate and dichloroisocyanurate in alkaline solution (HMSO, 1981). NO<sub>3</sub>-N and chloride were determined simultaneously in a high performance integrated ion chromatography system (DIONEX®, ICS-2000, ThermoFisher Scientific). An unfiltered aliquot of each sample was used to determine TRP and TP using a colorimetric method (Murphy and Riley, 1962) directly and following sample acid persulfate digestion at 120°C, respectively. All nutrient analyses were carried out in duplicates. Calibration standards and quality control standards for phosphorus analysis were prepared using stock solutions (1000 mg.L<sup>-1</sup>) prepared with analytical grade chemicals (ThermoFisher Scientific) (potassium phosphate, KH<sub>2</sub>PO<sub>4</sub>, and sodium phosphate Na<sub>3</sub>PO<sub>4</sub>, respectively). In the case of TP analysis, all standards were

digested in the same manner as the samples. Calibration standards and quality control standards for ammonium analysis were prepared using a certified standard solution (1000 mg.L<sup>-1</sup>; Inorganic<sup>TM</sup> Ventures) and a stock solution prepared with analytical grade ammonium chloride (NH<sub>4</sub>Cl; ThermoFisher Scientific) (1000 mg.L<sup>-1</sup>), respectively. Calibration standards and quality control standards for nitrate analysis were prepared using certified standard solutions (1000 mg.L<sup>-1</sup>) of sodium nitrate (NaNO<sub>3</sub>) purchased from different manufacturers (Inorganic<sup>TM</sup> Ventures and Reagecon, respectively). The limits of detection for nutrient analysis were 3  $\mu$ g.L<sup>-1</sup> for SRP and TRP, 4  $\mu$ g.L<sup>-1</sup> for TP, 0.017 mg.L<sup>-1</sup> for NH<sub>4</sub>-N and 0.015 mg.L<sup>-1</sup> for NO<sub>3</sub>-N. These were calculated for each method according to Armbruster and Pry (2008).

To determine TSS concentrations, 300 to 500 ml of each composite sample were filtered through previously dried and weighted 1.2 µm pore size glass microfiber filters (G/FC Sartorius, Germany), which were then dried for 24 hours at 105 °C, reweighted, and the differences in weights recorded. Finally, pH and conductivity were determined in each sample using a multiparameter probe (YSI Professional Plus, YSI Inc.).

# 6.2.3.2. Microbiological analysis

Due to time and material constraints, not all samples collected at each event were analysed for *E. coli* concentrations. In such cases, the camera images were inspected upon arrival to the laboratory to determine times of cattle in-stream activity and the corresponding samples were analysed. An aliquot of each sample was transferred into a sterile Duran bottle and used to determine water concentrations of *E. coli* using a membrane filtration technique. One to two 1:10 sequential dilutions were prepared from each original sample using Ringers solution as diluent. Each dilution was then filtered through a sterile cellulose esters membrane with 47 mm diameter and 0.45  $\mu$ m pore size with grids as described in ISO 9308-1:2000. The membranes were placed onto Petri dishes containing HarlequinTM *E*.

*coli*/Coliform medium (LabM, Lancashire, UK) and incubated at 37 °C for 18 – 24 hours according to the manufacturer's instructions. All green-blue colonies were counted as presumptive *E. coli* bacteria, and all purple/pink colonies were counted as Other Coliforms.

#### 6.2.4. Data processing

Data on stream discharge (m<sup>3</sup>.s<sup>-1</sup>, averaged hourly) and daily precipitation (mm day<sup>-1</sup>) in the 48 hours preceding the sampling events were provided by the Agricultural Catchments Programme where possible (see Table 6.4).

The camera images were inspected manually for periods of cattle in-stream activity, cattle numbers and direct excretion into stream waters. Due to the physical characteristics of the site and the limited possibilities for positioning the cameras, at times when a high number of animals (>5) accessed the site simultaneously, visibility within the group of cattle was limited despite having three angles; thus the total number of cattle in the stream on those occasions is the best estimate that was possible to obtain by inspecting the images collected from the three angles. Additionally it is possible that on such occasions, a small number of defecation or urination episodes may have not been recorded.

TSS concentrations were consistently disproportionally higher in the first sampling intervals of the events. It was most likely an indication of human disturbance of the stream sediment while preparing and starting the autosamplers, therefore the first observations for all parameters for both upstream and downstream (corresponding to 12 minutes from start) were excluded from analysis. Nutrient concentrations below the limit of detection for the method used were replaced with ½ of the limit of detection value for data analyses.

The differences in nutrient, TSS and *E. coli* concentrations downstream of the access site relative to the upstream site were calculated for each upstream-downstream sample pair. For this purpose, periods of cattle in-stream activity included the times at which cattle were

recorded accessing the stream, and also the 15 minutes period immediately following (i.e. the next sample), in order to account for the falling phase of the changes in freshwater parameters. The loads of nutrients, TSS and *E. coli* during and following periods of cattle instream activity were calculated by multiplying the change in concentration between the upstream and downstream sampling sites (which could be negative) by the average value of the two stream discharge (available as hourly averages) measured in the two points preceding and following each sample time interval. In the cases where the sampling interval coincided with the hour and thus with stream discharge measurement, the corresponding hour stream discharge was used in the calculations.

## 6.2.5. Statistical analysis

The difference in the distributions of the dataset for all calculated 15 minute loads of nutrients, TSS and *E. coli* for the downstream site compared to the same data for the upstream site was assessed using a two-sided Kolmogorov–Smirnov test. This test determines whether two distributions are significantly different from each other. Since it was likely that the calculated loads for sequential timesteps at the sampling locations and between sampling locations were not independent, bootstrap resampling was used to assign significance. The analysis was performed in R software (version 4.0.0; R Core Team, 2016) using the packages *Matching* (Sekhon, 2013) and *boot* (Canty and Ripley, 2020) and the ks.boots function, with nboots (number of resamplings) set to 1000. Empirical cumulative density functions (ECDF) were plotted for each pair of distributions.
#### 6.3. Results

In total, 16 events were sampled between August 2016 and June 2018 (Table 6.3). Eleven of these events are described in this chapter, selected based on the availability of data for a broader range of parameters and adequate camera data. Of these, seven events captured cattle in-stream activity (Table 6.4). All events with cattle activity also included periods before and after cattle activity when there was no cattle activity in the stream. The stream discharge in these 11 events varied from 0.003 m<sup>3</sup>.s<sup>-1</sup> to 0.129 m<sup>3</sup>.s<sup>-1</sup> (25<sup>th</sup> percentile = 0.013 m<sup>3</sup>.s<sup>-1</sup>; 75<sup>th</sup> percentile = 0.148 m<sup>3</sup>.s<sup>-1</sup>) (Table 6.4).

## 6.3.1. Background variation in water quality parameters

Background concentrations (i.e. those measured at the sampling point upstream of the access site) of nutrients, TSS and *E. coli* bacteria for the sampling events included in this chapter are shown in Table 6.5. Data for TRP, chloride, conductivity and Other Coliforms are not included in this chapter and are presented in Appendix A. Event 8, which took place on May 24, 2017 (Fig. 6.5), is presented here as an example of the variation in water quality parameters at the study site in the absence of cattle in the stream. During this event, although cattle were present in the fields, they did not visit the stream during the period of sampling based on the camera data.



**Fig.6.2.** Images taken during sampling events. a) fence pole view showing 3 animals in the stream in Event 2; b) side view of at least 4 animals in the stream in Event; c) in-stream urination and d) defecation in Event 9; e) one animal in Even10; f) side view of two animals and g) several animals in the stream in Event 16; h) front view of one animal in Event 15, after the sampling period.

Cattle **Reasons for** Available data Event Included Date exclusion/Comments access Different methodology (longer 24/08/2016 SRP, NH<sub>4</sub>-N, TON, TSS Yes No 1 periods of time between samples) 21/09/2016 SRP, NH<sub>4</sub>-N, TON, TSS Yes Yes 2 Fewer parameters 3 16/11/2016 NH4-N, TON, TSS No No 30/11/2016 NH<sub>4</sub>-N, TSS Fewer parameters 4 No No 5 06/04/2017 TSS, E. coli No No Fewer parameters 26/04/2017 SRP, TP, TSS, E. coli Yes No Poor quality of camera images 6 7 10/05/2017 SRP, TP, NH<sub>4</sub>-N, NO<sub>3</sub>-N, TSS, Cl<sup>-</sup>, *E. coli*, Other Coliforms Yes Yes 24/05/2017 SRP, TP, NH<sub>4</sub>-N, NO<sub>3</sub>-N, TSS, Cl<sup>-</sup>, *E. coli*, Other Coliforms No Yes 8 14/06/2017 SRP, TP, NH<sub>4</sub>-N, NO<sub>3</sub>-N, TSS, Cl<sup>-</sup>, *E. coli*, Other Coliforms Yes 9 Yes 10 Yes Yes Not possible to use data for E. coli **28/06/2017** SRP, TP, NH<sub>4</sub>-N, NO<sub>3</sub>-N, TSS, Cl<sup>-</sup>, conductivity

Table 6.3. List of sampling events conducted in this study. Events discussed in this chapter are in bold.

Table 6.3. (continued)

Event	Date	Available data	Cattle access	Included	Reasons for exclusion/Comments
11	08/08/2017	SRP, TP, NH <sub>4</sub> -N, NO <sub>3</sub> -N TSS, Cl <sup>-</sup> , <i>E. coli</i> , Other Coliforms, conductivity	Yes	Yes	
12	30/08/2017	SRP, TRP, TP NH <sub>4</sub> -N, NO <sub>3</sub> -N, TSS, Cl <sup>-</sup> , <i>E. coli</i> , Other Coliforms	Yes	Yes	
13	07/02/2018	SRP, TP, NH <sub>4</sub> -N, NO <sub>3</sub> -N, TSS, Cl <sup>-</sup> , <i>E. coli</i> , Other Coliforms, conductivity	No	Yes	
14	21/03/2018	SRP, TRP, TP, NH <sub>4</sub> -N, NO <sub>3</sub> -N, TSS, Cl <sup>-</sup> , <i>E.coli</i> , conductivity	No	Yes	
15	25/04/2018	SRP, TRP, TP, NH <sub>4</sub> -N, NO <sub>3</sub> -N, TSS, Cl <sup>-</sup> , <i>E. coli</i> , Other Coliforms, conductivity	No	Yes	Not possible to use data for TP
16	13/06/2018	SRP, NH <sub>4</sub> -N, NO <sub>3</sub> -N, TSS, Cl <sup>-</sup> , <i>E. coli</i> , Other Coliforms, conductivity	Yes	Yes	Not possible to use data for TP

Event	Date	Cattle in field	Access	Max cattle	Total access (min)	Cattle- minutes <sup>3</sup>	Defecations recorded	Urinations recorded	Daily mean stream discharge (m <sup>3</sup> .s <sup>-1</sup> ) <sup>1</sup>	Precipitation in previous 48 hours (mm) <sup>2</sup>	Precipitation on day (mm)
2	21/09/2016	Yes	Yes	5	18.8	41	1	1	0.013	0.0	7.0
7	10/05/2017	Yes	Yes	6	26.5	76	1	2	0.012	0.0	0.0
8	24/05/2017	Yes	No	-	-	-	-	-	0.007	0.4	0.0
9	14/06/2017	Yes	Yes	4	32.9	81	1	1	0.015	0.0	1.8
10	28/06/2017	Yes	Yes	1	2.0	2	0	0	0.021	32.1	6.8
11	08/08/2017	Yes	Yes	3	23.0	46	nd	1	0.005	1.0	0.0
12	30/08/2017	Yes	Yes	3	13.1	25	nd	nd	0.003	1.6	0.2
13	07/02/2018	No	No	-	-	-	-	-	0.129	1.8	0.0
14	21/03/2018	No	No	-	-	-	-	-	0.123	0.0	0.0
15	25/04/2018	Yes	No	-	-	-	-	-	0.056	8.0	0.0
16	13/06/2018	Yes	Yes	5	41.2	99	nd	nd	0.005	1.9	1.9

**Table 6.4.** Description of the sampling events presented in this chapter.

<sup>1</sup>Data provided by the Agricultural Catchments Programme, except for Event 16, for which this was calculated using a regression line based on ACP data and field flow measurements. <sup>2</sup>Data provided by the ACP except for Events 14 to 16, for which they were retrieved from data from Met Eireann's nearest meteorological station.<sup>3</sup>Cattle-minutes are calculated multiplying the number of minutes of access by the number of animals in the stream in each minute. nd – not detected

Stream nutrient concentrations measured in this event upstream of the access site varied from 125  $\mu$ g.L<sup>-1</sup> to 131  $\mu$ g.L<sup>-1</sup> for SRP, 169 to 194  $\mu$ g.L<sup>-1</sup> for TP, <0.02 to 0.05 mg.L<sup>-1</sup> for NH<sub>4</sub>-N and 4.12 to 4.24 mg.L<sup>-1</sup> for NO<sub>3</sub>-N. At the downstream site, nutrient concentrations ranged from 124  $\mu$ g.L<sup>-1</sup> to 129  $\mu$ g.L<sup>-1</sup> for SRP, 166 to 195  $\mu$ g.L<sup>-1</sup> for TP, <0.02 to 0.04 mg.L<sup>-1</sup> for NH<sub>4</sub>-N and 3.93 to 4.18 mg.L<sup>-1</sup> for NO<sub>3</sub>-N. Upstream TSS concentrations ranged from 3.4 to 8.6 mg.L<sup>-1</sup>, while downstream TSS concentrations varied from 2.6 to a maximum of 31.0 mg.L<sup>-1</sup>, which was a peak value observed in the composite sample for the time interval 162 to 177 minutes at the downstream site. For the upstream site, *E. coli* concentrations ranged from 2.7 x 10<sup>3</sup> to 1.3 x 10<sup>4</sup> CFU.100 ml<sup>-1</sup>. These changes in freshwater parameters concentrations reflect the background variability at the site both at each sampling point and between sampling points.

## 6.3.2. Changes in water quality parameters during cattle access to the stream

Events 7 and 12 are presented here as examples of the changes across all water physicochemical and microbiological parameters during periods of cattle in-stream activity of different intensities.

In Event 12 (Fig 6.4), which took place on August 30, 2017, cattle accessed the stream on five shortly spaced occasions for an average duration of 2.6 minutes and a total duration of 13.1 minutes. A maximum of three animals were observed in the stream simultaneously. During the period of cattle access, increases in the concentrations of TSS, TP, NH<sub>4</sub>-N and *E. coli* were observed downstream of the access site in comparison to upstream concentrations, in particular in the interval corresponding to 162 to 177 minutes from start time of sampling, after three animals accessed the stream simultaneously. TSS concentrations then increased to a maximum of 11.0 mg.L<sup>-1</sup> downstream of the access site, while concentrations upstream for the equivalent 15 min sampling interval were 3.8 mg.L<sup>-1</sup>.

Similarly, TP concentrations peaked at 349  $\mu$ g.L<sup>-1</sup> at the downstream site during cattle access, compared to a TP concentration upstream of 211  $\mu$ g.L<sup>-1</sup>. Both NH<sub>4</sub>-N and *E. coli* also showed upstream to downstream increases when cattle were in the stream. NH<sub>4</sub>-N concentrations peaked at 0.10 mg.L<sup>-1</sup> downstream at the sampling period of 162 to 177 minutes, compared to 0.02 mg.L<sup>-1</sup> upstream. *E. coli* concentrations reached a maximum of 5.1 x 10<sup>3</sup> CFU.100 ml<sup>-1</sup> downstream of the access site during in-stream activity, while upstream concentrations for the equivalent time were 1.7 x 10<sup>3</sup> CFU.100 ml<sup>-1</sup>. In contrast, both SRP and NO<sub>3</sub>-N concentrations did not increase downstream of the access site relative to upstream concentrations during or following cattle access to the stream. It is of note, however, that TSS concentrations downstream at the site peaked at 11.0 mg.L<sup>-1</sup> in the time interval 12 – 27 minutes from start, when no cattle were in the stream, while TSS concentrations upstream for the equivalent sample were 3.0 mg.L<sup>-1</sup>.

In Event 7 (Fig 6.5), which took place on May 10, 2017 cattle accessed the stream almost continuously for a period of 25.6 minutes (from minute 77 to minute 104), with a maximum of six animals standing in the stream simultaneously (Table 6.4). One in-stream defecation at (minute 94) and two in-stream urinations (minute 94 and 103) were registered during this period. During the period of cattle in-stream activity, concentrations of TSS and TP reached maximum values of 27.4 mg.L<sup>-1</sup> and 148  $\mu$ g.L<sup>-1</sup>, respectively, downstream of the site, while concentrations measured upstream for the same period were much lower at 3.4 mg.L<sup>-1</sup> and 84  $\mu$ g.L<sup>-1</sup>. Downstream concentrations of SRP showed only a modest increase during and following cattle in-stream activity, reaching a maximum of 81  $\mu$ g.L<sup>-1</sup> at interval 102 – 117 minutes from start, whereas the SRP concentration measured upstream at the same period of time was 73  $\mu$ g.L<sup>-1</sup>. NH<sub>4</sub>-N concentrations increased to 0.04 mg.L<sup>-1</sup> and 0.08 mg.L<sup>-1</sup> downstream in the sampling intervals of 102 – 117 and 117 – 132 minutes from start, which followed the two episodes of in-stream urination observed in this event. These changes in NH<sub>4</sub>-N corresponded to increases of +0.02 mg.L<sup>-1</sup> and +0.06 mg.L<sup>-1</sup> in comparison with upstream concentrations (0.02 mg.L<sup>-1</sup> for both). However, there was also an increase in

downstream NH<sub>4</sub>-N concentrations at the start of the sampling event (interval of 12 - 27 minutes from start) that could not be explained by cattle activity, when concentrations at the upstream and downstream sites were 0.02 mg.L<sup>-1</sup> and 0.06 mg.L<sup>-1</sup>, respectively. There was also a clear increase in *E. coli* concentrations that coincided with the time period when cattle were in the water (Fig.6.5). *E. coli* concentrations downstream of the access site were 1.0 x  $10^3$  CFU.100 ml<sup>-1</sup> at interval 62 - 87 minutes from start, during which cattle entered the stream, while upstream concentrations for the equivalent time were 4.9 x  $10^2$  CFU.100 ml<sup>-1</sup>. *E. coli* downstream concentrations peaked at 2.4 x  $10^4$  CFU.100 ml at interval 102 - 117 minutes from start, corresponding to an increase in two orders of magnitude relative to concentrations observed upstream of the access site (4.2 x  $10^2$  CFU.100 ml<sup>-1</sup>) at the same time.

Event	n	SRP (µg.L <sup>-1</sup> )	TP (μg.L <sup>-1</sup> )	NH₄-N (mg.L⁻¹)	NO₃-N (mg.L <sup>-1</sup> )	TSS (mg.L <sup>-1</sup> )	<i>E. coli</i> (CEU 100 ml <sup>-1</sup> )
2	14	163 + 3	ΝΑ	0.02 + 0.02	2 14 + 0 08*	04+02	
7	14	74 + 2	00.0	$0.02 \pm 0.02$	2.14 ± 0.00	$0.4 \pm 0.2$	$4.2 \times 10^2$ , EC (2 E)
1	14	14 ± 2	02 ± 0	$0.02 \pm 0.01$	$4.44 \pm 0.11$	$4.0 \pm 2.0$	$4.3 \times 10 \pm 50 (n=5)$
8	14	127 ±2	181 ±7	0.02 ± 0.01	4.19 ± 0.03	5.2 ± 1.7	$8.8 \times 10^3 \pm 5.6 \times 10^3 (n=7)$
9	19	101 ± 3	217 ± 50	0.02 ± 0.01	$3.09 \pm 0.26$	$4.5 \pm 0.9$	9.4 x 10 <sup>2</sup> ± 1.4 x 10 <sup>2</sup> (n=7)
10	18	244 ± 14	293 ± 27 (n=17)	$0.04 \pm 0.02$	2.72 ± 0.10	9.9 ± 12.1	NA
11	14	244 ± 7	276 ± 20	0.15 ± 0.07	$2.75 \pm 0.06$	6.4 ± 4.1	7.4 x 10 <sup>2</sup> ± 1.4 x 10 <sup>2</sup> (n=13)
12	14	209 ± 3	227 ± 10	$0.04 \pm 0.06$	2.73 ± 0.03	$2.6 \pm 0.9$	1.9 x 10 <sup>3</sup> ± 4.0 x 10 <sup>2</sup> (n=9)
13	14	32 ± 1	64 ± 4	$0.09 \pm 0.02$	8.31 ± 0.11	4.3 ± 1.5	3.5 x 10 <sup>2</sup> ± 1.2 x 10 <sup>2</sup> (n=13)
14	9	48 ± 3	77 ±6	0.07 ± 0.01	$7.50 \pm 0.07$	9.2 ± 1.6	$1.7 \times 10^3 \pm 1.2 \times 10^3$
15	14	103 ± 4	NA	0.33 ± 0.07	6.79 ± 0.10	17.2 ± 22.4	$2.8 \times 10^3 \pm 6.7 \times 10^2$
16	14	135 ± 2	NA	$0.05 \pm 0.03$	$5.44 \pm 0.10$	5.4 ± 1.6	$2.8 \times 10^3 \pm 2.4 \times 10^3$

**Table 6.5.** Concentrations of nutrients, total suspended solids (TSS) and *E. coli* bacteria measured upstream of the cattle access site during the sampling events (mean ± S.D.).

\*Measured as TON. n = number of 15 minute sampling times. NA – not available due to issues during sample collection or processing.





**Fig.6.3.** Variation in mean nutrients, TSS and *E. coli* bacteria concentrations upstream and downstream of the cattle access site during Event 8 (May 24, 2017), when cattle were present in the fields but did not access the stream during the sampling period. The dashed red line represents the limit of the detection for the analysis of the parameter.



**Fig.6.4.** Variation in nutrients, TSS and *E. coli* bacteria concentrations upstream and downstream of the cattle access site during Event 12 (August 30, 2017), when a maximum of 3 animals accessed the stream simultaneously. The dashed red line represents the limit of the detection for the analysis of the parameter.



**Fig.6.5.** Variation in nutrients, TSS and *E. coli* bacteria concentrations upstream and downstream of the cattle access site during Event 7 (May 10, 2017), when a maximum of 6 animals accessed the stream simultaneously. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. The dashed red line represents the limit of the detection for the analysis of the parameter.

	SR	SRP (µg.L⁻¹)		SRP TP NH <sub>4</sub> -N		NO <sub>3</sub> -N		TSS		E. coli		
Event	(µg.			(µg.L⁻¹)		(mg.L <sup>-1</sup> )		(mg.L <sup>-1</sup> )		(mg.L⁻¹)		(CFU.100 ml⁻¹)
	US	DS	US	DS	US	DS	US	DS	US	DS	US	DS
2	165	192	NA		0.05	0.14 2.04* 2.15*		2.15*	0.3	5.5	Ν	A
7	73	81	84	148	0.02	0.08	No inc	crease	3.4	27.4	4.2 x 10 <sup>2</sup>	2.4 x 10 <sup>4</sup>
9	93	99	155	309	< 0.02	0.05	No inc	crease	4.4	18.4	1.2 x 10 <sup>3</sup>	8.5 x 10 <sup>3</sup>
10	No inc	rease	299	306	0.06	0.07	No inc	crease	6.2	9.4	Ν	A
11	240	244	268	367	0.08	0.19	2.77	3.01	5.8	42.4	8.0 x 10 <sup>2</sup>	1.3 x 10 <sup>4</sup>
12	No inc	rease	217	349	0.02	0.10	No inc	crease	1.8	9.2	1.7 x 10 <sup>3</sup>	5.1 x 10 <sup>3</sup>
16	132	143	N	A	0.03	0.24	5.49	6.20	4.3	34.2	5.5 x 10 <sup>2</sup>	7.9 x 10 <sup>4</sup>

**Table 6.6.** Peak concentrations of nutrients, total suspended solids and *E. coli* bacteria measured downstream (DS) of the access site during periods of cattle access to the stream, and concentrations measured upstream (US) for the same sampling interval (15 minute composite samples).

\*Measured as TON. NA – Not available due to issues during sample collection or processing.

#### 6.3.2.1. Changes in total suspended solids (TSS) concentrations across all 11 events

Changes in total suspended solids (TSS) concentrations upstream and downstream of the access site during the sampling events are shown in Fig. 6.6 and Fig. 6.7. TSS concentrations were highly variable both upstream and downstream of the site, with occasional increases observed in events when there were no cattle present to access the stream (e.g. Events 8, 13 and 15) (Fig 6.6), as well as at times when cattle were present in the field but did not access the stream (Events 9 - 12) (Fig 6.7). These occasional increases ranged from +2.2 mg.L<sup>-1</sup> (Event 13) to +74.3 mg.L<sup>-1</sup> (Event 15) relative to TSS concentrations measured in the time interval immediately preceding increases. However, TSS concentrations also showed increases downstream of the access site during and following cattle in-stream activity.

In Event 10 (June 28, 2017), when one animal entered and remained in the stream for 2.0. minutes, downstream TSS concentrations peaked at 9.4 mg.L<sup>-1</sup> while concentrations measured upstream were 6.2 mg.L<sup>-1</sup>. However, these changes are not substantial in the context of the natural variation in TSS concentrations measured on this event. Similarly, in Event 12 (August 30, 2017), (maximum of three animals in the stream simultaneously; instream activity lasted 2.6 minutes on average and 13.1 minutes in total), TSS concentrations increased by a maximum of +7.2 mg.L<sup>-1</sup>, from 3.8 mg.L<sup>-1</sup> upstream to 11.0 mg.L<sup>-1</sup> downstream. Although these changes coincided with the period of cattle in-stream activity, they fell within the overall range of TSS concentrations measured for that event. For all other events when the intensity of cattle access to the stream was higher (i.e. higher number of animals visiting the stream and/or longer periods of in-stream activity), TSS concentrations increased downstream of the cattle access site during cattle in-stream activity (Table 6.6, Fig. 6.7). In Event 7 (May 10, 2018), when the highest number of animals observed in the stream simultaneously in this study was registered (6 animals), TSS concentrations reached a maximum of 27.4 mg.L<sup>1</sup>, eight times higher than the concentration measured upstream at

equivalent time (3.4 mg.L<sup>-1</sup>). However, the highest increase in TSS concentrations coinciding with cattle accessing the stream was measured during Event 11 (August 8, 2017), when a maximum of three animals visited the stream simultaneously, and the total duration of access was 22.0 minutes. On this occasion, an increase of +36.6 mg TSS.L<sup>-1</sup> was recorded, with a concentration of 5.8 mg.L<sup>-1</sup> at the upstream site and 42.4 mg.L<sup>-1</sup> at the downstream site.

## 6.3.2.2. Changes in total phosphorus (TP) water concentrations across all 11 events

Changes in TP water concentrations generally followed a similar trend to changes in TSS concentrations (Fig. 6.8 and 6.9). Although TP concentrations showed a relatively high variation in the absence of cattle (Table 6.5, Fig. 6.8), they consistently increased during or following cattle in-stream activity, with the exception of Event 10. In this latter event, TP concentrations increased and decreased markedly both upstream and downstream of the cattle access site, an observation that was not related to cattle access to the stream. , Although an increase of +7  $\mu$ g TP.L<sup>-1</sup> was observed downstream following cattle access, this was however negligible in the context of TP general variation in this event.





**Fig.6.6.** Variation in TSS concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales.

In the remaining sampling events, there were more substantial changes in TP concentrations coinciding with or following periods of cattle access to the stream (Table 6.6, Fig. 6.9). In Event 7 (May 10, 2017), in which a maximum of 6 animals accessed the stream for a total of 25.6 minutes, TP concentrations downstream of the site peaked at 148  $\mu$ g.L<sup>-1</sup> L, an increase of +64  $\mu$ g.L<sup>-1</sup> in comparison to upstream concentrations at the same time interval (84  $\mu$ g.L<sup>-1</sup>). In Event 9 (June 14, 2017), during which a maximum of 4 animals accessed the stream



Fig.6.7. Variation in TSS concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

and in-stream activity lasted on average 4.1 minutes and 32.9 minutes in total, TP concentrations downstream of the site peaked at 309  $\mu$ g.L<sup>-1</sup> whereas concentrations measured upstream in the same time interval were 155  $\mu$ g.L<sup>-1</sup>. This was the highest difference in concentrations for the downstream site relative to the upstream site measured during cattle in-stream activity in this study (+154  $\mu$ g.L<sup>-1</sup>),. It should be noted, however, that this was also due to a decrease in upstream concentrations which had been higher at 206  $\mu$ g.L<sup>-1</sup> in the previous sampling interval. In Event 11 (August 8, 2017), when a maximum of 5 animals visited the stream simultaneously and in-stream activity lasted on average 2.6 minutes and in total 13.1 minutes, downstream TP concentrations peaked at 320  $\mu$ g.L<sup>-1</sup>, while upstream concentrations at the same time period were 278  $\mu$ g.L<sup>-1</sup>. In Event 12 (August 30, 2017), TP downstream concentrations peaked at 349  $\mu$ g.L<sup>-1</sup>).

# 6.3.2.3. Changes in soluble reactive phosphorus (SRP) concentrations across all 11 events

The variation in SRP concentrations during the sampling events is shown in Fig. 6.10 and Fig. 6.11. Contrary to the patterns observed for TSS and TP, water concentrations of SRP did not show a consistent pattern of change during periods of cattle in-stream activity (Table 6.6, Fig. 6.11). During Events 10 (when one animal accessed the stream for 2.0 minutes) and 12 (when a maximum of 3 animals accessed the stream for a total of 13.1 minutes), SRP concentrations did not increase during or following periods of cattle in-stream activity. In Events 9 (June 14, 2017, when a maximum of 4 animals visited the stream simultaneously and access lasted 32.9 minutes in total) and 11 (August 8, 2017, with a maximum of 3 animals simultaneously in the stream and duration of access of 23 minutes), small increases of  $+6 \mu g.L^{-1}$  and  $+4 \mu g.L^{-1}$ , respectively, were registered. In Events 7 (May 10, 2017, when

there was a maximum of 6 animals in the stream and the total access duration was 25.6 minutes) and 16 (June 13, 2018, a maximum of 5 animals simultaneously in the stream with a total duration of access of 41.2 minutes), SRP concentrations downstream of the access site increased by a maximum of +11  $\mu$ g.L<sup>-1</sup> and +8  $\mu$ g.L<sup>-1</sup>, respectively (Table 6.6), following cattle in-stream activity.



Fig.6.8. Variation in TP concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales.



**Fig.6.9.** Variation in TP concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

Ammonium (NH<sub>4</sub>-N) concentrations in stream water at the study site were highly variable both in time and within sampling events (Table 6.5, Fig. 6.12 and 6.13). Nevertheless, while some increases in NH<sub>4</sub>-N concentrations downstream of the access site were registered at periods when cattle did not access the stream, changes in NH<sub>4</sub>-N concentrations were more pronounced during and following cattle in-stream activity. For example, NH<sub>4</sub>-N concentrations increased downstream of the access site during cattle activity during Event 12 (August 30, 2017), peaking at a concentration of 0.1 mg.L<sup>-1</sup>, an increase of +0.08 mg.L<sup>-1</sup> in relation to concentrations measured upstream. However, downstream concentrations that were higher than those measured upstream, by  $+0.03 \text{ mg.L}^{-1}$ , were also observed during the same sampling event during periods of no cattle access. In Event 11, NH<sub>4</sub>-N concentrations measured downstream of the site were higher than those measured upstream both during and following cattle in-stream activity, by a maximum of +0.11 mg.L<sup>-1</sup> (corresponding to 0.19 mg.L<sup>-1</sup> downstream), however this difference was also caused by a decrease in  $NH_4$ -N concentrations upstream of the access site (0.08 mg.L<sup>-1</sup>, from 0.13 mg.L<sup>-1</sup> in the previous sampling interval). In the same sampling event, NH<sub>4</sub>-N concentrations at the downstream site increased once again in comparison to upstream concentrations during a second period of cattle in-stream activity, but to a much lower extent (differences of +0.03 and +0.02 mg.L <sup>1</sup>). During Event 9 (June 14, 2017), ammonium concentrations increased and peaked at 0.05 mg.L<sup>-1</sup> during cattle in-stream activity, but also were higher than those measured upstream by +0.01 mg.L<sup>-1</sup> and +0.03 mg.L<sup>-1</sup> at sampling times both before and after the period of cattle access when no cattle were in the stream. In Event 16, NH<sub>4</sub>-N concentrations



Location - Downstream -- Upstream

**Fig.6.10.** Variation in SRP concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales.



Fig.6.11. Variation in SRP concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event. 177

increased markedly following a period of cattle in-stream activity and in-stream defecation, from 0.02 mg.L<sup>-1</sup> upstream to 0.21 mg.L<sup>-1</sup> downstream, and were again higher than upstream levels following a second period of cattle access (at minute 165) to the stream by +0.05 mg.L<sup>-1</sup>. In Event 7 (May 10, 2017), an increase in NH<sub>4</sub>-N concentrations was observed at the downstream site during cattle access and following two urinations and one defecation event, from 0.02 mg.L<sup>-1</sup> upstream to 0.08 mg.L<sup>-1</sup> downstream. NH<sub>4</sub>-N concentrations were at times higher upstream of the site than downstream in almost all sampled events, during both periods of cattle access and periods of no access, highlighting the high variability of this parameter. The highest difference in upstream concentrations were 0.22 mg.L<sup>-1</sup> upstream at the sampling interval of 207 to 222 minutes from start, while downstream concentrations were 0.02 mg.L<sup>-1</sup> downstream.

## 6.3.2.5. Changes in nitrate (NO<sub>3</sub>-N) water concentrations across all 11 events

Nitrate nitrogen (NO<sub>3</sub>-N) concentrations did not demonstrate any consistent response to cattle in-stream activity during the sampling events (Table 6.6., Fig. 6.14 and 6.15). For all seven events that captured cattle access to the stream, two events (Events 11 and 16) registered very small increases in NO<sub>3</sub>-N concentrations during and following cattle in-stream activity. In Event 8, NO<sub>3</sub>-N concentrations downstream of the site peaked at 3.01 mg.<sup>-1</sup> during cattle access to the stream, an increase of +0.24 mg.L<sup>-1</sup> in relation to upstream concentrations at the same time interval. In Event 16, NO<sub>3</sub>-N concentrations downstream of the stream, an increase of +0.20 mg.L<sup>-1</sup> in relation to upstream concentrations (5.42 mg.L<sup>-1</sup>). The downstream of the site reached 5.62 mg.L<sup>-1</sup> during a first period of cattle access to the stream, an increase of +0.20 mg.L<sup>-1</sup> in relation to upstream concentrations (5.42 mg.L<sup>-1</sup>). The downstream concentrations then peaked at 6.20 mg.L<sup>-1</sup> during a second period of cattle in-stream activity, an increase of +0.71 mg.L<sup>-1</sup> relative to upstream concentrations for the equivalent time interval (5.49 mg.L<sup>-1</sup>) (Table 6.6, Fig. 6.7). In this last sampling event, however, there was also a decrease of -0.34 mg.L<sup>-1</sup> in in downstream NO<sub>3</sub>-N concentrations (5.11 mg.L<sup>-1</sup>) in

comparison to upstream concentrations (5.45 mg.L<sup>-1</sup>) following the first period of cattle instream activity.



Location - Downstream -- Upstream

**Fig.6.12.** Variation in NH<sub>4</sub>-N concentrations upstream and downstream of the cattle access during events with no cattle access to the stream. Note the differences in scales. The dashed red line represents the limit of the detection for the analysis of the parameter



Fig.6.13. Variation in NH<sub>4</sub>-N concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. The dashed red line represents the limit of the detection for the analysis of the parameter. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event. 180

# 6.3.2.6. Changes in E. coli bacteria concentrations in stream water across all 11 events

Changes in E. coli concentrations during the sampling events are shown in Fig. 6.16 and Fig. 6.17. Concentrations of *E. coli* in stream waters showed a clear and consistent pattern of increase during periods of cattle in-stream activity (Table 6.6., Fig. 6.17). Increases in E. coli concentrations downstream of the access site were observed in all events that captured cattle in-stream activity, including those where no in-stream defecation was recorded. In Event 12, E. coli concentrations downstream of the site increased during and following three episodes of cattle access to 5.0 x  $10^3$  CFU.100 ml<sup>-1</sup>, 5.1 x  $10^3$  CFU.100 ml<sup>-1</sup> and 4.2 x  $10^3$ CFU.100 ml<sup>-1</sup> while upstream concentrations at the equivalent time intervals were 2.1 x  $10^3$ CFU.100 ml<sup>-1</sup>, 1.7 x 10<sup>3</sup> CFU.100 ml<sup>-1</sup> and 1.9 x 10<sup>3</sup> CFU.100 ml<sup>-1</sup>, respectively (Fig. 6.8). In Event 11 (August 8, 2017), *E. coli* concentrations peaked at 9.1 x 10<sup>3</sup> CFU.100 ml<sup>-1</sup> and 1.3 x 10<sup>4</sup> CFU.100 ml<sup>-1</sup> downstream of the access site during two periods of cattle in-stream, while upstream of the site concentrations measured for the corresponding sampling periods were 9.9 x 10<sup>2</sup> CFU.100 ml<sup>-1</sup> and 8.0 x 10<sup>2</sup> CFU.100 ml<sup>-1</sup>, respectively, representing a difference of approximately one order of magnitude. During Event 9, E. coli bacteria concentrations also increased during cattle in-stream activity, reaching a maximum of 8.5 x 10<sup>3</sup> CFU.100 ml<sup>-1</sup>, a near tenfold increase in comparison to upstream concentrations, which were at  $1.2 \times 10^3$ CFU.100 ml<sup>-1</sup> for the same time period. These increased *E. coli* concentrations at the downstream site were sustained for approximately 60 minutes, during which cattle visited the stream on several occasions. In Event 7, an increase of two orders of magnitude in E. coli bacteria concentrations from 4.2 x 10<sup>2</sup> CFU.100 ml<sup>-1</sup> upstream to 2.4 x 10<sup>4</sup> CFU.100 ml<sup>-1</sup> at the downstream site was observed during cattle access (maximum 6 animals) to the stream. Finally, in Event 16, downstream E. coli bacteria concentrations increased dramatically



Location - Downstream -- Upstream

**Fig.6.14.** Variation in NO<sub>3</sub>-N concentrations upstream and downstream of the cattle access during events with no cattle access to the stream.



**Fig.6.15.** Variation in  $NO_3$ -N concentrations upstream and downstream of the cattle access during events that captured cattle access to the stream. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

following cattle access to the stream, peaking at 7.9 x  $10^4$  CFU.100 ml<sup>-1</sup> from values of 5.5 x  $10^2$  CFU.100<sup>-1</sup> upstream of the access site, corresponding to a difference of more than two orders of magnitude. Following the period of stream disturbance by cattle during this event, downstream *E. coli* concentrations decreased and approached upstream concentrations, at 9.7 x  $10^3$  CFU.100 ml<sup>-1-</sup> after approximately 38 minutes. However, they increased again following a second period of cattle access (maximum of 4 animals) to the stream with a maximum concentration of  $1.9 \times 10^4$  CFU.100 ml<sup>-1</sup>, a period when upstream concentrations remained one order of magnitude lower at  $4.6 \times 10^3$  CFU.100 ml<sup>-1</sup>.

# 6.3.3. Loads of nutrients, TSS and E. coli bacteria during cattle access

The upstream-downstream difference in loads (g.15 minute<sup>-1</sup>) of phosphorus (SRP and TP),  $NH_4$ -N,  $NO_3$ -N, TSS and *E. coli* bacteria at peak concentrations, as well as the net upstream - downstream difference in loads during periods of cattle access to the stream are shown in Table 6.7.

The additional TSS loads during cattle in-stream activity corresponding to the time of the peak in recorded concentrations downstream of the access site ranged from +20.34 g.15 min.<sup>-1</sup> (Event 12, August 30, 2017, when there was a maximum of 3 animals accessing the stream for a total of 13.1 minutes) to +249.48 g.15 min<sup>-1</sup> (Event 7, May 10, 2017, which had a maximum of 6 animals and a total duration of access of 25.6 minutes). Additionally, there was a net increase in TSS loads downstream relative to upstream during cattle access for all access events, with the highest net difference registered in Event 9 (June 14, 2017 when a maximum of 4 cows accessed the stream for a total of 32.9 minutes) (+675.90 g TSS).

The additional TP loads at the downstream site at the time of peak concentrations ranged from +0.150 g.15 min<sup>-1</sup> (Event 10) to +2.177 g. 15 min<sup>-1</sup> (Event 9). The total added TP loads



Location - Downstream -- Upstream

**Fig.6.16.** Variation in *E. coli* bacteria concentrations upstream and downstream of the cattle access during events with no cattle access to the stream.



Fig.6.17. Variation in E. coli bacteria concentrations upstream and downstream of the cattle access during the sampling events. Note the differences in scales. Blue shading indicates times of in-stream activity. Brown dashed lines indicate times of defecation (D) and urination (U) episodes. Numbers indicate the maximum number of animals observed simultaneously in the stream for each event.

during cattle access ranged from +0.013 g (Event 10), to +4.442 g (Event 9). Maximum additional loads of NH<sub>4</sub>-N during cattle access to the stream varied from +0.099 g. 15 min<sup>-1</sup> (Event 10) to +1.017 g. 15 min<sup>-1</sup> (Event 16, June 13, 2018, with a maximum of 5 animals and a total duration of 41.2 minutes), with total net loads ranging from +0.055 g to +2.394 g corresponding to the same events. Again, the net variation in loads was lower than the added load at peak concentration for Event 10.

An increase in SRP loads downstream of the access site relative to upstream was estimated in all events with the exception of Events 10 and 12, events during which the intensity of cattle access to the stream was the lowest. However these increases were generally very small and the net variation in downstream SRP loads was actually negative for 5 out of 7 sampling events, when SRP loads were lower downstream of the access site than upstream during periods of cattle access. The exceptions were Event 2 (September 21, 2016, with a maximum of 5 animals accessed the stream simultaneously and access lasting 18.8 minutes in total) and Event 7. In these events, the additional SRP loads at the downstream site at the time of maximum concentrations were +0.316 g. 15 min<sup>-1</sup> and +0.083 g. 15 min<sup>-1</sup>, and net differences were +0.404 g and +0.135 g, respectively.

Nitrate loads downstream of the access site were actually lower than those measured upstream during cattle access for 4 out of the 7 sampling events when cattle were present, including at the time of maximum downstream concentrations. The exceptions were Events 2, 11 (August 8, 2017, with a maximum of 3 animals simultaneously in the stream and which had access that lasted 23.0 minutes) and Event 16, when the downstream loads at peak concentrations of +1.287 g. 15 min<sup>-1</sup> NO<sub>3</sub>-N, +1.197.g 15 min<sup>-1</sup> g and +3.384 g. 15 min<sup>-1</sup>, respectively.

Finally, there was a net increase in the load of *E. coli* bacteria (i.e. total additional CFU) for all events where cattle accessed the stream (Table 6.7). The additional *E. coli* loads at peak concentrations ranged from  $+9.3 \times 10^7$  CFU in Event 12 to  $+3.7 \times 10^9$  CFU in

Table 6.7. Calculated differences in nutrie	t, TSS and <i>E. coli</i> bacteri	ia loads at the downstream of	f the site relative to upstream	loads during periods of
cattle in-stream activity.				

Event	SRP (g)		TP (g)		NH₄-N (g)		NO <sub>3</sub> -N (g)		TSS (g)		<i>E. coli</i> (CFU)	
	Max	Net total	Max	Net total	Max	Net total	Max	Net total	Max	Net total	Max	Net total
2	0.316	0.404	Ν	NA	1.008	2.294	1.287	-2.867	61.830	135.946	N	IA
7	0.083	0.135	0.665	1.227	0.576	0.800	-3.015	-15.696	249.480	407.469	2.5 x 10 <sup>9</sup>	4.7 x 10 <sup>9</sup>
9	0.085	-0.551	2.177	4.442	0.558	0.678	-0.288	-0.688	197.910	675.898	1.0 x 10 <sup>9</sup>	3.7 x 10 <sup>9</sup>
10	-0.059	-0.209	0.150	0.013	0.099	0.055	-1.566	-3.504	68.850	115.882	Ν	IA
11	0.020	-0.333	0.493	1.823	0.567	0.621	1.197	-3.576	182.160	574.135	5.9 x 10 <sup>8</sup>	1.9 x 10 <sup>9</sup>
12	-0.003	-0.058	0.362	0.514	0.225	0.570	-0.027	-1.674	20.340	79.275	9.3 x 10 <sup>7</sup>	2.4 x 10 <sup>8</sup>
16	0.052	-0.029	Ν	١A	1.017	2.394	3.384	1.429	142.470	288.737	3.7 x 10 <sup>9</sup>	9.7 x 10 <sup>9</sup>

*Max* = difference in loads downstream relative to upstream corresponding to the highest concentration downstream during cattle access during that event. *Net total* = sum of differences in downstream loads relative to upstream loads during the total duration of cattle access to the stream. NA – data not available due to issues during sample collection or processing.

Event 16, and net differences in loads varied from +2.4 x  $10^8$  CFU and +9.7 x  $10^9$  CFU for the same events.

Despite these increases in loads when cattle accessed the stream, for all parameters with the exception of *E. coli*, the upstream-downstream difference in loads for periods of cattle access fell within the range of the upstream-downstream difference calculated for those times when no cattle were present (Fig. 6.18 and 6.19). Moreover, the difference in loads for *E. coli* during periods of access was always positive, while in periods of no access, they were close to zero (Fig. 6.18). *E. coli* was also the only parameter for which the magnitude of increase in loads coinciding with cattle access generally increased with the increasing intensity of access when quantified in cattle-minutes (Fig 6.19). Finally, the distribution of the dataset for the upstream-downstream difference in loads during periods of cattle access was significantly different from that during periods with no cattle access for three parameters only: *E. coli*, TSS and NH<sub>4</sub>-N (Table 6.8, Fig. 6.20). In contrast, there were no significant differences found for SRP, TP or NO<sub>3</sub>-N loads (Fig. 6.19).

**Table 6.8.** Statistical parameters for the Kolmogorov-Smirnov test with bootstrap resampling (n =1000). Significant effects are shown in bold.

Parameter	n (Access, No	Р	n
i arameter	access)	D	β
E. coli	21, 58	0.485	< 0.001
TSS	26, 121	0.488	< 0.001
TP	17, 84	0.276	0.232
SRP	26, 122	0.146	0.754
NH₄-N	26, 122	0.391	0.003
NO <sub>3</sub> -N	26, 122	0.189	0.431



**Fig.6.18.** Violin plots for the differences in loads downstream in relation to upstream of nutrients, TSS and *E. coli* bacteria during cattle access to the stream and during periods of no access.


**Fig.6.19.** Scatterplots showing the relationship between the differences in loads of nutrients, TSS and *E. coli* bacteria downstream relative to upstream during periods of cattle access to the stream (orange) and periods of no access (blue), intensity of access expressed in cattle-minutes.



**Fig.6.20.** Boxplots and empirical cumulative distribution functions (ECDF) for the difference in loads of a) TSS, b-e) nutrients and f) *E. coli* bacteria downstream of the access site relative to upstream during periods of cattle access to the stream and periods of no access (grey line on ECDF = Access, black = No Access); points are the values where there is a maximum difference between the two distributions.

#### 6.4. Discussion

Unrestricted cattle access to watercourses has been associated with water quality deterioration as a result of excess nutrient and fine sediment inputs and faecal contamination. However, there is a general paucity of research seeking to assess and quantify such impacts, in particular in the European context (O'Callaghan et al., 2018), and especially those that quantify these impacts at a high temporal resolution and/or concurrent with cattle access. The present study aimed at assessing these immediate impacts of cattle access to watercourses on a number of water quality parameters in a near real-time experiment (i.e. at a high temporal resolution that captured cattle activity). The study found that cattle in-stream activity consistently led to increased water concentrations of faecal bacteria (E. coli), even when no direct defecation in stream waters was registered. Moreover, these increases in concentration resulted in significantly higher loads to the stream. Additionally, cattle in-stream activity led to an increase in water total suspended solids concentrations and loads, and similarly significant but more variable increases in ammonium loads. In contrast, the study found that cattle access did not have a significant effect on water loads of SRP, TP or nitrate, a result which is consistent with the few other studies that are available in near-real time (e.g. Demal, 1982; Wilson and Everard, 2017).

#### 6.4.1. Impact of cattle access on total suspended solids in waters

The study hypothesised that total suspended solids would increase downstream of access sites whenever cattle entered the stream and the streambed sediment was disturbed. Factors potentially influencing the magnitude of such impacts include the number of animals in the stream at a given time, the intensity of activity (i.e. how frequently the animals move in the stream), as well as the particle size distribution of the sediment and the amount available for resuspension (Terry et al., 2014). The deposition of faeces in the stream water will also contribute to total suspended solids concentrations. Cattle faeces are mostly composed by

water (Bond et al., 2012), with 11% - 18% of the faeces fresh weight consisting of solids (Bond et al., 2012; James et al., 2007). Thus, a single deposit of 1.9 kg (James et al., 2007) would add to the stream waters roughly between 206 g and 342 g of fresh organic solids.

In the current study, TSS concentrations consistently increased downstream of the access site during cattle in-stream activity. However, the relationship between the magnitude of this increase and the intensity of cattle access in cattle-minutes was not clear, and occasional increases were also observed when no cattle were present. The total added TSS load at the downstream site ranged from 79.3 g in Event 12 (over 25 cattle-minutes) to 675.9 g in Event 9 (over 81 cattle-minutes), whereas the events with the lowest and highest intensity of access, Events 10 (2 cattle-minutes) and 16 (99 cattle-minutes), had total added loads during access of 115.3 g and 288.7 g, respectively, although in the former event, this was not substantially different from the TSS variation at the study site on that occasion. Although immediate increases in TSS water concentrations during cattle access have been reported in the limited number of studies where this has been quantified (e.g. Demal, 1982; Terry et al., 2014; Wilson and Everard, 2017), Terry et al. (2014) did not find a clear relationship between suspended solids concentrations and the total number of cattle feet in the stream. The authors hypothesised that this resulted from a combination of the sporadic nature of episodes of cattle in-stream activity together with varying intensities of access during the study, with in-between periods of minutes to days, and irregular flow patterns, leading to fluctuating sediment stores available for resuspension (Terry et al., 2014). Such processes will likely also be in operation in the current study.

It is of note that the variation in TSS loads downstream in relation to upstream during cattle access fell within the background range of variation observed during periods when cattle were absent from the stream, which included sporadic episodes of large increases in TSS concentrations either upstream or downstream of the site. In their study, Terry et al. (2014) also reported that 57.9% of the occasions when TSS concentrations exceeded the guideline

threshold of 0.025 mg.L<sup>-1</sup> defined by the EU Freshwater Fish Directive were caused by cattle in-stream activity, but that these only represented 3.6% of the total TSS exports, with flow being the main agent of sediment loss in the area. Additionally, wildlife visiting the stream could also have caused sporadic increases in TSS concentrations at either sampling locations, although in the current study such instances would have been captured on the cameras but were not.

#### 6.4.2. Impact of cattle access on E. coli bacteria in waters

Stream water concentrations of *E. coli* increased downstream of the cattle access site in all sampled events that captured cattle in-stream activity. Peak downstream concentrations were generally between one and two orders of magnitude higher than the *E. coli* concentrations registered upstream of the site at the equivalent time. The highest concentration of *E. coli* bacteria in waters downstream of the site during cattle access was observed in Event 16 (7.9 x  $10^4$  CFU.100 ml<sup>-1</sup>), which also corresponded to the highest added load ( $3.7 \times 10^9$  CFU) of *E. coli* relative to bacteria levels measured upstream. Wilson and Everard (2017) also reported significant increases in faecal coliforms concentrations downstream of their study site in relation to upstream during cattle access, however, their increases were rather modest (+85%) in comparison to the current study, whereas Demal (1982) observed increases in faecal coliforms downstream of access sites during cattle access to the stream in 5 out of 9 cattle access events sampled at 5 different sites, which also corresponded to increases of one to two orders of magnitude compared to upstream concentrations.

Increases in faecal bacteria concentrations downstream of the access site during cattle access are expected due to either in-stream defecation, and to disturbance of streambed sediments that are enriched in faecal matter as a result of cattle activity (Bragina et al., 2017). In addition, cattle often defecate at the edge of the stream, and it is possible that

some of the freshly deposited faecal matter is transported to the stream in cattle hooves due to cattle movement at the site, potentially leading to increased *E. coli* concentrations. Assuming cattle faeces contain roughly  $10^7$  CFU.g dry wt<sup>-1</sup> (Table 4.1) and an average weight of a faecal deposit of 1.9kg containing an average of 15% solids (James et al. 2007), a single in-stream defecation would deposit in the stream roughly 2.9 x  $10^9$  CFU *E. coli*. In Event 10, when an in-stream defecation episode was registered at minute 94, *E.* coli concentrations increased to  $1.7 \times 10^4$  and  $2.4 \times 10^4$  CFU.100 ml<sup>-1</sup>, between minutes 87 and 117, when upstream concentrations at equivalent times were  $3.5 \times 10^2$  and  $4.2.\times 10^2$  CFU.100 ml<sup>-1</sup>. When water flow is taken into account, these increases represented an added load of  $4.2 \times 10^9$  CFU *E. coli* in 30 minutes. Additionally, the highest increase in *E. coli* in water in Event 16 coincided with an episode of defecation at the edge of the stream (at minute 95), where the faecal deposit possibly reached the stream water due to the cattle movement.

Direct defecation in-stream waters or at the stream edge also causes the sediment to become enriched in faecal matter, and it has been widely demonstrated that faecal organisms can persist in sediments (Garzio-Hadzick et al., 2010; Ishii et al., 2007; Ishii and Sadowsky, 2008; Badgley et al., 2011; Shelton et al., 2014). Given the high concentration of *E. coli* in the stream sediment at the access site (Table 6.2), disturbance of the sediment by cattle is likely to cause a substantial increase in *E. coli* concentrations downstream of the site. For instance, in Event 12, when no direct defecations in stream waters were registered, the highest concentration of *E. coli* bacteria measured downstream of the access site was  $5.1 \times 10^3$  CFU.100 ml<sup>-1</sup>, which represented an increase in bacterial loads of  $9.3 \times 10^7$  CFU in relation to upstream at the equivalent sampling time. Downstream TSS concentrations at the same sampling interval were 11.0 mg.L<sup>-1</sup>, corresponding to an increase in TSS loads of 19.8 g in relation to upstream loads. Based on the average *E. coli* concentration in stream sediments at the study site (7.3  $\times 10^6$  CFU.g dry wt<sup>-1</sup> sediment), this increase in TSS loads

would correspond to an added bacterial load of  $1.4.x \ 10^8$  CFU, which is very similar to the observed increase in *E. coli* loads at that time.

In this study, similar total added loads of *E. coli* bacteria during cattle access were seen across sampling events, with the highest total loads observed with the access was most intense, as expressed by total cattle-minutes (Event 16, 99 cattle-minutes), and the lowest when access was the least intense (Event 12, 25 cattle-minutes). In the study conducted by Demal (1982) in Canada, the increases in faecal coliform levels did not appear to be correlated to the intensity of access or the number of in-stream defecations. For example, the author reported similar increases in faecal coliform levels in two different cattle access events at two different study sites, one in which 30 animals accessed the stream for 31 minutes and 7 defecations were registered, and the other with 9 animals in the stream, total duration of 11 minutes and no in-stream defecations. However, that study was conducted in sites that were very heterogeneous in physical characteristics (e.g. length, depth, degree of bank erosion), stocking density and type of access (e.g. unlimited versus limited and improved), thus it is difficult to draw conclusions from it.

#### 6.4.3. Impact of cattle access on NH<sub>4</sub>-N in waters

Cattle access also had a significant impact on NH<sub>4</sub>-N loads downstream of the access site, but the pattern of variation of NH<sub>4</sub>-N concentrations during access was less clear. During both periods of cattle access and non-access to the stream, NH<sub>4</sub>-N concentrations fluctuated at both sampling points. Ammonium is readily assimilated by macrophytes, algae and microbial biofilms, removing NH<sub>4</sub>-N from the water column (Birgand et al., 2017). Additionally, it can bind to organic and inorganic particles through ion exchange (Bernot and Dodds, 2005), and can be then removed from the water column through sedimentation. Conversely, NH<sub>4</sub>-N retained in sediments resulting from organic matter remineralisation can be returned to the water column upon sediment disturbance (Bernot and Dodds, 2005) or

diffusion. In this study, the differences in NH<sub>4</sub>-N loadings downstream of the access site in relation to upstream values during periods of cattle activity fell within the natural range of variation seen at periods of cattle absence from the stream. Nevertheless, positive differences, i.e. increases in loadings downstream of the site, were more frequent during cattle in-stream activity. Moreover, the largest increases in downstream loadings tended to coincide with times of in-stream urinations. Considering an average urine volume of 1.8 L (Misselbrook et al., 2016) and an average urine TN concentration of 8.2 g.L<sup>-1</sup> (Shelbie et al., 2015), a single urination episode would deposit in the stream 0.36 g of NH<sub>4</sub>-N. Urea, which is the predominant form of nitrogen in cattle urine (Selbie et al., 2015), will also yield free ammonium ions as a result of its decomposition in the aquatic system. However, in the absence of the enzyme urease, urea is a very stable molecule, with a half-life time of about 3.6 years (Ray et al., 2018). Of seven events that captured cattle access to the stream, direct urination episodes occurred in four events. In these events, the highest increases in NH₄-N concentrations downstream of the site relative to upstream concentrations generally coincided with the intervals in which urinations occurred, corresponding to increases in loads ranging from 0.62 g NH<sub>4</sub>-N (Event 11) to 1.18 g NH<sub>4</sub>-N (Event 2). Additionally, a peak in NH<sub>4</sub>-N downstream concentrations coinciding with an episode of urination in the edge of the stream (at minute 91), which may have reached the stream waters, was observed during Event 16, corresponding to an increase of 1.13 g of NH<sub>4</sub>-N downstream of the site. Demal (1982), reported increases in free ammonia concentrations in streamwater resulting from cattle in-stream activity in three of nine events sampled, of +0.13 mg.L<sup>-1</sup>, +0.16 mg.L<sup>-1</sup> and +0.34 mg.L<sup>-1,</sup> the latter two sampled at the same site. However, episodes of in-stream urination (10) were only registered on one of those events (when there was an increase of  $+0.13 \text{ mg}.\text{L}^{-1}$ ) (Demal, 1982).

#### 6.4.4. Impact of cattle access on SRP and TP in waters

Cattle access to the stream did not cause significant increases in loadings of SRP or TP. Both SRP and TP might be expected to increase during cattle in-stream activity mainly due to in-stream defecation and resuspension of nutrient- and organic matter-enriched sediment. In cattle faeces, phosphorus concentrations can be highly variable depending on type of diet, animal liveweight and reproductive status. Roughly, using the data reported by James et al. (2007) for heifers/dry cows, a single defecation episode would add 1.9 kg of fresh faeces into the stream, containing 1.9 g of TP, of which approximately 60% (Dou et al., 2002), or 1.14 g, would be in the form of readily soluble P. Increases in SRP loadings downstream of the access site during or following times of in-stream defecation were observed in Event 2 (+0.316 g SRP) and Event 7 (+0.083 g of SRP), but not in Event 9. Additionally, the differences in SRP loads observed at times of cattle in-stream activity fell within the range of differences observed at times of cattle absence. Similarly, loadings of TP downstream of the access site showed increases coinciding with times of cattle defecation, of +0.665 g in Event 7 and +2.177 g in Event 9. However, similar changes were also observed at times of no access, and, although the differences in loadings downstream of the site tended to be positive more often at cattle access times than at times of cattle absence, this effect was found to be nonsignificant. Demal (1982) reported only slight or negligible increases in filtered reactive phosphorus during and following cattle in-stream activity, with the exception of one of their study sites, where concentrations increased from 0.048 mg.L<sup>-1</sup> to 0.134 mg.L<sup>-1</sup> and 0.039 mg.L<sup>-1</sup> to 0.079 mg.L<sup>-1</sup> in two access events. In the same events, TP concentrations increased from 0.071 to 0.804 mg.L<sup>-1</sup> and 0.077 to 0.791 mg.L<sup>-1</sup>. Instream defecations were recorded in the first of these events (2 defecations over a 6 minutes period), however events sampled at other sites with a higher number of defecations and a longer period of in-stream activity did not result in increases in SRP or TP concentrations. The author attributed this to the sediment characteristics at the site, which was largely composed by fine sediment, and the low percentage of vegetative coverage, causing the sediment to be more prone to resuspension of particulate P and release of interstitial soluble P (Demal, 1982). Wilson and Everard (2017) also did not find a consistent relationship between cattle in-stream activity and increases in SRP concentrations downstream of the access site, with changes in SRP concentrations falling within the range of the background variation between sampling points. Phosphorus is a highly particle-reactive ion, undergoing reactions of sorption and desorption with particulate material and sediments, and, in its dissolved inorganic forms, it is readily assimilated by plants, algae and microorganisms, (Reddy et al., 1999; Jennings et al., 2003). Such characteristics may hinder the observation of measurable differences in water phosphorus concentrations that might result from cattle in-stream activity.

#### 6.4.5. Impact of cattle access on NO<sub>3</sub>-N in waters

Nitrate concentrations did not show a pattern of variation in response to cattle in-stream activity. Similar to this study, Demal (1982) reported negligible changes in nitrate concentrations in all monitored cattle access events. Nitrogen in cattle faeces is predominantly in organic forms, whereas urine nitrogen is predominantly urea-N (Selbie et al., 2015), which may explain the lack of an immediate effect of cattle in-stream activity on water nitrate concentrations. Nevertheless, decomposition of cattle faeces in sediments at the access site can contribute to nitrate water levels when organic nitrogen is remineralised (Birgand et al., 2007).

# 6.5. Implications

This study revealed that cattle access to watercourses has an immediate clear and significant impact on loads of *E. coli* bacteria and a less obvious but still significant impact on

TSS and NH<sub>4</sub>-N loads. Faecal bacteria such as *E. coli* are directly added to the water column during in-stream defecation, but E. coli would also become available through resuspension of viable faecal organisms from the stream sediment based on the concentrations described in Chapter 4. This contamination of water resources with potentially pathogenic faecal organisms represents a potential risk for human and animal health, including for E. coli O157, which has its main reservoirs in ruminant animals and in cattle in particular (Óhaiseadha et al., 2016). Furthermore, it is well established that faecal bacteria can survive and potentially grow in stream sediments, which have also been considered as genetic reactors where bacteria routinely exchange genetic material, including genes for antibiotic resistance (Jang et al., 2017). Additionally, excess sediment resulting from increased bank erosion and resuspension of bed sediment during cattle in-stream activity can also have important ecological impacts. Excess fine sediment can smother the substrate and clog substrate interstices (Wright and Berrie, 1987), leading to habitat loss and less diverse macroinvertebrate communities (Braccia and Voshell, 2006). Fine sediment can also clog the gills (Relyea et al., 2012) or filter-feeding organs of aquatic organisms (Lemly, 1982). Increased turbidity conditions can also reduce the feeding efficiency of predatory organisms (Jones et al., 2012) and have detrimental impacts on primary producers (Izagirre et al., 2009). Cattle access to watercourses therefore represents a route of systematic contamination of freshwaters with potentially pathogenic organisms and fine sediments, and exclusion of cattle from agriculture streams and riparian zones is desirable.

7. A short study investigating the effectiveness of fencing as mitigation measure

# Chapter 7. A short study investigating the effectiveness of fencing as mitigation measure

# 7.1. Introduction

Streamside fencing to exclude livestock from watercourses has been suggested as an effective method to mitigate the impacts of cattle access on freshwater ecosystems (Line, 2003; Scrimgeour and Kendall, 2003; Miller et al., 2010). One of the main effects of streamside fencing is preventing livestock from directly defecating and urinating within the stream channel (e.g. Miller et al., 2010). An additional result of fencing is reduction of inputs from runoff and erosion from those areas of un-vegetated or sparsely vegetated stream banks caused by livestock trampling and excess grazing (O'Callaghan et al., 2018). Exclusion fencing also promotes the establishment of the vegetation in riparian buffer areas which further aids in reduction of particulate input to streams through filtering overland discharge and retention in vegetation (Liu et al., 2008).

As described in Chapter 3, fencing to exclude livestock from watercourses as a water quality protection measure has been included in most European agri-environment schemes (AES) (Dworak et al., 2009), including in Ireland, where it has been part of the measures of all AES since the implementation of the first scheme in 1994. However, despite this common inclusion in policy, relatively few studies have evaluated the effectiveness of such measures. This is particularly apparent in Europe, and indeed in Ireland (Finn and Ó hUallacháin, 2012).

In a study in USA, Larsen et al. (1994) determined that a 0.61 m riparian fenced buffer had the potential to reduce faecal coliform concentrations entering a stream by 83%, while bacterial loads were reduced by 95% with a 2.31m buffer. In their study in North Carolina (USA), Line et al. (2000) reported reductions in stream water nitrate, total Kjeldahl nitrogen

(TKN) and total phosphorus (TP) concentrations of 33%, 78% and 76%, respectively. In a later study, Line (2003) observed statistically significant reductions in water levels of faecal coliforms and enterococci, of 65.9% and 57% respectively, following streamside fencing. Galeone (2000) also observed reductions in water nitrogen and phosphorus loads following the installation of livestock fencing in Pennsylvania (USA). Similarly, in a study in Vermont (USA), Meals (2001) observed reductions TKN and TP following the installation of measures to mitigate cattle access including fencing, improved stream crossings and bank stabilisation measures in the experimental catchment while observing increases in the control catchment. Additionally, the author detected decreases in E. coli, faecal coliform and faecal streptococcus levels in waters after cattle exclusion in the experimental catchment, while reporting an increase in faecal organism counts in the control catchment during the same period. Similarly, although not strictly related to fencing, Vidon et al. (2008) sampled above and below a stream section to which cattle had unrestricted access and observed a 36-fold increase in E. coli concentrations in stream waters and suggested that restricting access would result in improvements in water quality. Likewise, in Scotland, Kay et al. (2007) saw between a 66% and a 81% reduction in faecal indicator levels during high flows following remediation measures. While the effects of cattle exclusion on water levels of contaminants have been widely investigated, less attention has been given to the sediment compartment. In one such study, conducted in the Milltown Lake catchment in NE Ireland, Bragina et al. (2017), reported significant lower E. coli sediment concentrations in a stream that had been fenced to exclude cattle in comparison to those observed in an unfenced stream with similar characteristics in the same catchment.

Other studies, in contrast, have reported no significant improvement in water quality parameters following the installation of streamside fencing. For instance, Miller et al. (2010) found no difference in water quality variables (TP, TN, DO, temperature) in response to cattle exclusion. Similarly, despite the improvements regarding water faecal contamination, Line (2003) observed no significant changes in upstream to downstream ratio levels of

dissolved oxygen, pH, temperature and specific conductivity in cattle exclusion areas. In a study in the Milltown Lake catchment, Veerkamp (2019) reported that the estimated annual export rates of TP and TN were lower in the fenced stream (0.57 kg TP.ha<sup>-1</sup>.yr<sup>-1</sup> and 5.99 kg TN.ha<sup>-1</sup>.yr<sup>-1</sup>) than in the unfenced stream (0.77 kg TP.ha<sup>-1</sup>.yr<sup>-1</sup> and 7.24 kg TN.ha<sup>-1</sup>.yr<sup>-1</sup>). However, the author observed that this fencing was only effective during the grazing season, when the riparian vegetation was denser. Conversely, in the winter and early spring periods, when the riparian vegetation had died back, the TP loads in the fenced stream exceeded those in the unfenced stream (Veerkamp, 2019). Overall, it is apparent that while streamside fencing can have positive effects on freshwater faecal contamination, the effectiveness of such measures reducing livestock impacts on water physicochemical parameters remains unclear.

As discussed previously, despite the lack of empirical evidence as to their costeffectiveness, provisions for preventing cattle access have been included in many European agri-environment schemes, as well as in every Irish agri-environment scheme, to date (i.e. REPS, AEOS, GLAS). It has been acknowledged that in the absence of empirical evidence on the actual impact of cattle access and on the effectiveness of cattle exclusion, it is difficult to justify full riparian fencing of watercourses as a cost-effective approach to maintain or enhance freshwater ecosystems (Terry et al. 2014). Thus the aims of this short study were to:

- Assess changes in sediment physicochemical and microbial parameters in the bed sediment compartment at cattle access sites after ~ one year of fencing (short-term fencing), in a *before-after* comparison study;
- b. Assess the impact of longer term (nine years) cattle exclusion fencing on water physicochemical and microbial parameters (i.e. SRP, NH<sub>4</sub>-N, NO<sub>3</sub>-N, *E. coli*) using data from the Milltown Lake catchment in a *paired treatment-control* study. In this study, two similar streams (Table 7.3), one of which was fenced to exclude cattle in 2008, were sampled at monthly intervals during

approximately one year. A second aim of this study was to gather complementary data for a study on the effects of excluding cattle on macroinvertebrate communities under the COSAINT project.

#### 7.2. Site description and methods

#### 7.2.1. <u>Site description</u>

#### 7.2.1.1. Before-after study on the impacts of fencing on short-term

Six sites were assessed where cattle exclusion fencing had been installed by the famer or installed by the project team. This approach facilitated *before and after* analysis in these sites for sediment nutrient and sediment *E. coli* levels. A larger number of sites had been initially selected for this *before*-after comparison study, however due to non-compliance with fencing by a number of farmers at the time of re-sampling, study site selection had to be re-evaluated during the project. This resulted in a limited number of sites suitable for the study as well as microbiological data not available for one of the six sites.

The six cattle access points were located in five catchments (Fig. 7.1 and 7.2) in the east and south of Ireland and were sampled for bed sediment prior to, and one year following exclusion of cattle from streams via fencing, in October of 2016 and October of 2017, respectively. Four of the six sites are described in Chapter 3 and were part of the studies described in Chapters 4 and 5; these were sites MT1, CM1, BWA and BWC. Two extra sites, SN1 and SN2, were included in this study, which were located along the moderate status, second order Blacklion stream in Co. Carlow (Tables 7.1 and 7.2). The area is poorly drained and low lying lands are prone to flooding in the winter. The main soil group found is Peaty Gley, and the geology primarily consists of limestone with some sandstone and granite. The streams are tributaries of the River Slaney. Fencing of the access sites (two strand barbed-wire fencing for cattle exclusion, typically 1.5 m from the bank edge as per GLAS regulations) allowed for some visible recovery of the cattle access sites (reduction in visible fine sediment, some vegetation growth).



Milltown Lake catchment (MT)

Munster Blackwater catchment (BW)

**Fig.7.1.** Map of the study sites in the *before-after* study of fencing effectiveness as a water quality protection measure.

**Table 7.1.** Study sites in the Slaney catchment (not described previously in Chapter 3). \*Information derived predominantly from Edenireland.ie; information at time of sampling (2016/2017).

Catchment	Slaney		
River	Blacklion Stream		
County	Carlow		
Total appual precipitation (mm) <sup>1</sup>	591.6 (2016)		
	616 (2017)		
Ecological status (Sampling period)	Moderate		
Biological status(Sampling period)	NA		
Chemistry conditions(Sampling period)	Pass		
Recent chemistry trend(Sampling period)	Pass		
Nutrient Condition	Pass		
Ortho-P status quality (trend)	High (upwards)		
WFD Risk	At Risk		
Waterbody trend	No change		
Significant Pressure	Agriculture		

<sup>1</sup>Precipitation data derived from Tullow (Ardoyne Glebe) Met Eireann station.

	SN1	SN2		
Stream order	2	2		
Stream width (m)	1.72 1.71			
Reach gradient (%)	-4.70%	-3.10%		
Soil type	Poorly drained			
Geologic formations	Granite			
Site description	Open access site, vegetated banks, steep banks	Open access site, vegetated banks		

**Table 7.2.** Summary characteristics of the study sites in the Slaney catchment.

# 7.2.1.2. Paired treatment-control study on the effects of fencing on long-term

The longer-term efficacy of fencing as a means of improving some water quality parameters was also tested in the Milltown Lake catchment of Co. Monaghan. Here, one tributary in the catchment had been completely fenced in 2008, as part of a previous project that ran from 2005 - 2010 (the National Source Protection Pilot Project, NSPPP: Linnane et al. 2011) (see Fig. 7.2 and Tables 7.3 and 7.4). This western tributary, which is located in a sub-catchment fed from the Carnagh Lake in Tievnamara (TV), had fencing installed at a distance of 1.5 m from the stream edge at that time, excluding livestock from the water as well as allowing the re-establishment of streamside vegetation, thus providing a buffer strip. The other two tributaries in the Milltown Lake catchment, Gentle Owen (GO) (middle), and Tullycaghney (TH) (eastern) still had cattle access to the stream water. A survey conducted in 2010 by Veerkamp (2019) identified one unfenced cattle access site in the TV tributary, which was

relatively small in comparison to the remaining sites, serving as a crossing point. In the TH tributary (unfenced), 18 cattle access sites were identified. In total, 12 cattle access sites had been sampled in October 2008, six in the TV tributary and six in the TH tributary, prior to fencing, for a limited set of water column parameters (TRP, NH<sub>4</sub>-N and *E. coli*). These sites were re-sampled in the current study at approximately monthly frequency from December 2017 to November 2018, approximately nine years post fencing Data on discharge of the two tributaries were collected from the EPA Drumleek station approximately 500 m upstream (54°8'48"N; 6°42'15"W) from the inflow to Milltown Lake. Continuous stream discharge data (15 minute intervals) are available at this site from the EPA Hydronet website (hydronet.epa.ie).

Table 7.3. Characteristics of the streams used in the paired control-treatment study,	in the
Milltown Lake catchment (from Veerkamp, 2019).	

Characteristics	Catchment	Fenced tributary (TV)	Unfenced tributary (TH)	
Total area (km²) excluding Milltown Lake	28.8	9.6	10.1	
Stream length (km)	24.6	8.4	6.3	
Dwellings per tributary area (2010)	748	232	247	
Cattle per tributary area (2010)	5554	1722	1833	



**Fig.7.2.** Map of the study sites in the paired *treatment - control* study of fencing effectiveness as a water quality protection measure.

Name	Site description	Location	Stream order	Width (cm)		
		Loodiion		Mean	Max	Min
TV1	No cattle access	54.199751 -6.728785	1	NA	NA	NA
TV3	Cattle access site on fenced stream	54.186766, -6.735303	1	152	180	110
TV4	No cattle access	54.179181, -6.737842	2	226	305	200
TV5	No cattle access	54.172003, -6.738826	2	258	300	165
TV6	No cattle access	54.162057, -6.733004	2	NA	NA	NA
TV7	No cattle access	54.155433, -6.731612	2	251	285	200
TH1	Cattle access site	54.197612, -6.696669	1	NA	NA	NA
TH2	No cattle access	54.187583, -6.692207	1	167	285	62
TH3	Access site, steep field	54.183814, -6.692985	1	228	300	180
TH4	Access site	54.168668, -6.695717	1	189	240	97
TH5	Access site	54.157916, -6.690028	2	278	390	130
TH6	Access site	54.157000, -6.714875	2	221	253	140

**Table 7.4.** Description of the sites sampled in the fenced (TV) and unfenced (TH) tributaries in the Milltown Lake catchment (from Bragina, 2017).

# 7.2.2.1. Before-after study on the impacts of fencing on short-term: sediment nutrient and E. coli sampling

Sediment sampling and analysis techniques for nutrient and *E*. coli concentrations were similar to those described in Chapters 4 and 5. At each site, however, the stream bed sediment was sampled at three locations (location DS was not considered in this study): at the cattle access site, i.e. where cattle actively used the stream (CAS); upstream (20 - 322 m) of the access site, where animals had no access to the stream either due to fencing or natural physical barriers (US); and at the interface (edge) of the stream water level, at the access path used by cattle to enter the stream (INT). For microbiological analysis, three sediment samples were collected randomly at the cattle access sites (CAS) and three were collected upstream (US) (10 - 300 m) of these sites (i.e. total of six per site).

#### 7.2.2.2. Paired treatment-control study on the effects of fencing on long-term

One grab sample was taken at six sampling sites on the TV tributary and six sites on the TV tributary. Note that these sites were previously used for water sampling in the NSPPP (Linnane et al, 2010). Sample were analysed for SRP,  $NH_4$ -N and  $NO_3$ -N.

The samples were collected using a polyethylene 0.5 L beaker previously disinfected with 70% industrial methylated spirit (IMS), acid-washed with a 70% HCl solution and rinsed several times with ultrapure Milli-Q water. Water samples for nutrient analysis were transferred into polyethylene bottles previously acid-washed in the same manner, whereas water samples for microbiological analysis were transferred to previously autoclaved 100 ml Duran bottles. The sampling beaker was rinsed with IMS, acid-wash solution, ultrapure Milli-Q water and finally three times with stream water in between samples. Samples were placed

in cool boxes and transported to the laboratory where they were kept in the dark at 4°C until analysis.

# 7.2.3. Laboratory analysis

# 7.2.3.1. Before-after study on the impacts of fencing on short-term

Sediment samples were processed and analysed for *E. coli* and nutrient concentrations as described in Chapters 4 and 5, respectively.

# 7.2.3.2. Paired treatment-control study on the effects of fencing on long-term

The methods for analysis of water nutrient and *E. coli* concentrations were similar to those described in Chapter 6. Additionally, microbiological samples were also analysed using the Colilert Quanti-tray method (IDEXX Laboratories, 2012) to replicate the data collected for the NSPP project.

# 7.2.4. Data analysis

All *E. coli* data were log transformed before statistical analysis. Significant differences were assessed between the US and CAS locations (and between the US and INT locations) using a paired t-test in R (R Core Team, 2018). A Shapiro-Wilk test was used to test for the assumption of normality. Where this assumption was breached, a paired Wilcoxon text was used as an alternative to the paired t-test.

# 7.3. Results

#### 7.3.1. Before-after study on the impacts of fencing on short-term

There were no significant differences in sediment OC, TN or TP concentrations, or in OC:TN ratios, between the upstream sites (US) and the cattle access site (CAS) before fencing across all six sites (Fig. 7.3 and Table 7.5). However, prior to fencing, sediment concentrations for OC, TN and TP were significantly higher at the stream interface (INT) sites, where cattle would have been most likely to congregate, when compared to the US locations for all six sites (Table 7.6 and Fig. 7.4). These increases in concentration from US to INT locations were generally consistent across all six sites. In contrast, post-fencing, there was no significant difference between the US and INT sites for any of the three sediment nutrients assessed (Fig. 7.3, Table 7.6) It should be noted that no fine sediment (<2mm) was found at the INT site for one site (MT1) post-fencing.

Sediment *E. coli* levels did show an increase between the US and CAS locations at three of the five study sites in the Autumn of 2016 (prior to fencing) (n = 3), but the differences were not significant when all five sites were considered (Fig.7.5). This site-specific difference was most pronounced for the Milltown MT1 site, where the mean *E. coli* concentration increased from 2.1 x  $10^2$  CFU g.dry wt<sup>-1</sup> to 2.0 x  $10^5$  CFU g.dry wt.<sup>-1</sup> (t = -22.1, p < 0.001). At BWA, *E.coli* was not detected at the US location but had a mean value of 6.9 x  $10^2$  CFU g.dry wt.<sup>-1</sup> at the CAS location. Concentrations at the US and CAS locations for the MT1 site were also not significantly different from each other post-fencing, with mean values of  $6.2 \times 10^2$  CFU g.dry wt.<sup>-1</sup> respectively.

**Table 7.5.** Bed sediment concentrations of nutrients (mean  $\pm$  S.E., n = 6) and *E. coli* (mean  $\pm$  S.E., n = 5) at the study sites between upstream (US) and cattle access site (CAS): pre-fencing (2016) and post- fencing (2017).

	Pre-fe	Post-f	encing		
Parameter	US	CAS	US	CAS	
OC (mg.g dry wt <sup>-1</sup> )	11.997 <u>+</u> 6.177	8.794 <u>+</u> 2.571	12.180 <u>+</u> 4.440	10.466 <u>+</u> 3.719	
TN (mg.g dry wt <sup>-1</sup> )	1.023 <u>+</u> 0.392	0.796 <u>+</u> 0.198	1.358 <u>+</u> 0.537	1.021 ± 0.289	
OC:TN	10.0 <u>+</u> 1.4	10.8 <u>+</u> 0.7	9.3 <u>+</u> 1.0	9.4 <u>+</u> 1.0	
TP (mg.g dry wt <sup>-1</sup> )	275 <u>+</u> 80	412 <u>+</u> 115	351 <u>+</u> 65	375 <u>+</u> 105	
<i>E coli</i> (CFU.g dry wt⁻¹)	$2.0 \times 10^3$ ± 1.4 × 10 <sup>3</sup>	$4.1 \times 10^4 \pm 2.3 \times 10^4$	$9.9 \times 10^2 \pm 4.4 \times 10^2$	1.4 x 10 <sup>3</sup> ± 6.5 x 10 <sup>2</sup>	



**Fig.7.3.** Change in the mean sediment organic carbon (OC, row 1), total nitrogen (TN, row 2), and total phosphorus (TP, row 3) concentration at the upstream (US) control sites and at the cattle access site (CAS) and from the US sites versus the interface (INT) for the six sites (columns 1 and 3 = pre-fencing (2016) and columns 2 and 4 = post-fencing (2017).

**Table 7.6.** Bed sediment (mean  $\pm$  S.E., n=6) data for the study sites at the upstream control (US) and the interface (INT) at the cattle access site: pre-fencing (2017) and post-fencing (2018); data in bold were significantly different US versus INT. \*significant at the 0.05 level; \*\*significant at the 0.001 level.

	Pre-	fencing	Post-fencing		
Parameter	US	US INT		INT	
OC (mg.g dry wt <sup>-1</sup> )	11.997 <u>+</u> 6.177	23.416 <u>+</u> 6.637**	12.180 <u>+</u> 4.440	12.410 <u>+</u> 5.088	
TN (mg.g dry wt <sup>-1</sup> )	1.023 <u>+</u> 0.392	2.086 <u>+</u> 0.589**	1.358 <u>+</u> 0.537	1.181 <u>+</u> 0.440	
OC:TN	10.0 <u>+</u> 1.4	11.0 <u>+</u> 0.3	9.3 <u>+</u> 1.0	9.2 <u>+</u> 1.6	
TP (mg.g dry wt <sup>-1</sup> )	275 <u>+</u> 80	572 <u>+</u> 215*	351 <u>+</u> 65	349 <u>+</u> 110	



**Fig.7.4.** Change in the mean sediment *E. coli* concentration at the upstream (US) control sites and at the cattle access site (CAS) for the five sites (left = pre-fencing (autumn 2016) and right = post-fencing (autumn 2017).

#### 7.3.2. Paired treatment-control study on the effects of fencing on long-term

Stream water concentrations of nutrients and *E. coli* in the fenced and unfenced tributaries of the Drumleek River are presented in Table 7.7. For the stream water data collected over one annual cycle in the Milltown Lake catchment, there was no significant difference between the values for the TV tributary (fenced) and those for TH tributary (unfenced) for the concentrations of SRP, TP, NO<sub>3</sub>-N and NH<sub>4</sub>-N when all data were included, or when only data for months when cattle would be in the field (Apr - October) were included (Fig. 7.5, Fig. 7.6). Notably, nutrient concentrations in both tributaries followed a very similar pattern, and SRP, TP and NH<sub>4</sub>-N concentrations reached a peak in June 2018 (Fig 7.5). For water column *E. coli* concentrations which were measured (using the membrane filtration technique) four times over the annual cycle (December 2017, January/May/November 2018), there was again no significant difference between the fenced and unfenced tributaries

(Fig. 7.7). Similarly there was no significant difference for stream water *E. coli* concentrations measured using the Colilert method (Fig. 7.7).

Stream	Date	SRP (µg.L <sup>-1</sup> )	ΤΡ (μg.L <sup>-1</sup> )	NH₄-N (mg.L⁻¹)	NO₃-N (mg.L <sup>-1</sup> )	<i>E.coli</i> (CFU.100 ml <sup>-1</sup> )	<i>E. coli</i> (MPN.100 ml <sup>-1</sup> )	Mean monthly discharge (m <sup>3</sup> .s <sup>-1</sup> )
	16/11/2017	-	-	-	-	2.5 x 10 <sup>2</sup> ± 52	-	0.802
	04/12/2017	47 ± 5	77 ± 10	$0.07 \pm 0.02$	$0.59 \pm 0.03$	$1.4 \times 10^2 \pm 28$	149.2 ± 34.0	1.136
	24/01/2018	73 ± 4	158 ± 13	$0.06 \pm 0.02$	$0.47 \pm 0.07$	1.3 x 10 <sup>3</sup> ± 2.2 x 10 <sup>2</sup>	1238.8 ± 267.1	2.359
	16/03/2018	28 ± 2	87 ± 7	$0.10 \pm 0.02$	0.71 ± 0.05	-	-	0.952
	20/04/2018	39 ± 3	96 ± 3	-	-	-	-	0.701
Fenced	10/05/2018	$44 \pm 4$	115 ± 16	$0.09 \pm 0.03$	$0.47 \pm 0.06$	$3.7 \times 10^2 \pm 82$	351.6 ± 135.2	0.219
	28/06/2018	107 ± 20	211 ± 27	0.32 ± 0.15	0.76 ± 0.15	-	-	0.064
	27/07/2018	21 ± 7	48 ± 10	$0.07 \pm 0.03$	0.66 ± 0.13	-	-	0.058
	30/08/2018	21 ± 1	38 ± 2	$0.06 \pm 0.03$	$0.45 \pm 0.08$	-	-	0.144
	10/10/2018	48 ± 8	104 ± 13	0.13 ± 0.06	$0.41 \pm 0.07$	-	-	0.144
	20/11/2018	76 ± 2		$0.07 \pm 0.02$	1.55 ± 0.08	$2.2 \times 10^2 \pm 69$	216.1 ± 75.9	1.283
	16/11/2017	-	-	-	-	2.0 x 10 <sup>2</sup> ±1.0 x 10 <sup>2</sup>	-	0.802
	04/12/2017	30 ± 4	50 ± 7	0.03 ± 0.01	$0.44 \pm 0.08$	1.9 x 10 <sup>2</sup> ± 1.1 x 10 <sup>2</sup>	236.2 ± 122.1	1.136
	24/01/2018	40 ± 3	110 ± 5	0.04 ± 0.01	0.63 ± 0.10	5.2 x 10 <sup>2</sup> ± 68	552.2 ± 149.2	2.359
	16/03/2018	25 ± 3	66 ± 9	$0.14 \pm 0.05$	0.68 ± 0.13	-	-	0.952
	20/04/2018	26 ± 2	58 ± 4	-	-	-	-	0.701
Unfenced	10/05/2018	25 ± 4	80 ± 25	0.04 ± 0.01	$0.28 \pm 0.09$	7.1 x 10 <sup>2</sup> ± 4.9 x 10 <sup>2</sup>	701.7 ± 386.1	0.219
	28/06/2018	73 ± 10	217 ± 32	0.29 ± 0.13	0.86 ± 0.26	-	-	0.064
	27/07/2018	13 ± 3	57 ± 13	0.05 ± 0.02	0.63 ± 0.25	-	-	0.058
	30/08/2018	20 ± 2	40 ± 10	$0.02 \pm 0.004$	0.45 ± 0.09	-	-	0.144
	10/10/2018	36 ± 8	161 ± 84	0.07 ± 0.05	0.26 ± 0.07	-	-	0.144
	20/11/2018	66 ± 3		$0.03 \pm 0.003$	1.50 ± 0.31	$1.5 \times 10^2 \pm 41$	141.5 ± 29.8	1.283

**Table 7.7.** Stream water nutrients and *E. coli* concentrations in the fenced and unfenced streams in the Milltown Lake catchment (mean ± S.E.). Mean monthly discharge is calculated from daily mean discharge available from the EPA monitoring station in the Drumleek river.



**Fig.7.5.** Monthly total precipitation (mm) in the Milltown Lake catchment from December 2017 to December 2018 (data from the Met Eireann station Coose – Castleblayney), mean daily discharge (m<sup>3</sup>.s<sup>-1</sup>) in the Drumleek River (data from the nearest EPA monitoring station) and nutrient concentration variation in the TV tributary (fenced) (black) and TH tributary (unfenced) (grey).



**Fig.7.6.** Stream water nutrient concentrations in the fenced and unfenced streams. Top row: (December 2017 to November 2018, no sample in February and September 2018) (SRP: n = 60; NH<sub>4</sub>-N: n = 48; TP and NO<sub>3</sub>-N: n = 47). Bottom row: during the period of cattle grazing outdoors (April to October 2018, no sample in February 2018) (SRP: n = 36; TP: n = 35; NH<sub>4</sub>-N and NO<sub>3</sub>-N: n = 24).

,



**Fig.7.7.** Stream water *E coli* concentration the fenced and unfenced streams. a) and c) data obtained using the membrane filtration technique ( $\log_{10}$  CFU.100 ml<sup>-1</sup>); b) and d) data obtained using the Colilert method (Most Probable Number (MPN).100 ml<sup>-1-</sup>); a) and b) data collected in November and December 2017, January, May and November 2018 (MF: n = 59; Colilert method: n = 48, no data for November 2017); c) and d) data for May 2018 when cattle were grazing outdoors (n = 12).

#### 7.4. Discussion

While uncontrolled livestock access to watercourses can negatively impact the aquatic system and the riparian area, it has been suggested that such effects are localised and site-specific (Wilcock et al., 1999). Streamside fencing to exclude livestock from watercourses has been adopted as a water quality protection measure, or best management practice (BMP), in many regions globally; however the body of literature assessing its effectiveness is limited and studies have been predominantly undertaken in the USA and New Zealand (O'Callaghan et al., 2018). Furthermore, while many studies have focused on the effects of cattle exclusion fencing on stream water quality (O'Callaghan et al., 2018), few to date have assessed either the impacts of cattle access on nutrient and microbial contamination of stream bed sediment or assessed the effects on excluding cattle on these sediment characteristics. This short study aimed at providing a screening assessment of the effectiveness of fencing in the Irish context.

As seen in Chapter 5, prior to fencing bed sediment nutrient concentrations at edge of the stream, where cattle would congregate to drink (i.e. interface sites) were significantly higher than those for the bed sediments upstream of these sites. Interestingly, the sediments in the main channel (i.e. CAS) did not have significantly higher nutrient concentrations either in Chapter 5 or in this short study, suggesting an ongoing flushing of material downstream (Eder et al., 2014; Terry et al., 2014). These differences, however, were not apparent for the same comparison undertaken approximately one year post-fencing. While other factors may play a role in year to year variability, this likely reflects the effects of cattle absence from the stream during the grazing season of 2017.

The mechanisms by which these sediments at access sites can become nutrient-enriched include both the incorporation of additional organic matter generally from cattle faeces (e.g. Davies-Colley et al., 2004), and the adsorption of phosphate to ion exchange sites of fine

particles (McDowell and Sharpley, 2001). While the number of sites and the assessment time in the current study was limited, the results indicated that excluding cattle reduced the bed sediment nutrient concentrations in these access point stream margins. It is also possible, however, that in a longer study, increased vegetation in the stream margins of the fenced cattle access points could also act as a trap for fine sediment thus potentially increasing concentrations in these areas (Reddy et al., 1999; Sand-Jensen, 1998; Walling and Collins, 2016).

Regarding E. coli contamination, bed sediments were sampled in the main channel only. While the overall change across the six sites was not significant, there were substantially higher sediment E. coli concentrations at the cattle access site than upstream before fencing at individual sites, in particular the MT1 site (2.0 x 10<sup>5</sup> and 2.1 x 10<sup>2</sup> CFU.g dry wt<sup>-1</sup>, respectively). Again this difference was not apparent after one year of fencing, indicating that cattle exclusion likely had a positive impact on the levels of contamination of the bed sediment at the sites. This effect was similar to that reported for a larger two year study in the Milltown Lake catchment by Bragina et al. (2017) who observed similar concentrations and found significantly higher levels of *E. coli* in the bed sediments of the fenced TV tributary compared to the unfenced TH and GO tributaries. While high levels of E. coli in bed sediment can also persist over time, Jamieson et al. (2005b) pointed out that E. coli in bed sediments will be flushed and transported downstream. Bragina et al. (2017) also reported a decrease in bed sediment concentrations between October and April, which they suggested was due to flushing over the winter. Nevertheless, it is worth noting that the levels of E coli contamination even post-fencing and in the unfenced sites in the study by Bragina et al. (2017) were much higher (>10<sup>2</sup> CFU g.dry wt)<sup>,</sup> than those at a non-agricultural upland site that was also assessed in the latter study, suggesting on-going contamination of the stream bed from other sources of animal and/or human wastes.

While the stream water concentrations differed between sites and between times as would be expected for dynamic stream systems in agricultural catchments (e.g. Mellander et al.,
2015), there were no significant differences in the stream water concentrations of nutrients between the upstream reaches at the six fenced sites. There was also a high level of synchrony between the temporal patterns in the two sites. However, there was no significant difference between the nutrient concentrations for the fenced and unfenced streams over the annual cycle or over the months when cattle would have been in the field (April to October) for the two tributaries in the Milltown Lake catchment, which indicates a predominant role of diffuse pollution from the catchment area in determining the nutrient water levels. The impacts of intensification of agriculture in the Milltown Lake catchment since the 1950s and associated increased livestock numbers and application of fertilisers and slurries have been highlighted as the main drivers of water quality deterioration in paleolimnological studies by Carson et al. (2015) and Chique et al. (2018).

This short study does suggest, however, that streamside fencing can be beneficial in reducing sediment reservoirs of nutrients. Fencing eliminates nutrient contributions to watercourses resulting from cattle excretion within the stream channel, and would also promotes streambank recovery and vegetation establishment, thereby reducing bank erosion and the wash-out of particulate material into stream waters. In other words, cattle access sites in agricultural streams can be viewed as critical source areas of contamination (due to their potentially high reservoirs of contaminant that are susceptible to erosion/resuspension and transport downstream) (Thompson et al., 2013) which are reduced following cattle exclusion through fencing implementation. Reducing sediment nutrient reservoirs at these sites can reduce the potential for sediments to act as sources of contaminants to the water column, therefore mitigating potential legacy issues. In this light, although fencing may offer limited effectiveness in controlling diffuse pollution, particularly in winter periods when, in Ireland and other western European locations, rainfall is typically higher and vegetation in riparian buffer strips dies back, its implementation combined with other diffuse pollution measures can contribute to the success of such measures.

In other studies undertaken in the same sites under the COSAINT project, short-time streamside fencing was also demonstrated to reduce sediment deposits downstream of the access sites and site specific changes in the macroinvertebrate communities, potentially resulting from reduced impacts of excess fine sediment at the sites (Ó hUallacháin et al., 2020). Additionally, Ó hUallacháin et al. (2020) reported that long-term fencing in the Milltown Lake catchment had positive impacts on freshwater ecology with improvements in the abundances of Ephemeroptera, Plecoptera, Trichoptera (EPT) taxa observed in the fenced tributary in comparison to the unfenced tributary.

#### 7.5. Conclusions

This short study has highlighted the potential benefits of implementing cattle exclusion measures in reducing nutrient and *E. coli* concentrations in sediment reservoirs. Fencing can help mitigating future legacy effects through a reduction of the reservoirs of nutrients and faecal bacteria within the stream channel. However, this study also highlights the need to implement cattle exclusion measures and diffuse pollution mitigation measures in a concerted matter to successfully reduce agricultural pressures in aquatic systems. Notwithstanding the limitations of fencing as water quality protection measure, its reported benefits on freshwater nutrient and faecal contamination, along with the positive impacts it can have on freshwater ecology (e.g., Ó hUallacháin et al., 2020), favour its implementation. Thus, albeit limited, this short study provides relevant information to stakeholders, policy makers, and the local community suggesting fencing can be implemented as part of a concerted effort to manage nutrient and faecal contamination at the catchment scale.

## 8. Final Discussion

#### **Chapter 8. Final Discussion**

Freshwater systems constitute a vital natural resource, providing a wide range of ecosystem services. (Dudgeon et al., 2006: Pham et al., 2019). Moreover, they support approximately 6% of all plant and animal species described so far (Dudgeon et al., 2006). Yet, they represent one of the most endangered systems in the world, with declines in biodiversity surpassing those observed in most impacted terrestrial systems (Dudgeon et al., 2006; Schmidt-Kloiber et al., 2019). Freshwater systems have been insidiously impacted by urbanisation, damning and water abstraction, agriculture intensification, mining and industry activities, introduction of invasive species and climate change (Malmqvist and Rundle, 2002). As noted earlier, in the last decades several bodies of legislation have been introduced to protect freshwater systems, including the Clean Water Act in the US and several policies in the EU, in particular the over-arching Water Framework Directive, which aligns with the UN Sustainable Development Goals (Rouillard et al., 2017).

Agriculture has been identified as a major pressure on freshwater systems worldwide (Vorosmarty et al., 2010). Much of this pollution, which includes excess nitrogen and phosphorus from fertilisers, excess sediment loadings and microbial contaminants, reaches inland waters through diffuse pathways of contamination (Heathwaite, 2010; Muirhead and Monaghan, 2012; Deakin et al, 2016). Such pathways have been extensively studied and are well documented in literature (e.g. McDowell et al., 2006; McDowell and Wilcock, 2007; Mellander et al., 2012; Bowes et al., 2015; Lloyd et al., 2019) . Less consideration has been given, however, to agricultural point sources of pollution, for example the role of unrestricted access of livestock to watercourses and the potential contribution of contaminants to the aquatic system.

There has been a general paucity of research investigating the potential of cattle access and instream activity to contribute to water quality deterioration. This is particularly true in the European context, as most studies on this topic to date have been conducted in the US and Australasia, with a smaller number of contributions from the UK (e.g. Collins et al., 2010; Bond et al., 2012, 2014) and Ireland (e.g. Conroy et al., 2016; Bragina et al., 2017; Madden et al., 2019; O'Sullivan et al., 2019). The recent review by O'Callaghan et al. (2018) highlighted the uncertainty around the impact of cattle access to watercourses and the efficacy of fencing as water quality protection measure.

The focus of the research presented in this thesis was to contribute to an understanding of the effects of unrestricted cattle access to watercourses on freshwater systems, expanding the body of literature conducted on the topic globally. Specifically, the study aimed to assess the extent to which such practice impacts water and sediment levels of contaminants such as excess nutrients, suspended sediment, and *E. coli*. This topic is of paramount importance in Ireland, where 67.4% of the land is dedicated to agriculture (CSO, 2020) and beef and dairy production predominate (Teagasc, 2017b), and where, despite the paucity of empirical evidence, restricting cattle access to watercourses has been included in agri-environmental policy since the introduction of the Rural Environmental Protection Scheme (REPS) in 1994. This study is equally relevant to other European countries where beef and dairy production are important economic sectors, e.g. France, Germany, the UK and Luxembourg (Eurostat, 2019; Eurostat, 2020). An overview of how cattle access to watercourses may affect such parameters is presented in Fig. 8.1.



**Fig.8.1.** Diagram showing how unrestricted cattle access to watercourses can affect stream geochemical and microbial parameters. Parameters in bold are impacts demonstrated in this study.

#### 8.1. Key findings

The present study revealed new information on how unrestricted cattle access to watercourses can impact water quality in terms of pollutants such as sediment, excess nutrients and faecal contaminants. The results from this study are particularly important because as noted previously there is a very limited body of research conducted on the topic in Europe in particular, Moreover, in spite of the widely recognised role of bed sediments as sinks and sources of pollution in aquatic systems, the majority of research addressing freshwater pollution associated with cattle access to watercourses or the effectiveness of fencing as a mitigation measure tends to focus on concentrations of contaminants in the water column. This study specifically addressed the potential impacts of cattle access on stream bed sediment reservoirs of contaminants. In addition, the study investigated and quantified the changes in a broad range of water quality parameters in response to cattle access to stream waters in near-real time, which, to the knowledge of the authors, no other study has addressed.

The key findings of the present research can be summarised as follows:

- I. Unrestricted cattle access to watercourses can contribute to higher concentrations of *E. coli* of bed sediment
  - a. High levels of *E. coli* in stream sediments were found at all study sites indicating that this is widespread in Irish agricultural catchments
  - b. Cattle access to watercourses was found to significantly increase these stream bed sediment *E. coli* levels
  - c. Sediment *E. coli* contamination was reduced (but persisted) in postgrazing season when cattle had been removed from the fields

- d. Sediment contamination *E. coli* concentrations appeared to be governed by local (i.e. field-scale) management during grazing season, and catchment-scale factors in the absence of cattle
- II. Unrestricted cattle access to watercourses was found to influence nutrient concentrations in the stream sediment but to a lesser extent than *E. coli* 
  - Cattle access to watercourses contributed to stream sediment reservoirs of phosphorus, but had less impact on reservoirs of total nitrogen and organic carbon
  - b. The impact of nutrient contamination was more apparent in the smaller silt+clay fraction (< 63µm), as opposed to the < 2mm fraction</li>
  - c. Sediment nutrient levels for all three nutrients were more likely to be influenced by catchment-scale management i.e. cattle density, rather than local-scale (field-scale) management
  - d. Where there was an indication of cattle access effect, this was only apparent at the interface between stream and land, suggesting a rapid flushing of nutrients downstream in the main channel of the streams
- III. Cattle in-stream activity consistently resulted in increased the loads of total suspended solids and *E. coli* bacteria in the stream water while also increasing ammonium loads
  - Although increases in total phosphorus loads were observed during cattle in-stream activity, these were not significantly different from background variation at the site
  - Soluble reactive phosphorus loads were unaffected by cattle in-stream activity
  - c. Cattle in-stream activity had resulted in no increase in stream nitrate loads

## 8.1.1. <u>Unrestricted cattle access to watercourses contributed to sediment faecal</u> <u>contamination</u>

Pastoral agriculture can cause faecal contamination of watercourses through diffuse pathways, whereby high concentrations of faecal organisms are delivered to the watercourse (generally in surface run-off during storm events) (Kay et al., 2007; Muirhead and Monaghan, 2012). Nevertheless, previous research by Nagels et al. (2002) suggested that stream sediment reservoirs of faecal bacteria resulting from direct excretion in the stream channel by livestock can be of similar or greater importance than diffuse pathways of contamination in the overall faecal contaminant yield in agricultural streams.

The study described in Chapter 4 aimed at assessing the extent to which unrestricted cattle access to watercourses contributed to reservoirs of faecal contaminants in sediments of agricultural streams in Ireland. The study revealed that *E. coli* was present in all sediments in these Irish agricultural streams, and was detected at all but one of the sampled sites (the exception being site BWA and in post-grazing season only), including those sites where there was little or no upstream cattle activity. During the grazing season, E. coli concentrations were in the order of  $10^2 - 10^4$  CFU.g dry wt<sup>-1</sup> at stream reaches in headwater sites not used by cattle. As is typical for agricultural activity in Ireland these catchments were dominated by cattle-based farming, thus the findings highlight the potential significant impacts that agriculture (cattle-based) can have on stream sediment E. coli levels and possible faecal contamination, regardless of whether animals actually have access to the watercourses. E. coli contamination may result from wildlife and human septic tank wastes, but in the context of cattle-based agriculture, wash-out from slurry spreading and runoff from grazing fields where faeces have been deposited are the main diffuse pathways through which faecal contaminants are delivered to watercourses (Fenlon et al., 2000; Vinten et al., 2004; Murihead and Monaghan, 2012). It should also be noted that, as it is now widely accepted, E. coli can become naturalised in sediments (Perchec-Merien and Lewis, 2012;

Jang et al., 2017). Thus, the faecal contamination measured in this study does not necessarily reflect recent pollution events.

Additionally, stream reaches accessed by cattle were, in general, significantly more contaminated than reaches immediately upstream with no animal access. Sediment concentrations of *E. coli* bacteria at cattle access sites were one to three orders of magnitude higher, and four orders of magnitude in one instance, than concentrations at sites with no access. Although the faecal contamination reduced in the post-grazing season, the concentrations remained at  $10^2 - 10^3$  CFU.g dry wt<sup>-1</sup>, suggesting some level of bacteria persistence in the sediment matrix. These results indicate that unrestricted cattle access to watercourses has the potential to significantly exacerbate already high faecal contamination of agricultural streams, and that the threat to human and animal health should not be considered restricted to the summer months.

The increase in sediment *E. coli* contamination at cattle access sites (relative to upstream sites) was found to be associated with the degradation of the sites, as shown by the inverse relationship between *E. coli* bacteria concentrations and river habitat index (RHI) scores. These RHI scores, calculated at a local-scale, reflected the sites' general degradation, with lower RHI scores indicating more impacted sites (O'Sullivan et al., 2019). However, the current study found a relationship between stream sediment *E. coli* contamination and estimated cattle density per locality (estimated number of animals per ha at each sampled site) only in post-grazing season. This suggests that during grazing season, local-scale stream sediment *E. coli* contamination is governed by localised (i.e. field-scale) factors – for example animal behaviour, faecal bacterial concentrations, and stream flushing effects – which dominate over diffuse contamination pathways. Faecal bacteria in agricultural soils to which animal slurries and manures have been applied have been shown to have longer decay rates when UV and temperature values are lower (Hodgson et al., 2016), which could contribute to the observation that cattle density per locality per locality influenced *E. coli* bed sediment

concentrations in post-grazing season (i.e. early winter) but not in mid-grazing season (i.e. summer).

Faecal contamination of stream waters has implications not only for human health, due to the risk of zoonotic infection, but to animal health and performance (Willms et al., 2002). Where infected animals defecate into waters, the stream can serve as a vehicle of disease or parasite transmission to animals downstream. For instance, calves infected with the parasite *Cryptosporidium* sp. excrete large numbers of oocysts (a highly resistant spore-like stage of the parasite) in faeces, which may remain infective in waters for many months, particularly in norther climates where surface water temperatures can be low but remain above freezing (Fayer, 2004).

Another public health issue that has been repeatedly found to be associated with cattle density is human verocytotoxigenic *E. coli* (VTEC) infection (e.g. Brehony et al., 2018; Óhaiseadha et al., 2016). Ireland, in particular, has consistently reported the highest rate of VTEC infection in the EU, several times higher than the EU average (Garvey et al., 2016; Ohaiseadha et al., 2017; Brehony et al., 2018). Previous research has linked high cattle density farming and consumption of water retrieved from private wells, which are exempt of regulations, to increased incidence of VTEC infection (Garvey et al., 2016; Ohaiseadha et al., 2017; Brehony et al., 2018). In this light, the findings of this present study are of importance, as they indicate that unrestricted cattle access to watercourses plays a role in this public health issue, contributing to the pathways for animal and human infection, thus favouring the implementation of cattle restriction measures.

## 8.1.2. <u>Cattle access to watercourses potentially contributes to stream sediment reservoirs</u> of phosphorus

Chapter 5 investigated the extent to which unrestricted cattle access to watercourses contributed to the sediment reservoirs of total phosphorus (TP), total nitrogen (TN) and organic carbon (OC). Nutrient concentrations in sediments at all study sites were assessed in the fine fraction (<2mm particle size) and in the silt+clay fraction only (<63µm particle size), as this latter fraction is generally considered the most chemically reactive fraction. The silt+clay fraction was enriched in all three assessed nutrients, and analysis of this fraction revealed a pattern whereby the highest nutrient concentrations were observed in the catchments located in regions with traditionally more intensive cattle agriculture, i.e. the Commons River (CM) and the Milltown Lake (ML) catchments and a significant effect of estimated cattle density per region on sediment concentrations of all three nutrients in this fraction. This is one of the major results from this thesis and has implications for issues related to legacy phosphorus (Jarvie et al., 2013; Sharpley et al., 2013) in streams and for livestock management in relation to water quality.

Contrary to what was observed for sediment faecal bacteria concentrations, sediment nutrient concentrations did not vary significantly within the study sites (i.e. between the access site itself and upstream sites with no access), indicating an apparent lack of effect of direct cattle access to watercourses on sediment nutrient concentrations. Nevertheless, the GAM analysis suggested a potentially synergistic effect between cattle access (at the interface between land and watercourse) and estimated cattle density on total phosphorus concentrations in the silt+clay fraction. Additionally, and in contrast to sediment *E. coli* levels, there was no relationship between overall sediment nutrient levels at the study sites and the sites' RHI. These findings suggest that sediment nutrient reservoirs in agricultural streams are predominantly governed by catchment-scale activities. Nevertheless, unrestricted cattle access to watercourses potentially contributes to stream sediment reservoirs of phosphorus,

which is of note given the role of P in eutrophication (Jennings et al., 2003) and deterioration of water quality and its potential to cause legacy issues (Sharpley et al., 2013).

The apparent contrasting nature of the mechanisms governing sediment faecal and nutrient contamination might be related to the distinct predominant pathways whereby nutrients and bacteria reach agricultural watercourses and their fate once in the aquatic system. Cattle faeces contain *E. coli* loads of 10<sup>7</sup> CFU (Avery et al., 2004; Davies-Colley et al., 2004; Weaver et al., 2005) and it is widely accepted that, once deposited, bacteria can become quickly incorporated in soils and sediment, where they can persist (Ishii et al., 2007; Oliver et al., 2007; Ishii and Sadowsky, 2008; Garzio-Hadzick et al., 2010; Badgley et al., 2011; Shelton et al., 2014; Hodgson et al., 2016) and multiply (Perchec-Merien and Lewis, 2012). In contrast, nutrient loads in cattle faeces may be relatively lower (Bond et al., 2012). In the aquatic system, nutrients also undergo biogeochemical reactions, including mineralisation, biotic uptake, and transport downstream in both particulate and dissolved forms, which may hinder localised nutrient accumulation in the sediment. The observation of a potential effect of cattle access on TP levels at the interface between land and watercourse, but not within the stream channel at the cattle access (CAS) might also indicate a rapid flush downstream of sediment TP within the stream channel with higher flows (Eder et al., 2014).

## 8.1.3. <u>Cattle in-stream activity consistently results in increased concentrations of total</u> suspended solids and *E. coli* bacteria in stream waters

Chapter 6 investigated the impacts of direct cattle access to watercourses and cattle instream activity on water quality at real-time. It was observed that cattle in-stream activity consistently resulted in increased concentrations of total suspended solids (TSS) and *E. coli* bacteria in stream waters, and that the increase in loads when cattle were present was significantly higher than when they were absent. Cattle in-stream activity was also significantly associated with increased loads of ammonium, while having no apparent effect on nitrate loads. Interestingly, it did not have a significant effect on stream loads of SRP or TP.

The increase in *E. coli* bacteria concentrations in stream waters during cattle in-stream activity occurred irrespective of the number of in-stream defecations. There was also an increase in TSS when cattle were in the water. These results indicate that the cattle disturbance of stream sediment, which can be enriched in faecal bacteria as observed in Chapter 4, can contribute through resuspension into the water column of high numbers of viable bacteria. Nevertheless the defecations which were recorded will result in increased numbers of *E. coli*, and faecal material, in the water. The findings indicate that cattle access to, and activity in, watercourses has an impact on water faecal contamination, and are line with the observations of Nagels et al. (2002), who suggested that disturbance of sediment enriched in faecal matter could be as important as diffuse contamination pathways in the overall bacterial yield in an agricultural stream in New Zealand.

The increase in TSS associated with cattle access is in line with the findings of Terry et al. (2014) and supports the evidence found by O'Sullivan et al. (2019), who reported significantly greater mass of deposited sediment downstream of cattle access sites than upstream of these sites for the sites used in this study. The lack of any immediate effect of cattle in-stream activity on phosphorus loads is less surprising for SRP, as P is a very reactive ion and exhibited high variation in concentrations in the study stream. Total phosphorus loads, however, would be expected to be influenced by bed sediment disturbance and direct defecation in waters. Generally, TP concentrations did increase in association with increases in TSS concentrations; however, due to the high background variability in TP loads at the study site, these increases were not significant.

# 8.1.4. <u>Fencing of watercourses can have positive impacts on sediment nutrients and *E.coli* <u>reservoirs</u></u>

Chapter 7 described a small study assessing the effectiveness of fencing as a mitigation measure, including an assessment of potential improvements in sediment concentrations of nutrients and E. coli bacteria at cattle access sites after short-term fencing (~ one year) (in a before-after experiment), and an investigation of the effects of long-term fencing (nine years) on selected water quality parameters (nutrients and E. coli) (in a paired control-treatment sub-study). Although limited, the short-term experiment showed that fencing can lead to significant reductions on sediment nutrient concentrations (<2mm faction) at the stream-land interface in cattle access sites, an effect that can result from the physical recovery of bank and consequent elimination of these more impacted areas. Despite the statistically nonsignificant differences in sediment E. coli levels at the study sites before and after fencing, which might be related to time of sampling (i.e. post-grazing season, when, as seen in Chapter 4, E. coli sediment concentrations had been reduced), the findings of this short-term experiment also suggest that fencing can have positive effects on sediment E. coli reservoirs for individual sites. The longer-term sub-study did not reveal significant differences in water nutrient or E. coli bacteria concentrations. Overall, however, the results of this study add to the findings of the previous chapters in supporting the implementation of fencing as a mitigation measure.

#### 8.2. Implications for management

This research has shown that unrestricted cattle access to watercourses can exacerbate contamination of stream bed sediments with *E. coli* and consequently of stream waters, increase suspended sediment levels, and potentially contribute to sediment phosphorus reservoirs. While the effects of direct cattle access on *E. coli* contamination of watercourses

appear to be localised and thus strongly influenced by field-scale characteristics and practices, instream sediment nutrient levels seem to be mainly governed by catchment-scale activities. Thus, whilst there is evidence to support the implementation of measures aimed at excluding cattle from watercourses (e.g. fencing), such measures should be combined with larger scale, targeted mitigation measures to reduce diffuse losses of nutrients, sediments and faecal organisms, and effectively protect water quality.

Fencing off watercourses has been reported to have positive effects on water quality (see O'Callaghan et al., 2018). Under the COSAINT project, it was observed that short-term (~1 year) fencing of watercourses led to significant reductions in fine sediment deposits downstream of cattle access sites, with associated positive effects on macroinvertebrate communities in some cases (Ó hUallacháin et al., 2020). Additionally, an assessment of stream recovery following long-term fencing (~9 years) in paired fenced-unfenced streams in the Milltown Lake catchment, showed increased EPT (Ephemeroptera – Plecoptera – Trichoptera) richness in the fenced stream while the unfenced stream showed deterioration in water quality (Ó hUallacháin et al., 2020). In a study of faecal contamination in the same paired streams, Bragina et al. (2017) reported significantly lower streambed sediment concentrations of *E. coli* bacteria in the fenced stream in comparison to the unfenced stream during grazing season.

Also under the COSAINT project, Kilgarriff et al. (2020) determined the spatial distribution of cost-effectiveness of fencing watercourses in Ireland (using as metrics the reduction in faecal matter deposition within watercourses). They reported that the cost-effectiveness of fencing watercourses as a water quality protection measure is highest in regions with high agriculture intensity and low density of watercourses, which should therefore be prioritised (Kilgarriff et al., 2020). This is in line with the 4<sup>th</sup> NAP recommendation of fencing watercourses according to agricultural intensity (i.e. in derogation farms, which have stocking rates above 170 kg N.ha<sup>-1</sup>), which will come into operation in Ireland in 2021.

Fencing watercourses eliminates the direct impacts of cattle access and cattle in-stream activity, but also promotes the establishment of a riparian buffer strip, which can further contribute to water quality protection by intercepting diffuse pathways of contamination. Conversely, the establishment of hedgerows in the riparian area can also be adopted as a mitigation measure, which would initially require fencing, but would later offset the costs of fencing maintenance (fences have a life spam of ~ 5 years) (Ó hUallacháin et al., 2020).

The need for supplementing cattle exclusion measures concurrently with diffuse pollution mitigation measures has been highlighted in Ireland (e.g. Bragina et al., 2017; Veerkamp, 2019) and elsewhere (e.g. Ranganath et al., 2009). Such measures often build on the concept of critical source area management, whereby pollution sources or activities are dissociated in space or time from hydrologically active areas (Easton et al., 2017). They can include nutrient management plans aimed at reducing excess nutrient runoff (i.e. reducing residual nutrient transfers), limiting the times of slurry spreading to periods of lower hydrological risk to reduce incidental nutrient transfers to watercourses, and, where possible, adopting rotational grazing, which has been shown to have beneficial effects in reducing agricultural pollution in comparison with continuous grazing systems (e.g. Sovell et al., 2010). In order to be effective, diffuse pollution mitigation measures must be implemented on a case by case basis, taking into account specific landscape and management conditions that can potentially combine to exacerbate diffuse losses to watercourses (Collins and McGonigle, 2008; Deakin et al., 2016).

#### 8.3. Implications for policy

Ensuring the protection of water resources in Ireland is particularly relevant in the context of the recent and ongoing policies aimed at increasing the productivity of the agri-food sector,

such as the Food Wise 2025 strategy. The second cycle of the current Irish River Basin Management Plan (RBMP) 2018 – 2021, maintains and strengthens key water quality protection measures. Under the current Green Low Carbon Agri-environment Scheme (GLAS), more than 21 000 farmers have fenced 16 000 km of watercourses. Furthermore, the 4<sup>th</sup> Nitrates Action Plan (NAP), which came into operation in 2017, requires derogation farmers to fence watercourses 1.5 m from the top of the banks from January 1<sup>st</sup>, 2021 (DoHPLG, 2018a). The 4<sup>th</sup> NAP also included strengthened measures to tackle diffuse sediment and nutrient losses from agriculture, such as a requirement for farmers to prevent direct runoff from farm roadways to watercourses and new conditions for slurry spreading derogation farmers (DoHPLG, 2018a).

The EU's legislative proposals for the post-2020 Common Agricultural Practice (CAP) (2021-2027) aim at delivering a higher level of climate and environmental ambition, while placing greater emphasis in the achievement of results at the regional and national scale. The policy comprises nine specific goals to be achieved by each Member State (MS). Three of these goals are directly related to protection and environment and climate, including to "foster sustainable development and efficient management of natural resources such as water, soil and air" (EC, 2019a). Each MS is responsible to delineate a CAP Strategic Plan, which will include specific targets and objectives for its territory and present actions to achieve them (EC, 2019a). The CAP 2021-2027 is interlinked with other EU policies for the protection of the environment and the agri-food sector such as the Water Framework Directive and the recent Farm to Fork Strategy and Biodiversity Strategy 2030, which are part of the climate and environment-focused European Green Deal (EC, 2019b). The Farm to Fork Strategy specifically requires the development of integrated nutrient management action plans in MS CAP Strategic Plans to tackle "nutrient pollution at source and increase the sustainability of the livestock sector" and the application of precise fertilisation techniques and sustainable agricultural practices particularly in areas of intensive livestock farming (EC, 2020b).

The current study provides information that is relevant for current policy as well as new policy developments. The findings of Chapter 4 and 6 relative to microbial contamination of waters show that unrestricted cattle access to watercourses can have implications in compliance with the Bathing Water Directive as well as the Drinking Water Directive. Additionally, the findings of Chapter 5 suggest cattle access can potentially contribute to sediment nutrient contamination which could impact on compliance with the WFD. Overall, the study supports the inclusion of fencing on-farm watercourses under agri-environment policy as part of the new 2021 – 2027 Rural Development Plan (RDP). Additionally, as seen in Chapter 5 where sediment nutrient reservoirs increased above cattle density rates of 1.2 (TP) to 1.6 (N and OC) animals per h, they favour the changes included in the 4<sup>th</sup> NAP regarding the targeting of intensive agriculture farms. This is further supported by Kilgarriff et al. (2020) who have shown that fencing would be a cost-effective water quality protection measure in intensive agriculture farms, including derogation farms, and recommended a vertical, targeted approach in agri-environment policy design as opposed to horizontal, onesize-fits-all regulations and incentives. Cullen et al. (2018) stated that the targeted approach adopted in GLAS resulted in higher farmer participation rates in regions with higher numbers of environmental features in need of protection or conservation, making this scheme more financially effective than its predecessors. This is because with the horizontal, flat-rate REPS and AEOS, farmers in extensive agriculture regions, who theoretically would be required to adopt fewer practical changes under the AES (but would receive the same payments as intensive farmers) would be more likely to adhere to the AES (Cullen et al., 2018). However, Madden et al. (2019) highlighted that in Ireland, high-status waterbody sites frequently coincide with designated Natura 2000 sites, which were also granted priority in entering GLAS. Farmers in Natura 2000 sites could select measures related to this feature, which do not necessarily include cattle exclusion measures, over water quality measures, which may be less financially rewarding (Madden et al., 2019). This highlights the need to ensure, in future AES, that incentives for adopting cattle exclusion measures are sufficient to cover the costs of the measure (i.e. fencing costs, provision of alternative drinking water sources, etc.),

in order to make it more financially attractive for farmers and encourage its adoption where it is more beneficial. Where fencing of watercourses to completely exclude cattle is not a feasible or cost-effective measure, other measures such as the provision of alternative water sources (e.g. mains water, nose pumps) can be incentivised. Several studies have reported that providing cattle with alternative water sources resulted in cattle spending significantly less time in the watercourses (e.g. Clawson, 1993; Miner et al., 1992; Sheffield et al., 1997).

The current study highlights the need to combine cattle exclusion with diffuse pollution control measures where there are larger catchment-scale challenges. The need to adopt a more holistic approach rather than local reach scale measures to effectively mitigate agricultural impacts on water quality was emphasised in previous research (e.g. Weigel et al., 2000; Ranganath et al., 2009; Bragina et al., 2017). Research conducted under the recently concluded MARS Project launched by the EU also highlighted that restoring single, short stretches of rivers is insufficient to effectively address the environmental pressures currently faced by European rivers, which predominantly consist of diffuse pollution and hydromorphological degradation (Hering and Birk, 2018). Amongst the MARS Project recommendations is the implementation of riparian buffer strips along a substantial proportion of the European rivers network as a means to mitigate diffuse pollution impacts and allowing the regeneration of riparian and in-stream habitats (Hering and Birk, 2018).

Lastly, providing greater knowledge to farmers on the potential environmental and public health impacts of unrestricted cattle access to watercourses can help increasing awareness, understanding available measures and enhancing farmers' confidence in their ability to undertake water quality protection actions. Engaging farmers in water quality management through not only regulation but also knowledge transfer and participative approaches is crucial to more robust and effective management of water resources (Winter et al., 2011). This need was highlighted as part of the COSAINT project (Ó hUallacháin et al., 2020) and acknowledged in the Irish River Basin Management Plan (RBMP) 2018 - 2021.

#### 8.4. Limitations of the study and future research recommendations

#### 8.4.1. Uncertainty associated with the study

With empirical research such as the research reported in this study, there are uncertainties associated with the lack of control of numerous factors that can influence the variables studied, namely agricultural management practices as well as the effect of climatic conditions and site-specific factors. In this study, some attempts were made to conduct a more controlled field investigation of the impacts of cattle access on water quality parameters (i.e. by controlling number of animals in the watercourse, time spent in the watercourse, avoiding direct excretion to assess the impact of sediment disturbance only), however this proved unfeasible. Studies such as this are also dependent on the cooperation of farmers and landowners, which is not always guaranteed.

#### 8.4.2. Influence of site-specific characteristics and management

This study did not specifically investigate several variables that can influence the extent to which unrestricted cattle access can impact freshwater systems, including site-specific management variables (stocking density, animal breed and age, density of cattle access point per length of watercourse). Future studies could further investigate the relationship between stocking density and impact of unrestricted cattle access, and investigate any differences in behaviour and impact caused by dairy and beef cattle.

#### 8.4.3. Sediment faecal contamination and public health risk

Faecal indicator bacteria (FIB) such as *E. coli* have been widely used to detect recent environmental faecal contamination, under the assumption that the bacterium does not persist or multiply outside the host, and estimate the risk of exposure to faecal pathogens (Zhi et al., 2017). However, it is widely recognised that *E. coli* comprises environmental strains (Ishii et al., 2006; Ishii and Sadowsky, 2008; Perchec-Merien and Lewis, 2012; Jang et al., 2017). Another limitation of the use of FIB to investigate faecal contamination is that, since *E. coli* is ubiquitous in warm-blooded organisms, it does not provide any information on the origin of such contamination, i.e. human origin, cattle origin, other animal origin, or indeed environmental *E. coli*. Bradshaw et al. (2016) have recommended a combination of FIB assessment and the use of faecal source trackers (FST) to determine the origin of contamination in order to better estimate the extent of environmental pollution. Incorporating such methods in further studies would allow for a better understanding of the extent of the impacts of cattle-based agriculture on faecal contamination of streams in rural areas, but also provide insight on the mechanisms that govern faecal freshwater pollution.

Some studies have also reported that *E. coli* bacteria levels may correlate poorly with the presence of certain pathogens and thus might provide inaccurate estimates of the public health risk (e.g. Abdelzaher et al., 2010; Abia et al., 2016; Bradshaw et al., 2016). Furthermore, the current study assessed sediment reservoirs of *E. coli* bacteria using plate count methods that depend on the culturability of bacteria in the samples. However, it has been shown that prolonged contact with surfaces including sediments may induce bacteria to enter a viable but non-culturable (VBNC) state, in which they will not grow in solid selective microbiological media, but may resuscitate under favourable conditions (Hassard et al., 2016). Studies have shown that *E. coli* bacteria, including *E. coli* O157:H7 (Poulsen et al., 1995; Wu et al., 2009), can leave the host already in a VBNC state, as bacteria in this state possess higher resistance to environmental stressors (Cheville et al., 1996; Saby et al.,

2009). For these reasons, the current study might not provide an accurate estimate the true extent of faecal contamination and the public health risk in the study catchments. It would therefore be of interest to assess faecal contamination using methods not dependent on the culturability of microbial cells, e.g. quantitative PCR (qPCR). Targeting of specific pathogens in agricultural sediment reservoirs (e.g. VTEC, *Campylobacter jejuni, Crypstosporidium* sp., *Giardia* sp., and several viruses) could also provide a better estimate of the human and animal health risk posed by cattle-based agriculture and unrestricted access to watercourses.

#### 8.4.4. Effectiveness of fencing as a mitigation measure

A further study could be undertaken on a more detailed investigation on the effectiveness of fencing in reducing nutrient and faecal contamination in streams. This would include sampling of a higher number of sites across a range of agricultural intensities and hydrological conditions, both in short-term and long-term, and at a higher temporal frequency. Where *before-after* studies are conducted, similarly to the study described Chapter 7, sampling at different points of the agricultural management cycle is recommended.

Such a study would also benefit of an assessment of the effectiveness of fencing on reducing nutrient sediment contents on the silt and clay fraction. Additionally, a study on the effects of cattle exclusion on the sediment nutrient reservoirs, particularly TP, similarly to the paired study on water quality described in Chapter 7, would provide a deeper insight of the effectiveness of the measure on such parameters, allowing a better understanding of the extent to which nutrient sediment reservoirs are controlled by diffuse pollution mechanisms.

#### 8.4.5. Studies on phosphorus fractions at cattle access sites

It has been reported that, although stream banks and stream bed sediments might have large pools of TP, these do not necessarily represent an environmental concern. Of more concern will be the proportion of labile P at such sites (McDowell and Sharpley, 2001; Palmer-Felgate et al., 2009; Fox et al., 2016). Therefore, investigating the different fractions of P in bed sediments at cattle access sites would be of interest to better understand the extent to which these sites may affect water quality.

#### 8.4.6. Studies on cattle behaviour

Although not described in detail in this study, the research provided some insight on the behaviour of cattle, in terms of watercourse usage, in Irish conditions. Cattle behaviour in terms of access to watercourses is influenced by factors such as temperature, provision of alternative water sources, provision of shade areas, and riparian vegetation (Belsky et al., 1999; Haan et al., 2010). Understanding how cattle behave (i.e. frequency of cattle access to the watercourse, duration of access, frequency of in-stream defecation and urination) in the Irish or western-European setting would provide relevant information to an understanding of the extent to which unrestricted access to watercourses might impact freshwater systems at the catchment scale , for example in modelling studies.

#### 8.5. Overall conclusion

Unrestricted cattle access to watercourses has been recognised by policy makers as potentially detrimental for freshwater systems and water quality. However, although cattle exclusion measures have been included in Irish agri-environment policy for more than twenty years, few attempts at quantifying such impacts, and indeed at assessing the effectiveness of those measures, have been made to date. This study established that unrestricted cattle access to watercourses can negatively impact freshwater systems. Cattle access creates areas where nutrients, in particular phosphorus, can potentially accumulate. It is also apparent that systematic defecation within the stream channel at cattle access sites results in the accumulation of faecal organisms in stream sediments. The high density of cattle access sites found in several Irish agricultural streams (e.g. c. 8 access sites per km) (Jordan and Ryan, 2011; Jordan and Smietanka, 2013) is likely to have an overall significant effect on water quality. The current study therefore favours the use of policy tools to incentivise farmers, particularly those in regions of high agricultural intensity, to implement measures aiming at eliminating or reducing cattle access to watercourses, while emphasising a need for a holistic approach in agricultural pollution mitigation plans.

# Appendices

**Appendix A1.** Concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 2 (September 21, 2016).

Location	Minutoc	Cattle	Мах	Urination/	SPD (ug 1 <sup>-1</sup> )	NH₄-N	TON	TSS
Location	winutes	activity	cattle	Defecation	SKP (µg.L)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	no	-	-	166	0.01	2.23	0.5
	42	no	-	-	163	0.01	2.21	0.5
	57	no	-	-	168	0.01	2.33	0.3
	72	no	-	-	162	0.01	2.13	0.5
	87	no	-	-	165	0.01	2.10	0.6
	102	no	-	-	162	0.01	2.11	0.3
	117	no	-	-	161	0.01	2.12	0.4
Upstream	132	no	-	-	158	0.01	2.16	0.2
	147	no	-	-	159	0.01	2.16	0.4
	162	no	-	-	160	0.01	2.14	0.1
	177	yes	5	no	161	0.01	2.09	0.4
	192	yes	2	U+D	160	0.05	2.09	0.3
	207	yes	2	no	165	0.06	2.07	0.9
	222	yes	3	no	167	0.05	2.06	0.7
	237	no	-	-	162	0.04	2.04	0.4

**Appendix A2**. Concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 2 (September 21, 2016).

Minutos	Cattle	Max aattla	Urination/	SPD (ug L <sup>-1</sup> )	NH₄-N	TON	TSS
Minutes	activity	Max calle	Defecation	3κr (μg.L )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L⁻¹)
27	no	-	-	175	0.02	2.11	0.6
42	no	-	-	171	0.02	2.09	1.0
57	no	-	-	168	0.02	2.09	0.4
72	no	-	-	166	0.02	2.14	0.2
87	no	-	-	163	0.02	2.16	0.4
102	no	-	-	165	0.02	2.14	0.3
117	no	-	-	166	0.02	2.13	0.2
132	no	-	-	165	0.02	2.09	0.3
147	no	-	-	164	0.02	2.06	0.3
162	no	-	-	163	0.02	2.04	0.1
177	yes	5	no	162	0.06	1.97	0.4
192	yes	2	U+D	162	0.14	1.97	5.5
207	yes	2	no	192	0.08	2.03	1.8
222	yes	3	no	169	0.07	1.99	5.3
237	no	-	-	164	0.05	2.15	1.2
	Minutes 27 42 57 72 87 102 117 132 147 162 177 192 207 222 237	Cattle           Activity           27         no           42         no           57         no           57         no           72         no           87         no           102         no           117         no           132         no           147         no           147         yes           192         yes           207         yes           222         yes           237         no	Minutes         Cattle activity         Max cattle           27         no         -           42         no         -           57         no         -           57         no         -           72         no         -           72         no         -           87         no         -           102         no         -           117         no         -           132         no         -           147         no         -           147         no         -           147         yes         5           192         yes         2           207         yes         2           222         yes         3           237         no         -	Minutes         Cattle activity         Max cattle Defecation           27         no         -           42         no         -           42         no         -           57         no         -           57         no         -           72         no         -           72         no         -           73         no         -           74         no         -           75         no         -           72         no         -           73         no         -           74         no         -           102         no         -           117         no         -           132         no         -           147         no         -           147         no         -           147         yes         5           192         yes         2           192         yes         2           207         yes         3           222         yes         3           237         no         -	Minutes         Cattle         Max cattle         Urination/ Defecation         SRP ( $\mu$ g.L <sup>-1</sup> )           27         no         -         175           42         no         -         171           57         no         -         168           72         no         -         168           72         no         -         168           72         no         -         166           87         no         -         163           102         no         -         165           117         no         -         165           117         no         -         165           132         no         -         165           147         no         -         163           152         no         -         163           147         no         -         163           152         no         -         163           177         yes         5         no         162           192         yes         2         U+D         162           207         yes         3         no         169	Minutes         Cattle activity         Max cattle         Urination/ Defecation         SRP (µg.L <sup>-1</sup> ) (µg.L <sup>-1</sup> )         NH <sub>4</sub> -N (mg.L <sup>-1</sup> )           27         no         -         -         175         0.02           42         no         -         -         171         0.02           57         no         -         -         168         0.02           57         no         -         -         168         0.02           72         no         -         -         168         0.02           72         no         -         -         166         0.02           87         no         -         -         166         0.02           102         no         -         -         165         0.02           117         no         -         -         166         0.02           1132         no         -         -         166         0.02           147         no         -         -         164         0.02           152         no         -         163         0.02           162         no         -         164         0.05           177	Marcattle activityMarcattle $Max cattleUrination/DefecationSRP (\mug.L-1)NH4-NTON(mg.L^{-1})27no1750.022.1142no1710.022.0957no1680.022.0957no1660.022.1487no1660.022.14102no1630.022.14103no1660.022.14117no1660.022.14117no1660.022.14117no1660.022.14117no1660.022.14117no1660.022.14117no1660.022.09117no1660.022.09117no1630.022.04117no1630.022.04118no1630.022.04119yes5no1620.141.971192yes2N1920.082.031207yes3no1690.071.991219no164<$

Location I	Minutee	Cattle	Max	University /Defendetion	NH <sub>4</sub> - N	SRP	TP	NO <sub>3</sub> – N	CI	TSS
Location	winutes	activity	cattle	Urination/Defecation	(mgN.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	-	-	0.02	73	83	4.43	20.65	2.8
	42	No	-	-	0.02	74	77	4.42	21.25	2.2
	57	No	-	-	0.02	74	76	4.47	21.69	4.4
	72	No	-	-	0.02	73	79	4.44	20.15	3.2
	87	Yes	5	No	0.02	73	73	4.48	21.01	2.4
	102	Yes	6	U+D	0.03	73	76	4.46	20.74	2.6
l la stas sus	117	Yes	1	U	0.02	73	84	4.42	20.61	3.4
Opstream	132	No	-	-	0.02	73	NA	4.41	22.01	4.2
	147	No	-	-	0.02	73	75	4.46	21.17	3.4
	162	No	-	-	0.04	76	NA	4.36	23.52	6.2
	177	No	-	-	0.04	79	96	4.12	23.45	9.2
	192	No	-	-	0.03	76	99	4.65	20.34	6.0
	207	No	-	-	0.03	76	85	4.45	21.87	4.2
	222	No	-	-	0.02	74	80	4.53	20.65	2.4

Appendix A3. Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 7 (May 10, 2017).

Location	Minutes	Cattle	Max	Uringtion /Defection	NH <sub>4</sub> - N	SRP	TP	NO <sub>3</sub> - N	CI	TSS
Location	minutes	activity	cattle	Urination/Defecation	(mg.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L⁻¹)	(mg.L <sup>-1</sup> )
	27	No	-	-	0.03	72	71	3.98	20.73	3.8
	42	No	-	-	0.03	71	72	4.03	20.70	2.6
	57	No	-	-	0.03	71	67	4.06	20.91	3.4
	72	No	-	-	0.03	72	68	4.07	20.77	2.8
	87	Yes	5	No	0.02	72	73	4.05	20.78	4.2
	102	Yes	6	U+D	0.04	79	130	4.03	21.81	14.8
Downstroom	117	Yes	1	U	0.08	81	148	4.13	21.89	27.4
Downstream	132	No	-	-	0.03	73	79	4.05	21.09	5.4
	147	No	-	-	0.02	70	72	4.06	22.37	2.6
	162	No	-	-	0.02	71	77	4.00	21.02	3.4
	177	No	-	-	0.03	77	104	4.04	22.65	5.2
	192	No	-	-	0.04	78	93	4.07	20.55	5.8
	207	No	-	-	0.03	73	87	4.06	20.80	5.4
	222	No	-	-	0.03	72	76	4.01	21.24	2.4

Appendix A4. Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 7 (May 10, 2017).

Location	Minutos	Cattle	Max	Defection/Urination	E. coli	00
Location	winnutes	activity	cattle	Delecation/ormation	Log <sub>10</sub> CFU.100ml	Log₁₀CFU.100ml
	27	No	-	-	NA	NA
	42	No	-	-	NA	NA
	57	No	-	-	NA	NA
	72	No	-	-	2.60	3.20
	87	Yes	5	No	2.69	3.06
	102	Yes	6	U+D	2.54	2.67
l la stas sus	117	Yes	1	U	2.62	2.46
Opstream	132	No	-	-	2.67	2.45
	147	No	-	-	NA	NA
	162	No	-	-	NA	NA
	177	No	-	-	NA	NA
	192	No	-	-	NA	NA
	207	No	-	-	NA	NA
	222	No	-	-	NA	NA

**Appendix A5.** Water concentrations of *.E. coli* and Other Coliforms upstream of the cattle access site on Event 7 (May 10, 2017).

**Appendix A6.** Water concentrations of *.E. coli* and Other Coliforms downstream of the cattle access site on Event 7 (May 10, 2017).

Location	Minutos	Cattle	Мах	Defection/Urination	E. coli	00
Location	WIIIIule5	activity	cattle	Delecation/ormation	Log <sub>10</sub> CFU.100ml	Log₁₀CFU.100ml
	27	No	-	-	NA	NA
	42	No	-	-	NA	NA
	57	No	-	-	NA	NA
	72	No	-	-	2.61	3.56
	87	Yes	5	No	3.00	3.15
	102	Yes	6	U+D	4.24	3.23
Downstroom	117	Yes	1	U 4.38		3.46
Downstream	132	No	-	-	3.61	2.76
	147	No	-	-	NA	NA
	162	No	-	-	NA	NA
	177	No	-	-	NA	NA
	192	No	-	-	NA	NA
	207	No	-	-	NA	NA
	222	No	-	-	NA	NA

**Appendix A7.** Water concentrations of nutrients, total suspended solids, *E coli* and Other Coliforms upstream of the cattle access site on Event 8 (May 24, 2017).

Location	Minutos	Cattle	NH4 - N	SRP	TP	NO3 - N	CI <sup>-1</sup>	TSS	E.coli	OC
Location	Minutes	activity	(mgN.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L <sup>-1</sup> )	(mgN.L <sup>-1</sup> )	(mg.L⁻¹)	(mg.L <sup>-1</sup> )	log₁₀CFU.100ml	log₁₀CFU.100ml
	27	No	0.05	131	194	4.21	23.65	4.8	4.11	3.72
	42	No	0.04	128	191	4.19	28.47	4.8	4.01	3.78
	57	No	0.03	127	180	4.16	22.53	3.4	3.67	3.36
	72	No	0.02	125	175	4.12	23.87	5.0	3.51	3.30
	87	No	< L.O.D	126	182	4.15	22.33	4.4	3.41	3.67
	102	No	< L.O.D	128	180	4.16	23.92	4.4	4.25	4.04
L In atria ana	117	No	0.02	125	187	4.15	25.32	8.5	NA	NA
Opstream	132	No	0.02	126	180	4.20	23.06	4.6	NA	NA
	147	No	0.02	128	174	4.18	29.98	7.5	NA	NA
	162	No	0.02	127	179	4.20	24.83	4.7	NA	NA
	177	No	0.02	127	174	4.20	23.14	8.6	NA	NA
	192	No	0.02	127	177	4.22	22.68	3.2	NA	NA
	207	No	< L.O.D	125	180	4.22	23.31	4.6	NA	NA
	222	No	0.02	128	169	4.24	23.12	4.2	NA	NA

**Appendix A8.** Water concentrations of nutrients, total suspended solids, *E coli* and Other Coliforms downstream of the cattle access site on Event 8 (May 24, 2017).

Location I	Minutoc	Cattle	NH4 - N	SRP	TP	NO3 - N	Cl⁻¹	TSS	E.coli	00
	winnutes	activity	(mgN.L <sup>-1</sup> )	(µg.L <sup>-1</sup> )	(µg.L <sup>-1</sup> )	(mgN.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	log10CFU.100ml	log10CFU.100ml
	27	No	0.03	128	189	4.06	22.92	3.4	3.82	3.49
	42	No	0.03	128	191	4.14	22.45	4.6	4.10	3.89
	57	No	0.04	127	181	4.18	22.54	4.2	3.91	3.57
	72	No	0.03	129	182	4.16	22.34	4.6	3.72	3.51
	87	No	0.03	127	171	4.13	25.70	6.6	3.43	3.39
	102	No	< L.O.D	126	181	3.99	23.23	4.4	3.44	3.39
Downstroom	117	No	< L.O.D	126	175	4.14	26.54	3.8	NA	NA
Downstream	132	No	< L.O.D	126	172	4.13	22.86	4.8	NA	NA
	147	No	< L.O.D	127	169	4.15	22.65	4.0	NA	NA
	162	No	0.02	126	168	4.12	30.13	2.6	NA	NA
	177	No	0.02	125	171	4.11	32.59	31.0	NA	NA
	192	No	0.02	124	169	4.10	23.62	4.0	NA	NA
	207	No	0.02	126	167	4.09	25.52	3.4	NA	NA
	222	No	0.02	126	166	4.15	23.19	5.2	NA	NA

Location	Minutes	Cattle	Max	Urination/	NH <sub>4</sub> - N	SRP	TP	NO <sub>3</sub> - N	Cl	TSS
		activity	cattle	Defecation	(mgN.L ')	(µg.L ')	(µg.L ')	(mgN.L ')	(mg.L ')	(mg.L ')
	27	No	-	-	0.02	100	243	3.05	23.34	5.2
	42	No	-	-	0.02	102	174	3.09	23.41	4.0
	57	No	-	-	< L.O.D.	102	154	3.22	23.31	5.4
	72	No	-	-	0.02	100	269	3.11	23.26	5.0
	87	No	-	-	< L.O.D.	100	217	3.30	24.56	4.8
	102	No	-	-	0.02	101	171	3.19	23.59	4.2
	117	No	-	-	0.02	100	348	3.17	23.10	5.0
	132	No	-	-	< L.O.D.	101	229	3.28	23.67	3.8
	147	Yes	2	No	< L.O.D.	101	217	3.15	24.07	3.8
Upstream	162	Yes	4	U	0.02	99	206	3.15	24.70	2.8
	177	Yes	4	No	< L.O.D.	101	155	2.04	23.97	4.4
	192	Yes	3	D	< L.O.D.	101	160	3.14	26.23	4.6
	207	Yes	1	No	0.02	101	208	3.09	24.88	5.2
	222	No	-	-	0.03	92	153	3.14	22.06	4.6
	237	No	-	-	0.04	101	225	3.11	22.76	3.4
	252	No	-	-	0.02	101	244	3.08	23.35	2.8
	267	No	-	-	0.03	105	232	3.14	22.76	5.2
	282	No	-	-	0.03	109	245	3.10	22.81	6.0
	297	No	-	-	0.02	105	266	3.13	23.06	5.6

Appendix A9. Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 9 (June 14, 2017).

Location	Minutes	Cattle	Max	Urination/	NH <sub>4</sub> - N	SRP	ТР	NO <sub>3</sub> - N	Cl	TSS
	Minutes	activity	cattle	Defecation	(mgN.L⁻¹)	(µg.L⁻¹)	(µg.L⁻¹)	(mgN.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	-	-	< L.O.D	97	205	2.72	31.08	12.6
	42	No	-	-	0.03	90	245	3.10	23.26	38.8
	57	No	-	-	0.02	102	234	2.95	24.16	7.6
	72	No	-	-	< L.O.D.	97	219	2.98	24.41	13.2
	87	No	-	-	< L.O.D.	100	226	2.91	24.54	3.0
	102	No	-	-	< L.O.D.	101	225	3.03	23.30	6.4
	117	No	-	-	< L.O.D.	97	414	3.00	22.80	2.6
	132	No	-	-	< L.O.D.	97	236	3.03	22.99	5.2
	147	Yes	2	No	< L.O.D.	96	199	2.70	24.03	6.8
Downstream	162	Yes	4	U	0.04	87	166	2.80	23.79	14.0
	177	Yes	4	No	0.05	89	309	3.11	23.90	18.4
	192	Yes	3	D	< L.O.D.	92	231	2.87	22.54	12.6
	207	Yes	1	No	< L.O.D.	94	259	3.07	23.02	12.2
	222	No	-	-	0.02	99	249	3.10	22.34	9.2
	237	No	-	-	0.05	98	248	3.13	22.15	7.6
	252	No	-	-	0.05	99	249	3.11	21.69	5.6
	267	No	-	-	0.05	98	243	3.10	22.58	4.8
	282	No	-	-	0.02	101	234	3.09	22.04	8.4
	297	No	-	-	< L.O.D.	102	279	3.026	21.45	6.0

Appendix A10. Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 9 (June 14, 2017).
Lecation					E. coli	OC
Location	Minutes	Cattle activity	Max cattle	Urination/Defecation	Log₁₀CFU.100ml	Log₁₀CFU.100mI
	27	No	-	-	NA	NA
	42	No	-	-	NA	NA
	57	No	-	-	NA	NA
	72	No	-	-	NA	NA
	87	No	-	-	NA	NA
	102	No	-	-	NA	NA
	117	No	-	-	NA	NA
	132	No	-	-	2.90	2.99
	147	Yes	2	No	2.91	3.02
Upstream	162	Yes	4	U	2.98	2.99
	177	Yes	4	No	3.07	3.15
	192	Yes	3	D	3.02	3.17
	207	Yes	1	No	2.94	3.28
	222	No	-	-	2.95	3.11
	237	No	-	-	NA	NA
	252	No	-	-	NA	NA
	267	No	-	-	NA	NA
	282	No	-	-	NA	NA
	297	No	-	-	NA	NA

Appendix A11. Water concentrations of *E. coli* and Other Coliforms upstream of the cattle access site on Event 9 (June 14, 2017).

Location	Minutaa		Max aattla	Uringtian/Defension	E. coli	OC
Location	winutes	Cattle activity	Max cattle	Unnation/Delecation	Log₁₀CFU.100ml	Log₁₀CFU.100ml
	27	No	-	-	NA	NA
	42	No	-	-	NA	NA
	57	No	-	-	NA	NA
	72	No	-	-	NA	NA
	87	No	-	-	NA	NA
	102	No	-	-	NA	NA
	117	No	-	-	NA	NA
	132	No	-	-	3.03	3.08
	147	Yes	2	No	3.20	3.16
Downstream	162	Yes	4	U	3.88	3.69
	177	Yes	4	No	3.93	3.90
	192	Yes	3	D	3.68	3.74
	207	Yes	1	No	3.77	3.88
	222	No	-	-	3.57	3.56
	237	No	-	-	NA	NA
	252	No	-	-	NA	NA
	267	No	-	-	NA	NA
	282	No	-	-	NA	NA
	297	No	-	-	NA	NA

Appendix A12. Water concentrations of *E. coli* and Other Coliforms downstream of the cattle access site on Event 9 (June 14, 2017).

Lesstian	Minutes	Cattle	Max actile	Urination/	NH <sub>4</sub> - N	SRP	ТР	NO <sub>3</sub> - N	CI	TSS
Location	winutes	activity	wax cattle	Defecation	(mg.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	-	-	0.02	233	281	2.85	24.28	12.0
	42	No	-	-	0.02	243	289	2.79	25.85	6.0
	57	No	-	-	0.05	263	325	2.51	26.18	7.2
	72	No	-	-	0.08	272	340	2.78	25.33	11.2
	87	No	-	-	0.08	269	342	2.75	25.68	8.8
	102	No	-	-	0.07	261	320	2.69	25.08	9.2
	117	Yes	1	No	0.06	253	319	2.70	25.69	8.2
	132	No	-	-	0.06	249	299	2.69	24.75	6.2
	147	No	-	-	0.05	241	294	2.70	24.28	8.2
Opsiream	162	No	-	-	0.04	239	295	2.71	24.74	5.6
	177	No	-	-	0.03	234	277	2.46	25.52	6.8
	192	No	-	-	0.02	233	262	2.70	26.66	6.0
	207	No	-	-	0.02	233	264	2.76	26.35	4.0
	222	No	-	-	0.02	232	271	2.81	24.47	5.6
	237	No	-	-	0.02	232	270	2.86	23.92	57.4
	252	No	-	-	0.02	231	262	2.79	23.62	5.0
	267	No	-	-	0.03	229	271	2.71	23.74	4.7
	282	No	-	-	0.02	236	214	2.74	31.35	6.0

Appendix A13. Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 10 (June 28, 2017).

		Cattle		Urination/	NH4 - N	SRP	TP	NO <sub>3</sub> - N	Cl	TSS
Location	Minutes	activity	Max cattle	Defecation	(mg.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L⁻¹)	(mg.L <sup>-1</sup> )
	27	No	-	-	0.07	233	291	2.75	30.39	12.2
	42	No	-		0.04	222	273	2.75	28.74	8.8
	57	No	-	-	0.06	247	314	2.70	29.23	9.0
Doursettreese	72	No	-	-	0.08	255	334	2.67	25.89	12.8
	87	No	-	-	0.10	266	336	2.62	47.74	9.4
	102	No	-	-	0.09	262	319	2.64	29.54	12.8
	117	Yes	1	No	0.07	250	312	2.62	25.76	10.6
	132	No	-	-	0.06	242	306	2.60	31.64	9.4
	147	No	-	-	0.04	236	283	2.58	31.15	9.0
Downstream	162	No	-	-	0.03	239	282	2.62	34.03	9.6
	177	No	-	-	0.03	234	298	2.67	31.22	9.8
	192	No	-	-	0.03	238	276	2.65	25.91	8.2
	207	No	-	-	0.03	229	267	2.64	30.63	9.0
	222	No	-	-	0.03	235	265	2.76	28.65	7.6
	237	No	-	-	0.03	226	263	2.73	26.24	6.6
	252	No	-	-	0.04	227	251	2.73	29.84	5.2
	267	No	-	-	0.03	226	266	2.67	27.98	5.8
	282	No	-	-	0.03	231	268	2.69	32.19	4.0

Appendix A14. Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 10 (June 28, 2017).

Appendix A15. Water conductivity upstream of the cattle

access site on Event 10 (June 28, 2017)

	Minutes	Conductivity	Temperature	
Location	winutes	(µS.cm⁻¹)	(°C)	
	27	603.5	14.7	
	42	608.2	12.8	
	57	594.2	12.2	
	72	594.8	14.0	
	87	7 596.5 1	14.7	
	102	589.7	12.7	
	117	593.4	11.8	
	132	595.8	13.3	
Linstroom	147	594.9	12.3	
Opsilean	162	600.1	13.2	
	177	605.1	12.2	
	192	610.1	11.6	
	207	602.8	12.5	
	222	608.2	13.5	
	237	604.7	14.1	
	252	604.5	14.1	
	267	604.5	13.1	
	282	595.5	13.2	

Appendix A16. Water conductivity downstream of the cattle

access site on Event 10 (June 28, 2017).

Location	27	(µS.cm⁻¹)	(°C)
	27		
		610.5	12.4
	42	599.7	12.0
	57	593.7	12.8
	72	598.6	11.8
	87	589.5	11.5
	102	588.3	14.6
	117	585.0	11.7
	132	587.8	12.3
Downstroom	147	591.8	11.7
Downstream	162	595.9	11.0
	177	600.7	11.8
	192	595.4	11.4
	207	604.9	11.8
	222	600.9	12.2
	237	609.1	11.5
	252	604.7	12.9
	267	594.1	13.3
	282	604.8	12.6

Location	Minutes	Cattle	Max	Defecation/	NH₄ - N	SRP	TP	NO3 - N	Cl	TSS
Location	winutes	activity	cattle	Urination	(mg.L⁻¹)	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	-	-	0.15	241	265	2.81	23.03	3.4
	42	No	-	-	0.23	243	306	2.80	23.30	5.2
	57	Yes	3	No	0.13	240	279	2.80	22.58	4.0
	72	Yes	2	U	0.08	239	332	2.78	22.57	5.8
	87	Yes	1	No	0.12	242	268	2.79	22.71	5.0
	102	No	-	-	0.06	242	264	2.80	22.97	4.2
Lingtroom	117	No	-	-	0.13	238	260	2.78	22.64	5.0
Opstream	132	No	-	-	0.33	235	263	2.76	23.89	12.0
	147	No	-	-	0.16	227	277	2.65	22.44	16.0
	162	No	-	-	0.17	233	264	2.74	29.87	3.8
	177	Yes	2	No	0.24	225	279	2.63	23.63	13.0
	192	No	-	-	0.13	238	263	2.77	22.73	3.6
	207	Yes	1	No	0.18	238	278	2.69	23.05	4.0
	222	No	-	-	0.07	244	264	2.74	22.54	4.0

Appendix A17. Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 11 (August 8, 2017).

Leastion	Minutaa	Cattle	Max	Defecation/	NH <sub>4</sub> - N	SRP	TP	NO <sub>3</sub> - N	CI	TSS
Location	winutes	activity	cattle	Urination	(mg.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	0	-	0.15	241	263	2.80	23.26	8.2
	42	No	0	-	0.09	236	281	2.81	22.26	4.6
	57	Yes	3	No	0.18	245	308	2.83	23.75	15.0
	72	Yes	2	U	0.19	256	401	2.89	22.72	42.4
	87	Yes	1	No	0.16	247	367	2.84	22.93	32.2
	102	No	0	-	0.15	250	296	2.77	23.87	12.0
Dourotroom	117	No	0	-	0.13	244	276	2.77	22.54	6.6
Downstream	132	No	0	-	0.22	248	285	2.83	23.24	11.8
	147	No	0	-	0.12	234	261	2.84	23.68	6.4
	162	No	0	-	0.14	235	257	2.80	22.58	5.6
	177	Yes	2	No	0.10	238	274	2.84	23.47	7.2
	192	No	0	-	0.16	256	311	3.01	24.18	23.4
	207	Yes	1	No	0.08	243	320	2.80	22.51	17.0
	222	No	0	-	0.09	240	316	2.74	22.44	9.6

Appendix A18. Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 11 (August 8, 2017).

Location	Minutes	Cattle activity	Max cattle	Defecation/Urination	<i>E. coli</i> log <sub>10</sub> CFU.100ml	OC log <sub>10</sub> CFU.100ml
	27	No	-	-	NA	NA
	42	No	-	-	2.91	3.69
	57	Yes	3	No	2.86	NA
	72	Yes	2	U	3.00	3.10
	87	Yes	1	No	2.90	3.53
	102	No	-	-	2.94	3.28
L la atra ana	117	No	-	-	2.70	2.61
Opstream	132	No	-	-	2.91	NA
	147	No	-	-	2.81	3.35
	162	No	-	-	2.93	NA
	177	Yes	2	No	2.79	3.43
	192	No	-	-	2.90	3.66
	207	Yes	1	No	2.76	3.58
	222	No	-	-	2.76	3.53

Appendix A19. Water concentrations of *E. coli* and Other Coliforms upstream of the cattle access site on Event 11 (August 8, 2017).

Location	Minutoo		Max aattle	Defenction/Uringtion	E. coli	OC
Location	WIIIIules	Callie activity		Delecation/ormation	log₁₀CFU.100ml	log₁₀ CFU.100ml
	27	No	-	-	NA	NA
	42	No	-	-	2.96	3.85
	57	Yes	3	No	3.36	2.70
	72	Yes	2	U	3.96	3.66
	87	Yes	1	No	3.88	3.53
Downstroom	102	No	-	-	3.49	3.40
	117	No	-	-	3.33	3.30
Downstream	132	No	-	-	3.01	2.52
	147	No	-	-	2.76	1.78
	162	No	-	-	3.16	3.68
	177	Yes	2	No	3.27	3.26
	192	No	-	-	4.10	2.71
	207	Yes	1	No	3.62	2.73
	222	No	-	-	3.62	4.09

Appendix A20. Water concentrations of *E. coli* and Other Coliforms downstream of the cattle access site on Event 11 (August 8, 2017).

					•	•	
Location	Minutes	Conductivity	Temperature	Location	Minutes	Conductivity	Temperature
		(µ5.cm)	(0)			(µ5.cm)	(0)
	27	670.0	12.8		27	667.0	13.7
	42	671.0	13.2		42	670.0	14.2
	57	664.0	13.4		57	672.0	14.3
	72	681.0	13.5		72	680.0	15.6
	87	671.0	14.2		87	676.0	14.8
	102 675.0   117 673.0	13.9		102	674.0	14.8	
		673.0	14.7	Deverenters are	117	678.0	15.0
Upstream	132	665.0	14.9	Downstream	132	663.0	15.3
	147	677.0	13.9		147	664.0	15.3
	162	675.0	15.3		162	668.0	15.1
	177	667.0	14.9		177	673.0	15.8
	192	671.0	15.9		192	693.0	15.8
	207	675.0	15.6		207	667.0	15.9
	222	672.0	15.5		222	674.0	16.8

**Appendix A21**. Water conductivity upstream of the cattle access site on Event 11 (August 8, 2017).

**Appendix A22**.Water conductivity downstream of the cattle access site on Event 11 (August 8, 2017).

Lesstian		Cattle	Мах	Defecation/	NH₄-N	SRP	TRP	TP	NO <sub>3</sub> -N	CI	TSS
Location	winutes	activity	cattle	Urination	(mg.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L⁻¹)
	27	No	-	-	0.02	210	224	242	2.74	31.87	3.0
	42	No	-	-	< L.O.D.	208	225	234	2.72	35.31	3.4
	57	No	-	-	< L.O.D.	205	224	235	2.70	35.98	3.2
	72	No	-	-	0.09	213	231	245	2.72	33.83	2.4
	87	No	-	-	0.03	209	225	222	2.72	32.60	3.6
	102	No	-	-	< L.O.D.	205	223	224	2.68	31.88	1.8
Upstroom	117	No	-	-	< L.O.D.	206	219	240	2.75	33.70	1.4
opstream	132	Yes	1	No	< L.O.D.	208	221	226	2.79	34.24	1.4
	147	No	-	-	< L.O.D.	207	221	219	2.74	36.69	1.8
	162	Yes	3	No	0.02	211	222	218	2.76	34.31	3.4
	177	No	-	-	0.02	211	223	217	2.75	32.65	3.8
	192	Yes	3	No	0.02	208	224	217	2.70	36.19	1.6
	207	No	-	-	0.03	211	222	216	2.69	32.72	2.8
	222	No	-	-	0.22	216	220	218	2.72	35.13	2.2

Appendix A23. Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 12 (August 30, 2017).

Location Minu	Minuteo	Cattle	Max	Defecation/	NH <sub>4</sub> -N	SRP	TRP	TP	NO <sub>3</sub> -N	Cl	TSS
Location	winutes	activity	cattle	Urination	(mg.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	-	-	0.03	223	254	279	2.55	45.13	11.0
	42	No	-	-	0.02	222	228	362	2.60	35.13	6.0
	57	No	-	-	0.02	207	214	234	2.60	33.63	6.2
	72	No	-	-	< L.O.D.	207	214	221	2.62	32.52	2.8
	87	No	-	-	< L.O.D.	211	221	213	2.63	34.23	2.8
	102	No	-	-	0.04	222	253	261	2.62	41.92	3.0
Douvoatrooro	117	No	-	-	< L.O.D.	204	219	198	2.60	35.69	3.2
Downstream	132	Yes	1	No	0.02	200	215	217	2.59	35.69	4.4
	147	No	-	-	0.05	204	214	238	2.62	37.93	9.2
	162	Yes	3	No	0.02	207	214	230	2.64	35.90	2.0
	177	No	-	-	0.10	210	219	349	2.60	35.79	11.0
	192	Yes	3	No	0.05	207	217	230	2.69	34.60	7.8
	207	No	-	-	0.07	207	208	236	2.68	35.35	9.2
	222	No	-	-	0.02	204	211	217	2.64	41.19	3.0

Appendix A24. Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 12 (August 30, 2017).

Location	Minutaa		Mox oottlo	Defecation/	E. coli	OC
Location	winutes	Cattle activity		Urination	Log₁₀CFU.100ml	Log <sub>10</sub> CFU.100ml
	27	No	-	-	NA	NA
	42	No	-	-	NA	NA
	57	No	-	-	NA	NA
	72	No	-	-	NA	NA
	87	No	-	-	NA	NA
	102	No	-	-	3.13	3.01
	117	No	-	-	3.26	3.15
Opstream	132	Yes	1	No	3.29	2.67
	147	No	-	-	3.32	3.20
	162	Yes	3	No	3.32	3.17
	177	No	-	-	3.23	3.13
	192	Yes	3	No	3.45	2.95
	207	No	-	-	3.27	2.85
	222	No	-	-	3.23	2.89

Appendix A25. Water concentrations of *E. coli* and Other Coliforms upstream of the cattle access site on Event 12 (August 30, 2017).

	Minutos	Cattle activity	Max aattla	Defecation/	E. coli	OC
Location	winnutes	Calle activity		Urination	Log <sub>10</sub> CFU.100ml	Log₁₀CFU.100ml
	27	No	-	-	NA	NA
	42	No	-	-	NA	NA
	57	No	-	-	NA	NA
	72	No	-	-	NA	NA
	87	No	-	-	NA	NA
	102	No	-	-	2.98	3.08
Downotroom	117	No	-	-	3.08	3.01
Downstream	132	Yes	1	No	3.20	3.14
	147	No	-	-	3.70	2.93
	162	Yes	3	No	3.37	3.05
	177	No	-	-	3.71	3.34
	192	Yes	3	No	3.49	2.78
	207	No	-	-	3.62	2.83
	222	No	-	-	3.29	2.72

**Appendix A26.** Water concentrations of *E. coli* and Other Coliforms downstream of the cattle access site on Event 12 (August 30, 2017).

		Cattle	NH₄ - N	SRP	TP	NO <sub>3</sub> -N	CI	TSS
Location	Minutes	activity	(mg.L <sup>-1</sup> )	(µg.L <sup>-1</sup> )	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	0.07	32	70	8.29	26.36	9.2
	42	No	0.07	31	64	8.27	25.93	4.0
	57	No	0.08	30	63	8.18	26.73	3.2
	72	No	0.06	32	64	8.30	26.79	4.6
	87	No	0.15	32	60	8.47	27.08	4.0
	102	No	0.08	32	62	8.41	26.66	3.2
	117	No	0.10	30	69	8.41	25.90	4.6
Opstream	132	No	0.09	31	72	8.25	26.65	3.4
	147	No	0.08	33	62	8.18	25.68	4.0
	162	No	0.08	32	64	8.27	25.45	4.8
	177	No	0.10	33	62	8.16	26.24	4.2
	192	No	0.10	33	62	8.25	26.00	4.0
	207	No	0.07	32	60	8.31	26.65	4.0
	222	No	0.10	33	60	8.53	26.69	3.2
	222	No	0.10	33	60	8.53	26.69	3.2

**Appendix A27.** Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 13 (February 7, 2018)

Location	Minutoo	Cattle	NH <sub>4</sub> - N	SRP	TP	NO <sub>3</sub> -N	Cl	TSS
Location	winutes	activity	(mg.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )
	27	No	0.08	30	70	8.08	25.72	3.6
	42	No	0.08	31	66	8.03	26.29	4.2
	57	No	0.12	29	67	8.07	26.04	3.0
	72	No	0.07	31	77	8.12	26.45	4.4
	87	No	0.06	31	74	8.46	26.36	4.0
	102	No	0.07	31	62	8.10	26.42	4.4
Deverenteere	117	No	0.09	29	68	8.20	26.32	3.2
Downstream	132	No	0.09	30	62	8.12	25.45	3.6
	147	No	0.09	31	65	8.13	26.35	3.6
	162	No	0.08	33	69	8.14	25.81	3.0
	177	No	0.09	32	63	8.23	25.68	2.6
	192	No	0.10	31	67	8.07	25.64	3.6
	207	No	0.07	33	103	8.08	25.66	15.8
	222	No	0.09	32	59	8.12	25.71	4.8

**Appendix A28.** Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 13 (February 7, 2018).

Location	Minutos	Cattle activity	E. coli	00
Location	Windles	Calle activity	Log10CFU.100ml	Log10CFU.100ml
	27	No	2.61	2.34
	42	No	2.43	1.70
	57	No	2.56	2.32
	72	No	2.73	2.18
	87	No	2.81	2.45
	102	No	2.54	2.11
Unstroom	117	No	2.43	2.36
Opstream	132	No	2.48	2.59
	147	No	2.48	2.86
	162	No	2.40	2.81
	177	No	2.54	2.86
	192	No	2.43	2.53
	207	No	NA	NA
	222	No	2.30	2.08

**Appendix A29**. Water concentrations of *E. coli* and Other Coliforms upstream of the cattle access site on Event 13 (February 7, 2018).

Location	Minutos	Cattle activity	E. coli	00
Location	WITTULES	Calle activity	Log₁₀CFU.100ml	Log <sub>10</sub> CFU.100ml
	27	No	2.40	2.51
	42	No	2.46	4.19
	57	No	2.40	2.30
	72	No	2.70	2.11
	87	No	NA	NA
	102	No	2.45	2.45
Linetroom	117	No	2.43	2.23
Opsiream	132	No	2.41	2.64
	147	No	2.43	2.72
	162	No	2.43	2.69
	177	No	2.36	2.69
	192	No	2.56	2.68
	207	No	2.56	2.54
	222	No	2.40	2.20

**Appendix A30**. Water concentrations of *E. coli* and Other Coliforms downstream of the cattle access site on Event 13 (February 7, 2018).

Appendix A31. Water conductivity upstream of the cattle access
site on Event 13 (February 7, 2018).

**Appendix A32**. Water conductivity downstream of the cattle access site on Event 13 (February 7, 2018).

Location	Minutos	Conductivity	Temperature	Location	Minutes	Conductivity	Temperature
Location	Minutes	(µS.cm⁻¹)	(°C)	Location	Winnutes	(µS.cm⁻¹)	(°C)
	27	649.6	11.0		27	638.7	12.5
	42	636.3	10.5		42	645.1	11.5
	57	648.6	10.1		57	641.8	11.2
	72	640.0	10.1		72	655.3	11.2
	87	651.4	10.3		87	650.4	10.9
	102	653.3	10.6		102	653.4	10.5
Unotroom	117	640.3	12.1	Downstroom	117	651.9	10.8
Opsileani	132	646.1	12.2	Downstream	132	648.3	10.8
	147	644.4	11.1		147	644.5	11.5
	162	NA	12.1		162	671.5	10.8
	177	653.6	10.3		177	641.0	10.3
	192	655.6	9.9		192	648.9	10.1
	207	640.7	10.3		207	652.5	9.9
	222	637.2	11.3		222	638.5	10.3

Lesstion	Minutes		NH₄-N	SRP	TRP	TP	NO <sub>3</sub> -N	Cl	
Location	Minutes	Cattle activity	(mgN.L <sup>-1</sup> )	(µg.L⁻¹)	(µg.L <sup>-1</sup> )	(µg.L⁻¹)	(mg.L <sup>-1</sup> )	(mg.L⁻¹)	155 (mg.L)
	27	No	0.08	49	53	80	7.58	27.88	9.8
	42	No	0.07	44	51	76	7.42	27.82	10.3
	57	No	0.07	44	51	63	7.64	27.73	7.5
	72	No	0.07	45	55	83	7.44	27.71	11.8
Upstream	87	No	0.07	49	56	82	7.46	27.60	8.3
	102	No	0.09	51	57	82	7.50	27.59	11.1
	117	No	0.08	53	59	76	7.52	27.80	7.4
	132	No	0.07	51	57	77	7.45	27.70	8.7
	147	No	0.06	50	60	78	7.46	27.82	7.7
	27	No	0.07	49	53	79	7.52	37.43	11.5
	42	No	0.05	44	48	64	7.68	32.97	11.0
	57	No	0.05	47	50	75	7.55	29.27	12.0
	72	No	0.04	47	53	66	7.54	28.13	12.0
Downstream	87	No	0.06	49	52	76	7.70	29.26	9.7
	102	No	0.06	53	59	78	7.54	27.66	9.5
	11	No	0.06	51	61	92	7.59	28.10	5.2
	132	No	0.06	49	59	93	7.50	27.84	9.0
	147	No	0.05	51	56	91	7.61	28.17	7.6

Appendix A33. Water concentrations of nutrients and total suspended solids on Event 14 (March 21, 2018).

Location	Minutes	Cattle activity	<i>E coli</i> Log₁₀CFU.100mI
	27	No	3.65
	42	No	3.39
	57	No	3.27
	72	No	3.20
Upstream	87	No	3.15
	102	No	2.86
	117	No	3.05
	132	No	2.94
	147	No	2.90
	27	No	3.60
	42	No	3.41
	57	No	3.15
	72	No	3.22
Downstream	87	No	3.13
	102	No	3.10
	117	No	2.99
	132	No	3.01
	147	No	3.03

Appendix A34. Water concentrations of *E. coli* on Event 14 (March 21, 2018).

Location	Minutes	Conductivity (μS.cm <sup>-1</sup> )	Temperature (°C)
	27	691.4	10.2
	42	696.0	10.5
	57	694.0	11.3
	72	695.1	10.2
Upstream	87	689.4	9.6
	102	693.7	9.8
	117	693.6	9.7
	132	695.0	11.3
	147	704.0	11.3
	27	701.3	8.7
	42	692.2	9.4
	57	711.0	9.9
	72	695.8	9.8
Downstream	87	692.1	9.5
	102	694.4	9.0
	117	696.8	8.8
	132	700.0	10.6
	147	693.8	10.2

Appendix A35. Water conductivity on Event 14 (Mar	rch 21, 2018)
---	---------------

**Appendix A36**. Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 15 (April 25, 2018).

Location	Minutoo	Cattle	NH₄-N	SRP	TRP	NO <sub>3</sub> -N	CI	<b>TEE</b> (mg   <sup>-1</sup> )
Location	winutes	activity	(mgN.L <sup>-1</sup> )	(µgP.L⁻¹)	(µgP.L⁻¹)	(mgN.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	133 (mg.L )
	27	No	0.27	100	145	6.58	29.07	7.7
	42	No	0.28	106	161	6.68	29.23	9.7
	57	No	0.21	105	147	6.74	29.27	8.7
	72	No	0.29	101	160	6.80	30.46	8.7
	87	No	0.27	97	158	6.77	29.83	9.7
	102	No	0.31	101	165	6.82	30.07	9.3
Unotroom	117	No	0.30	101	168	6.83	31.00	9.3
Opsileam	132	No	0.31	104	174	6.72	34.58	12.3
	147	No	0.30	108	172	6.83	35.75	11.3
	162	No	0.38	104	178	6.85	34.16	16.7
	177	No	0.48	104	173	6.96	35.41	19.7
	192	No	0.37	100	118	6.85	32.55	94.0
	207	No	0.40	104	97	6.90	31.92	10.0
	222	No	0.40	111	106	6.74	31.21	13.3

**Appendix A37.** Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 15 (April 25, 2018).

Location	Minutos	Cattle	NH₄-N	SRP	TRP	NO <sub>3</sub> -N	Cl	<b>TSS</b> (mg   <sup>-1</sup> )
Location	winnutes	activity	(mgN.L <sup>-1</sup> )	(µg.L <sup>-1</sup> )	(µgP.L⁻¹)	(mgN.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	133 (IIIg.L )
	27	No	0.24	106	89	6.80	29.67	NA
	42	No	0.23	106	74	6.53	28.74	21.0
	57	No	0.24	115	101	6.61	28.86	9.0
	72	No	0.25	111	101	6.64	29.61	6.3
	87	No	0.24	110	93	6.63	29.53	6.3
	102	No	0.25	108	93	6.62	29.95	9.5
Deuratraam	117	No	0.25	109	85	6.59	30.14	6.0
Downstream	132	No	0.29	112	97	6.57	33.70	7.7
	147	No	0.29	103	87	6.50	34.84	7.7
	162	No	0.34	113	109	6.60	33.58	8.0
	177	No	0.31	107	124	6.63	31.14	9.7
	192	No	0.41	110	109	6.86	32.10	10.0
	207	No	0.33	106	117	6.86	31.79	11.0
	222	No	0.40	109	115	6.66	30.89	10.0

_ocation	Minutes	Cattle activity	<i>E. coli</i> Log <sub>10</sub> CFU.100ml	OC Log₁₀CFU.100ml
	27	No	3.38	3.34
	42	No	3.60	3.56
	57	No	3.48	3.72
	72	No	3.54	3.83
	87	No	3.30	3.90
	102	No	3.36	3.82
Instragm	117	No	3.30	3.79
pstream	132	No	3.40	3.88
	147	No	3.34	4.08
	162	No	3.56	4.00
	177	No	3.58	4.18
	192	No	3.41	4.15
	207	No	3.48	3.94
	222	No	3.40	3.84

**Appendix A38.** Water concentrations of *E. coli* and Other Coliforms upstream of the cattle access site on Event 15 (April 25, 2018).

Location	Minutes	Cattle activity	<i>E. coli</i> Log <sub>10</sub> CFU.100ml	OC Log <sub>10</sub> CFU.100ml
	27	No	3.36	2.90
	42	No	3.20	2.78
	57	No	3.54	2.70
	72	No	3.46	2.95
	87	No	3.51	3.30
	102	No	3.56	3.34
Downstroom	117	No	3.32	3.34
Downstream	132	No	3.40	3.32
	147	No	3.46	3.40
	162	No	3.57	3.32
	177	No	3.18	3.20
	192	No	3.32	3.15
	207	No	3.52	3.38
	222	No	3.40	3.34

**Appendix A39.** Water concentrations of *E. coli* and Other Coliforms downstream of the cattle access site on Event 15 (April 25, 2018).

	10 (April 20, 2				
Location	Minutoo	Conductivity	Temperature	Location	
Location	Minutes	(µS.cm⁻¹)	(°C)	Location	
	27	640.3	10.7		
	42	646.6	10.4		
	57	636.6	10.6		
	72	631.1	11.6		
	87	636.6	11.5		
	102	633.5	12.0		
Instragm	117	647.9	12.3	Downstroom	
opstream	132	653.2	11.4	Downstream	
	147	656.5	12.4		
	162	654.2	11.5		
	177	649.3	11.0		
	192	656.5	10.6		
	207	652.3	11.3		
	222	655.8	12.3		

**Appendix A40.** Water conductivity upstream of the cattle access site on Event 15 (April 25, 2018).

**Appendix A41.** Water conductivity downstream of the cattle access site on Event 15 (April 25, 2018).

Leastian	Minutes	Conductivity	Temperature
Location	minutes	(µS.cm⁻¹)	(°C)
	27	NA	NA
	42	NA	NA
	57	636.3	12.9
	72	644.3	8.3
	87	648.6	8.5
	102	642.0	9.9
Downstroom	117	644.4	10.0
Downstream	132	659.8	9.1
	147	664.2	9.5
	162	653.9	9.4
	177	648.8	9.8
	192	645.4	10.4
	207	657.7	9.9
	222	649.0	10.7

Lootion	Minuteo	Cattle	Мах	Defecation/	NH₄-N	SRP	NO <sub>3</sub> -N	Cl	TSS
Location	minutes	activity	cattle	Urination	(mgN.L <sup>-1</sup> )	(µgP.L⁻¹)	(mgN.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	(mg.L⁻¹)
	27	No	-	-	0.06	137	5.42	27.82	7.0
	42	No	-	-	0.03	135	5.40	27.92	8.3
	57	No	-	-	0.03	137	5.44	28.51	3.3
	72	No	-	-	< L.O.D.	136	5.56	27.85	7.5
	87	Yes	5	No	0.02	135	5.41	28.14	5.3
	102	Yes	5	U+D	0.02	132	5.45	27.98	4.3
l la stus sus	117	Yes	4	No	0.03	133	5.42	28.18	3.0
Upstream	132	No	-	-	0.06	135	5.45	28.41	4.7
	147	No	-	-	0.12	138	5.51	28.63	6.0
	162	No	-	-	0.11	138	5.51	28.14	5.3
	177	Yes	4	No	0.07	137	5.55	28.25	5.3
	192	Yes	3	No	0.04	134	5.49	28.69	6.7
	207	No	-	-	0.06	137	5.43	28.20	4.0
	222	No	-	-	0.06	132	5.16	27.92	4.7

**Appendix A42**. Water concentrations of nutrients and total suspended solids upstream of the cattle access site on Event 16 (June 13, 2018).

**Appendix A43**. Water concentrations of nutrients and total suspended solids downstream of the cattle access site on Event 16 (June 13, 2018).

Location	Minutos	Cattle	Max	Defecation/	NH₄-N	SRP	NO <sub>3</sub> -N	Cl	<b>TSS</b> (mg   <sup>-1</sup> )
Location	Minutes	activity	cattle	Urination	(mgN.L <sup>-1</sup> )	(µgP.L <sup>-1</sup> )	(mgN.L <sup>-1</sup> )	(mg.L <sup>-1</sup> )	100 (ilig.L )
	27	No	-	-	0.02	137	5.37	27.73	10.0
	42	No	-	-	< L.O.D.	128	5.37	27.32	1.7
	57	No	-	-	< L.O.D.	132	5.39	27.40	6.7
	72	No	-	-	< L.O.D.	131	5.39	27.65	4.7
	87	Yes	5	No	< L.O.D.	129	5.01	26.93	7.7
	102	Yes	5	U+D	0.21	143	5.55	28.42	34.2
Downotroom	117	Yes	4	No	0.24	139	5.62	29.01	16.0
Downstream	132	No	-	-	0.07	132	5.11	27.91	8.7
	147	No	-	-	0.11	131	5.48	28.19	4.0
	162	No	-	-	0.12	132	5.51	28.23	7.7
	177	Yes	4	No	0.12	133	5.51	28.17	9.3
	192	Yes	3	No	0.09	131	6.20	31.13	9.7
	207	No	-	-	0.06	130	5.50	28.29	8.3
	222	No	-	-	0.04	129	5.45	27.65	6.7

			Max asttle		E. coli	OC	
Location	Minutes	Cattle activity	Max cattle	Urination/Defecation	Log₁₀CFU.100ml	Log₁₀CFU.100ml	
	27	No	-	-	2.99	3.70	
	42	No	-	-	2.92	3.15	
	57	No	-	-	2.91	3.48	
	72	No	-	-	2.82	3.30	
	87	Yes	5	No	2.89	3.53	
	102	Yes	5	U+D	2.74	2.62	
Instroom	117	Yes	4	No	3.05	2.60	
Jpstream	132	No	-	-	3.71	NA	
	147	No	-	-	3.90	2.59	
	162	No	-	-	3.69	NA	
	177	Yes	4	No	3.73	2.58	
	192	Yes	3	No	3.66	2.54	
	207	No	-	-	3.30	NA	
	222	No	-	-	3.51	2.82	

Lesstian	Minutes		Mex estile		E. coli	00
Location	winutes	Cattle activity	wax cattle	Urination/Defecation	Log₁₀CFU.100mI	Log₁₀CFU.100mI
	27	No	-	-	3.61	4.45
	42	No	-	-	3.56	3.79
	57	No	-	-	3.41	3.94
	72	No	-	-	3.36	3.67
	87	Yes	5	No	3.13	3.85
	102	Yes	5	U+D	4.90	NA
Downstroom	117	Yes	4	No	4.79	NA
Downstream	132	No	-	-	4.31	3.23
	147	No	-	-	3.99	3.00
	162	No	-	-	4.02	3.45
	177	Yes	4	No	4.40	3.95
	192	Yes	3	No	4.28	NA
	207	No	-	-	4.25	NA
	222	No	-	-	2.75	NA

Appendix A45. Water concentrations of *E. coli* and Other Coliforms upstream of the cattle access site on Event 16 (June 13, 2018)

access site o	n Event 16 (	June 13, 2018).		access site on Event 16 (June 13, 2018).					
Location	Minutos	Conductivity	Temperature	Location	Minutos	Conductivity	Temperature		
Location	Minutes	(µS.cm⁻¹)	(°C)	Location	WIIIIules	(µS.cm⁻¹)	(°C)		
	27	633.6	10.1		27	631.2	10.5		
	42	631.8	9.9		42	624.9	10.7		
	57	620.3	9.8		57	645.2	11.7		
	72	638.5	10.6		72	630.6	10.8		
	87	623.6	11.0		87	633.5	10.5		
	102	626.5	10.8		102	634.8	10.8		
	117	631.9	11.7	Deurectroere	117	641.2	10.9		
opstream	132	627.5	11.6	Downstream	132	632.4	10.9		
	147	634.3	11.6		147	632.4	10.8		
	162	634.2	11.5		162	638.1	10.8		
	177	621.0	11.2		177	633.5	10.7		
	192	628.7	11.2		192	631.1	11.7		
	207	NA	NA		207	629.5	11.6		
	222	627.2	11.3		222	627.6	10.8		

Appendix A47. Water conductivity downstream of the cattle

**Appendix A46.** Water conductivity upstream of the cattle access site on Event 16 (June 13, 2018).

294

Catchment	Site	Type and number of animals	Cattle introduced in field	Cattle removed from field	Period of grazing	Slurry spreading	Nutrient management plan
Brackan River (BK)	BK1A				NA		
	BK1B	Mixed livestock, 25 – 30 cows plus calves	Early April	End of October 2016	7 – 10 days in field and off for one month	Farmyard manure in autumn	Yes
	BK2A/BK3B	Beef, 30 cows plus 15 calves Mixed livestock 25	April	September	Animals in field for summer months	No slurry spreading	No
	BK2B	– 30 cows plus calves, also silage/hay	Early April	October 2016	7 – 10 days in field and off for one month	No slurry spreading	Yes
	BK3A				NA		
Commons River (CM)	CM1	Replacements for dairy herd, 1.5 LU/Ha	Mid-March	Mid-October		No slurry spreading	Yes
	CM2				NA		
	CM3	Dairy, 1.6 LU/Ha	April 1	December 1		One slurry spreading	Yes
Milltown Lake (MT)	MT1	50 dairy cows	April	Mid-November	2 days in field, 3 weeks off	Slurry spreading once every 2 – 3 years	No
	MT2	20 beef suckler cows	April	Mid-November	4 – 5 weeks out, same period off	No slurry spreading	No
	MT3				NA		

**Appendix B1.** Summary of the information obtained in the survey carried out with farmers involved in the study.

Catchment	Site	Type and number of animals	Cattle introduced in field	Cattle removed from field	Period of grazing	Slurry spreading	Nutrient management plan
Munster	BWA	Mixed livestock, 10 cows	Mid-April	September/ October	2 weeks, 3 times per year	Slurry spreading once in end of summer	No
Blackwater	BWB				NA		
(BW)	BWC	Beef, calves and weenlings, 20 animals	Mid-July	End of October	Rotations depending on grass growth	Slurry spreading once in July, not always	Yes
Douglas River	DGA	Mixed livestock, 20 – 25 animals	June	Mid- October/Mid- November	1 week in field, 1 week off	Slurry spreading once a year in July	No
(DG)	DGB				NA		
	DGC				NA		

Appendix B1 (continued): Summary of the information obtained in the survey carried out with farmers involved in the study.

## References

Abdelzaher, A.M., Wright, M.E., Ortega, C., Solo-Gabriele, H.M., Miller, G., Elmir, S., Newman, X., Shih, P., Bonilla, J., Bonilla, T.D., Palmer, C.J., Scott, T., Lukasik, J., Harwood, V.J., McQuaig, S., Sinigalliano, C., Gidley, M., Plano, L.R.W., Zhu, X., Wang, J.D., Fleming, L.E., 2010. Presence of pathogens and indicator microbes at a non-point source subtropical recreational marine beach. Appl. Environ. Microbiol. 76, 724–732. <u>https://doi.org/10.1128/AEM.02127-09</u>

Abia, A.L.K., Ubomba-Jaswa, E., Momba, M.N.B., 2016. Prevalence of pathogenic microorganisms and their correlation with the abundance of indicator organisms in riverbed sediments. Int. J. Environ. Sci. Technol. 13, 2905–2916. <u>https://doi.org/10.1007/s13762-016-1116-y</u>

Agouridis, C.T., Workman, S.R., Warner, R.C., Jennings, G.D., 2005. Livestock Grazing Management Impacts on Stream Water Quality. J. Am. Water Resour. Assoc. 41, 591–606 <u>https://doi.org/10.1111/j.1752-1688.2005.tb03757.x</u>

Alm, E.W., Burke, J., Spain, A., 2003. Fecal indicator bacteria are abundant in wet sand at freshwater beaches. Water Res. 37, 3978–3982. <u>https://doi.org/10.1016/S0043-1354(03)00301-4</u>

Anderson, K.L., Whitlock, J.E., Harwood, V.J., 2005. Persistence and differential survival of fecal indicator bacteria in subtropical waters and sediments. Appl. Environ. Microbiol. 71, 3041–8. <u>https://doi.org/10.1128/AEM.71.6.3041-3048.2005</u>

Arnscheidt, J., Jordan, P., Li, S., McCormick, S., McFaul, R., McGrogan, H. J., Neal, M., Sims, J. T., 2007. Defining the sources of low-flow phosphorus transfers in complex catchments. Sci. Total Environ., 382(1), 1–13. <u>doi:10.1016/j.scitotenv.2007.03.036</u>

Avery, S.M., Moore, A., Hutchison, M.L., 2004. Fate of *Escherichia coli* originating from livestock faeces deposited directly onto pasture. Lett. Appl. Microbiol. 38, 355–359. <u>https://doi.org/10.1111/j.1472-765X.2004.01501.x</u>

Badgley, B.D., Thomas, F.I.M., Harwood, V.J., 2011. Quantifying environmental reservoirs of fecal indicator bacteria associated with sediment and submerged aquatic vegetation. Environ. Microbiol. 13, 932–942. <u>https://doi.org/10.1111/j.1462-2920.2010.02397.x</u>
Belsky, A.J., Matzke, A., Uselman, S., Uselman, M. and S., 1999. Survey of livestock influences on stream and riparian ecosystems in the western United States. J. Soil Water Conserv. 54, 419–431.

Bilotta, G.S., Brazier, R.E., 2008. Understanding the influence of suspended solids on water quality and aquatic biota. Water Res. 42, 2849–2861. https://doi.org/10.1016/j.watres.2008.03.018

Bilotta, G. S., Brazier, R.E., 2008. Understanding the influence of suspended solids on water quality and aquatic biota. Water Res. 42, 2849–2861. https://doi.org/10.1016/j.watres.2008.03.018

Birgand, F., Skaggs, R.W., Chescheir, G.M., Gilliam, W., 2007. Nitrogen Removal in Streams of Agricultural Catchments — A Literature Review. Crit. Rev. Environ. Sci. Technol. 37, 381–487. <u>https://doi.org/10.1080/10643380600966426</u>

Boehm, A.B., Griffith, J., Mcgee, C., Edge, T.A, Whitman, R., Getrich, M., Jay, J.A., Ferguson, D., Goodwin, K.D., Lee, C.M., Madison, M., Weisberg, S.B., 2010. Faecal indicator bacteria enumeration in beach sand: a comparison study of extraction methods in medium to coarse sands. J Appl Microbiol. 107(5), 1740–1750. <u>https://doi.org/10.1111/j.1365-2672.2009.04440.x.Faecal</u>

Bond, T.A., Sear, D., Edwards, M., 2012. Temperature-driven river utilisation and preferential defecation by cattle in an English chalk stream. Livest. Sci. 146, 59–66. <u>https://doi.org/10.1016/j.livsci.2012.02.022</u>

Bond, T., Sear, D., Sykes, T., 2014. Estimating the contribution of in-stream cattle faeces deposits to nutrient loading in an English Chalk stream. Agric. Water Manag. 131, 156–162. https://doi.org/10.1016/j.agwat.2013.08.015

Boulton, A.J., Findlay, S., Marmonier, P., Stanley, E.H., Valett, H.M., 1998. The Functional Significance of the Hyporheic Zone in Streams and Rivers. Annu. Rev. Ecol. Syst. 29, 59–81. https://doi.org/10.1146/annurev.ecolsys.29.1.59Bowes, M.J., Jarvie, H.P., Halliday, S.J., Skeffington, R.A., Wade, A. J., Loewenthal, M., Gozzard, E., Newman, J.R., Palmer-Felgate, E.J., 2015. Characterising phosphorus and nitrate inputs to a rural river using high-frequency

concentration–flow relationships. Sci. Total Environ. 511, 608–620. https://doi.org/10.1016/j.scitotenv.2014.12.086

Braccia, A., Voshell, J. R., 2006. Environmental factors accounting for benthic macroinvertebrate assemblage structure at the sample scale in streams subjected to a gradient of cattle grazing. Hydrobiologia, 573, 55-73.

Braccia, A., Voshell, J. R., 2007. Benthic Macroinvertebrate Responses to Increasing Levels of Cattle Grazing in Blue Ridge Mountain Streams, Virginia, USA. Environ. Monit. Assess., 131, 185-200.

Bradshaw, J.K., Snyder, B.J., Oladeinde, A., Spidle, D., Berrang, M.E., Meinersmann, R.J., Oakley, B., Sidle, R.C., Sullivan, K., Molina, M., 2016. Characterizing relationships among fecal indicator bacteria, microbial source tracking markers, and associated waterborne pathogen occurrence in stream water and sediments in a mixed land use watershed. Water Res. 101, 498–509. <u>https://doi.org/10.1016/j.watres.2016.05.014</u>

Bragina, L., 2017. Assessing microbial contamination of stream sediments in a rural agricultural catchment. PhD Thesis. Dundalk Institute of Technology, Ireland.

Bragina, L., Sherlock, O., van Rossum, A.J., Jennings, E., 2017. Cattle exclusion using fencing reduces *Escherichia coli* (*E. coli*) level in stream sediment reservoirs in northeast Ireland. Agric. Ecosyst. Environ. 239, 349–358. <u>https://doi.org/10.1016/j.agee.2017.01.021</u>

Brehony, C., Cullinan, J., Cormican, M., Morris, D., 2018. Shiga toxigenic *Escherichia coli* incidence is related to small area variation in cattle density in a region in Ireland. Sci. Total Environ. 637–638, 865–870. <u>https://doi.org/10.1016/j.scitotenv.2018.05.038</u>

Bremner, K., Gordon, R.J., Powers, J., Rooney, N., Madani, A., 2016. PARTIAL OR FULLY RESTRICTED CATTLE WATERING ACCESS: WATER QUALITY CONSIDERATIONS. Appl. Eng. Agric. 32(6), 811–821. <u>https://doi.org/10.13031/aea.32.11491</u>

Butturini, A., Battin, T.O.M.J., Sabater, F., 2000. NITRIFICATION IN STREAM SEDIMENT BIOFILMS : THE ROLE OF AMMONIUM CONCENTRATION AND DOC QUALITY. Water Res. 34, 629–639. Byers, H.L., Cabrera, M.L., Matthews, M.K., Franklin, D.H., Andrae, J.G., Radcliffe, D.E., McCann, M.A, Kuykendall, H.A, Hoveland, C.S., Calvert, V.H., 2005. Phosphorus, sediment, and *Escherichia coli* loads in unfenced streams of the Georgia Piedmont, USA. J. Environ. Qual. 34, 2293–2300. <u>https://doi.org/10.2134/jeq2004.0335</u>

Canty, A., & Ripley, B. (2020). Package 'boot'.

Camargo, J. a., Alonso, Á., 2006. Ecological and toxicological effects of inorganic nitrogen pollution in aquatic ecosystems: A global assessment. Environ. Int. 32, 831–849. https://doi.org/10.1016/j.envint.2006.05.002

Capece, J.C., Campbell, K.L., Bohlen, P.J., Graetz, D.A., Portier, K.M., 2007. Soil phosphorus, cattle stocking rates, and water quality in subtropical pastures in Florida, USA. Rangel. Ecol. Manag. 60, 19–30. <u>https://doi.org/10.2111/05-072R1.1</u>

Carpenter, S.R., 2008. Phosphorus control is critical to mitigating eutrophication. Proc. Natl. Acad. Sci. U. S. A. 105, 11039–11040.

Carter, M. R., Gregorich, E. G. (Ed.). (1993). Soil sampling and methods of analysis. CRC press.

Carson, A.,2010. A temporal and spatial analysis of eutrophication and its drivers in Milltown Lake and its catchment. Co. Monaghan, Ireland. PhD Thesis. Dundalk Institute of Technology, Ireland.

Carson, A., Jennings, E., Linnane, S., Jordan, S.N., 2015. Clearing the muddy waters: using lake sediment records to inform agricultural management. J. Paleolimnol. 53, 1–15. <u>https://doi.org/10.1007/s10933-014-9803-5</u>

Chalar, G., Garcia-Pesenti, P., Silva-Pablo, M., Perdomo, C., Olivero, V., Arocena, R., 2017. Weighting the impacts to stream water quality in small basins devoted to forage crops, dairy and beef cow production. Limnologica 65, 76–84. <u>https://doi.org/10.1016/j.limno.2017.06.002</u>

Chique, C., Potito, A.P., Molloy, K., Cornett, J., 2018. Tracking recent human impacts on a nutrient sensitive Irish lake : integrating landscape to water linkages. Hydrobiologia 807, 207–231. <u>https://doi.org/10.1007/s10750-017-3395-9</u>

Cho, K.H., Pachepsky, Y.A., Kim, J.H., Guber, A.K., Shelton, D.R., Rowland, R., 2010. Release of *Escherichia coli* from the bottom sediment in a first-order creek: Experiment and reach-specific modeling. J. Hydrol. 391, 322–332. <u>https://doi.org/10.1016/j.jhydrol.2010.07.033</u>

Clarendon, S.D. V, Weaver, D.M., Davies, P.M., Coles, N.A., 2019. The influence of particle size and mineralogy on both phosphorus retention and release by streambed sediments. J. Soils Sediments19(5), 2624-2633

Clary, W.P., 1999. Stream channel and vegetation responses to late spring cattle grazing. J. Range Manag. 52, 218–227.

Croxen, M. A., Law, R. J., Scholz, R., Keeney, K. M., Wlodarska, M., Finlay, B. B. (2013). Recent advances in understanding enteric pathogenic *Escherichia coli*. Clinical Microbiology Reviews, 26(4), 822-880.

Cloutier, D.D., Mclellan, S.L., 2017. Distribution and Differential Survival of Traditonal and Alternative Indicators of Fecal Pollution at Freshwater Beaches. Appl. Environ. Microbiol. 83, 1–17. <u>https://doi.org/10.1128/AEM.02881-16CE</u>

Collins, A.L., McGonigle, D.F., 2008. Monitoring and modelling diffuse pollution from agriculture for policy support: UK and European experience. Environ. Sci. Policy 11, 97–101. https://doi.org/10.1016/j.envsci.2008.01.001

Collins, R., Mcleod, M., Hedley, M., Donnison, A., Close, M., Hanly, J., Horne, D., Ross, C., Davies-Colley, R., Bagshaw, C., Matthews, L., 2007. Best management practices to mitigate faecal contamination by livestock of New Zealand waters. New Zeal. J. Agric. Res. 50, 267–278. <u>https://doi.org/10.1080/00288230709510294</u>

Conroy, E., Turner, J.N., Rymszewicz, A., O'Sullivan, J.J., Bruen, M., Lawler, D., Lally, H., Kelly-Quinn, M., 2016. The impact of cattle access on ecological water quality in streams: Examples from agricultural catchments within Ireland. Sci. Total Environ. 547, 17–29. https://doi.org/10.1016/j.scitotenv.2015.12.120 Cooper, C.M., 1993. Biological Effects of Agriculturally Derived Surface Water Pollutants on Aquatic Systems—A Review. J. Environ. Qual. 22, 402–408. https://doi.org/10.2134/jeq1993.00472425002200030003x

Corner-Warner, S.A., Gooddy, D.C, Ullah, S., Glover, L., Kettrige, N., Wexler, S.K., Kaiser, J., Krause, S., 2020. Seasonal variability of sediment controls of nitrogen cycling in an agricultural stream. Biogeochemistry 148, 31 – 48. <u>https://doi.org/10.1007/s10533-020-00644-z</u>

Correll, D.L., 1998. The Role of Phosphorus in the Eutrophication of Receiving Waters: A Review. J. Environ. Qual. 27, 261–266. https://doi.org/10.2134/jeq1998.00472425002700020004x

Craig, D.L., Fallowfield, H.J., Cromar, N.J., 2004. Use of microcosms to determine persistence of *Escherichia coli* in recreational coastal water and sediment and validation with in situ measurements. J. Appl. Microbiol. 96, 922–930. <u>https://doi.org/10.1111/j.1365-2672.2004.02243.x</u>

Crowther, J., Kay, D., Wyer, M.D., 2002. Faecal-indicator concentrations in waters draining lowland pastoral catchments in the UK: Relationships with land use and farming practices. Water Res. 36, 1725–1734. <u>https://doi.org/10.1016/S0043-1354(01)00394-3</u>

Central Statistics Office, 2020. https://www.cso.ie/en/

Cullen, P., Ryan, M., O'Donoghue, C., Kilgarriff, P., 2018. The economics of Agri-Environment Scheme desing: an Irish case study. 92<sup>nd</sup> Annual Conference of the Agricultural Economics Society, University of Warwick, England, 16-18 April 2018

Dai, X., Karring, H., 2014. A Determination and Comparison of Urease Activity in Feces and Fresh Manure from Pig and Cattle in Relation to Ammonia Production and pH Changes. PLoS One, 9(11), e110402. <u>https://doi.org/10.1371/journal.pone.0110402</u>

Dean, W.E., 1974. Determination of carbonate and organic matter in calcareous sediments and sedimentary rocks by loss on ignition: comparison with other methods. J. Sed. Petrol. 44: 242–248. <u>https://doi.org/10.1306/74D729D2-2B21-11D7-8648000102C1865D</u>

Deakin, J., Flynn, R., Archbold, M., Daly, D., O'Brien, R., Orr, A., Misstear, B., 2016. Understanding pathways transferring nutrients to streams: Review of a major Irish study and its implications for determining water quality management strategies. Biol. Environ. 116B. <u>https://doi.org/10.3318/BIOE.2016.19</u>

Department of Housing, Planning and Local Government, 2018a. River Basin Management Plan for Ireland 2018 – 2021. Retrieved online on January 27, 2020 from <u>https://www.housing.gov.ie/sites/default/files/publications/files/rbmp\_full\_reportweb.pdf</u>

Department of Housing, Planning and Local Governmen, 2018b. Strategic Environmental Assessment Statement for Ireland's Nitrates Action Programme. Retrieved on Agust 17, 2020 from

https://www.housing.gov.ie/sites/default/files/publications/files/mdr1387rp0005f01\_sea\_state ment\_nap.pdf

Desmarais, T.R., Solo-Gabriele, H.M., Palmer, C.J., 2002. Influence of soil on fecal indicator organisms in a tidally influenced subtropical environment. Appl. Environ. Microbiol. 68, 1165–1172. <u>https://doi.org/10.1128/AEM.68.3.1165-1172.2002</u>

Dijkstra, J., Oenema, O., van Groenigen, J.W., Spek, J.W., van Vuuren, A.M., Bannink, A., 2013. Diet effects on urine composition of cattle and N<sub>2</sub>O emissions. Animal 7, 292–302. <u>https://doi.org/10.1017/S1751731113000578</u>

Dodds, W.K., Oakes, R.M., 2008. Headwater Influences on Downstream Water Quality. Environ. Manage. 41, 367–377. <u>https://doi.org/10.1007/s00267-007-9033-y</u>

Dodds, W.K., Smith, V.H., 2016. Nitrogen , phosphorus , and eutrophication in streams. Inl. Waters 6, 155–164. <u>https://doi.org/10.5268/IW-6.2.909</u>

Dou, Z., Knowlton, K.F., Kohn, R.A, Wu, Z., Satter, L.D., Zhang, G., Toth, J.D., Ferguson, J.D., 2002. Phosphorus characteristics of dairy feces affected by diets. J. Environ. Qual. 31, 2058–2065. <u>https://doi.org/10.2134/jeq2002.2058</u>

Douglas, R.W., Menary, W., Jordan, P., 2007. Phosphorus and sediment transfers in a grassland river catchment. Nutr. Cycl. Agroecosystems 77, 199–212. https://doi.org/10.1007/s10705-006-9057-3 Draganova, I., Yule, I., Stevenson, M., Betteridge, K., 2015. The effects of temporal and environmental factors on the urination behaviour of dairy cows using tracking and sensor technologies. Precis. Agric. <u>https://doi.org/10.1007/s11119-015-9427-4</u>

Drummond, J.D., Davies-Colley, R.J., Stott, R., Sukias, J.P., Nagels, J.W., Sharp, A., Packman, A.I., 2014. Retention and remobilization dynamics of fine particles and microorganisms in pastoral streams. Water Res. 66, 459–472. https://doi.org/10.1016/j.watres.2014.08.025

Dudgeon, D., Arthington, A.H., Gessner, M.O., Kawabata, Z.-I., Knowler, D.J., Lévêque, C., Naiman, R.J., Prieur-Richard, A.-H., Soto, D., Stiassny, M.L.J., Sullivan, C.A., 2006. Freshwater biodiversity: importance, threats, status and conservation challenges. Biol. Rev. Camb. Philos. Soc. 81, 163–82. <u>https://doi.org/10.1017/S1464793105006950</u>

Durand, P., Breuer, L., Johnes, P. J., Billen, G., Butturini, A., Pinay, G., van Grinsven, H., Garnier, J., Rivett, M., Reay, D. S., Curtis, C., Siemens, J., Maberly, S., Kaste, O., Humborg, C., Loeb, R., de Klein, J., Hejzlar, J., Skoulikidis, N., Kortelainen, P., Lepisto, A., Wright, R., 2011. Nitrogen processes in aquatic ecosystems. In: Sutton, M. A., Howard, C. M., Erisman, J. W., Billen, G., Bleeker, A., Grennfelt, P., van Grinsven, H., Grizzetti, B. (eds.) European Nitrogen Assessment. Cambridge University Press, Cambridge, pp. 126-146. ISBN 9781107006126 Available at <u>http://centaur.reading.ac.uk/20855/</u>

Dworak, T.,Berglund, M. Grandmougin, B., Mattheiss, V. Holen, S., 2009. International review on payment schemes for wet buffer strips and other types of wet zones along privately owned land. Study for RWS-Waterdienst. Ecologic Institute, Berlin/Wien.

Eder, A., Exner-Kittridge, M., Strauss, P., Blöschl, G., 2014. Re-suspension of bed sediment in a small stream–results from two flushing experiments. Hydrol. Earth Syst, Sci, 18(3), 1043-1052. <u>https://doi.org/10.5194/hess-18-1043-2014</u>

Edwards, A.C., Watson, H.A., Cook, Y.E.M., 2012. Source strengths, transport pathways and delivery mechanisms of nutrients, suspended solids and coliforms within a small agricultural headwater catchment. Sci. Total Environ. 434, 123–129. https://doi.org/10.1016/j.scitotenv.2011.10.055 Emerson, H.J., Gillmor, D.A., 1999. The Rural Environment Protection Scheme of the Republic of Ireland. Land use policy 16, 235–245. <u>https://doi.org/10.1016/S0264-8377(99)00018-6</u>

Environmental Protection Agency, 2007. The provision and quality of drinking water in Ireland: a report for the years 2006 - 2007. Retrieved on October 27, 2015 from <a href="https://www.epa.ie/pubs/reports/water/drinking/DW%20Report%2020062.pdf">https://www.epa.ie/pubs/reports/water/drinking/DW%20Report%2020062.pdf</a>

Environmental Protection Agency, 2009. The provision and quality of drinking water in Ireland: a report for the years 2007 - 2008. Retrieved on October 27, 2015 from <a href="http://www.epa.ie/pubs/reports/water/drinking/Final%20DW%20Report%202007%20(2)2.pdf">http://www.epa.ie/pubs/reports/water/drinking/Final%20DW%20Report%202007%20(2)2.pdf</a>

Environmental Protection Agency, 2011. Water treatment manual: Disinfection. Retrieved on October 27, 2015 from <u>http://www.epa.ie/pubs/advice/drinkingwater/Disinfection2\_web.pdf</u>

Environmental Protection Agency, 2015. Cryptosporidiosis: human, animal and environmental interface. Retrieved on October 27, 2015 from <u>http://www.epa.ie/pubs/reports/research/health/Research%20152%20Report%20Web\_versi</u> on.pdf

Environmental Protection Agency, 2016. Drinking Water Report for Public Water Supplies 2015. Retrieved on July 3, 2017 from

http://www.epa.ie/pubs/reports/water/drinking/2015%20DW%20Report%20Public%20Suppli es\_web.pdf

Environmental Protection Agency, 2019. Water Quality in Ireland 2013 – 2018. Retrieved on April 5, 2020 from

https://www.epa.ie/pubs/reports/water/waterqua/waterqualityinireland2013-2018.html

Environmental Protection Agency, 2020. https://www.catchments.ie/data/#/catchment/25A?\_k=gj6taf

European Centre for Disease Prevention and Control. Shiga-toxin/verocytotoxin-producing *Escherichia coli* (STEC/VTEC) infection. In: ECDC. Annual epidemiological report for 2018. Stockholm: ECDC; 2020.

European Commission, 2010a. The EU Nitrates Directive. Retrieved November 19, 2015 from <a href="http://ec.europa.eu/environment/pubs/pdf/factsheets/nitrates.pdf">http://ec.europa.eu/environment/pubs/pdf/factsheets/nitrates.pdf</a>

European Commission , 2010b. Water is for life: how the Water Framework Directive helps safeguard EU's resources. Retrieved November 19, 2015 from <a href="http://www.eurosfaire.prd.fr/7pc/doc/1301470143\_kh3109164enc\_002.pdf">http://www.eurosfaire.prd.fr/7pc/doc/1301470143\_kh3109164enc\_002.pdf</a>

European Commission , 2015a. The history of the CAP. Retrieved November 10, 2015 from <u>http://ec.europa.eu/agriculture/cap-history/index\_en.htm</u>

European Commission, 2015b. The Nitrates Directive. Retrieved November 19, 2015 from <a href="http://ec.europa.eu/environment/water/water-nitrates/index\_en.html">http://ec.europa.eu/environment/water/water-nitrates/index\_en.html</a>

European Commission, 2015c. CAP Context Indicators 2014 - 2020. Retrieved online on December 13, 2018 from <u>https://ec.europa.eu/agriculture/cap-indicators/context/2015/full-text\_en.pdf</u>

European Commision, 2016. CAP in Ireland - Agriculture at a glance. Retrieved online on December 13, 2018 from <u>https://ec.europa.eu/info/sites/info/files/food-farming-fisheries/by\_country/documents/cap-in-your-country-ie\_en.pdf.</u>

European Commission, 2019a. The Post-2020 Common Agricultural Policy: Environmental Benefits. Retrieved online on Agust 22, 2020 from <a href="https://ec.europa.eu/info/sites/info/files/food-farming-fisheries/key\_policies/documents/cap-post-2020-environ-benefits-simplification\_en.pdf">https://ec.europa.eu/info/sites/info/files/food-farming-fisheries/key\_policies/documents/cap-post-2020-environ-benefits-simplification\_en.pdf</a>

European Commission, 2019b. The European Green Deal. Communication from the Commissiom to the European Parliament, the European Council, the Council, the Eueopean Economic and Social Committee and the Committee of the Regions. Retrieved online on August 22, 2020 from <a href="https://ec.europa.eu/info/sites/info/files/european-green-deal-communication\_en.pdf">https://ec.europa.eu/info/sites/info/files/european-green-deal-communication\_en.pdf</a>

European Commission, 2020a. Introduction to the EU Water Framework Directive. Accessed online on August 13, 2020 at <a href="https://ec.europa.eu/environment/water/water-framework/info/intro\_en.htm">https://ec.europa.eu/environment/water/water-framework/info/intro\_en.htm</a>

European Commission, 2020b. Farm to Fork Strategy. Retrieved online on August 22, 2020 from <u>https://ec.europa.eu/food/sites/food/files/safety/docs/f2f\_action-plan\_2020\_strategy-info\_en.pdf</u>

European Commission, 2021a. Drinking Water Legislation. Assessed online on May 28th

## 2021 at https://ec.europa.eu/environment/water/water-drink/legislation\_en.html

European Commision, 2021b. Bathing Water Quality. Assessed online on May 28<sup>th</sup> 2021 at <u>https://ec.europa.eu/environment/water/water-bathing/summary.html</u>

European Environment Agency, 2018a. Agricultural land: nitrogen balance. Accessed online on Agust 24, 2020 at <u>https://www.eea.europa.eu/airs/2018/natural-capital/agricultural-land-nitrogen-balance</u>

European Environment Agency, 2018b. European Waters. Assessment of status and pressures 2018. Retrieved online on August 22, 2020 from <a href="https://www.eea.europa.eu/publications/state-of-water">https://www.eea.europa.eu/publications/state-of-water</a>

European Environment Agency, 2019. The European environment - state and outlook 2020. ISBN: 978-92-9480-090-9. Retrieved online on August 23, 2020 from <u>https://www.eea.europa.eu/publications/soer-2020</u>

Eurostat, 2019. Agriculture, forests and fisheries statistics – 2019 edition. Retrieved online on May 5, 2020 from <u>https://ec.europa.eu/eurostat/web/products-statistical-books/-/KS-FK-19-001</u>.

Eurostat, 2020. Agri-environmental indicator – livestock patterns. Accessed online on May 5, 2020 at <a href="https://ec.europa.eu/eurostat/statistics-explained/index.php/Agri-environmental\_indicator\_-livestock\_patterns">https://ec.europa.eu/eurostat/statistics-explained/index.php/Agri-environmental\_indicator\_-livestock\_patterns</a>

Evans, M.R., Ribeiro, C.D., Salmon, R.L., 2003. Hazards of healthy living: bottled water and salad vegetables as risk factors for Campylobacter infection. Emerg. Infect. Dis. 9, 1219–1225. <u>https://doi.org/10.3201/eid0910.020823</u>

Eyles, R., Niyogi, D., Townsend, C., Benwell, G., Weinstein, P., 2003. Spatial and Temporal Patterns of Campylobacter Contamination Underlying Public Health Risk in the Taieri River, New Zealand. J. Environ. Qual. 32, 1820–1828.

Falkengren-Grerup, U., Brink, D., Brunet, J., 2006. Land use effects on soil N, P, C and pH persist over 40 – 80 years of forest growth on agricultural soils. For. Ecol. Manage. 225, 74–81. <u>https://doi.org/10.1016/j.foreco.2005.12.027</u>

Fanning, A., Craig, M., Webster, P., Bradley, C., Tierney, D., Wilkes, R., Mannix, A., 2017. Water Quality in Ireland 2010-2015. Environmental Protection Agency. <u>https://doi.org/10.1186/s12904-017-0245-5</u>

Fayer, R., 2004. Cryptosporidium : a water-borne zoonotic parasite. Vet. Parasitol. 126, 37– 56. <u>https://doi.org/10.1016/j.vetpar.2004.09.004</u>

Fenlon, D.R., Ogden, I.D., Vinten, A., Svoboda, I., 2000. The fate of *Escherichia coli* and *E. coli* O157 in cattle slurry after application to land. J. Appl. Microbiol. 88, 149S-156S. https://doi.org/10.1111/j.1365-2672.2000.tb05343.x

Finn, J. A., Ó hUallacháin, D., 2012.. A review of evidence on the environmental impact of Ireland's rural environment protection scheme (REPS). Biology and Environment: Proceedings of the Royal Irish Academy 112B(1), 11-34.

Forshay, K.J., Dodson, S.I., 2011. Macrophyte presence is an indicator of enhanced denitrification and nitrification in sediments of a temperate restored agricultural stream. Hydrobiologia 668, 21-34. DOI 10.1007/s10750-011-0619-2

Fox, G.A., Purvis, R.A., Penn, C.J., 2016. Streambanks : A net source of sediment and phosphorus to streams and rivers. J. Environ. Manage. 181, 602–614. <u>https://doi.org/10.1016/j.jenvman.2016.06.071</u>

Foy, R.H., Lennox, S.D., Gibson, C.E., 2003. Changing perspectives on the importance of urban phosphorus inputs as the cause of nutrient enrichment in Lough Neagh. Sci. Total. Environ. 310, 87–99.

Freeman, M.C., Pringle, C.M., Jackson, C.R., 2007. Hydrologic connectivity and the contribution of stream headwaters to ecological integrity at regional scales. J. Am. Water Resour. Assoc. 43, 5–14. <u>https://doi.org/10.1111/j.1752-1688.2007.00002.x</u>

Frey, S.K., Gottschall, N., Wilkes, G., Gregoire, D.S., Topp, E., Pintar, K.D.M., Sunohara, M., Marti, R. and Lapen, D.R., 2015. Rainfall-Induced Runoff from Exposed Streambed Sediments: An Important Source of Water Pollution. J. Environ. Qual. 44, 236-247. <u>https://doi.org/10.2134/jeq2014.03.0122</u> Galeone, D.G., 2000. Preliminary Effects of Streambank Fencing of Pasture Land on the Quality of Surface Water in a Small Watershed in Lancaster County, Pennsylvania. Water-Resources Investigations Report 00-4205. U.S. Department of the Interior and U.S. Geological Survey.

Garvey, P., Carroll, A., McNamara, E., McKeown, P.J., 2015. Verotoxigenic *Escherichia coli* transmission in Ireland: A review of notified outbreaks, 2004-2012. Epidemiol. Infect. 144, 917–926. <u>https://doi.org/10.1017/S0950268815002034</u>

Garzio-Hadzick, A., Shelton, D.R., Hill, R.L., Pachepsky, Y.A., Guber, A.K., Rowland, R., 2010. Survival of manure-borne E. coli in streambed sediment: Effects of temperature and sediment properties. Water Res. 44, 2753–2762. https://doi.org/10.1016/j.watres.2010.02.011

Georgakakos, C.B., Morris, C.K., Walter, M.T., 2018. Challenges and Opportunities With On-Farm Research: Total and Soluble Reactive Stream Phosphorus Before and After Implementation of a Cattle-Exclusion, Riparian Buffer. Front. Environ. Sci. 6, 1–13. <u>https://doi.org/10.3389/fenvs.2018.00071</u>

Gray, N.F. 2008. Drinking water quality: Problems and solutions. 2nd ed. Cambridge Univ. Press, New York. <u>https://doi.org/10.1017/CBO9780511805387</u>

GLAS Farm Planner, 2020. Some farmers may be forced to fence off watercourses by 2021. Accessed online on September 10, 2020 at <a href="https://glasfarmplanners.ie/some-farmers-may-be-forced-to-fence-off-watercourses-by-2021/">https://glasfarmplanners.ie/some-farmers-may-be-forced-to-fence-off-watercourses-by-2021/</a>

Gonda H.L. and Lindberg J.E. 1994. Evaluation of dietary nitrogen utilization in dairy cows based on urea concentrations in blood, urine and milk, and on urinary concentration of purine derivatives. Acta Agriculturae Scandinavica, Section A – Animal Science 44, 236–245.

Government of Ireland, 2021. Nitrates Directive. Assessed online on June 1<sup>st</sup> 2021 at https://www.gov.ie/en/publication/b87ad-nitrates-directive/

Haan, A.M.M., Russell, J.R., Davis, J.D., Morrical, D.G., Haan, M.M., Russell, J.R., Davis, J.D., Morrical, D.G., 2010. Grazing Management and Microclimate Effects on Cattle Distribution Relative to a Cool Season Pasture Stream. Rangel. Ecol. Manag. 63, 572–580. https://doi.org/10.2111/REM-D-09-00045.1 Hanlon, E. A., Johnson, G. V., Jones Jr, J. B. Karla, Y. P., O. Miller, R., Soltanpour, P. N., Tucker, M. R., Warnke, D.D., and Watson, M., 2000. 'Soil analysis handbook of reference methods' Soil and Plant Analysis Council Inc: U.S.

Harris, D., Horwáth, W. R., and Van Kessel, C., 2001. Acid fumigation of soils to remove carbonates prior to total organic carbon or carbon-13 isotopic analysis. Soil Sci. Soc. Am. J. *65*(6), 1853-1856.

Harrison, S.S.C., and Harris, I.T., 2002. The effects of bankside management on chalk stream invertebrate communities. Freshw. Biol. 47:2233–2245. https://doi.org/10.1046/j.1365-2427.2002.00939.x

Hassard, F., Andrews, A., Jones, D.L., Parsons, L., Jones, V., Cox, B.A., Daldorph, P., Brett, H., McDonald, J.E., Malham, S.K., 2017. Physicochemical factors influence the abundance and culturability of human enteric pathogens and fecal indicator organisms in estuarine water and sediment. Front. Microbiol. 8, 1–18. <u>https://doi.org/10.3389/fmicb.2017.01996</u>

Hassard, F., Gwyther, C.L., Farkas, K., Andrews, A., Jones, V., Cox, B., Brett, H., Jones, D.L., McDonald, J.E., Malham, S.K., 2016. Abundance and distribution of enteric bacteria and viruses in coastal and estuarine sediments-A review. Front. Microbiol. 7. https://doi.org/10.3389/fmicb.2016.01692

Heathwaite, A.L., 2010. Multiple stressors on water availability at global to catchment scales: Understanding human impact on nutrient cycles to protect water quality and water availability in the long term. Freshw. Biol. 55, 241–257. <u>https://doi.org/10.1111/j.1365-</u> 2427.2009.02368.x

Hedrick, L.B., Anderson, J.T., Welsh, S.A., Lin, L., 2013. Sedimentation in Mountain Streams: A Review of Methods of Measurement. Natural Resources 4, 92 – 104. http://dx.doi.org/10.4236/nr.2013.41011

Heiman, K.E., Mody, R.K., Johnson, S.D., Griffin, P.M., Gould, L.H., 2015. *Escherichia coli* O157 Outbreaks in the United States , 2003 – 2012. Emerg Infect Dis 21, 1293–1301. <u>https://doi.org/10.3201/eid2108.141364</u> Hennessy, T., Buckley, C., Cushion, M., Kinsella, A., Moran, B., 2011. National Farm Survey of Manure Application and Storage Practices on Irish Farms. Agriculture Economics and Farm Surveys Department. Teagasc.

Herbst, D.B., Bogan, M.T., Roll, S.K., Safford, H.D., 2012. Effects of livestock exclusion on in-stream habitat and benthic invertebrate assemblages in montane streams. Freshw. Biol. 57, 204–217. <u>https://doi.org/10.1111/j.1365-2427.2011.02706.x</u>

Hering, D., Birk., S., 2018. MARS Project – Managing Aquatic systems and water Resources under multiple Stressors. Policy Brief - Relevance of multiple pressures, related stressors and their interactions in European River Basin Management. Retrieved online on August 22, 2020 from <u>http://www.mars-</u>

project.eu/files/download/policy\_briefs/MARS\_Policy\_Brief\_Relevance\_of\_multiple\_pressure s.pdf

House, W.A., 2003. Geochemical cycling of phosphorus in rivers. Appl. Geochemistry 18, 739–748. <u>https://doi.org/10.1016/S0883-2927(02)00158-0</u>

Hodgson, C. J., Oliver, D. M., Fish, R. D., Bulmer, N. M., Heathwaite, A. L., Winter, M., Chadwick, D. R., 2016. Seasonal persistence of faecal indicator organisms in soil following dairy slurry application to land by surface broadcasting and shallow injection. J. Environ. Manage. 183(1), 325-332. <u>https://doi.org/10.1016/j.jenvman.2016.08.047</u>

.Hickey, C.W., Vickkers, M.L., 1994. Toxicity of ammonia to nine native New Zealand freshwater invertebrate species. Arch. Environ. Contam. Toxicol. 26:292–298. <u>https://doi.org/10.1007/BF00203554</u>

House, W.A., Denison, F.H., 2002. Exchange of Inorganic Phosphate between River Waters and Bed-Sediments. Environ. Sci. Technol. 36, 4295–4301. <u>https://doi.org/10.1021/es020039z</u>

Howard-Williams, C., Davies, J., Pickmere, S., 1982. The dynamics of growth, the effects of changing area and nitrate uptake by watercress (*Nasturtium officinale* R. Br). in a New Zealand stream. J. App. Ecol. 19, 589-601.

Howard-Williams, C., Davies-Colley, R., Rutherford, K., Wilcock, R., 2010. Diffuse pollution and freshwater degradation: New Zealand perspectives. Issues and solutions to diffuse

pollution, Selected papers from the 14<sup>th</sup> International Conference of the IWA Diffuse Pollution Specialist Group, 126-140.

Ishii, S., Hansen, D.L., Hicks, R.E., Sadowsky, M.J., 2007. Beach sand and sediments are temporal sinks and sources of *Escherichia coli* in Lake Superior. Environ. Sci. Technol. 41, 2203–2209. <u>https://doi.org/10.1021/es0623156</u>

Ishii, S., Sadowsky, M.J., 2008. *Escherichia coli* in the Environment: Implications for Water Quality and Human Health. Microbes Environ. 23, 101–108. <u>https://doi.org/10.1264/jsme2.23.101</u>

Izagirre, O., Serra, A., Guasch, H., Elosegi, A., 2009. Effects of sediment deposition on periphytic biomass , photosynthetic activity and algal community structure. Sci. Total Environ. 407, 5694–5700. <u>https://doi.org/10.1016/j.scitotenv.2009.06.049</u>

Jamieson, R.C., Joy, D.M., Lee, H., Kostaschuk, R., Gordon, R.J., 2005a. Resuspension of Sediment-Associated in a Natural Stream. J. Environ. Qual. 34, 581. <u>https://doi.org/10.2134/jeq2005.0581</u>

Jamieson, R., Joy, D.M., Lee, H., Kostaschuk, R., Gordon, R., 2005b. Transport and deposition of sediment-associated *Escherichia coli* in natural streams. Water Res. 39, 2665–2675. <u>https://doi.org/10.1016/j.watres.2005.04.040</u>

Jang, J., Hur, H., Sadowsky, M., Byappanahalli, M.N., Yan, T., Ishii, S., 2017. Environmental *Escherichia coli*: Ecology and Public Health Implications - A Review. ARPN J. Eng. Appl. Sci. 12, 3218–3221. <u>https://doi.org/10.1111/jam.13468</u>

Jarvie, H.P., Sharpley, A.N., Spears, B., Buda, A.R., May, L., Kleinman, P.J.A., 2013. Water Quality Remediation Faces Unprecedented Challenges from "Legacy Phosphorus ". Environ. Sci. Technol. 47, 8997–8998. <u>https://doi.org/10.1021/es403160a</u>

Jenkins, M.B., Adams, M.P., Endale, D.M., Fisher, D.S., Lowrance, R., Newton, G.L., Vellidis, G., 2015. Storm flow dynamics and loads of fecal bacteria associated with ponds in southern piedmont and coastal plain watersheds with animal agriculture. Agric. Water Manag. 148, 97–105. <u>https://doi.org/10.1016/j.agwat.2014.09.022</u> Jenkins, M.B., Endale, D.M., Fisher, D.S., Adams, M.P., Lowrance, R., Newton, G.L, Vellidis, G., 2012. Survival dynamics of fecal bacteria in ponds in agricultural watersheds of the Piedmont and Coastal Plain of Georgia. Water Res. 46, 176–186. https://doi.org/10.1016/j.watres.2011.10.049

Jennings, E., Mills, P., Jordan, P., Jensen, J., Sondergaard, M., Barr, A., Glasgow, G., Irvine, K., 2003. Eutrophication from agricultural sources: seasonal patterns and effects of phosphorus. Environmental Protection Agency. ISBN:1-84095-106-0

Jordan, P., Melland, A.R., Mellander, P.E., Shortle, G., Wall, D., 2012. The seasonality of phosphorus transfers from land to water: Implications for trophic impacts and policy evaluation. Sci. Total Environ. 434, 101–109. <u>https://doi.org/10.1016/j.scitotenv.2011.12.070</u>

Kallis, G., Butler, D., 2001. The EU water framework directive: measures and implications. Water Policy 3, 125–142. <u>https://doi.org/10.1016/S1366-7017(01)00007-1</u>

Kaper, J.B., Nataro, J.P., Mobley, H.L.T., 2004. Pathogenic *Escherichia coli*. Nat. Rev. Microbiol. 2, 123–140. <u>https://doi.org/10.1038/nrmicro818</u>

Karmali, M.A., Gannon, V., Sargeant, J.M., 2010. Verocytotoxin-producing *Escherichia coli* (VTEC). Vet. Microbiol. 140, 360–370. <u>https://doi.org/10.1016/j.vetmic.2009.04.011</u>

Kauffman, J.B., Krueger, W.C., Vavra, M., 1983. Impacts of cattle on stream banks in northeastern Oregon. J. Range Manag. 36, 683–685.

Kay, D., Aitken, M., Crowther, J., Dickson, I., Edwards, A.C., 2007. Reducing fluxes of faecal indicator compliance parameters to bathing waters from diffuse agricultural sources : The Brighouse Bay study, Scotland. Environ. Pollut. 147, 138–149. https://doi.org/10.1016/j.envpol.2006.08.019

Kay, D., Crowther, J., Stapleton, C.M., Wyer, M.D., 2018. Faecal indicator organism inputs to watercourses from streamside pastures grazed by cattle: Before and after implementation of streambank fencing. Water Res. 143, 229–239.

https://doi.org/10.1016/j.watres.2018.06.046

Kilgarriff, P., Ryan, M., Donoghue, C.O., Green, S., Ó hUallacháin, D., 2020. Livestock exclusion from watercourses : Policy e ff ectiveness and implications. Environ. Sci. Policy 106, 58–67. <u>https://doi.org/10.1016/j.envsci.2020.01.013</u>

Kim, J.W., Pachepsky, Y.A., Shelton, D.R., Coppock, C., 2010. Effect of streambed bacteria release on E. coli concentrations: Monitoring and modeling with the modified SWAT. Ecol. Modell. 221, 1592–1604. <u>https://doi.org/10.1016/j.ecolmodel.2010.03.005</u>

Kirchmann, H., Witter, E., 1992. Composition of fresh, aerobic and anaerobic farm animal dungs. Bioresour. Technol. 40, 137–142. <u>https://doi.org/10.1016/0960-8524(92)90199-8</u>

Kondolf, G.M. 1995. Five elements for effective evaluation of stream restoration. Restor. Ecol. 3:133–136. <u>https://doi.org/10.1111/j.1526-100X.1995.tb00086.x</u>

Kool, D.M., Hoffland, E., Hummelink, E.W.J., Willem, J., Groenigen, V., 2006. Increased hippuric acid content of urine can reduce soil  $N_2O$  fluxes. Soil. Biol. 38, 1021–1027. <u>https://doi.org/10.1016/j.soilbio.2005.08.017</u>

Kronvang, B., Audet, J., Baattrup-Pedersen, A., Jensen, H.S., Larsen, S.E., 2012. Phosphorus Load to Surface Water from Bank Erosion in a Danish Lowland River Basin. J. Environ. Qual. 41, 304–313. <u>https://doi.org/10.2134/jeq2010.0434</u>

Kurz, I., O'Reilly, C.D., Tunney, H., 2006. Impact of cattle on soil physical properties and nutrient concentrations in overland flow from pasture in Ireland. Agric. Ecosyst. Environ. 113, 378–390. <u>https://doi.org/10.1016/j.agee.2005.10.004</u>

Kuypers, M. M. M., Marchant, H. K., Kartal, B., 2018. The microbial nitrogen-cycling network. Nature Reviews Microbiology, 16(5), 263–276. <u>doi:10.1038/nrmicro.2018.9</u>

Lantinga, E.A., Keuning, J.A., Groenwold, J., Deenan, P.J.A.G., 1987. Distribution of excreted nitrogen by grazing cattle and its effects on sward quality, herbage production and utilization. In: van der Meer HG, Unwin RJ, van Dijk TA, Ennik GC, editors. Animal Manure on Grassland and Fodder Crops Fertilizer or Waste? Dordrecht7 Martinus Nijhoff Publishers; 1987. p. 103–7

Lardner, H.A., Kirychuk, B.D., Braul, L., Willms, W.D., Yarotski, J., 2005. The effect of water quality on cattle performance on pasture. Aust. J. Agric. Res. 56, 97–104. <u>https://doi.org/10.1071/AR04086</u>

Larsen, R.E., Miner, J.R., Buckhouse, J.C., Moore, J.A., 1994. Water-quality benefits of having cattle manure deposited away from streams. Bioresour. Technol. 48:113–118. <u>https://doi.org/10.1016/0960-8524(94)90197-X</u>

Lassaletta, L., García-Gómez, H., Gimeno, B.S., Rovira, J. V., 2010. Headwater streams: Neglected ecosystems in the EU Water Framework Directive. Implications for nitrogen pollution control. Environ. Sci. Policy 13, 423–433. <u>https://doi.org/10.1016/j.envsci.2010.04.005</u>

Latacz-Lohmann, U., Hodge, I., 2003. European agri-environmental policy for the 21st century. Aust. J. Agric. Resour. Econ. 47, 123–139. <u>https://doi.org/10.1111/1467-8489.00206</u>

Laubel, A., Kronvang, B., Hald, A.B., Jensen, C., 2003. Hydromorphological and biological factors influencing sediment and phosphorus loss via bank erosion in small lowland rural streams in Denmark. Hydrol. Processes 17:3443–3463. <u>https://doi.org/10.1002/hyp.1302</u>

Leip, A., Billen, G., Garnier, J., Grizzetti, B., Lassaletta, L., Reis, S., 2015. Impacts of European livestock production : nitrogen , sulphur , phosphorus and greenhouse gas emissions , land-use , water eutrophication and biodiversity. Environ. Res. Lett. 10, 1–13. https://doi.org/10.1088/1748-9326/10/11/115004

Lemly, A.D., 1982. Modification of benthic insect communities in polluted streams : combined effects of sedimentation and nutrient enrichment 245, 229–245.

Lenihan, M.H., Brasier, K.J., 2009. Scaling down the European model of agriculture: the case of the Rural Environmental Protection Scheme in Ireland. Agric. Human Values 26, 365–378. <u>https://doi.org/10.1007/s10460-008-9169-2</u>

Line, D.E., 2003. Changes in a stream's physical and biological conditions following livestock exclusion. Trans. ASAE 46, 287–293. <u>https://doi.org/10.13031/2013.12979</u>

Line, D.E., Harman, W.A., Jennings, G.D., Thompson, E.J, Osmond, D.L., 2000. Nonpointsource pollutant load reductions associated with livestock exclusion. J. Environ. Qual. 29, 1882-1890. <u>https://doi.org/10.2134/jeq2000.00472425002900060022x</u>

Line, D. E., Osmond, D. L., Childres, W., 2016. Effectiveness of livestock exclusion in a pasture of central North Carolina. J. Environ. Qual. 45(6), 1926-1932.

Linnane, S., Jordan, S., McCarthy, V., Jennings, E., Carson, A., Sweeny, N., Wynne, C., McDonald, B., 2011. National Source Protection Pilot Project at the Churchill and Oram Group Water Scheme Co., Monaghan. Final report. Centre for Freshwater and Environmental Studies, Dundalk Institute of Technology, Dundalk, Ireland.

Ling, T.Y., Azyyati, Z.I.M., Lesley, M.B., 2012. Temporal and spatial variations and decay rates of *E. coli* in river sediment. J. Sustain. Sci. Manag. 7, 16–22.

Liu, X., Zhang, X. and Zhang, M., 2008. Major factors influencing the efficacy of vegetated buffers on sediment trapping: a review and analysis. J. Environ. Qual. 37, 1667–1674.

Lloyd, C.E.M., Johnes, P.J., Freer, J.E., Carswell, A.M., Jones, J.I., Stirling, M.W., Hodgkinson, R.A., Richmond, C., Collins, A.L., 2019. Determining the sources of nutrient fl ux to water in headwater catchments : Examining the speciation balance to inform the targeting of mitigation measures. Sci. Total Environ. 648, 1179–1200. <u>https://doi.org/10.1016/j.scitotenv.2018.08.190</u>

Lucci, G.M., McDowell, R.W., Condron, L.M., 2010. Phosphorus and sediment loads/ transfer from potential source areas in New Zealand dairy pasture. Soil Use Manag. 26, 44– 52

Lucy, F.E., Graczyk, T.K., Tamang, L., Miraflor, A., Minchin, D., 2008. Biomonitoring of surface and coastal water for *Cryptosporidium*, *Giardia*, and human-virulent microsporidia using molluscan shellfish. Parasitol. Res. 103, 1369–1375. <u>https://doi.org/10.1007/s00436-008-1143-9</u>

Madden, D., Harrison, S., Finn, J.A., 2019. The impact of cattle drinking points on aquatic macroinvertebrates in streams in south-east Ireland. Irish J. Agriculuture Food Res. 2017, 13–20. <u>https://doi.org/10.2478/ijafr-2019-0002</u>

Magilligan, F.J., McDowell, P.F.. 1997. Stream channel adjustments following elimination of cattle grazing. J. Am. Water Resour. Assoc. 33:867–878. <u>https://doi.org/10.1111/j.1752-1688.1997.tb04111.x</u>

Malmqvist, B., Rundle, S., 2002. Threats to the running water ecosystems of the world. Environ. Conserv. 29(2), 134–153. <u>https://doi.org/10.1017/S0376892902000097</u>

McCabe, E., Burgess, C.M., Lawal, D., Whyte, P., Duffy, G., 2018. An investigation of shedding and super-shedding of Shiga toxigenic *Escherichia coli* O157 and *E. coli* O26 in cattle presented for slaughter in the Republic of Ireland. Zoonoses Public Health 1–9. https://doi.org/10.1111/zph.12531

McDowell, R. W., Wilcock, R.J., 2007. Sources of Sediment and Phosphorus in Stream Flow of a Highly Productive Dairy Farmed Catchment. J. Environ. Qual. 36, 540–548. <u>https://doi.org/10.2134/jeq2006.0352</u>

McDowell, R.W., Sharpley, A.N., 2001. A comparison of fluvial sediment phosphorus (P) chemistry in relation to location and potential to influence stream P concentrations. Aquat. Geochemistry 7, 255–265. <u>https://doi.org/10.1023/A:1015274426411</u>

McGarrigle, M.L., Bowman, J.J., Clabby, K.J., Lucey, J., Cunningham, P., MacCarthaigh, M., Keegan, M., Cantrell, B., Lehane, M., Clenaghan, C., Toner, P.F., 2002. Water Quality in Ireland 1998–2000. Environmental Protection Agency, Ireland.

McGechan, M.B., Lewis, D.R., Hooda, P.S., 2005. Modelling through-soil transport of phosphorus to surface waters from livestock agriculture at the field and catchment scale. Sci. Total Environ. 344, 185–199. <u>https://doi.org/10.1016/j.scitotenv.2005.02.015</u>

McKergow, L.A., Weaver, D.M., Prosser, I.P., Grayson, R.B., Reed, A.E.G., 2003. Before and after riparian management: Sediment and nutrient exports from a small agricultural catchment, Western Australia. J. Hydrol. 270, 253–272. <u>https://doi.org/10.1016/S0022-1694(02)00286-X</u>

Melland, A.R., Mellander, P., Murphy, P.N.C., Wall, D.P., Mechan, S., Shine, O., Shortle, G., Jordan, P., 2012. Stream water quality in intensive cereal cropping catchments with

## regulated nutrient management. Environ, Sci. Policy 24, 58-70. https://doi.org/10.1016/j.envsci.2012.06.006

Mellander, P.E., Melland, A.R., Jordan, P., Wall, D.P., Murphy, P.N.C., Shortle, G., 2012. Quantifying nutrient transfer pathways in agricultural catchments using high temporal resolution data. Environ. Sci. Policy 24, 44–57. <u>https://doi.org/10.1016/j.envsci.2012.06.004</u>

Mendonça, C.M.P.N. de, Almeida, A.S., Castro, A., de Lurdes Delgado, M., Soares, S., da Costa, J.M.C., Canada, N.B., 2007. Molecular characterization of Cryptosporidium and Giardia isolates from cattle from Portugal. Vet. Parasitol. 147, 47–50. https://doi.org/10.1016/j.vetpar.2007.03.019

Mesa, L.M., Maldini, C., Mayora, G., Saigo, M., 2016. Decomposition of cattle manure and colonization by macroinvertebrates in sediment of a wetland of the Middle Paraná River. J. Soils Sediments. <u>https://doi.org/10.1007/s11368-016-1465-7</u>

Miller, J., Chanasyk, D., Curtis, T., Entz, T., Willms, W., 2010. Influence of streambank fencing with a cattle crossing on riparian health and water quality of the Lower Little Bow River in Southern Alberta, Canada. Agric. Water Manag. 97, 247–258. https://doi.org/10.1016/j.agwat.2009.09.016

Miller, J.J., Bremer, E., Curtis, T., Chanasyk, D.S., 2017. An assessment of nutrient dynamics in streambank soils of the Lower Little Bow River in southern Alberta using ion exchange membranes. Water Qual. Res. J. 52(3), 196–208. https://doi.org/10.2166/wqrj.2017.007

Misselbrook, T., Fleming, H., Camp, V., Umstatter, C., Duthie, C., Nicoll, L., Waterhouse, T., 2016. Automated monitoring of urination events from grazing cattle. Agric. Ecosyst. Environ. 230, 191–198.

Mockler, E.M., Deakin, J., Archbold, M., Daly, D., Bruen, M., 2016. Nutrient load apportionment to support the identification of appropriate water framework directive measures. Biol. Environ. 116B, 245–263. <u>https://doi.org/10.3318/bioe.2016.22</u>

Money, P., Kelly, A.F., Gould, S.W.J., Denholm-Price, J., Threlfall, E.J., Fielder, M.D., 2010. Cattle, weather and water: Mapping *Escherichia coli* O157:H7 infections in humans in England and Scotland. Environ. Microbiol. 12, 2633–2644. <u>https://doi.org/10.1111/j.1462-2920.2010.02293.x</u>

Muirhead, R.W., Davies-Colley, R.J., Donnison, A. M., Nagels, J.W., 2004. Faecal bacteria yields in artificial flood events: Quantifying in-stream stores. Water Res. 38, 1215–1224. https://doi.org/10.1016/j.watres.2003.12.010

Muirhead, R.W., Monaghan, R.M., 2012. A two reservoir model to predict *Escherichia coli* losses to water from pastures grazed by dairy cows. Environ. Int. 40, 8–14. <u>https://doi.org/10.1016/j.envint.2011.11.009</u>

Mulholland, B., Fullen, M.A., 1991. Cattle trampling and soil compaction on loamy sands. Soil Use Manag. 7, 189–193.

Murphy, J. and Riley, J. P. (1962). A modified single solution method for the determination of phosphate in natural waters. Analytica chimica acta, *27*, 31-36.

Murphy, B.P., McCabe, E., Murphy, M., Buckley, J.F., Crowley, D., Fanning, S., Duffy, G., 2016. Longitudinal study of two Irish dairy herds: Low numbers of Shiga toxin-producing *Escherichia coli* O157 and O26 super-shedders identified. Front. Microbiol. 7, 1–8. <u>https://doi.org/10.3389/fmicb.2016.01850</u>

Murphy, P.N.C., Mellander, P., Melland, A.R., Buckley, C., Shore, M., Shortle, G., Wall, D.P., Treacy, M., Shine, O., Mechan, S., Jordan, P., 2015. Variable response to phosphorus mitigation measures across the nutrient transfer continuum in a dairy grassland catchment. Agric. Ecosyst. Environ. 207, 192–202. <u>https://doi.org/10.1016/j.agee.2015.04.008</u>

Murphy, D. V., Macdonald, A.J., Stockdale, E.A., Goulding, K.W.T., Fortune, S., Gaunt, J.L., Poulton P R, Wakefield, J.A., Webster, C.P., Wilmer, W.S., 2000. Soluble organic nitrogen in agricultural soils. Biol. Fertil. Soils 30, 374–387.

Naden, P.S., Murphy, J.F., Old, G.H., Newman, J., Scarlett, P., Harman, M., Duerdoth, C.P., Hawczak, A., Pretty, J.L., Arnold, A., Laizé, C., Hornby, D.D., Collins, A.L., Sear, D.A., Jones, J.I., 2016. Understanding the controls on deposited fine sediment in the streams of agricultural catchments. Sci. Total Environ. 547, 366–381. https://doi.org/10.1016/j.scitotenv.2015.12.079

Nagels, J.W., Davies-Colley, R.J., Donnison, A.M., Muirhead, R.W., 2002. Faecal contamination over flood events in a pastoral agricultural stream in New Zealand. Water Sci. Technol. 45, 45–52.

Neumann, K., Elbersen, B. S., Verburg, P. H., Staritsky, I., Pérez-Soba, M., de Vries, W., Rienks, W. A., 2009. Modelling the spatial distribution of livestock in Europe. Landsc. Ecol. , 24(9), 1207-1222.

Neidhardt, H., Achten, F., Kern, S., Schwientek, M., Oelmann, Y., 2019. Phosphorus Pool Composition in Soils and Sediments of Transitional Ecotones under the Influence of Agriculture. J. Environ. Qual. 48, 1325–1335. <u>https://doi.org/10.2134/jeq2019.01.0012</u>

Nickerson, Z.L., Mortazavi, B., Atkinson, C.L., 2019. Using functional traits to assess the influence of burrowing bivalves on nitrogen-removal in streams. Biogeochemistry 146, 125–143. <u>https://doi.org/10.1007/s10533-019-00612-2</u>

Nixon, S.W., 1995. Coastal marine eutrophication : A definition, social causes , and future concerns. Ophelia 41, 199–219. <u>https://doi.org/10.1080/00785236.1995.10422044</u>

O'Boyle, S., Wilkes, R., Mcdermott, G., Ni Longphuirt, S., 2020. Will recent improvements in estuarine water quality in Ireland be compromised by plans for increased agricultural production ? A case study of the Blackwater estuary in southern Ireland. Ocean Coast. Manag. 143, 87–95. <u>https://doi.org/10.1016/j.ocecoaman.2016.12.020</u>

O'Callaghan, P., Kelly-Quinn, M., Jennings, E., Antunes, P., O'Sullivan, M., Fenton, O., hUallacháin, D.Ó., 2018. The Environmental Impact of Cattle Access to Watercourses: A Review. J. Environ. Qual. 0, 0. <u>https://doi.org/10.2134/jeq2018.04.0167</u>

Ó hUallacháin, D., Jennings, E., Antunes, P. Green, S., Kilgarriff, P., Linnane, S., O'Callaghan, P., O'Sullivan, M., Regan, F., Ryan, M., Kelly-Quinn, M., 2020. COSAINT: Cattle exclusion from watercourses: environmental and socio-economic implications.EPA Research Report No.330. Available online at

## http://www.epa.ie/pubs/reports/research/water/Research\_Report\_330.pdf

O'Mullan, G., Juhl, A., Reichert, R., Schneider, E., Martinez, N., 2019. Patterns of sedimentassociated fecal indicator bacteria in an urban estuary: Benthic-pelagic coupling and implications for shoreline water quality. Sci. Total Environ. 656, 1168–1177. <u>https://doi.org/10.1016/j.scitotenv.2018.11.405</u>

O'Sullivan, M., Ó hUallacháin, D., Antunes, P.O., Jennings, E., Kelly-Quinn, M., 2019. The impacts of cattle access points on deposited sediment levels in headwater streams in Ireland. River Res. Appl. 1–13. <u>https://doi.org/10.1002/rra.3382</u>

Óhaiseadha, C., Hynds, P.D., Fallon, U.B., O'Dwyer, J., 2016. A geostatistical investigation of agricultural and infrastructural risk factors associated with primary verotoxigenic *E. coli* (VTEC) infection in the Republic of Ireland, 2008-2013. Epidemiol. Infect. 145, 95–105. https://doi.org/10.1017/S095026881600193X

Oladeinde, A., Bohrmann, T., Wong, K., Purucker, S.T., Bradshaw, K., Brown, R., Snyder, B., Molina, M., 2014. Decay of fecal indicator bacterial populations and bovine-associated source-tracking markers in freshly deposited cow pats. Appl. Environ. Microbiol. 80, 110–118. <u>https://doi.org/10.1128/AEM.02203-13</u>

Oliver, D. M., Clegg, C. D., Heathwaite, A. L., Haygarth, P. M., 2007. Preferential attachment of *Escherichia coli* to different particle size fractions of an agricultural grassland soil. Water Air Soil Pollut., 185(1-4), 369-375. <u>http://doi.org/10.1007/s11270-007-9451-8</u>

Oliver, D.M., Page, T., 2016. Effects of seasonal meteorological variables on *E. coli* persistence in livestock faeces and implications for environmental and human health. Sci. Rep. 6, 1–10. <u>https://doi.org/10.1038/srep37101</u>

Oliver, D.M., Page, T., Heathwaite, A.L., Haygarth, P.M., 2010. Re-shaping models of *E*. *coli* population dynamics in livestock faeces : Increased bacterial risk to humans? Environ. Int. 36, 1–7. <u>https://doi.org/10.1016/j.envint.2009.08.006</u>

Orr, R.J., Griffith, B.A., Champion, R.A., Cook, J.E., 2012. Defaecation and urination behaviour in beef cattle grazing semi-natural grassland. Appl. Anim. Behav. Sci. 139, 18–25. <u>https://doi.org/10.1016/j.applanim.2012.03.013</u> Ouattara, N.K., Passerat, J., Servais, P., 2011. Faecal contamination of water and sediment in the rivers of the Scheldt drainage network. Environ. Monit. Assess. 183, 243–257. https://doi.org/10.1007/s10661-011-1918-9

Oudshoorn, F.W., Kristensen, T., Shahrak, E., 2008. Dairy cow defecation and urination frequency and spatial distribution in relation to time-limited grazing. Livest. Sci. 113, 62–73. https://doi.org/10.1016/j.livsci.2007.02.021

Pachepsky, Y., Shelton, D.R., 2011. *Escherichia coli* and Fecal Coliforms in Freshwater and Estuarine Sediments. Crit. Rev. Environ. Sci. Technol. 41, 1067–1110. <u>https://doi.org/10.1080/10643380903392718</u>

Palmer-Felgate, E.J., Jarvie, H.P., Withers, P.J.A., Mortimer, R.J.G., Krom, M.D., 2009. Stream-bed phosphorus in paired catchments with different agricultural land use intensity. Agric. Ecosyst. Environ. 134, 53–66. <u>https://doi.org/10.1016/j.agee.2009.05.014</u>

Park, Y., Pachepsky, Y., Hong, E.-M., Shelton, D., Coppock, C., 2017. Release from Streambed to Water Column during Baseflow Periods: A Modeling Study. J. Environ. Qual. 46, 219. <u>https://doi.org/10.2134/jeq2016.03.0114</u>

Pärn, J., Pinay, G., Mander, Ü., 2012. Indicators of nutrients transport from agricultural catchments under temperate climate: A review. Ecol. Indic. 22, 4–15. <u>https://doi.org/10.1016/j.ecolind.2011.10.002</u>

Pennington, H., 2010. *Escherichia coli* O157. Lancet 376, 1428–1435. https://doi.org/10.1016/S0140-6736(10)60963-4

Perchec-Merien, A.M., Lewis, G.D., 2012. Naturalized *Escherichia coli* from New Zealand wetland and stream environments. FEMS Microbiol. Ecol. 83, 494–503. <u>https://doi.org/10.1111/1574-6941.12010</u>

Petit, F., Clermont, O., Delannoy, S., Servais, P., Gourmelon, M., Fach, P., Oberlé, K., Fournier, M., Denamur, E., Berthe, T., 2017. Change in the structure of *Escherichia coli* population and the pattern of virulence genes along a rural aquatic continuum. Front. Microbiol. 8, 1–14. <u>https://doi.org/10.3389/fmicb.2017.00609</u>

Pham, V., Torresan, S., Critto, A., Marcomini, A., 2019. Alteration of freshwater ecosystem services under global change – A review focusing on the Po River basin (Italy) and the Red River basin (Vietnam). Sci. Total Environ. 652, 1347–1365. https://doi.org/10.1016/j.scitotenv.2018.10.303

Pinheiro , J., Bates, D., DebRoy, S., Sarkar, D., R Core Team (2019). nlme: Linear and Nonlinear Mixed Effects Models. R package version 3.1-149, <u>https://CRAN</u>.

Platts, W.S., Nelson, R.L., 1985. Impacts of Rest-Rotation Grazing on Stream Banks in Forested Watersheds in Idaho. North Am. J. Fish. Manag. 5, 547–556. <u>https://doi.org/10.1577/1548-8659(1985)5<547</u>

Ranganath, S.C., Hession, W.C., Wynn, T.M., 2009. Livestock exclusion influences on riparian vegetation, channel morphology, and benthic macroinvertebrate assemblages. J. Soil Water Conserv. 64, 33–42.

Ray, H., Saetta, D., Boyer, T.H., 2018. Characterisation of urea hydrolysis in fresh human urine and inhibition by chemical addition. Environ. Sci. Water Res. Technol. 4, 87–98. <u>https://doi.org/10.1039/c7ew00271h</u>

Records, R.M., Wohl, E., Arabi, M., 2016. Phosphorus in the river corridor. Earth Sci. Rev. 158, 65–88. <u>https://doi.org/10.1016/j.earscirev.2016.04.010</u>

Reddy, K.R., Kadlec, R.H., Flaig, E., Gale, P.M., 1999. Phosphorus Retention in Streams and Wetlands : A Review. Crit. Rev. Environ. Sci. Technol. 29, 83–146.

Rehmann, C.R., Soupir, M.L., 2009. Importance of interactions between the water column and the sediment for microbial concentrations in streams. Water Res. 43, 4579–4589. https://doi.org/10.1016/j.watres.2009.06.049

Relyea, C.D., Minshall, G.W., Danehy, R.J., 2012. Development and Validation of an Aquatic Fine Sediment Biotic Index. Environ. Manage. 49, 242–252. <u>https://doi.org/10.1007/s00267-011-9784-3</u>

Reynolds, C.S., Davies, P.S., 2001. Sources and bioavailability of phosphorus fractions in freshwaters: a British perspective. Biol. Rev. Camb. Philos. Soc. 76, 27–64. <u>https://doi.org/10.1111/j.1469-185X.2000.tb00058.x</u>

Rice, T., Quinn, N., Sleator, R.D., Lucey, B., 2016. Changing diagnostic methods and increased detection of verotoxigenic *Escherichia coli*, Ireland. Emerg. Infect. Dis. 22, 1656–1657. <u>https://doi.org/10.3201/eid2209.160477</u>

Rouillard, J., Lago, M., Abhold, K., Roeschel, L., Kafyeke, T., Klimmek, H., Mattheiß, V., 2018. Protecting and Restoring Biodiversity across the Freshwater, Coastal and Marine Realms: Is the existing EU policy framework fit for purpose? Environ. Policy Gov. 28, 114–128. <u>https://doi.org/10.1002/eet.1793</u>

Ryan, U.M., Bath, C., Robertson, I., Read, C., Elliot, A., Mcinnes, L., Traub, R., Besier, B., 2005. Sheep may not be an important zoonotic reservoir for Cryptosporidium and Giardia parasites. Appl. Environ. Microbiol. 71, 4992–4997. <u>https://doi.org/10.1128/AEM.71.9.4992-4997.2005</u>

Sand-Jensen, K.A.J., 1998. Influence of submerged macrophytes on sediment composition and near-bed flow in lowland streams. Freshw. Biol. 39, 663–679.

Schmidt-Kloiber, A., Bremerich, V., De Wever, A., Jahnig, S., Martens, K., Strackbein, J., Tockner, K., Hering, D., 2019. The Freshwater Information Platform : a global online network providing data , tools and resources for science and policy support. Hydrobiologia 838, 1–11. https://doi.org/10.1007/s10750-019-03985-5

Schulte, R.P.O., Melland, A.R., Fenton, O., Herlihy, M., Richards, K., Jordan, P., 2010. Modelling soil phosphorus decline: Expectations of Water Framework Directive policies. Environ. Sci. Policy 13, 472–484. <u>https://doi.org/10.1016/j.envsci.2010.06.002</u>

Scrimgeour, G.J., Kendall, S., 2003. Effects of livestock grazing on benthic invertebrates from a native grassland ecosystem. Freshw. Biol. 48, 347–362.

Sekhon, J. S., 2008. Multivariate and propensity score matching software with automated balance optimization: the matching package for R. J. Stat. Softw. 42(7), 1-47.

Selbie, D.R., Buckthought, L.E., Shepherd, M.A., 2015. The Challenge of the Urine Patch for Managing Nitrogen in Grazed Pasture Systems, Adv. Agron. 29, 229-292. <u>https://doi.org/10.1016/bs.agron.2014.09.004</u>

Sharpley, A., Jarvie, H.P., Buda, A., May, L., Spears, B., Kleinman, P., 2013. Phosphorus Legacy: Overcoming the Effects of Past Management Practices to Mitigate Future Water Quality Impairment. J. Environ. Qual. 42, 1308-1326. https://doi.org/10.2134/jeq2013.03.0098

Sheffield, R.E., Mostaghimi, S., Vaughan, D.H., Collins Jr., E.R., Allen, V.G., 1997. Off-Stream Water Sources for Grazing Cattle as Stream Bank Stabilization and Water Quality BMP. Trans. ASAE 40, 595–604. <u>https://doi.org/10.13031/2013.21318</u>

Shelton, D.R., Pachepsky, Y.A., Kiefer, L.A., Blaustein, R.A., McCarty, G.W., Dao, T.H., 2014. Response of coliform populations in streambed sediment and water column to changes in nutrient concentrations in water. Water Res. 59, 316–324. https://doi.org/10.1016/j.watres.2014.04.019

Sherriff, S.C., Ó hUallacháin, D., Rowan, J.S., Fenton, O., Jordan, P., 2019. Influence of land management on soil erosion, connectivity, and sediment delivery in agricultural catchments : Closing the sediment budget. L. Degrad. Dev. 1–15. <u>https://doi.org/10.1002/ldr.3413</u>

Shore, M., Jordan, P., Melland, A.R., Mellander, P., Mcdonald, N., Shortle, G., 2016. Incidental nutrient transfers : Assessing critical times in agricultural catchments using highresolution data. Sci. Total Environ. 553, 404–415. https://doi.org/10.1016/j.scitotenv.2016.02.085

Shore, M., Murphy, P.N.C., Jordan, P., Mellander, P., Kelly-quinn, M., Cushen, M., Mechan, S., Shine, O., Melland, A.R., 2013. Evaluation of a surface hydrological connectivity index in agricultural catchments. Environ. Model. Softw. 47, 7–15. <u>https://doi.org/10.1016/j.envsoft.2013.04.003</u>

Shore, M., Murphy, S., Mellander, P., Shortle, G., Melland, A.R., Crockford, L., Flaherty, V.O., Williams, L., Morgan, G., Jordan, P., 2017. Influence of storm flow and base flow

phosphorus pressures on stream ecology in agricultural catchments. Sci. Total Environ. 591, 469–483. <u>https://doi.org/10.1016/j.scitotenv.2017.02.100</u>

Spek, J.W., Bannink, A., Gort, G., Hendriks, W.H., Dijkstra, J., 2012. Effect of sodium chloride intake on urine volume, urinary urea excretion, and milk urea concentration in lactating dairy cattle. J. Dairy Sci. 95, 7288–7298. <u>https://doi.org/10.3168/jds.2012-5688</u>

Stutter, M.I., Chardon, W.J., Kronvang, B., 2012. Riparian Buffer Strips as a Multifunctional Management Tool in Agricultural Landscapes: Introduction. J. Environ. Qual. 41, 297. https://doi.org/10.2134/jeq2011.0439

Taylor, D., Dalton, C., Leira, M., Jordan, P., Chen, G., Leon-Vintro, L., Irvine, K., Bennion, H., Nolan, T., 2006. Recent histories of six productive lakes in the Irish Ecoregion based on multiproxy palaeolimnological evidence. Hydrobiologia 571, 237–259. <u>https://doi.org/10.1007/s10750-006-0243-8</u>

Teagasc, 2017a Agricultural Catchment Programme – Catchments. Accessed online at <a href="https://www.teagasc.ie/environment/water-quality/agricultural-catchments/catchments/">https://www.teagasc.ie/environment/water-quality/agricultural-catchments/catchments/</a>

Teagasc, 2017b. Agriculture in Ireland. Accessed online on May 5, 2020 at <a href="https://www.teagasc.ie/rural-economy/rural-economy/agri-food-business/agriculture-in-ireland/">https://www.teagasc.ie/rural-economy/rural-economy/agri-food-business/agriculture-in-ireland/</a>

Ternouth, J.H., 1990. Phosphorus and beef production in northern Australia. 3. Phosphorus in cattle - a review. Trop. Grasslands 24, 159-169.

Terry, J.A, McW.H. Benskin, C., Eastoe, E.F., Haygarth, P.M., 2014. Temporal dynamics between cattle in-stream presence and suspended solids in a headwater catchment. Environ. Sci. Process. Impacts 16, 1570-1577. <u>https://doi.org/10.1039/c3em00686g</u>

Thoma, D.P., Gupta, S.C., Bauer, M.E., Kirchoff, C.E., 2005. Airborne laser scanning for riverbank erosion assessment. Remote Sens. Environ. 95, 493–501. <u>https://doi.org/10.1016/j.rse.2005.01.012</u>

Thompson, J., Cassidy, R., Doody, D.G., Flynn, R., 2013. Predicting critical source areas of sediment in headwater catchments. Agric. Ecosyst. Environ. 179, 41–52. <u>https://doi.org/10.1016/j.agee.2013.07.010</u> Tietzsch-Tyler, D., Sleeman, A.G., 1994. Geology of South Wexford: A Geological Description of South Wexford and Adjoining Parts of Waterford, Kilkenny and Carlow to Accompany the Bedrock Geology 1: 100,000 Scale Map Series, Sheet 23, South Wexfor*d*. Geological Survey of Ireland.

Trimble, S.W., 1994. Erosional effects of cattle on streambanks in Tennessee, U.S.A. Earth Surf. Process. Landforms 19, 451–464. <u>https://doi.org/10.1002/esp.3290190506</u>

Trimble, S.W., Mendel, A.C., 1995. The cow as a geomorphic agent — A critical review A2 -Hupp, C.R. Biogeomorphology, Terr. Freshw. Syst. 13, 233–253. <u>https://doi.org/https://doi.org/10.1016/B978-0-444-81867-6.50019-8</u>

Trimmer, M., Grey, J., Heppell, C. M., Hildrew, A. G., Lansdown, K., Stahl, H., Yvon-Durocher, G., 2012. River bed carbon and nitrogen cycling: State of play and some new directions. Sci. Total Environ. 434, 143-158. <u>doi:10.1016/j.scitotenv.2011.10.074</u>

Tufekcioglu, M., 2010. Stream Bank Soil and Phosphorus Losses within Grazed Pasture Stream Reaches in the Rathbun Watershed in Southern Iowa (Ph.D. dissertation). Iowa State University, Aimes, IA.

United States Environmental Protection Agency (USEPA), 2007. METHOD 3051A. Microwave assisted acid digestion of sedimens, sludges, soils and oils. Available online at <u>https://www.epa.gov/sites/production/files/2015-12/documents/3051a.pdf</u>

Ulén, B., Bechmann, M., Folster, J., Jarvie, H.P., Tunney, H., 2007. Agriculture as a phosphorus source for eutrophication in the north-west European countries, Norway, Sweden, United Kingdom and Ireland: a review. Soil Use Manag. 23, 5–15.

Veerkamp, V., 2019. An investigation into the influence of nutrient export and climatic effects on the trophic status of a small inter-drumlin lake. PhD Thesis. Dundalk Institute of Technology. Ireland.

Vidon, P., Campbell, M.A., Gray, M., 2008. Unrestricted cattle access to streams and water quality in till landscape of the Midwest. Agric. Water Manag. 95, 322–330. https://doi.org/10.1016/j.agwat.2007.10.017 Vinten, A.J.A., Douglas, J.T., Lewis, D.R., Aitken, M.N., Fenlon, D.R., 2004. Relative risk of surface water pollution by *E. coli* derived from faeces of grazing animals compared to slurry application. Soil Use Manag. 20, 13–22. <u>https://doi.org/10.1079/SUM2004214</u>

Vörösmarty, C.J., McIntyre, P.B., Gessner, M.O., Dudgeon, D., Prusevich, A., Green, P., Glidden, S., Bunn, S.E., Sullivan, C. a, Liermann, C.R., Davies, P.M., 2010. Global threats to human water security and river biodiversity. Nature 467, 555–561. https://doi.org/10.1038/nature09549

Watanabe, F. S., Olsen, S.R., 1965. Test of an ascorbic acid method for determining phosphorus in water and NaHCO<sub>3</sub> extracts from soil. Soil Sci. Soc. Am. J. 29(6), 677-678.

Weaver, R.W., Entry, J.A., Graves, A., 2005. Numbers of fecal streptococci and *Escherichia coli* in fresh and dry cattle, horse, and sheep manure. Can. J. Microbiol. 51, 847–851. <u>https://doi.org/10.1139/w05-071</u>

Wells, B., Shaw, H., Hotchkiss, E., Gilray, J., Ayton, R., Green, J., Katzer, F., Wells, A., Innes, E., 2015. Prevalence, species identification and genotyping *Cryptosporidium* from livestock and deer in a catchment in the Cairngorms with a history of a contaminated public water supply. Parasit. Vectors 8, 66. <u>https://doi.org/10.1186/s13071-015-0684-x</u>

Whelan, J., Fry, J., 2010. The Irish Rural Environmental Protection Scheme and lack of Strategic Environmental Assessment. Working Paper 10-WP-RE-10. The Rural Economy Research Centre.

White, S.L., Sheffield, R.E., Washburn, S.P., King, L.D., Green, J.T., 2001. Spatial and Time Distribution of Dairy Cattle Excreta in an Intensive Pasture System. J. Environ. Qual. 30, 2180–2187. <u>https://doi.org/10.2134/jeq2001.2180</u>

Wilcock, R.J., Nagels, J.W., Rodda, H.J., O'Connor, M.B., Thorrold, B.S., Barnett, J.W., 1999. Water quality of a lowland stream in a New Zealand dairy farming catchment. New Zealand journal of marine and freshwater research, 33(4), 683-696. <u>https://doi.org/10.1080/00288330.1999.9516911</u>

Wilcock, R.J., Betteridge, K., Shearman, D., Fowles, C.R., Scarsbrook, M.R., Thorrold, B.S., Costall, D., 2009. Riparian protection and on-farm best management practices for restoration of a lowland stream in an intensive dairy farming catchment: A case study. New Zeal. J. Mar. Freshw. Res. 43, 803–818. <u>https://doi.org/10.1080/00288330909510042</u>

Wilcock, R.J., Monaghan, R.M., Quinn, J.M., Srinivasan, M.S., Houlbrooke, D.J., Duncan, M.J., Wright-Stow, A.E., Scarsbrook, M.R., 2013. Trends in water quality of five dairy farming streams in response to adoption of best practice and benefits of long-term monitoring at the catchment scale. Mar. Freshw. Res. 64, 401–412. <u>https://doi.org/10.1071/MF12155</u>

Willat, S.T., Pullar, D.M., 1983. Changes in Soil Physical Properties under Grazed Pastures. Aust. J. Soil Res. 22, 343–348.

Williams, A. P., Gordon, H., Jones, D.L., Strachan, N.J.C., Avery, L.M., Killham, K., 2008. Leaching of bioluminescent *Escherichia coli* O157:H7 from sheep and cattle faeces during simulated rainstorm events. J. Appl. Microbiol. 105, 1452–1460. <u>https://doi.org/10.1111/j.1365-2672.2008.03898.x</u>

Willms, W.D., Kenzie, O.R., Mcallister, T.A., Colwell, D., Veira, D., Wilmshurst, J.F., Entz, T., Olson, M.E., 2002. Effects of water quality on cattle performance. J. Range Manag. 55, 452–460. <u>https://doi.org/Doi 10.2307/4003222</u>

Wilson, J.L., Everard, M., 2017. Real-time consequences of riparian cattle trampling for mobilisation of sediment , nutrients and bacteria in a British lowland river. Int. J. River Basin Manag. 16, 231–244. <u>https://doi.org/10.1080/15715124.2017.1402778</u>

Winter, M., Oliver, D.M., Fish, R., Heathwaite, A.L., Chadwick, D., Hodgson, C., 2011. Catchments, sub-catchments and private spaces: Scale and process in managing microbial pollution from source to sea. Environ. Sci. Policy 14, 315–326. <u>https://doi.org/10.1016/j.envsci.2010.10.011</u>

Wironen, M.B., Bennett, E.M., Erickson, J.D., 2018. Phosphorus fl ows and legacy accumulation in an animal-dominated agricultural region from 1925 to 2012. Glob. Environ. Chang. 50, 88–99. <u>https://doi.org/10.1016/j.gloenvcha.2018.02.017</u>

Withers, P.J.A, Jarvie, H.P., 2008. Delivery and cycling of phosphorus in rivers: A review. Sci. Total Environ. 400, 379–395. <u>https://doi.org/10.1016/j.scitotenv.2008.08.002</u>

Wood, P.J. and Armitage, P.D., 1997. Biological effects of fine sediment in the lotic environment. Environmental Management 21, 203–217

Wood, S., 2015. Package 'mgcv'. R package version, 1, 29.

Wright, J.F., Berrie, A.D., 1987. ECOLOGICAL EFFECTS OF GROUNDWATER PUMPING AND A NATURAL DROUGHT ON THE UPPER REACHES OF A CHALK STREAM. Regul. Rivers Res. Manag. 1, 145–160.

Wynne, C., 2012. A Water Quality Assessment of Milltown Lake (Muckno Mill) Catchment, Co. Monaghan using Macrophyte and Macroinvertebrate Biological Indicators. PhD Thesis. Dundalk Institute of Technology, Ireland.

Zaimes, G.N., Schultz, R.C., Isenhart, T.M., 2008. Total phosphorus concentrations and compaction in riparian areas under different riparian land-uses of Iowa. Agric., Ecosyst. Environ. 127, 22–30. <u>https://doi.org/10.1016/j.agee.2008.02.008</u>

Zaimes, G.N., Schultz, R.C., 2011. Stream bed substrate composition adjacent to different riparian land-uses in Iowa, USA. Ecol. Eng. 37, 1692–1699. https://doi.org/10.1016/j.ecoleng.2011.06.029

Zuur, A., Ieno, E., Walker, N., Saveliev, A., Smith, G., 2009. Mixed Effect Models and Extensions in Ecology with R. Springer, New York.